TRUPANION INC.

Form 3

October 27, 2014

FORM 3

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

INITIAL STATEMENT OF BENEFICIAL OWNERSHIP OF

OMB APPROVAL

OMB Number:

3235-0104

Expires:

January 31, 2005

0.5

Estimated average burden hours per

response...

SECURITIES

Filed pursuant to Section 16(a) of the Securities Exchange Act of 1934, Section 17(a) of the Public Utility Holding Company Act of 1935 or Section 30(h) of the Investment Company Act of 1940

(Print or Type Responses)

1. Name and Address of Reporting 2. Date of Event Requiring 3. Issuer Name and Ticker or Trading Symbol Person * Statement TRUPANION INC. [TRUP] **ANDREW ALISON** (Month/Day/Year) 10/27/2014 (Last) (First) (Middle) 4. Relationship of Reporting 5. If Amendment, Date Original Person(s) to Issuer Filed(Month/Day/Year) C/O TRUPANION, INC., 907 (Check all applicable) NW BALLARD WAY (Street) 6. Individual or Joint/Group 10% Owner Director _X__ Officer Other Filing(Check Applicable Line) (give title below) (specify below) _X_ Form filed by One Reporting Chief Marketing Officer Person SEATTLE, WAÂ 98107 Form filed by More than One Reporting Person (City) (State) (Zip) Table I - Non-Derivative Securities Beneficially Owned 1. Title of Security 2. Amount of Securities 4. Nature of Indirect Beneficial Beneficially Owned Ownership Ownership (Instr. 4) (Instr. 4) Form: (Instr. 5) Direct (D) or Indirect (I) (Instr. 5) Â 0 Common Stock D Reminder: Report on a separate line for each class of securities beneficially SEC 1473 (7-02) owned directly or indirectly. Persons who respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB control number.

Table II - Derivative Securities Beneficially Owned (e.g., puts, calls, warrants, options, convertible securities)

1. Title of Derivative Security 2. Date Exercisable and 3. Title and Amount of 5. 6. Nature of Indirect (Instr. 4) **Expiration Date** Securities Underlying Conversion Ownership Beneficial Ownership (Month/Day/Year) **Derivative Security** or Exercise Form of (Instr. 5) (Instr. 4) Price of Derivative Derivative Security: Title Direct (D) Security

	Date Exercisable	Expiration Date		Amount or Number of Shares		or Indirect (I) (Instr. 5)	
Stock Option (right to buy)	(1)	02/04/2023	Common Stock	120,000	\$ 4.05	D	Â

Reporting Owners

Reporting Owner Name / Address	Relationships				
• 0	Director	10% Owner	Officer	Other	
ANDREW ALISON C/O TRUPANION, INC. 907 NW BALLARD WAY SEATTLE Â WAÂ 98107	Â	Â	Chief Marketing Officer	Â	

Signatures

/s/ Charlotte Sim-Warner as attorney-in-fact for Alison
Andrew
10/27/2014

**Signature of Reporting Person Date

Explanation of Responses:

- * If the form is filed by more than one reporting person, see Instruction 5(b)(v).
- ** Intentional misstatements or omissions of facts constitute Federal Criminal Violations. See 18 U.S.C. 1001 and 15 U.S.C. 78ff(a).

The option vests as to 25% of the total shares on October 1, 2013, and then 2.0833% of the total shares vest monthly thereafter, with (1) 100% of the total shares vested and exercisable on October 1, 2016, subject to the reporting person's provision of service to the issuer on each vesting date.

Note: File three copies of this Form, one of which must be manually signed. If space is insufficient, *See* Instruction 6 for procedure. Potential persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB number. size:10pt">

Counterparts A-70

11.08
Entire Agreement A-71

11.09
Amendments A-71

11.10
Severability A-71

11.11
Governing Law; Venue A-71

Reporting Owners 2

Attorney s Fees A-72	11.12
Audiley STees A-72	11.13
Waiver of Jury Trial A-72	11.14
Remedies A-72	11.14
	11.15
Waiver A-72	11.16
Construction A-73	
Waivers; Terminations A-73	11.17
	11.18
Non-Recourse A-74	11.19
Time of Essence A-74	
Further Assurances A-74	11.20
	11.21
Parent Guaranty A-74	11.22
Release A-75	

SCHEDULES AND EXHIBITS

Annex A Brand Business
Annex B Gaia Business

Schedules

Schedule 1.01(a)	Funded Indebtedness
Schedule 1.01(b)	Knowledge Groups
Schedule 1.01(c)	Gaia Agreements
Schedule 1.01(d)	Affiliate Agreements
Schedule 1.01(e)	Material Registered Company Intellectual Property
Schedule 2.05(b)	Pro Forma Closing Balance Sheet
Schedule 7.01	Conduct of Business
Schedule 7.06(b)	Material Consents

Schedule 7.01 Conduct of Business
Schedule 7.06(b) Material Consents
Schedule 8.01(a) Antitrust Laws
Schedule 8.02(d)(iii) Resigning Persons

Exhibits

Exhibit A	Contribution Agreement
Exhibit B	Form of Escrow Agreement
Exhibit C	Form of IP License Agreement
Exhibit D	Form of Transition Services Agreement
Exhibit E	Form of SVOD Rights Assignment Agreement
Exhibit F	Form of SVOD Rights Sub-License Agreement
Exhibit G	Form of Reception Services Agreement
Exhibit H	Form of Third Amendment and Assignment of Lease

-iv-

MEMBERSHIP INTEREST PURCHASE AGREEMENT

This Membership Interest Purchase Agreement, dated as of May 10, 2016 (this <u>Agreement</u>), is by and among (i) STRETCH & BEND HOLDINGS, LLC, a Delaware limited liability company (<u>Purchaser</u>); (ii) SEQUENTIAL BRANDS GROUP, INC., a Delaware corporation (<u>Parent</u>), solely for the purposes <u>of Section 7.17 and Article XI</u>, and (iii) GAIAM, INC., a Colorado corporation (<u>Seller</u>). Certain terms used in this Agreement are defined <u>in Section 1.01</u>.

RECITALS

- A. Seller directly owns all of the issued and outstanding Equity Securities of Gaiam Brand Holdco, LLC, a Delaware limited liability company (the <u>Company</u>), and indirectly owns all of the issued and outstanding Equity Securities of each of the Company subsidiaries as set forth in <u>Section 5.02 of the Disclosure Schedule</u> (collectively, the <u>Company Subsidiaries</u> and together with the Company, the <u>Brand Companies</u>).
- B. Seller operates the business described on <u>Annex A</u> hereto (the <u>Brand Business</u>), in addition to its other business described on <u>Annex B</u> hereto (the <u>Gaia Business</u>).
- C. Prior to the date hereof, Seller contributed the assets of the Brand Business to the Company (the <u>Contribution</u>), pursuant to that certain Contribution Agreement, a copy of which is attached hereto as <u>Exhibit A</u> (the <u>Contribution Agreement</u>).
- D. Concurrently with the execution and delivery of this Agreement, Seller entered into that certain Asset Purchase Agreement, dated as of the date hereof (the <u>Gaiam-FFL AP</u>A), pursuant to which, immediately prior to the Closing, Seller will cause the Brand Companies to sell and transfer to Fit For Life LLC (<u>FFL</u>), and FFL will purchase and assume from the Brand Companies, certain assets and liabilities relating to the Brand Business (the <u>FFL Acquired Assets and Assumed Liabilities</u>) pursuant to the terms and conditions of the Gaiam-FFL APA.
- E. Seller desires to sell and transfer to Purchaser, and Purchaser desires to purchase and acquire from Seller, the Brand Business (other than the FFL Acquired Assets and Assumed Liabilities) and all of the issued and outstanding Equity Securities of the Company (the Purchased Interest), on the terms and conditions and as more specifically provided in this Agreement.

AGREEMENT

In consideration of the covenants and agreements contained herein and the other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Seller, Purchaser, and Parent (solely for purposes of Article XI) agree as follows:

ARTICLE I

DEFINITIONS

1.01 Definitions. Capitalized terms and other terms used in this Agreement have the following respective meanings:

338(g) Subsidiaries has the meaning set forth in Section 7.10(g).

Accounting Referee has the meaning set forth in Section 2.06(c)(i).

Adjustment Escrow Account has the meaning set forth in Section 2.03,

A-1

Adjustment Escrow Amount means \$5,010,000.

Adverse Recommendation Change has the meaning set forth in Section 7.21(d).

Affiliate of any Person, means another Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such first Person; *provided*, *however*, that an Affiliate of any Person shall also include (a) any Person that directly or indirectly owns, or in which such Person directly or indirectly owns more than fifteen percent (15%) of any class of capital stock or other equity interest of such Person, (b) in the case of a corporation, any officer or director of such corporation, (c) in the case of a partnership, any general partner of such partnership, (d) in the case of a trust, any trustee or beneficiary of such trust, (e) any spouse, parent, sibling or child or lineal descendant of any individual described in clauses (a) through (d) above, and (f) any trust for the benefit of any individual described in clauses (a) through (e) above. For purposes hereof, the term control, under common control with or controlled by shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of any such Person or the power to veto major policy decisions of any such Person, whether through the ownership of Equity Securities, by contract or otherwise.

Affiliate Agreements means each of the Contracts set forth on Schedule 1.01(d).

<u>Agreement</u> has the meaning set forth in the opening paragraph of this Agreement.

<u>Allocation</u> has the meaning set forth in Section 7.11.

Alternate Debt Financing has the meaning set forth in Section 7.14(e).

Alternative Acquisition Agreement has the meaning set forth in Section 7.21(e)(ii).

Alternative Transaction means any of the following transactions: (a) any acquisition or purchase of a majority of either the voting power or economic value of the Equity Securities of Seller, the Company or the Company Subsidiaries by any Person other than Purchaser; (b) any merger, consolidation, business combination or similar transaction involving Seller or the Company (after the contribution of the Brand Business thereto pursuant to the Contribution Agreement(s)) and any Person other than Purchaser; or (c) any sale, lease, exchange, transfer, license, acquisition or disposition of (x) all or substantially all of the consolidated assets of Seller, the Company and the Company Subsidiaries (which shall include the Brand Business) or (y) all or substantially all of the assets of the Brand Business, in each case, in any single transaction or series of related transactions, in each case, other than pursuant to this Agreement and the Contemplated Transactions; *provided*, *however*, that in no event shall Alternative Transaction include any transaction solely involving or solely relating to the Gaia Business.

<u>Alternative Transaction Proposal</u> means any offer, proposal, letter of intent or indication of interest, written or oral (whether binding or non-binding), to the Company or Seller, relating to an Alternative Transaction.

<u>Antitrust Authorities</u> means the Antitrust Division of the United States Department of Justice, the United States Federal Trade Commission or the antitrust or competition law authorities of any other jurisdiction (whether federal, state, foreign or multinational).

Antitrust Laws means any Applicable Law relating to antirust or competition, including the HSR Act.

<u>Applicable La</u>w means, with respect to any Person, any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, code, regulations, enacted, adopted, approved, promulgated, made,

A-2

implemented or otherwise put into effect (and all judicial interpretations thereof), in each case, as of the date of this Agreement, by any Governmental Entity that applies to such Person, its business and its properties.

<u>Australia Joint Venture</u> means Gaiam Pty, Ltd., an Entity formed under the laws of Australia.

<u>Banker</u> means Stifel Nicolaus & Company, Incorporated.

Board means the board of directors of Seller.

Boulder Road Lease means that certain Lease Agreement, dated January 1, 2015, by and between Boulder Road LLC and Gaiam Americas, as amended by that certain First Amendment to Lease dated as of September 3, 2015, as further amended by that certain Second Amendment to Lease dated as of January 1, 2016 and as may be further amended, modified, or supplemented from time to time.

Brand Business has the meaning set forth in the Recitals.

Brand Companies has the meaning set forth in the Recitals.

Basket Amount has the meaning set forth in Section 10.04(a).

<u>Brand Directors</u> has the meaning set forth <u>in Section 7</u>.09.

<u>Brand Employees</u> has the meaning set forth in Section 5.16.

Brand Insurance Policies has the meaning set forth in Section 5.12.

<u>Business Day</u> means any day other than a Saturday, Sunday or a day on which commercial banks in Denver, Colorado or New York, New York are required or authorized to be closed for business.

<u>Cap Amount</u> has the meaning set forth in Section 10.04(b).

<u>Cash on Hand</u> means the cash and cash equivalents of the Brand Companies as of the Closing Date, as determined in accordance with GAAP; *provided, however*, that none of the cash and cash equivalents of the Australia Joint Venture shall be included in the calculation of Cash on Hand.

<u>Change of Control Payment</u> means any bonus payment that is due and payable, or will become due and payable, by any Brand Company to any current or former officer, director, manager, employee, consultant or independent contractor of any Brand Company upon, or as a result of, the consummation of the Contemplated Transactions, together with the portion of any applicable payroll Taxes for which any Brand Company is liable or gross-ups incurred and any related matching contributions required to be made under any applicable retirement plans by any Brand Company in respect thereof.

<u>Claim Expiration Date</u> has the meaning set forth in Section 10.03(c).

<u>Claim Notice</u> has the meaning set forth in Section 10.06(a).

<u>Closing</u> has the meaning set forth in Section 3.01.

<u>Closing Balance Sheet</u> has the meaning set forth <u>in Section 2.06(a)</u>.

<u>Closing Date</u> has the meaning set forth in Section 3.01.

<u>Closing Indebtedness</u> means the Indebtedness of the Brand Companies as of the Closing Date.

A-3

Closing Statement has the meaning set forth in Section 2.06(a).

<u>Code</u> means the U.S. Internal Revenue Code of 1986, as amended from time to time, or any successor statute, and the rules and regulations promulgated thereunder. Any reference herein to a particular provision of the Code shall mean, where appropriate, the corresponding provision in any successor statute.

<u>Collateral Agreements</u> means the documents, instruments, certificates and agreements executed in connection with this Agreement, including without limitation, the Escrow Agreement and the Gaia Agreements.

<u>Colorado Facility</u> has the meaning set forth <u>in Section 7.08(a)</u>.

<u>Company</u> has the meaning set forth in the Recitals.

<u>Company Benefit Plans</u> has the meaning set forth in Section 5.15(a).

<u>Company Intellectual Property</u> means any and all Intellectual Property that is owned by or licensed to the Brand Companies or any of the Intellectual Property that is material to the operation of the Brand Business which is used by or held for use by the Brand Companies.

<u>Company IT Assets</u> shall mean the Software (whether in object or source code form), hardware, databases, and servers, and all other information technology equipment and assets, in each case owned by or licensed to the Brand Companies solely for the purpose of operating the Brand Companies e-commerce platforms.

Company Material Adverse Effect means any change, event, effect, condition or circumstance, or series of related changes, events, effects, conditions, or circumstances, that, individually or when taken together, (a) has or would reasonably be likely to have a materially adverse effect on the business, assets, Liabilities, results of operations or condition (financial or otherwise) of the Brand Companies, taken as a whole; or (b) would reasonably be expected to materially prevent the consummation of the Contemplated Transactions; provided, however, that such change, event, effect, condition or circumstance or series of related changes, events, effects, conditions or circumstances shall not be deemed to constitute a Company Material Adverse Effect to the extent that any change, event, effect, condition or circumstance or series of related changes, events, effects, conditions or circumstances described in clause (a) or (b) occurred after the date hereof is related to or arises from: (i) actions or inactions expressly required or expressly permitted by this Agreement; (ii) a change in general political, economic, or financial market conditions (except if such conditions have a disproportionately adverse effect on the Brand Companies relative to other Persons operating in the industry in which the Brand Companies operate); (iii) a change that affected the industries in which the Brand Companies operate generally (except if such change has a disproportionately adverse effect on the Brand Companies relative to other Persons operating in the industry in which the Brand Companies operate); (iv) the announcement or pendency of this Agreement and the Contemplated Transactions; (v) any changes after the date of this Agreement in GAAP or Applicable Law (except if such changes have a disproportionately adverse effect on the Brand Companies relative to other Persons operating in the industry in which the Brand Companies operate); (vi) natural disaster, sabotage, acts of terrorism or war (whether or not declared) or other outbreak of hostilities, (except if such conditions have a disproportionately adverse effect on the Brand Companies relative to other Persons operating in the industry in which the Brand Companies operate); or (vii) the failure of the Brand Companies to meet their financial projections; provided, however, that this clause (vii) shall not be construed as implying that Seller is making any representation or warranty herein with respect to any financial projections for any period; and provided, further, that the underlying reason for the failure of the Brand Companies to meet their financial projections shall not be excluded pursuant to this clause (vii).

<u>Company Subsidiaries</u> has the meaning set forth in the Recitals.

<u>Company Trademarks</u> has the meaning set forth <u>in Section 7</u>.19.

A-4

Company Transaction Expenses means all Transaction Expenses incurred or otherwise payable by or on behalf of the Brand Companies, including (a) any severance payments owed to employees of the Brand Companies whose employment has been terminated by the Brand Companies in the period between the signing of this Agreement and the Closing Date (including any Taxes relating to such severance payments) unless such termination(s) were requested in writing by Purchaser, (b) any annual bonuses payable to employees of the Brand Companies with respect to the fiscal year ended December 31, 2015 which have not been paid prior to the Closing Date (including any Taxes relating to such annual bonus payments), (c) an amount equal to Seller s Portion of the Representation and Warranty Policy Premium Amount, and (d) any amounts paid pursuant to a voluntary disclosure agreement with any of the Taxing Authorities detailed in Schedule 7.01.

<u>Confidential Information</u> means all information (written or oral) that is confidential or proprietary to the Brand Companies or the Brand Business or is not otherwise generally available to the public regarding the Brand Companies or the Brand Business. The term <u>Confidential Information</u> shall not include information that is or becomes generally available to the public, other than as a result of disclosure by Seller or its Representatives in violation of this Agreement.

<u>Confidentiality Agreement</u> means the Confidentiality Agreement dated as of July 15, 2015, between Gaiam, Inc. and Sequential Brands Group, Inc., as modified, amended and supplemented from time to time.

<u>Contemplated Transactions</u> means Purchaser s acquisition of the Purchased Interests and all other transactions contemplated by this Agreement.

<u>Contract</u> means, with respect to any Person, any legally binding agreement, contract, lease, license, understanding, arrangement, commitment or other instrument or obligation (whether written or oral) to which such Person is a party, and any amendments thereto.

<u>Contributed Assets</u> means the assets contributed (directly or indirectly) to the Company pursuant to the terms of the Contribution Agreement.

<u>Contributed Liabilities</u> means the liabilities contributed (directly or indirectly) to the Company pursuant to the terms of the Contribution Agreement.

<u>Contribution Agreement</u> has the meaning set forth in the Recitals.

Damages of any Person means (a) any and all actual out-of-pocket claims, actions, causes of action, judgments, awards, Liabilities, losses, costs and damages (including the reasonable fees and expenses of outside counsel, accountants and other professional advisors), whether involving a dispute solely between the Parties hereto or otherwise, incurred or suffered by such Person; and (b) any losses or costs incurred by such Person in investigating, defending or settling any claim, action or cause of action described in clause (a), whether or not the underlying claim, action or cause of action is actually asserted or is merely alleged or threatened; *provided, however*, that solely for purposes of Section 10.01, exemplary, punitive, special, incidental, consequential damages (including any claim for lost profits, diminution in value, loss of revenue, or income) speculative, treble, remote, indirect damages, and loss of business reputation or opportunity are excluded from this definition of Damages (for the avoidance of doubt, exemplary, punitive, special, incidental and consequential damages (including any claim for lost profits, diminution in value, loss of revenue, or income) shall not be excluded from the definition of Damages for purposes of calculating Damages arising out of or in connection with a breach the representations and warranties set forth in Article IV and Article V); *provided, further*, that all exemplary, consequential, punitive and special damages actually paid to a third party shall constitute direct Damages notwithstanding the characterization of such damages *vis-à-vis* the third party.

For purposes of computing the amount of Damages incurred by any Person, (i) there shall be deducted an amount equal to the amount of any insurance proceeds (other than proceeds from the Representation and Warranty Policy), indemnification payments, contribution payments, or reimbursements directly or indirectly actually received by such Person or

A-5

any of such Person s Affiliates as compensation specifically for such Damages or the circumstances giving rise thereto (net of any reasonable costs and expenses incurred by such Person in investigating the underlying claim or collecting such proceeds, payments or reimbursements), and (ii) any qualifications in the representations, warranties and covenants with respect to a Company Material Adverse Effect, materiality, material or similar terms shall be disregarded and will not have any effect with respect to the calculation of the amount of any Damages attributable to a breach of any representation, warranty or covenant set forth in this Agreement.

<u>Debt Commitment Letters</u> has the meaning set forth <u>in Section 6.04(a)</u>.

<u>Debt Financing</u> has the meaning set forth in Section 6.04(a).

<u>Designated Employee Accrued PTO Amount</u> means all accrued but unpaid and unused paid time off for all Designated Employees as of the Closing Date.

<u>Designated Employees</u> has the meaning set forth in Section 7.08(b).

<u>Disclosure Schedule</u> has the meaning set forth in the introduction to Article IV.

<u>Disputed Items</u> has the meaning set forth in Section 2.06(b).

Employment Practices has the meaning set forth in Section 5.16.

<u>Entity</u> means any corporation, partnership, limited liability company, professional association, trust or other entity.

<u>Environment</u> means soil, land surface or subsurface strata, surface waters (including navigable waters and ocean waters), groundwaters, drinking water supply, stream sediments, ambient air (including indoor air), plant and animal life and any other environmental medium or natural resource.

Environmental Law means any Applicable Law that requires or relates to: (a) advising appropriate authorities, employees or the public of intended or actual Releases of pollutants or hazardous substances or materials, violations of discharge limits or other prohibitions and the commencement of activities, such as resource extraction or construction, that could have significant impact on the Environment; (b) preventing or reducing to acceptable levels the Release of pollutants or hazardous substances or materials into the Environment; (c) reducing the quantities, preventing the Release or minimizing the hazardous characteristics of wastes that are generated; (d) protecting resources, species or ecological amenities; (e) reducing to acceptable levels the risks inherent in the transportation of any Hazardous Material or other potentially harmful substance; (f) cleaning up a Release of pollutants, preventing the threat of Release or paying the costs of such clean up or prevention; (g) making responsible parties pay private parties, or groups of them, for damages done to their health or the Environment or permitting self-appointed representatives of the public interest to recover for injuries done to public assets; (h) the authority of any Governmental Entity regulating, or creating any liability for, Hazardous Materials, including but not limited to CERCLA, federal Solid Waste Disposal Act, federal Clean Air Act, federal Clean Water Act, federal Toxic Substances Control Act, federal Hazardous Materials Transportation Act, federal Occupational Safety and Health Act, federal Oil Pollution Act, federal Endangered Species Act, federal Emergency Planning and Community Right-to-Know Act, and their state counterparts; and (i) the protection or preservation of public health or the Environment.

<u>Equity Security</u> means (a) any common, preferred, or other capital stock, limited liability company interest, unit or membership interest, partnership interest or similar equity security; (b) any warrants, options, or other rights to, directly or indirectly, acquire any equity security described in clause (a); (c) any other security containing equity

features or profit participation features; (d) any security or instrument convertible or exchangeable directly or indirectly, with or without consideration, into or for any equity security described in

A-6

clauses (a) through (c) above or another similar security (including convertible notes); and (e) any security carrying any warrant or right to subscribe for or purchase any security described in clauses (a) through (d) above or any similar security.

ERISA has the meaning set forth in Section 5.15(a).

Escrow Agent means UMB Bank, N.A.

<u>Escrow Agreement</u> has the meaning set forth in Section 2.03.

<u>Estimated Cash on Hand</u> has the meaning set forth <u>in Section 2.05(b)</u>.

<u>Estimated Closing Indebtedness</u> has the meaning set forth <u>in Section 2.05(b)</u>.

<u>Estimated Closing Statement</u> has the meaning set forth <u>in Section 2.05(b)</u>.

<u>Estimated Company Transaction Expenses</u> has the meaning set forth <u>in Section 2.05(b)</u>.

<u>Fee Letter</u> has the meaning set forth in Section 6.04(a).

<u>FFL</u> has the meaning set forth in the Recitals.

<u>FFL Acquired Assets and Assumed Liabilities</u> has the meaning set forth in the Recitals.

FFL Closing Date means the Closing Date under the Gaiam-FFL APA.

<u>Final Net Closing Cash Consideration</u> means an amount equal to (a) the Purchase Price, minus (b) the Adjustment Escrow Amount, minus (c) the Closing Indebtedness (as finally determined pursuant to <u>Section 2.06</u>), plus (d) the Cash on Hand (as finally determined pursuant to <u>Section 2.06</u>), minus (e) the Company Transaction Expenses (as finally determined pursuant to <u>Section 2.06</u>).

<u>Financial Statements</u> has the meaning set forth in Section 5.05.

<u>Fundamental Representations</u> means the representations and warranties set forth in Section 4.01 (Organization), Section 4.02(a) and Section 4.02(b) (*Authorization; Execution and Delivery*), Section 4.04 (*Title to Purchased Interests*), Section 4.05 (*Judgments*), Section 4.07 (*Brokers*), Sections 5.02(a), 5.02(b), 5.02(c), and 5.02(d) (*Capitalization*), Section 5.03(a) (*Authority*), Section 5.10(b) (*Title to Material Intellectual Property*), Section 5.14(b) (*Judgments*), Section 5.20 (*Brokers*), Sections 6.01(a) and 6.01(b) (*Authority; Execution and Delivery*), Section 6.03 (*Solvency*), Section 6.05 (*Judgments*) and Section 6.07 (*Brokers*).

<u>Funded Indebtedness</u> shall mean all Indebtedness listed <u>on Schedule 1.01(a)</u>.

<u>GAAP</u> means United States generally accepted accounting principles in effect as of the Closing Date applied on a consistent basis with the past practice of the Brand Companies.

<u>Gaia Agreements</u> means each of the Contracts set forth on Schedule 1.01(c).

<u>Gaia Business</u> has the meaning set forth in the Recitals.

<u>Gaiam-FFL APA</u> has the meaning set forth in the Recitals.

<u>Gaiam Americas</u> means Gaiam Americas, Inc., a Colorado corporation.

A-7

Government Person has the meaning set forth in Section 5.19(a).

<u>Governmental Entity</u> means any federal, state, local or foreign government or any court of competent jurisdiction, administrative or regulatory body, agency, bureau or commission in any domestic or foreign jurisdiction, and any appropriate division of any of the foregoing.

<u>Guaranteed Obligations</u> has the meaning set forth in Section 11.21(a).

Guaranty has the meaning set forth in Section 11.21(a).

<u>Hazardous Material</u> means: (a) any petroleum products, petroleum by-products or breakdown products, waste oil, crude oil, asbestos, pesticides, urea formaldehyde or polychlorinated biphenyl; (b) any waste, gas or other substance or material that is explosive or radioactive; and (c) any hazardous substance, pollutant, contaminant, waste, hazardous, regulated substance, hazardous chemical or toxic chemical as designated, listed, defined (whether expressly of by reference), or regulated in any statute, regulation or other Environmental Law.

<u>HSR Act</u> means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended from time to time, and the rules and regulations promulgated thereunder.

<u>Indebtedness</u> of any Person means, without duplication, (a) all outstanding indebtedness of such Person for borrowed money or indebtedness issued or incurred in substitution or exchange for indebtedness for borrowed money, (b) amounts owing as deferred purchase price of property or services, including all seller notes and earn-out payments, whether or not matured (other than (i) trade payables incurred in the ordinary course of business of such Person and (ii) employee compensation incurred in the ordinary course of business), (c) any liability of such Person in respect of banker s acceptances or letters of credit issued for the account of such Person (to the extent drawn), (d) indebtedness evidenced by any note, bond, debenture, mortgage or other debt instrument, debt security or other similar instrument, (e) indebtedness secured by a Lien on assets or properties of such Person, (f) obligations under any interest rate swap agreement, (g) all obligations of such Person as lessee under leases that have been or should be, in accordance with GAAP, recorded as capital leases, (h) obligations or commitments to repay deposits or other amounts advanced by and owing to third parties, (i) all obligations of such Person to purchase, redeem, retire, defease or otherwise acquire for value any capital stock of such Person or any warrants, rights or options to acquire such capital stock, valued, in the case of redeemable preferred stock, at the greater of its voluntary or involuntary liquidation preference plus accrued and unpaid dividends, (j) direct or indirect guarantees of any indebtedness of any other Person of a type described in clauses (a) through (i) above, and (k) with respect to any indebtedness of a type described in clauses (a) through (i) above, all accrued and unpaid interest, premiums, penalties, breakage costs, unwind costs, fees, termination costs, redemption costs, expenses and other charges with respect thereto.

<u>Indemnified Party</u> has the meaning set forth in Section 10.06(a).

<u>Indemnifying Party</u> has the meaning set forth <u>in Section 10.06(a)</u>.

<u>Information Statement</u> means, collectively, the information statement to be provided to the shareholders of Seller in connection with the Shareholder Consent, including the sale of the Purchased Interests and the amendment to the Articles of Incorporation of Seller as contemplated by <u>Section 7.19</u>, or any of the amendments or supplements thereto.

<u>Intellectual Property</u> means any and all of the following, in any jurisdiction throughout the world: (a) all patents and patent applications, together with all reissuances, continuations, continuations-in-part, divisionals, provisionals, non-provisionals, revisions, extensions and re-examinations thereof and foreign equivalents thereof; (b) all trademarks

and service marks (including both word marks and design marks), trade

A-8

dress, and other indicia of origin, whether registered or unregistered, together with all goodwill associated therewith, and all applications for registration and registrations in connection therewith and all renewals thereof (collectively, Trademarks); (c) all copyrights, whether registered or unregistered, and all applications for registration and registrations in connection therewith and all renewals thereof, and all works of authorship, moral rights and similar rights of authors; (d) all trade secrets recognized under Applicable Law; (e) Internet domain names (including associated registrations); (f) rights of publicity (including all rights in a Person s name, voice, signature, biography, likeness, image and persona); and (g) rights in Software and inventions (whether or not patentable), know-how, concepts, processes, methods, techniques, technology, logos, brand names, slogans, trade names, websites, advertising material, displays, designs, design archives, patterns, prototypes, prints, samples and other works of authorship, in each case as they arise under one or more of the categories of intellectual property referenced in clauses (a) through (f) above; and (h) all other intellectual property and industrial property rights recognized under Applicable Law.

<u>Interim Balance She</u>et has the meaning set forth <u>in Section 5</u>.05.

<u>Interim Balance Sheet Date</u> means December 31, 2015.

<u>IP License Agreement</u> means that certain Intellectual Property License and Coexistence Agreement in substantially the form attached hereto as <u>Exhibit C</u>.

<u>IP Title Corrective Actions</u> has the meanings set forth <u>in Section 7.18</u>.

<u>Judgment</u> means any order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, writ or award issued, made, entered, rendered or otherwise put into effect by or under the authority of any Governmental Entity or any arbitrator.

Knowledge of Purchaser and other phrases of like substance mean the knowledge of the individuals comprising the Purchaser Knowledge Group, after reasonable investigation and inquiry on the part of such individuals

<u>Knowledge of Seller</u> and other phrases of like substance mean the knowledge of the individuals comprising the Seller Knowledge Group, after reasonable investigation and inquiry on the part of such individuals.

<u>Leased Property</u> has the meaning set forth in Section 5.09.

<u>Liabilities</u> shall mean any and all Indebtedness, liabilities and obligations of any kind, whether accrued or fixed, known or unknown, absolute or contingent, or matured or unmatured.

<u>License Agreements</u> has the meaning set forth in Section 5.10(e).

<u>Licensee</u> shall mean each third-party licensee of Purchaser or, after the Closing, of any Brand Company, in each case, designated by Purchaser. As of the date hereof, Purchaser has designated FFL and Highlife Apparel LLC.

<u>Licensee 401(k) Plan</u> means the 401(k) Plan maintained by the applicable Licensee in effect as of the Closing Date in which the Transferred Employees will participate pursuant to <u>Section 7.08(i)</u> of this Agreement.

<u>Liens</u> has the meaning set forth in Section 4.04.

<u>Material Contracts</u> has the meaning set forth <u>in Section 5.11</u>(a).

A-9

<u>Material Registered Company Intellectual Property</u> means the registered U.S. Trademarks set forth <u>on Schedule 1.01(e)</u>.

Money Laundering Laws has the meaning set forth in Section 5.19(b).

<u>Net Closing Cash Consideration</u> means an amount equal to (a) the Purchase Price, minus (b) the Adjustment Escrow Amount, minus (c) the Estimated Closing Indebtedness, plus (d) the Estimated Cash on Hand, minus (e) the Estimated Company Transaction Expenses.

Net Deficit Amount has the meaning set forth in Section 2.05(c)(ii).

Net Overage Amount has the meaning set forth in Section 2.05(c)(i).

New Debt Commitment Letter has the meaning set forth in Section 7.14(e).

Notice Date has the meaning set forth in Section 7.08(a).

Objection Notice has the meaning set forth in Section 2.06(b).

Occurrence shall mean any accident, happening or event that is caused or allegedly caused by any hazard or defect in manufacture, design, materials or workmanship including any failure or alleged failure to warn or any breach or alleged breach of express or implied warranties or representations with respect to a product manufactured, shipped, sold or delivered by or on behalf of any of the Brand Companies which results or is alleged to have resulted in injury or death to any Person or damage to or destruction of property (including damage to or destruction of the product itself).

Organizational Documents means, with respect to any particular Entity, (a) if a corporation, the articles or certificate of incorporation and the bylaws; (b) if a general partnership, the partnership agreement and any statement of partnership; (c) if a limited partnership, the limited partnership agreement and the articles or certificate of limited partnership; (d) if a limited liability company, the articles of organization or certificate of formation and operating agreement, limited liability company agreement, or company agreement; (e) if another type of Entity, any other charter or similar document adopted or filed in connection with the creation, formation or organization of the Entity; and (f) any amendment or supplement to any of the foregoing.

Outside Date has the meaning set forth in Section 9.01(b).

<u>Party</u> or <u>Parties</u> means Purchaser, Parent or Seller.

<u>Pay-Off Letters</u> has the meaning set forth in Section 8.02(d)(xii).

Payroll Provider means PayCom Payroll, LLC.

<u>Permit</u> means any permit, license, certificate, clearance, certification, registration, qualification, exemption or authorization issued or granted by any Governmental Entity or pursuant to any Applicable Law.

<u>Permitted Liens</u> means (a) Liens imposed by Applicable Law related to the sale, transfer, pledge or other disposition of securities, (b) Liens for Taxes not yet due and payable or for Taxes being contested in good faith through appropriate proceedings and for which appropriate reserves have been established in accordance with GAAP,

(c) purchase money Liens and Liens securing rental payments incurred in the ordinary course of business, (d) Real Estate Liens, (e) Liens set forth on <u>Section 5.08 of the Disclosure Schedule</u>, and (f) other Liens incurred in ordinary course of business and not incurred in connection with the borrowing of money, if any,

A-10

the date hereof.

Table of Contents

which do not, individually or in the aggregate, materially impair the continued use and operation of any real property or tangible personal property of the Brand Companies to which they relate as currently used or operated. <u>Person</u> means any individual, Entity or Governmental Entity. <u>Post-Closing Period</u> means any taxable year or period that begins after the Closing Date and, in the case of any Straddle Period, the portion of such Straddle Period beginning immediately after the Closing Date. <u>Pre-Closing Period</u> means any taxable year or period that ends on or before the Closing Date and, in the case of any Straddle Period, the portion of such Straddle Period ending on the Closing Date. <u>Proceeding</u> means any formal action, arbitration, claim, hearing, inquiry, investigation, litigation or suit (whether civil, criminal, administrative or judicial, whether public or private, at law or in equity) commenced, brought or conducted by or before any Governmental Entity or arbitrator. <u>Purchase Price</u> has the meaning set forth in Section 2.02. <u>Purchased Interests</u> has the meaning set forth in the Recitals. Purchaser has the meaning set forth in the opening paragraph of this Agreement. <u>Purchaser Knowledge Group</u> means each of the individuals set forth on Schedule 1.01(b). <u>Purchaser Party</u> has the meaning set forth in Section 10.01. Real Estate Liens has the meaning set forth in Section 5.09. Real Property Leases has the meaning set forth in Section 5.09. Reception Services Agreement means that certain Reception Services Agreement in substantially the form attached hereto as Exhibit G. Registered Company Intellectual Property has the meaning set forth in Section 5.10(a). Related Party means, with respect to any Person, (a) any officer, director or shareholder of such Person, and (b) any other Entity controlled, directly or indirectly, by one or more of such officer, director or shareholder. Release means any release, spill, emission, leaking, pumping, pouring, dumping, emptying, injection, deposit, disposal, discharge, dispersal, leaching or migration on or into the Environment or into or out of any property. Representation and Warranty Policy means that certain buyer side representations and warranties insurance policy in

25

Representation and Warranty Policy Insurer means, collectively, the insurance under the Representation and Warranty

favor of Purchaser issued by the Representation and Warranty Policy Insurer that is being conditionally bound as of

Policy in accordance with their respective quota share percentage.

<u>Representative</u> means, with respect to any Person, such Person s officers, directors, managers, executive employees, financial advisors, legal counsel, accountants, consultants, and other representatives and agents.

A-11

<u>SEC</u> means the Securities and Exchange Commission.

<u>Section 338(h)(10) Election</u> has the meaning set forth in Section 7.10(g).

<u>Securities Act</u> has the meaning set forth <u>in Section 6</u>.08.

<u>Seller</u> has the meaning set forth in the opening paragraph of this Agreement.

Seller 401(k) Plan means the Gaiam, Inc. 401(k) Plan.

<u>Seller Knowledge Group</u> means each of the individuals set forth on Schedule 1.01(b).

Seller Party has the meaning set forth in Section 10.02.

Seller s Portion of the Representation and Warranty Policy Premium Amount means \$2,246,357.

<u>Severance Amount</u> means \$1,345,361; *provided, however*, if the Closing occurs after June 10, 2016, the Severance Amount shall be increased by the amount of increased severance that is due and payable to any Designated Employee due to such Designated Employee achieving an employment anniversary with any Brand Company following June 10, 2016.

<u>Shareholder Consent</u> means the written consent of the shareholders of Seller holding the requisite number of votes required to approve this Agreement and the transactions contemplated by this Agreement in accordance with Seller s Organizational Documents and Applicable Law.

<u>Software</u> means any software, programs and databases in any form, including compilers, middleware, development tools, websites (including the content thereon), firmware, operating systems and specifications, platforms, interfaces, APIs, test specifications and scripts, source code and object code.

Straddle Period has the meaning set forth in Section 7.10(a).

<u>Sub-Basket</u> has the meaning set forth in Section 10.04(a).

<u>Subsidiary</u> of any Person means another Person, an amount of the voting securities, other voting ownership or voting partnership interests of which is sufficient to elect at least a majority of its board of directors or other governing body, or, if there are no such voting interests, 50% or more of the Equity Securities of which is owned directly or indirectly by such first Person or by another subsidiary of such first Person.

Superior Proposal means any unsolicited bona fide written Alternative Transaction Proposal not attributable to or arising from a breach of Section 7.21 or to any material breach of any other provision of this Agreement, which the Board determines in good faith (after consultation with its outside legal counsel and an independent financial advisor of nationally recognized reputation), taking into account all legal, financial, regulatory, timing and other aspects of the Alternative Transaction Proposal (including financing and breakup fee provisions) and the Person making such Alternative Transaction Proposal: (a) is reasonably capable of being completed on substantially the same or superior terms and conditions as set forth in this Agreement; (b) to the extent financing is required, is supported by fully committed financing; (c) results in Seller receiving a purchase price of which the portion attributable to the Brand Business is at least five percent (5%) greater than the Purchase Price; and (d) is otherwise on terms that the Board has determined would be, if such Alternative Transaction is consummated, more favorable to the shareholders of Seller

than the Contemplated Transactions; *provided, however*, that for purposes of this definition of Superior Proposal, the term Alternative Transaction Proposal shall have the meaning assigned to such term herein, except that the reference to majority of either the voting power or economic value of the Equity Securities in the definition of Alternative Transaction Proposal

A-12

shall be deemed to be a reference to eighty percent (80%) or more of its total voting power or economic value , and the reference to a (x) all or substantially all of the consolidated assets of Seller, the Company and the Company Subsidiaries (which shall include the Brand Business) or (y) all or substantially all of the assets of the Brand Business, in each case shall be deemed to be a reference to all or substantially all of the consolidated assets of Seller, the Company and the Company Subsidiaries (which shall include the Brand Business) .

Survival Date has the meaning set forth in Section 10.03(a).

<u>SVOD Rights Assignment Agreement</u> means that certain SVOD Rights Assignment Agreement, to be dated as of the Closing Date and attached hereto as <u>Exhibit E</u>.

<u>SVOD Rights Sub-License Agreement</u> means that certain SVOD Rights Sub-License Agreement, to be dated as of the Closing Date and attached hereto as <u>Exhibit F</u>.

<u>Tail Policy</u> has the meaning set forth <u>in Section 7</u>.09.

Tax means (a) all forms of foreign, federal, state or local income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, escheat, windfall profits, environmental, customs duties, estimated, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or added-on minimum tax or other tax of any kind whatsoever imposed, or required to be collected or withheld, including charges, together with any related interest, penalties or other additional amounts, and (b) any liability for items described in clause (a) of any Person under U.S. Treasury Regulations §1.1502-6 (or any similar provisions of state, local or foreign law) or as a transferee or successor, by contract, or otherwise.

<u>Tax Controversy</u> has the meaning set forth in Section 7.10(c).

<u>Tax Return</u> means any return, filing, report, claim, declaration, refund request, information statement or other document required to be filed, including any schedule, attachments or amendments that may be filed, for any taxable period with any Taxing Authority (whether or not a payment is required to be made with respect to such filing).

<u>Taxing Authority</u> means any Governmental Entity exercising any authority to impose, regulate or administer the imposition of Taxes.

<u>Termination Fee</u> has the meaning set forth in Section 9.04.

<u>Third Amendment and Assignment of Lease</u> means that certain Third Amendment to and Assignment and Assumption of Lease in substantially the form attached hereto as <u>Exhibit H</u>.

<u>Third Party Claim</u> has the meaning set forth in Section 10.07(a).

Third Party Claim Notice has the meaning set forth in Section 10.07(a).

<u>Trademarks</u> has the meaning set forth in the definition of Intellectual Property.

<u>Transaction Expenses</u> means, with respect to any Person, all fees, costs and expenses (including all legal and attorneys fees and expenses, all fees and expenses payable to any broker, advisor or finder, and all fees and expenses of any audit firm or accountants) that have been incurred in connection with the Contemplated Transactions (including, the

Shareholder Consent and the Information Statement) and, with respect to the Brand Companies, the Change of Control Payments, if any (including the employer portion of any Taxes relating to such payments).

A-13

<u>Transferred Employee Accrued PTO Amount</u> means all accrued but unpaid and unused paid time off for all Transferred Employees as of the Closing Date

<u>Transferred Employees</u> has the meaning set forth in Section 7.08(b).

<u>Transferred Liabilities</u> means the liabilities expressly retained by Seller pursuant to the terms of the Contribution Agreement.

<u>Transition Services Agreement</u> means that certain Transition Service Agreement in substantially the form attached hereto as <u>Exhibit D</u>.

<u>WARN Act</u> has the meaning set forth in Section 7.08(e).

<u>WARN Amount</u> means the aggregate amount, if any, to be paid to the Designated Employees terminated as of the Closing Date for the balance of the applicable WARN notice period, if such WARN notice period is less than sixty (60) calendar days.

<u>WARN Notice</u> has the meaning set forth in Section 7.08(e).

ARTICLE II

PURCHASE AND SALE

- **2.01 Purchase and Sale of Purchased Interests.** Subject to the terms and conditions of this Agreement, Seller hereby agrees to sell, convey, transfer and assign to Purchaser at Closing, and Purchaser hereby agrees to purchase and acquire from Seller at Closing, all right, title and interest in and to the Purchased Interests, free and clear of all Liens (other than Liens imposed by Applicable Law related to the sale, transfer, pledge or other disposition of securities).
- **2.02 Purchase Price.** Subject to (a) any adjustment made in accordance with Sections 2.05, and 2.06 and (b) any adjustment in accordance with Section 10.09, the purchase price for the Purchased Interests shall be an amount equal to One Hundred Forty-Five Million Seven Hundred Thousand Dollars (\$145,700,000.00) (such amount, the Purchase Price) which shall be payable by Purchaser at Closing in accordance with Section 3.02.
- **2.03 Adjustment Escrow Account.** At the Closing, a portion of the Purchase Price equal to the Adjustment Escrow Amount shall be deposited by Purchaser into an escrow account designated by the Escrow Agent (the account into which the Adjustment Escrow Amount is deposited being referred to herein as the <u>Adjustment Escrow Account</u>), in accordance with <u>Section 3.02(a)(i)</u>, to be held in escrow by the Escrow Agent pursuant to the terms and conditions of the Escrow Agreement in substantially the form attached hereto as <u>Exhibit B</u> (the <u>Escrow Agreement</u>).
- **2.04 Closing Indebtedness.** All of the Funded Indebtedness of the Brand Companies as of the Closing Date, if any, shall be repaid in full at Closing out of the Purchase Price in accordance with <u>Section 3.02(a)(ii)</u>.

2.05 Purchase Price Adjustment.

(a) General. The Purchase Price shall be subject to adjustment at the Closing, and the Net Closing Cash Consideration shall be subject to adjustment following the Closing, in each case, in accordance with this <u>Section 2.05</u> and <u>Section 2.06</u>.

(b) Closing Adjustment. As soon as practicable before the Closing, but in no event later than 5:00 p.m. (Mountain Time) on the date that is two Business Days prior to the Closing Date, Seller shall deliver to

A-14

Purchaser, (i) an estimated consolidated balance sheet of the Brand Companies as of the Closing Date, and (ii) a statement (the <u>Estimated Closing Statement</u>) setting forth in reasonable detail Seller s (A) estimate of the Cash on Hand, if any, (such estimate, the <u>Estimated Cash on Hand</u>), (B) estimate of the Closing Indebtedness, if any (such estimate, the <u>Estimated Closing Indebtedness</u>), (C) estimate of the Company Transaction Expenses that have not been paid as of the Closing (such estimate, the <u>Estimated Company Transaction Expenses</u>), and (D) calculation of the Net Closing Cash Consideration based on the foregoing, in each case prepared by Seller in good faith in accordance with the Brand Companies historical practices and presented in a manner consistent with the illustrative example attached hereto as <u>Schedule 2.05(b)</u>. In preparing the Estimated Closing Statement, Seller shall employ the same judgments, accounting methods, policies, practices, procedures, classifications or estimation methodologies employed by Seller and Purchaser in preparing <u>Schedule 2.05(b)</u>.

- (c) **Post-Closing Adjustment.** After the final calculation of Final Net Closing Cash Consideration (as determined pursuant to <u>Section 2.06</u> below):
- (i) If the Final Net Closing Cash Consideration minus the Net Closing Cash Consideration is greater than \$0 (such difference, the Net Overage Amount), then, within three Business Days following the determination of the Final Net Closing Cash Consideration in accordance with Section 2.06, Purchaser shall deliver to Seller the Net Overage Amount by wire transfer of immediately available funds to the account or accounts designated by Seller, and Seller and Purchaser shall, subject to the final resolution of the purchase price adjustment pursuant to Section 2.05 of the Gaiam-FFL APA, deliver a joint instruction to the Escrow Agent to distribute all funds remaining in the Adjustment Escrow Account to Seller by wire transfer of immediately available funds to the account or accounts designated by Seller; and
- (ii) If the Final Net Closing Cash Consideration minus the Net Closing Cash Consideration is less than \$0 (such difference a Net Deficit Amount), then, within three (3) Business Days following the determination of the Final Net Closing Cash Consideration in accordance with Section 2.06, Seller and Purchaser shall deliver a joint instruction to the Escrow Agent to distribute from the Adjustment Escrow Account, (A) to Purchaser the Net Deficit Amount by wire transfer of immediately available funds to the account or accounts designated by Purchaser, and (B) subject to the final resolution of the purchase price adjustment pursuant to Section 2.05 of the Gaiam-FFL APA, to Seller any funds remaining in the Adjustment Escrow Account by wire transfer of immediately available funds to the account or accounts designated by Seller; provided, however, that in no event shall the amount payable by Seller under this Section 2.05(c)(ii), if any, exceed the Adjustment Escrow Amount.
- (d) Acknowledgement. The Parties agree that the purpose of determining the Closing Indebtedness, the Cash on Hand, the Company Transaction Expenses, and the Final Net Closing Cash Consideration pursuant to Section 2.06 is to correct for any differences between the amounts of the foregoing as set forth on the Estimated Closing Statement and the actual amounts thereof as of the Closing Date, and the adjustment procedures set forth in this Section 2.05 and Section 2.06 are not intended to permit the introduction of new or different judgments, accounting methods, policies, practices, procedures, classifications, or estimation methodologies for the purpose of determining the Final Net Closing Cash Consideration or the components thereof. For the avoidance of doubt, Seller acknowledges and agrees that any payments to be made by Seller pursuant to Section 2.05(c) or Section 2.06(c) shall not constitute an indemnification payment or a claim for indemnification under Article X, and shall not be subject to or affect any limitations set forth therein. Purchaser further acknowledges and agrees that that the calculation of Final Net Closing Cash Consideration shall be made in accordance with this Section 2.05 and Section 2.06, and shall not take into account any breach of Seller s representations, warranties or covenants hereunder which, for the avoidance of doubt, shall be addressed as claims for indemnification under Article X.

2.06 Purchase Price Adjustment Procedures.

(a) **Preparation of Closing Balance Sheet.** As promptly as practicable, but no later than 5:00 p.m. (Mountain Time) on the 75th day after the Closing Date or such later date as Purchaser and Seller agree in

A-15

writing, Purchaser will prepare or cause to be prepared a consolidated balance sheet of the Brand Companies as of the Closing Date (the <u>Closing Balance Sheet</u>), prepared in accordance with the Brand Companies historical practices and presented in a manner consistent with <u>Schedule 2.05(b)</u> to be delivered to Seller, together with a statement (the <u>Closing Statement</u>) setting forth in reasonable detail Purchaser is calculation of (i) the Cash on Hand, (ii) the Closing Indebtedness, (iii) the Company Transaction Expenses that were not paid as of the Closing, in each case, prepared by Purchaser in good faith in accordance with the Brand Companies historical practices and presented in a manner consistent with the proforma example attached hereto as <u>Schedule 2.05(b)</u>, and (iv) the Final Net Closing Cash Consideration based on the foregoing. In the event that Purchaser does not deliver the Closing Balance Sheet and the Closing Statement within such 75-day period, Purchaser shall be conclusively deemed to have accepted the Estimated Cash on Hand, the Estimated Closing Indebtedness and the Estimated Company Transaction Expenses as the Cash on Hand, the Closing Indebtedness, and the Company Transaction Expenses, respectively. Purchaser acknowledges and agrees that, for the purposes of Seller is review of the Closing Balance Sheet delivered by Purchaser pursuant to this <u>Section 2.06(a)</u>, Purchaser shall afford, and shall cause the Company to afford, Seller commercially reasonable access during normal business hours to the books and records (other than privileged documents) of Purchaser, the Brand Companies and their respective Representatives.

(b) Disagreement by Seller. If Seller disagrees with Purchaser's calculation of the Cash on Hand, the Closing Indebtedness and/or the Company Transaction Expenses as set forth on the Closing Statement, Seller may, within 20 days after receipt of the Closing Balance Sheet and the Closing Statement, deliver a notice to Purchaser disagreeing with such calculation(s) and setting forth in reasonable detail Seller's (i) objection(s) to such calculation(s) of the Cash on Hand, the Closing Indebtedness, and/or the Company Transaction Expenses as set forth on the Closing Statement, and (ii) calculation(s) thereof and of the Final Net Closing Cash Consideration (an Objection Notice of Notice shall specify those items or amounts as to which Seller disagrees (the Disputed Items). If Seller fails to deliver an Objection Notice during such 20-day period, the Cash on Hand, the Closing Indebtedness, and the Company Transaction Expenses determined by Purchaser and delivered to Seller in accordance with Section 2.06(a) shall be deemed final, binding, and conclusive for purposes of this Agreement.

(c) Dispute Resolution.

- (i) If an Objection Notice shall have been timely delivered by Seller to Purchaser pursuant to Section 2.06(b), Seller and Purchaser shall, during the 10 days following such delivery, use commercially reasonable efforts to reach agreement on the Disputed Items in order to determine the Cash on Hand, the Closing Indebtedness, and/or the Company Transaction Expenses (in each case, to the extent disputed in the Objection Notice); provided, that the amount of any agreed upon Disputed Item shall fall between (or be equal to either) Purchaser s calculation thereof delivered pursuant to Section 2.06(a) and Seller s calculation thereof delivered pursuant to Section 2.06(b). If, during such 10-day period, Seller and Purchaser agree as to any Disputed Item, then Seller and Purchaser shall execute a written acknowledgement of the amount of such Disputed Item and any Disputed Item so agreed upon shall be deemed final, binding, and conclusive for purposes of this Agreement. If, during such period, Seller and Purchaser are unable to reach agreement on all Disputed Items, then the Disputed Items that remain in dispute shall be resolved by KPMG LLP or, if such firm is conflicted or declines to act in such capacity by the office of such other nationally recognized major firm of independent accountants having no material relationship with any Party and reasonably acceptable to both Seller and Purchaser (the Accounting Referee).
- (ii) Within 15 days of the retention of the Accounting Referee, Purchaser and Seller shall jointly submit the matter to the Accounting Referee and instruct the Accounting Referee that it (A) shall act as an expert in accounting, and not as an arbitrator, to resolve, in accordance with the Brand Companies historical practices and presented in a manner consistent with Schedule 2.05(b), only the Disputed Items that remain in dispute; (B) shall adjust the calculation(s) of the Cash on Hand, the Closing Indebtedness, and/or the Company Transaction Expenses based thereon to reflect such

resolution; (C) shall determine the amount of any Disputed

A-16

Item to be an amount that falls between (or is equal to either) Purchaser's calculation thereof delivered pursuant to Section 2.06(a) and Seller's calculation thereof delivered pursuant to Section 2.06(b); and (D) shall deliver to Purchaser and Seller a written decision as promptly as practicable and in any event within 30 days following the submission of the remaining Disputed Items to the Accounting Referee for resolution. The Accounting Referee shall deliver to Purchaser and Seller, as promptly as practicable, a report setting forth such calculation. The Parties acknowledge and agree that, if any Disputed Item is submitted to the Accounting Referee pursuant to this Section 2.06(c), the Cash on Hand, the Closing Indebtedness, and/or the Company Transaction Expenses, as determined by the Accounting Referee, shall be, absent manifest error, final, binding, and conclusive for purposes of this Agreement.

- (iii) If requested by the Accounting Referee, Purchaser and Seller agree to execute a reasonable engagement letter. The cost of such review and report (including any retainer) shall be borne (A) by Purchaser if Seller s calculation of the Final Net Closing Cash Consideration (based on Seller s position with respect to the Disputed Items submitted to the Accounting Referee) is closer to the Final Net Closing Cash Consideration calculated by the Accounting Referee (based on the Accounting Referee s determination of such Disputed Items), (B) from the Adjustment Escrow Account, if Purchaser s calculation of the Final Net Closing Cash Consideration (based on Purchaser s position with respect to the Disputed Items submitted to the Accounting Referee) is closer to the Final Net Closing Cash Consideration calculated by the Accounting Referee (based on the Accounting Referee s determination of such Disputed Items), and (C) otherwise equally by Purchaser and from the Adjustment Escrow Account. However, initially, any retainer charged by the Accounting Referee shall be paid 50% by Purchaser and 50% from the Adjustment Escrow Account.
- (d) Cooperation. Seller and Purchaser will reasonably cooperate and assist as requested in the preparation of the Closing Balance Sheet and the calculations of the Cash on Hand, the Closing Indebtedness, and the Company Transaction Expenses and in the conduct of the audits and reviews referred to in this <u>Section 2.06</u>, including the making available to the extent necessary of books, records, work papers, and personnel.
- **2.07 Withholding.** Purchaser shall be permitted to withhold from the consideration otherwise payable pursuant to this Agreement such amounts as are required to be withheld with respect to the making of such payment under Applicable Law (e.g., the applicable payroll Taxes for which the applicable Brand Company will be liable), and shall duly remit any amounts withheld to the proper Governmental Entity. Such withheld and remitted amounts shall be treated for all purposes of this Agreement as having been paid to Seller.

ARTICLE III

CLOSING

- **3.01 Closing Date.** Subject to the fulfillment of the conditions precedent specified in <u>Sections 8.01</u>, <u>8.02</u> and <u>8.03</u>, the purchase and sale of the Purchased Interests shall be consummated by the exchange of signatures by facsimile or other electronic transmission or, if such exchange is not practicable, at a closing (the <u>Closing</u>) to be held at the offices of Brownstein Hyatt Farber Schreck, LLP, 410 Seventeenth Street, Suite 220, Denver, Colorado 80202 at 10:00 a.m. (Mountain Time) within three Business Days following the satisfaction or waiver of the conditions precedent set forth in <u>Sections 8.01</u>, <u>8.02</u> and <u>8.03</u>, or on such other date or at such other location as Seller and Purchaser shall mutually agree; *provided, however*, that in no event shall the Closing occur prior to the 45th day following the date hereof (such date on which the Closing occurs, the <u>Closing Date</u>).
- **3.02 Deliveries at Closing.** Subject to the terms and conditions of this Agreement, at the Closing, the following Persons shall deliver or cause to be delivered the following:

- (a) Purchaser shall deliver the Purchase Price as follows:
- (i) Purchaser shall deliver the Adjustment Escrow Amount to the Escrow Agent by wire transfer of immediately available funds for deposit in the Adjustment Escrow Account in accordance with the terms of the Escrow Agreement;

A-17

- (ii) Purchaser shall repay, on behalf of the Brand Companies, the Funded Indebtedness outstanding as of the Closing, by wire transfer to the holder(s) thereof, in accordance with the Pay-Off Letters;
- (iii) Purchaser shall pay, on behalf of the Brand Companies, the Company Transaction Expenses (other than the Change of Control Payments which are addressed below) not paid by Seller prior to the Closing by wire transfer to the payees thereof, in accordance with the invoices delivered pursuant to Section 8.02(d)(xiii);
- (iv) Purchaser shall deliver, on behalf of the Brand Companies, an amount equal to the aggregate amount of Change of Control Payments by wire transfer of immediately available funds directly to the Payroll Provider for payment through a special payroll on the Closing Date (subject to withholding for Taxes, which amounts shall be paid to the Company for disbursement to the applicable taxing authority); and
- (v) Purchaser shall deliver to Seller the Net Closing Cash Consideration by wire transfer of immediately available funds to an account or accounts designated by Seller (such designation to occur no less than two Business Days prior to the Closing).
- **(b)** Purchaser shall deliver, or cause to be delivered, to Seller the deliveries set forth in <u>Section 8.03(c)</u> and <u>Section 8.03(d)</u>.
- (c) Seller shall deliver, or cause to be delivered, to Purchaser the deliveries set forth in <u>Section 8.02(d)</u>.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES RELATING TO SELLER

Seller hereby represents and warrants to Purchaser, except as set forth in the disclosure schedule delivered by Seller to Purchaser on the date hereof (as modified, amended and supplemented, the <u>Disclosure Schedule</u>), as follows:

4.01 Organization. Seller is a corporation duly incorporated and is validly existing and in good standing under the laws of the State of Colorado.

4.02 Authority; Execution and Delivery; Enforceability.

- (a) Seller has the requisite power and authority to execute and deliver this Agreement and the Collateral Agreements to which Seller is or will be a party, to perform its obligations hereunder and thereunder and, subject to obtaining the Shareholder Consent, to consummate the Contemplated Transactions.
- (b) The execution, delivery and performance by Seller of this Agreement and the Collateral Agreements to which Seller is or will be a party, and the consummation by Seller of the Contemplated Transactions, have been duly authorized by all necessary action on the part of Seller, and, subject to obtaining the Shareholder Consent, no other action on the part of Seller is or will be necessary to authorize the execution, delivery and performance of this Agreement and the Collateral Agreements to which Seller is or will be a party and the Contemplated Transactions. This Agreement has been, and each of the Collateral Agreements to which Seller is a party will be, duly executed and delivered by Seller.
- (c) This Agreement constitutes, and each of the Collateral Agreements to which Seller is a party will constitute, the legal, valid and binding obligations of Seller, enforceable against Seller in accordance with their terms, except to the extent that their enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or

other similar laws affecting the enforcement of creditors rights generally and to general equitable principles.

A-18

(d) The Board, by resolutions duly adopted by vote at a meeting of the Board duly called and held at which a quorum was present, has (i) determined that this Agreement and the Contemplated Transactions are advisable and in the best interests of Seller and its shareholders, (ii) authorized, adopted and approved the execution, delivery and performance of this Agreement and the Contemplated Transactions, (iii) approved and declared advisable this Agreement and the Contemplated Transactions, and (iv) resolved to recommend that the shareholders of Seller adopt and approve the consummation of the Contemplated Transactions, and such resolutions have not been rescinded, withdrawn or modified in any way as of the date hereof.

4.03 Noncontravention.

- (a) Neither the execution and delivery of any of this Agreement or the Collateral Agreements to which Seller is or will be a party, nor the consummation or performance of any of the Contemplated Transactions, will, subject to obtaining the Shareholder Consent, (i) in any material respect contravene, conflict with or result in a violation of any Applicable Law or any Judgment to which Seller is subject or by which any of its properties or assets are bound, assuming that all filings and notifications in accordance with the HSR Act have been made and the applicable waiting period has expired or been terminated; (ii) in any material respect contravene, conflict with or result in a breach of, or constitute a default under, or result in the acceleration of any obligation or loss of any benefits under, the provisions of any material Contract to which Seller is subject or by which any of its properties or assets are bound; (iii) result in the loss of use or diminishment in any material respect of any rights of any of the Brand Companies to use any Material Registered Company Intellectual Property; or (iv) contravene, conflict with or result in a violation of any of the provisions of the Organizational Documents of Seller.
- (b) Neither Seller nor any of its Affiliates owns an interest in any Person which reasonably could be expected to cause a Governmental Entity to seek to prevent or materially impair or delay the consummation of the Contemplated Transactions pursuant to the HSR Act.
- **4.04 Title to Purchased Interests.** Seller is the record and beneficial owner of the Purchased Interests set forth on Section 4.04 of the Disclosure Schedule, and holds good and valid title to such Purchased Interests free and clear of any and all mortgages, security interests, charges, easements, rights, options, claims, restrictions, encumbrances, encroachments, indentures, deeds of trust, title irregularities, licenses or leases to third parties, or other liens or limitations of any kind (collectively, <u>Liens</u>), other than Permitted Liens. As of the Closing Date, Seller has the sole power and authority to sell, transfer, assign and deliver the Purchased Interests as provided in this Agreement, and such delivery will convey to Purchaser good and valid title to the Purchased Interests, free and clear of any and all Liens (other than Liens imposed by Applicable Law related to the sale, transfer, pledge or other disposition of securities). Seller is not a party to any voting trust or other voting agreement with respect to any of the Purchased Interests or to any agreement relating to the issuance, sale, redemption transfer or other disposition of the Purchased Interests.
- **4.05 Judgments.** There is no outstanding Judgment to which Seller is a party or is subject that prohibits or impairs the consummation of the Contemplated Transactions by Seller.
- **4.06 Solvency.** As of the Closing and immediately after consummating the Contemplated Transactions, Seller will not (a) be insolvent (either because its financial condition is such that the sum of its debts is greater than the fair value of its assets or because the present fair salable value of its assets will be less than the amount required to pay its probable liability on its debts as they become absolute and matured), (b) have unreasonably small capital with which to engage in its business, or (c) have incurred debts beyond its ability to repay such debts as they become absolute and matured.

4.07 Brokers. Other than the Banker, Seller has not retained any Person to act as a broker or agreed or become obligated to pay, or has taken any action that might result in any Person claiming to be entitled to receive, any brokerage commission, finder s fee or similar commission or fee in connection with any of the Contemplated Transactions.

A-19

ARTICLE V

REPRESENTATIONS AND WARRANTIES RELATING TO THE BRAND COMPANIES

Seller hereby represents and warrants to Purchaser, except as set forth in the Disclosure Schedule, as follows:

5.01 Organization and Standing. Each Brand Company is an Entity duly formed or organized and validly existing and in good standing under the laws of the jurisdiction of its formation or organization, as set forth on Section 5.01 of the Disclosure Schedule. Each Brand Company has the requisite power and authority to own, lease or otherwise hold its properties and assets and to carry on its business as presently conducted, except where the failure to have such power and authority has not had and would not reasonably be expected to have a Company Material Adverse Effect. Except as would not reasonably be expected to have a Company Material Adverse Effect, each Brand Company is duly qualified and in good standing to do business as a foreign Entity in each jurisdiction in which such qualification to carry on its business as presently conducted, and to hold its properties and assets, is necessary. Seller has made available to Purchaser accurate and complete copies of the Organizational Documents of each Brand Company.

5.02 Capitalization.

- (a) Section 5.02(a) of the Disclosure Schedule accurately and completely sets forth for each Brand Company (i) each class and series of Equity Securities, (ii) the aggregate number of shares of each such class and series of Equity Securities that are authorized for issuance, (iii) the aggregate number of shares of each such class and series of Equity Securities that are outstanding, and (iv) a list of the names of each record and beneficial owner of such Equity Securities, and opposite the name of each such owner, the number, class, and series of Equity Securities owned by each such owner. Except for the outstanding Equity Securities set forth on Section 5.02(a) of the Disclosure Schedule, there are no Equity Securities of any of the Brand Companies that are issued, reserved for issuance or outstanding. There are no accrued but unpaid dividends in respect of any Equity Securities of any of the Brand Companies. No shares of capital stock are held in treasury by the Brand Companies. All outstanding shares of capital stock of the Brand Companies are duly authorized and at the time of issuance were validly issued, fully paid and nonassessable and are not subject to, issued or held in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under Applicable Law or the Organizational Documents of the Brand Companies.
- (b) None of the Brand Companies are a party to any outstanding or authorized option, warrant, right (including any preemptive right), subscription, claim of any character, agreement, obligation, convertible or exchangeable securities, or other commitments contingent or otherwise, relating to the Equity Securities of such Brand Companies, pursuant to which any of the Brand Companies are or may become obligated to issue, deliver or sell or cause to be issued, delivered or sold, Equity Securities of the Brand Companies. There is no outstanding or authorized stock appreciation, phantom stock, profit participation or similar rights with respect to the Equity Securities of any of the Brand Companies.
- (c) The Company has good and valid title to all of the outstanding Equity Securities of the Company Subsidiaries, in each case free and clear of all Liens (other than Liens imposed by Applicable Law related to the sale, transfer, pledge or other disposition of securities).
- (d) None of the Brand Companies owns, directly or indirectly, any Equity Securities in any Entity, other than the Company Subsidiaries set forth on Section 5.02(a) of the Disclosure Schedule.

(e) There are no restrictions of any kind that prevent the payment of dividends or other distributions by any of the Brand Companies, other than those imposed by Applicable Law.

A-20

5.03 Authority; Execution and Delivery; Enforceability.

- (a) Each Brand Company has the requisite power and authority to execute and deliver the Collateral Agreements to which it is or will be a party, to perform its obligations hereunder and thereunder and to consummate the Contemplated Transactions.
- (b) The execution, delivery and performance by each Brand Company of the Collateral Agreements to which it is or will be a party and the consummation by such Brand Company of the Contemplated Transactions, have been duly authorized by all necessary corporate action on the part of such Brand Company, and no other action on the part of such Brand Company is or will be necessary to authorize the execution, delivery and performance of the Collateral Agreements to which it is or will be a party and the Contemplated Transactions.
- (c) The Collateral Agreements to which each Brand Company is or will be a party will be duly executed and delivered by such Brand Company. The Collateral Agreements to which each Brand Company is or will be a party will constitute the legal, valid and binding obligation of such Brand Company, enforceable against it in accordance with their terms, except to the extent that their enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors rights generally and to general equitable principles.
- **5.04 Noncontravention.** Neither the execution and delivery of this Agreement, nor the consummation or performance of any of the Contemplated Transactions, will (a) in any material respect contravene, conflict with or result in a violation of any Applicable Law or any Judgment to which any Brand Company is subject or by which any of its properties or assets are bound, assuming that all filings and notifications in accordance with the HSR Act have been made and the applicable waiting period has expired or been terminated; (b) contravene, conflict with or result in a breach of, or constitute a default under, or result in the acceleration of any obligation or loss of any benefits under, the provisions of any Material Contract or any material Permit of any Brand Company; (c) result in the loss of use or diminishment of any rights of any of the Brand Companies to use any Material Registered Company Intellectual Property; (d) contravene, conflict with or result in a violation of any of the provisions of any of the Organizational Documents of each Brand Company; or (e) result in the creation of any Lien upon or with respect to any of the properties or assets of the Brand Companies.
- **5.05 Financial Statements.** Attached to Section 5.05 of the Disclosure Schedule are accurate and complete copies of the following pro forma financial statements (collectively, the <u>Financial Statements</u>): (i) the unaudited consolidated balance sheet of the Brand Companies for the fiscal year ended December 31, 2014, and related unaudited consolidated statement of operations of the Brand Companies for the fiscal year then ended, and (ii) the unaudited consolidated balance sheet of the Brand Companies for the 12 months ended December 31, 2015 (the <u>Interim Balance</u> Sheet), and the related unaudited consolidated statement of operations of the Brand Companies for such period. The Financial Statements have been derived from the consolidated books and records of Seller which are prepared in accordance with GAAP applied on a consistent basis. The Financial Statements have been prepared in accordance with the Brand Companies historical practices and fairly present, in all material respects, the financial position, and results of operations of the Brand Companies as of the respective dates thereof and for the respective periods indicated therein, in each case in the aggregate, subject to the pro forma adjustments and the elimination of inter-company accounts. The Financial Statements include an allocation or invoicing of certain expenses to Seller and certain of its Subsidiaries, including, among others, general corporate expenses related to finance, legal, information technology, human resources, compliance, shared services, insurance, employee benefits and incentives, and stock-based compensation. Purchaser acknowledges that these expenses have been allocated or invoiced to Seller or certain of its Subsidiaries on the basis of direct usage when identifiable, with the remainder allocated on a basis of operating income, headcount, or square footage. Purchaser further acknowledges that actual costs that would have been incurred

if the Brand Companies had been a stand-alone company would depend on multiple factors, including organizational structure and strategic decisions made in various areas, including information technology and infrastructure and that Seller is not making any representations with respect to these actual costs or their allocations.

A-21

5.06 No Undisclosed Liabilities.

- (a) No Brand Company has any claims or Liabilities of any nature, other than Liabilities (i) specifically identified on the Interim Balance Sheet, (ii) incurred by such Brand Company in the ordinary course of business consistent with past practice since the Interim Balance Sheet Date, (iii) relating to the performance under Material Contracts that have not yet been fully performed and under which such Brand Company is not in breach or default (without giving effect to any grace period or requirement with respect to the giving of notice contained therein), (iv) incurred by any Brand Company in connection with the Contemplated Transactions and satisfied on or prior to the Closing Date, (v) as set forth in Section 5.06 of the Disclosure Schedule, or (vi) less than \$50,000 individually and \$250,000 in the aggregate. None of the Brand Companies is a party to, or has any commitment to become a party to, any off balance sheet arrangement, including any off balance sheet arrangement (as defined in Item 303(a) of Regulation S K promulgated by the Securities and Exchange Commission).
- (b) There are no pending claims and, to the Knowledge of Seller, no facts that would reasonably entitle any director or officer of Seller or the Brand Companies to indemnification by any of the Brand Companies under Applicable Law, the Organizational Documents of any Brand Company, any Brand Insurance Policies, or any indemnity agreements of any Brand Company or similar agreements to which any Brand Company is a party or by which any of its properties or assets is or may be bound.
- **5.07 Absence of Changes or Events.** Since the Interim Balance Sheet Date, each Brand Company has (and Seller, in so far as it operated any aspect of the Brand Business or owned any asset constituting a Contributed Asset prior to the date hereof, has) (i) conducted its business in the ordinary course consistent with past practice, and there has not occurred any change, effect, event, condition or circumstance that, individually or in the aggregate, has resulted in or would reasonably be expected to result in a Company Material Adverse Effect, (ii) used its commercially reasonable efforts to preserve intact its business organization and goodwill and relationships with customers, suppliers, licensors, licensees, distributors, Governmental Entities, and other third parties and to keep available the services of its current officers and employees, and (iii) used its commercially reasonable efforts to protect its Intellectual Property, and to avoid infringing, violating or misappropriating any Intellectual Property of any third party (assuming that the practices, customs, actions and omissions in effect on or taken prior to the Interim Balance Sheet Date were sufficient to avoid the same). Since the Interim Balance Sheet Date, except in the ordinary course of business consistent with past practice or as would not be material and adverse to the Brand Companies:
- (a) none of the Brand Companies has sold, leased, transferred, or assigned any of its assets, tangible or intangible, to a third party for more than \$100,000 or caused or permitted to exist any Lien (other than Permitted Liens) on any of its assets, including Company Intellectual Property;
- (b) none of the Brand Companies has acquired by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof;
- (c) none of the Brand Companies has entered into any Contract outside of the ordinary course of business requiring aggregate expenditures of more than \$50,000;
- (d) none of the Brand Companies has made or committed to make any capital expenditure in excess of \$50,000, individually, or \$150,000, in the aggregate, outside of the ordinary course of business;
- (e) none of the Brand Companies has issued, created, incurred, assumed, or guaranteed any Indebtedness involving more than \$50,000:

(f) none of the Brand Companies has changed any of its payment policies with landlords, vendors, suppliers or other creditors or collection policies with respect to customers;

(g) none of the Brand Companies has amended its Organizational Documents;

A-22

- (h) none of the Brand Companies has issued, sold, or otherwise disposed of any of its capital stock, or granted any options, warrants, or other rights to purchase or obtain (including upon conversion, exchange, or exercise) any of its capital stock other than in connection with the Contribution;
- (i) other than in accordance with the Brand Companies Organizational Documents, none of the Brand Companies has declared, set aside, or paid any dividend or made any distribution with respect to its capital stock (whether in cash or in kind) or redeemed, purchased, or otherwise acquired any of its capital stock;
- (j) none of the Brand Companies has experienced any damage, destruction, or loss (whether or not covered by insurance) to its property in excess of \$250,000;
- (k) none of the Brand Companies has granted any increase in the base compensation or wages of any of its directors or officers or employees outside of the ordinary course of business;
- (l) none of the Brand Companies has (1) made, revoked or amended any Tax election, changed any accounting method for Tax purposes, filed any Tax Return outside the ordinary course of business or amended any Tax Return, (2) settled any claim for Taxes or entered into any Tax sharing or similar, contract, or (3) surrendered any right to claim a refund of Taxes;
- (m) none of the Brand Companies has settled or compromised any dispute, claim or assessment regarding a Tax liability in excess of \$50,000;
- (n) none of the Brand Companies has discharged or settled any Proceeding or Liability, other than respect to any of the matters set forth in <u>Section 5.14 of the Disclosure Schedule</u>;
- (o) none of the Brand Companies has made any change in its accounting methods or principles except as required by or in a manner consistent with GAAP and Applicable Laws;
- (p) none of the Brand Companies has sold, transferred, leased, abandoned, cancelled, licensed or otherwise disposed of any Company Intellectual Property, other than (i) non-exclusive licenses to customers granted in the ordinary course of business consistent with past practice, (ii) to other Brand Companies, and (iii) pursuant to the Contribution Agreement or the Gaia Agreements; and
- (q) other than as set forth above, none of the Brand Companies has entered into or agreed to enter into any Contract relating to any of the foregoing.
- **5.08** Certain Assets. Other than with respect to real property or interests in real property, such matters being the subject of Section 5.09, and Intellectual Property, such matters being the subject of Section 5.10, each Brand Company has good and marketable title to, or in the case of leased assets, a valid and enforceable leasehold interests in all tangible and intangible personal property and assets reflected in the Financial Statements or thereafter acquired (except for properties and assets disposed of in the ordinary course of business, consistent with past practice). Except for Permitted Liens (including those set forth in Section 5.08 of the Disclosure Schedule), all of such owned assets are owned, and such leased assets are leased, free and clear of all Liens. Except as set forth in the Collateral Agreements, none of Seller or any of its Affiliates (other than the Brand Companies) possesses, or has a right to use, any tangible or intangible personal properties and assets owned or leased by the Brand Companies. All of the tangible personal property used in the Brand Business is in good operating condition (ordinary wear and tear excepted) and is adequate and suitable for the purpose for which it is presently being used.

5.09 Real Property. Section 5.09 of the Disclosure Schedule sets forth an accurate and complete list of all real property in which any Brand Company has a leasehold interest (each, a <u>Leased Property</u>), the address of the Leased Property and the amount of the lease payment for the each period with respect to which such amount

A-23

is due. Seller has made available to Purchaser true, complete and correct copies of all real property leases relating to the Leased Property (collectively, the <u>Real Property Leases</u>), including all amendments thereto. No Brand Company holds fee title to any real property. Each Brand Company has good and valid title to the leasehold estates in all Leased Property, in each case free and clear of all Liens, except for: (a) the following Liens (collectively, <u>Real Estate Liens</u>): (i) easements, covenants, rights-of-way and other similar restrictions of record, (ii) zoning, building and other similar restrictions, (iii) Liens that have been placed by any developer, landlord or other third party on property over which any Brand Company has easement rights or on any Leased Property and subordination or similar agreements relating thereto and (iv) unrecorded easements, covenants, rights-of-way and other similar restrictions; or (b) any Permitted Lien. None of the Brand Companies is in default under any Real Property Lease, and, to the Knowledge of Seller, no other party to any such Real Property Lease is in default thereunder, with or without the giving of notice, the lapse of time or both. Since January 1, 2014, no party to any Real Property Lease has exercised any option or right to (1) terminate such Real Property Lease prior to the end of the term thereof, (2) lease additional premises pursuant to such Real Property Lease, (3) reduce or relocate the premises demised by such Real Property Lease or (4) purchase any real property pursuant to such Real Property Lease.

5.10 Intellectual Property.

- (a) All Intellectual Property owned by the Brand Companies that is the subject of a patent or registration for protection under Applicable Law, or a pending application for such a patent or registration, is set forth in Section 5.10(a) of the Disclosure Schedule (Registered Company Intellectual Property). To the extent indicated as Registered on Section 5.10(a) of the Disclosure Schedule, the Intellectual Property contained therein has been duly registered in, filed in or issued by the United States Patent and Trademark Office, the United States Copyright Office, a duly accredited and appropriate domain name registrar, or the appropriate Governmental Entity of other jurisdictions (foreign and domestic). To the extent indicated as Registered on Section 5.10(a) of the Disclosure Schedule, all Registered Company Intellectual Property is valid, subsisting and enforceable. There are no actions that must be taken or payments that must be made by the Brand Companies within ninety (90) days following the date hereof that, if not taken, will materially and adversely affect any of the Registered Company Intellectual Property.
- (b) The Brand Companies own, free and clear of all Liens (other than (i) the Permitted Liens, and (ii) non-exclusive licenses granted in the ordinary course of business consistent with past practice), all Material Registered Company Intellectual Property.
- (c) The Brand Companies own, free and clear of all Liens, all Registered Company Intellectual Property. No Related Party or Affiliate of the Company which is not also a Brand Company (i) owns any Company Intellectual Property, or (ii) has any right to use any Company Intellectual Property; *provided* that, the foregoing subsection (ii) shall not apply to the rights to use any Company Intellectual Property that are granted to Gaia, Inc. pursuant to the Gaia Agreements.
- (d) No Company Intellectual Property has been found to be invalid or unenforceable under Applicable Law. Neither the Company Intellectual Property (including any use thereof) nor the conduct of the Brand Business (including any of its products and services) infringes upon or misappropriates the valid Intellectual Property of any Person. The Company Intellectual Property is not subject to any pending Proceeding by any Person which Proceeding is currently ongoing, and no Proceeding has been threatened in writing in the last three (3) years. No Person is infringing or misappropriating any Company Intellectual Property in any material respect. Without limiting the foregoing, as of the date hereof, neither Seller nor any Brand Companies has received in the past three years any notice or claim (including threats or offers to license and cease and desist communications) from any Person challenging the right of any of the Brand Companies, or any of their respective manufacturers, contractors or licensees to use any of the Company Intellectual Property.

(e) <u>Section 5.10(e)</u> of the <u>Disclosure Schedule</u> sets forth an accurate and complete list of all Contracts under which any Brand Company has (i) acquired or obtained, or has been licensed or otherwise granted, any

A-24

license, permission or other right to utilize any Intellectual Property that is owned by a Person other than Seller or the Brand Companies pursuant to which any Brand Company paid, or is obligated to pay, in excess of \$100,000 annually (collectively, the <u>License Agreements</u>); or (ii) licensed or otherwise granted any Person any license, permission or other right to utilize any Material Registered Company Intellectual Property other than non-exclusive licenses granted to customers and digital partners in the ordinary course of business consistent with past practice. Each Contract set forth on <u>Section 5.10(e)</u> of the <u>Disclosure Schedule</u> is binding against the applicable Brand Company and, to the Knowledge of Seller, against the other party thereto.

- **(f)** The Brand Companies own or possess a license to use all Intellectual Property that is necessary and sufficient for the operation of the Brand Business as currently conducted and will continue to do so immediately following the Closing.
- (g) With respect to any and all services rendered by (i) the employees of Seller or any of the Brand Companies within the scope of their employment or (ii) all independent contractors engaged by Seller or any of the Brand Companies prior to the date hereof within the scope of their engagement, Seller and the Brand Companies are the sole and exclusive owners of all of the results and proceeds of such services (including all developments, contributions and creations relating to Company Intellectual Property, and any materials embodying, disclosing or constituting the same), by operation of law or by valid assignment; and insofar as those employees and independent contractors are concerned, Seller and the Brand Companies have complete, unencumbered and unrestricted ownership of and rights to use and exploit all such results and proceeds, throughout the world and in perpetuity.
- (h) To the extent that any of the Brand Companies has sold, transferred, leased, licensed or otherwise disposed of any Company Intellectual Property to Gaia, Inc. pursuant to the Affiliate Agreements or the Gaia Agreements, except as set forth in the Gaia Agreements, such Company Intellectual Property (i) is not used, directly or indirectly, by any of the Brand Companies, or (ii) has been licensed back for use by the applicable Brand Companies consistent with their use prior to such sale, transfer, lease, license or other disposition.
- (i) Gaiam Americas has all necessary rights to (i) assign the rights provided for in the SVOD Rights Assignment Agreement and (ii) grant the sublicenses provided for in the SVOD Rights Sub-License Agreement, respectively, and doing so will not breach any agreement between Gaiam Americas and any other Person.
- (j) Except as set forth in Section 5.10(j) of the Disclosure Schedule, none of the Brand Companies are a party to any Contract pursuant to which a Brand Company has (i) the right to use or exploit (A) the services, name and/or likeness of any individual (including, without limitation, audio and/or audiovisual recordings embodying the performances of any yoga and/or fitness talent) or (B) Intellectual Property owned by a Person other than Seller or the Brand Companies and (ii) an obligation to pay advances, minimum guarantees or any other payment which is not contingent on sales, but in each case, excluding (y) royalty payment obligations, which are contingent on sales, and (z) the Gaia Agreements.

5.11 Material Contracts.

(a) Section 5.11(a) of the Disclosure Schedule sets forth an accurate and complete list of each of the following Contracts to which any Brand Company is a party or by which any such Brand Company and its properties and assets are bound (together with all Contracts under which any Brand Company has (x) acquired or obtained, or has or has been licensed or otherwise granted, any license, permission or other right to utilize any Intellectual Property that is owned by a Person other than Seller or the Brand Companies or (y) licensed or otherwise granted any Person any license, permission or other right to utilize any Material Registered Company Intellectual Property, including the License Agreements, collectively, the Material Contracts):

(i) Contracts with a customer of any Brand Company that generated net revenue for the Company in excess of \$1,000,000 during the 12-month period ended December 31, 2015;

A-25

- (ii) Contracts pursuant to which any Brand Company paid to any supplier, vendor or similar Person in excess of \$1,000,000 during the 12-month period ended December 31, 2015;
- (iii) Contracts relating to the rental or use of tangible personal property, equipment, vehicles, other personal property or fixtures, except for any Contract individually involving payment of annual rental sums less than \$150,000 annually;
- (iv) Contracts pursuant to which any Brand Company is bound by any (A) covenant not to compete with any Person or in any geographical area, (B) covenant not to engage in a specific line of business, (C) covenant not to use, exploit or enforce any Company Intellectual Property in any capacity;
- (v) Contracts pursuant to which any Brand Company has incurred any Indebtedness in excess of \$50,000 or granted a Lien (other than Permitted Liens) on any property or asset of any Brand Company;
- (vi) Contracts relating to any joint venture, partnership, strategic alliance, shareholders agreement, co-marketing, co-promotion, co-packaging, joint development or similar arrangement pursuant to which any Brand Company either receives or makes payments in excess of \$50,000 annually;
- (vii) Contracts for the employment, hire or retention of any officer, employee, consultant, or independent contractor of any Brand Company (on a full-time or part-time basis) (A) for which such Brand Company is obligated as of the date of this Agreement to make payments in excess of \$100,000 annually or (B) providing for the payment of cash or other compensation or benefits upon the consummation of the Contemplated Transactions;
- (viii) Contracts involving any resolution or settlement of any actual or threatened Proceeding which involve (A) payments in excess of \$150,000 which have not yet been paid or (B) any restrictive covenants that are currently binding upon any Brand Company;
- (ix) other than pursuant to the Gaiam-FFL APA, Contracts for the sale of any of the properties or assets of the Brand Companies, other than in the ordinary course of business consistent with past practice, for consideration in excess of \$150,000 which were entered into within the last twenty-four (24) months and pursuant to which a Brand Company has any ongoing obligations thereunder; and
- (x) Contracts, not otherwise identified above, pursuant to which any of the Brand Companies has a non-contingent obligation as of the date of this Agreement to make payments in excess of \$150,000 individually during the 12-month period following the date of this Agreement.
- (b) Seller has delivered or made available to Purchaser complete and accurate copies of each written Contract (other than purchase orders) set forth on Section 5.11(a) of the Disclosure Schedule (including all written amendments, modifications and supplements thereto). All Material Contracts are valid, binding and enforceable against the applicable Brand Company and against the other parties thereto (except in each case as the enforceability thereof may be limited by any applicable bankruptcy, reorganization, insolvency or other Applicable Laws affecting creditors rights generally or by general principles of equity), and is in full force and effect. Each Brand Company has performed all obligations required to be performed by it to date under the Material Contracts to which it is a party, and it is not in breach or default in any material respect thereunder. No other party to any Material Contract is in breach or default thereunder and there exists no change, event, effect, condition or circumstance which, with the giving of notice, the lapse of time or the happening of any other event or condition, would become a default or event of default thereunder with respect to any Material Contract. Each Material Contract set forth in Section 5.11(a) of the Disclosure Schedule (or required to be set forth in Section 5.11(a) of the Disclosure Schedule) has not been terminated or been repudiated.

5.12 Insurance. Section 5.12 of the Disclosure Schedule sets forth an accurate and complete list of all insurance policies with policy periods in effect on the date hereof maintained for the benefit of the Brand

A-26

Companies (the <u>Brand Insurance Policies</u>), it being agreed that following the Closing, the Brand Companies will not have any benefits, rights, obligations, or Liabilities under any Brand Insurance Policies.

5.13 Taxes.

- (a) Each of Seller and each of the Brand Companies has filed when due (taking into account properly obtained extensions) all federal income Tax Returns with respect to income, assets, and operations of any Brand Company or the Purchased Interests and all other material Tax Returns that it was required to file under Applicable Laws. All such Tax Returns are true, correct and complete in all material respects. No extension of time in which to file any such Tax Return is currently in effect.
- (b) All Taxes with respect to income of any Brand Company or the Purchased Interests and all material Taxes with respect to assets and operations of any Brand Company or the Purchased Interests due and owing by Seller and any Brand Company (whether or not shown on such Tax Returns) have been timely paid by the due date thereof.
- (c) Seller and the Brand Companies have withheld and collected all amounts required by Applicable Law to be withheld or collected by it on account of Taxes, and has duly and timely paid all such withheld or collected amounts (or set aside for payment when due) to the appropriate Governmental Entity.
- (d) There are no Liens for Taxes (other than Taxes not yet due and payable) upon any of the assets of any Brand Company.
- (e) At all times since its formation, for federal income Tax purposes, the Company has been treated as an entity disregarded as separate from its owner within the meaning of U.S. Treasury Regulations §301.7701-3.
- (f) No audits, examinations or other proceedings are in progress, threatened in writing by any Governmental Entity or pending with regard to any Taxes or Tax Returns of Seller or any of the Brand Companies. All deficiencies asserted, or assessments made, against Seller or any Brand Company as a result of any audit, examination or other Proceeding by a Governmental Entity have been timely paid in full. No extensions or waivers of statutes of limitations have been given or requested with respect to any Taxes, or the filing of any Tax Return, of Seller or any Brand Company.
- (g) No Brand Company is liable for any Taxes of any other Person (other than any Brand Company) pursuant to U.S. Treasury Regulations §1.1502-6 (or any comparable provision of state, local or foreign law), as a transferee or successor, by contract or otherwise, or is a party to, or bound by, any Tax indemnity, Tax-sharing or Tax allocation agreement.
- (h) Neither Seller nor any Brand Company has executed or entered into any written agreement with, or obtained or applied for any written consents or written clearances or any other Tax rulings from, nor has there been any written agreement executed or entered into on behalf of any of them with any Governmental Entity, relating to Taxes, which such ruling or agreement will have continuing effect after the Closing Date.
- (i) Neither Seller nor any Brand Company will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any of the following that occurred or existed on or prior to the Closing Date: (i) a closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or non-U.S. income Tax law), (ii) an installment sale or open transaction, (iii) a prepaid amount, (iv) an intercompany item under U.S. Treasury Regulations §1.1502-13 or an excess loss account under U.S. Treasury Regulations §1.1502-19, (v) a change in the accounting method of Seller or any Brand Company pursuant to Section 481 of the Code or any similar

provision of the Code or the corresponding Tax laws of any nation, state or locality, or (vi) an election pursuant to Section 108(i) or the Code.

A-27

- (j) During the five (5) year period ending on the date hereof, neither Seller nor any Brand Company was a distributing corporation or a controlled corporation in a transaction intended to be governed by Section 355 of the Code.
- (k) Neither Seller nor any Brand Company has engaged in a reportable transaction within the meaning of U.S. Treasury Regulations §1.6011-4(b).
- (I) Since its formation, each of the Company Subsidiaries has been validly classified for U.S. federal income tax purposes as an association taxable as a corporation within the meaning of U.S. Treasury Regulations §301.7701-2.
- (m) No written claim has been made by any Taxing Authority in a jurisdiction where any Brand Company does not file Tax Returns that any Brand Company is or may be subject to taxation by that jurisdiction.
- (n) Except for Taxes relating to or arising from the transactions contemplated by the Contribution Agreement, there are no other Taxes incurred by or attributable to any Brand Company relating to any transactions attributable to the Pre-Closing Period that are or may become due after the Closing.
- (o) No Brand Company is, and has been at any time during the five (5) years preceding the date hereof, a United States real property holding company, as defined in Section 897(c)(2) of the Code.
- (p) Gaiam Americas is a consolidated target within the meaning of Treasury Regulations Section 1.338(h)(10)-1(b). Seller is the common parent of the selling consolidated group within the meaning of Treasury Regulations Section 1.338(h)(10)-1(b) that will be treated as selling the stock of Gaiam Americas pursuant to the Contemplated Transactions for U.S. federal income Tax purposes.

5.14 Proceedings; Judgments.

- (a) <u>Section 5.14(a)</u> of the <u>Disclosure Schedule</u> sets forth an accurate and complete list of all Proceedings before or against any Governmental Entity or any other Person pending or threatened (in writing) against any Brand Company, or any properties, assets, rights, officers or directors (in such capacity as officers or directors) of such Brand Company.
- (b) There is no outstanding Judgment to which any Brand Company is a party or subject (i) that materially and adversely affects any Brand Company, its properties, assets, business, or operations, (ii) that has had or could reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, or (iii) that prohibits or impairs, or could prohibit or impair, the consummation of the Contemplated Transactions.
- (c) As of the date hereof, to the Knowledge of Seller, there is no valid basis for any action, proceeding or investigation of the type described in the foregoing clauses (a) or (b).

5.15 Benefit Plans.

(a) Section 5.15(a) of the Disclosure Schedule sets forth an accurate and complete list of each employee benefit plan (as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>)) and each other compensation, bonus, pension, profit sharing, deferred compensation, stock ownership, stock purchase, stock option, phantom stock, retirement, employment, change-in-control, welfare, collective bargaining, severance, disability, death benefit, hospitalization and medical plan, program, policy and arrangement maintained or contributed to (or required to be contributed to) for the benefit of any Brand Employee, or former employee, officer or director who primarily performed service for or at the behest of Seller, any Brand Company or their respective Affiliates or

with respect to which any Brand Company would reasonably be expected to have direct, indirect, joint and several or contingent liability (the <u>Company Benefit Plans</u>).

A-28

- (b) Each Company Benefit Plan intended to qualify under Code Section 401(a) is the subject of a favorable determination, advisory or opinion letter issued by the Internal Revenue Service as to its qualified status under the Code, which determination, advisory or opinion letter may still be relied upon as to the qualified status of the Company Benefit Plan, and no circumstances have occurred since the date of such favorable determination or opinion letter that would result in the loss of the tax-qualified status of any such Company Benefit Plan.
- (c) None of the Company Benefit Plans that are welfare benefit plans as defined in ERISA Section 3(1) provide for continuing benefits or coverage for any participant or beneficiary of a participant after such participant s termination of employment, except to the extent required by Applicable Law.
- (d) No employee pension benefit plan (as defined under Section 3(2) of ERISA) sponsored, maintained or contributed to, at any time, by Seller, any Brand Company, or any of their respective Affiliates: (i) provides or provided any benefit guaranteed by the Pension Benefit Guaranty Corporation; (ii) is or was a multiemployer plan as defined in Section 4001(a)(3) of ERISA; or (iii) is or was subject to the minimum funding standards of Section 412 of the Code or Section 302 of ERISA.
- (e) Seller has made available to Purchaser, with respect to each Company Benefit Plan: (i) a copy of the Company Benefit Plan and all amendments (including any amendment that is scheduled to take effect in the future) and, in the case of any Company Benefit Plan not set forth in writing, a written description thereof; (ii) a copy of each Contract (including any trust agreement, funding agreement, service provider agreement, insurance agreement, investment management agreement or recordkeeping agreement) relating to the Company Benefit Plan; (iii) a copy of any summary plan description for the Company Benefit Plan; (iv) a copy of any Form 5500 for the last three Company Benefit Plan years; and (v) a copy of any determination, advisory, or opinion letter related to the Company Benefit Plan and any notice or other document that has been issued by, or that has been received by the sponsor of such Company Benefit Plan from any Governmental Entity with respect to the Company Benefit Plan.
- (f) Each Company Benefit Plan has been operated and administered in all material respects in accordance with its terms and in compliance with Applicable Law, including the Code and ERISA.
- (g) Each contribution or other payment that is required to have been accrued or made to, under, or with respect to any Company Benefit Plan has been duly accrued or made on a timely basis.
- (h) No prohibited transaction within the meaning of Code Section 4975 or ERISA Section 406 or 407, and not otherwise exempt under ERISA Section 408, has occurred with respect to a Company Benefit Plan that would reasonably be expected to subject the Company to any material liability.
- (i) There are no Proceedings pending or threatened with respect to any Company Benefit Plan or the assets of any Company Benefit Plan or any related trust (other than routine claims for benefits).
- (j) Other than in the ordinary course of business, there has been no amendment to, announcement by Seller or any of its Affiliates relating to, or change in employee participation or coverage under, any Company Benefit Plan that would increase the annual expense of maintaining such plan above the level of the expense incurred for the most recently completed fiscal year with respect to any Brand Employee. Other than in the ordinary course of business, Seller does not have any commitment or obligation and has not made any representations to any Brand Employee, whether or not legally binding, to adopt, amend or modify any Company Benefit Plan.
- (k) Neither the execution and delivery of this Agreement, nor the consummation of the Contemplated Transactions could (i) entitle any Brand Employee to material severance pay or any increase in severance pay, (ii) accelerate the

time of payment or vesting, or increase the amount of compensation due to any such Brand

A-29

Employee, (iii) directly or indirectly cause Seller to transfer or set aside any assets to fund any benefits under any Company Benefit Plan, (iv) otherwise give rise to any liability under any Company Benefit Plan, or (v) or require a gross-up or other payment to any disqualified individual within the meaning of Section 280G(e) of the Code due to the imposition of the excise tax under Section 4999 of the Code on any payment by a Brand Company to such disqualified individual or due to the failure of any payment to such disqualified individual to be deductible under Section 280G of the Code. There is no Contract to which Seller or the Brand Companies is a party that could, individually or collectively, result (or has resulted) in the payment of any amount that would not be deductible by reason of Section 280G of the Code.

- (I) Any other representation or warranty contained in this Agreement notwithstanding, the representations and warranties contained in this <u>Section 5.15</u> constitute the sole representations and warranties of Seller relating to employee benefit plan matters.
- **5.16 Employees and Labor Matters.** No Brand Company is party to or bound by (a) any written contract related to the employment of any individual or (b) a collective bargaining or other agreement with any labor union or other labor organization, and no Brand Company is negotiating or required to negotiate any such agreement with a labor union or other labor organization. No Brand Company is subject to any pending or threatened representation or certification proceedings or demand for recognition or certification. There are no pending or threatened strikes, lockouts, union organization activities (including, but not limited to, union organization campaigns or requests for representation), pickets, slowdowns, stoppages, or labor disputes in respect of any Brand Company. Purchaser has been provided access to a list containing the name, job title, rate of compensation, classification as exempt or non-exempt from overtime pay and vacation, sick time and personal leave accruals as of the date hereof, of each individual who, as of the date hereof, primarily performs services to the Brand Business and including all employees on authorized leave as of the date hereof (together with any employees or independent contractors hired or engaged by any Brand Company following the date hereof and prior to the Closing Date, the Brand Employees). Each Brand Company has at all times complied in all material respects with all Applicable Laws relating to labor or employment practices, terms, and conditions of employment, equal employment opportunity, nondiscrimination, sexual harassment, immigration, wages, hours, benefits, collective bargaining and similar requirements, the payment of Social Security and similar Taxes, and occupational safety and health (collectively, Employment Practices), and no Brand Company is engaged in any unfair labor practice, except where the failure to so comply has not had and would not reasonably be expected to have, individually or in the aggregate, a material liability. There are no pending or threatened, Proceedings or investigations relating to the Employment Practices of any Brand Company. No Brand Company employee is in violation of any noncompete, nonsolicitation, nondisclosure, confidentiality, employment, consulting or similar agreement with a third party in connection with his or her employment with a Brand Company. Any other representation or warranty contained in this Agreement notwithstanding, the representations and warranties contained in this <u>Section 5.16</u> constitute the sole representations and warranties of Seller relating to employee and labor matters.

5.17 Compliance with Applicable Laws; Permits.

(a) Since January 1, 2013, the Brand Companies have, and Seller has, conducted their respective business, and operated and used all of their respective assets, in compliance in all material respects with all Applicable Laws or Judgments. Since January 1, 2013, no Brand Company has, and Seller has not, received any notice (written or otherwise) from any Governmental Entity or any other Person alleging the violation of, or failure to comply with, any Applicable Law or Judgment. For the avoidance of doubt, this <u>Section 5.17</u> does not apply to (i) Tax matters, such matters being the subject of <u>Section 5.13</u>, (ii) employee benefit matters, such matters being the subject of <u>Section 5.15</u>, (iii) employee and labor matters, such matters being the subject of <u>Section 5.16</u>, (iv) unlawful payments, such matters being the subject of Section 5.26.

A-30

- (b) The Brand Companies possess all material federal, state, local and foreign Permits that are necessary for the operation of the Brand Business as presently conducted. All such Permits are valid and have not lapsed, been cancelled, terminated or withdrawn, nor has any of the Brand Companies received any notice (written or otherwise) from any Governmental Entity for the cancellation, termination or withdrawal of such Permits. The Brand Companies are in compliance in all material respects with all such Permits.
- **5.18** Agreements with Related Parties. None of (x) Seller or any of its Subsidiaries (other than a Brand Company) and/or (y) any Related Party of Seller or any of its Subsidiaries or any Brand Company, in each case, (a) is party to any agreements, transactions, arrangements or understandings with any Brand Company (other than with respect to an officer or director of such Brand Company pursuant to the Organizational Documents of such Company or with respect to an officer, director or employee of such Brand Company pursuant to Company Benefit Plans, in each case as in effect as of the date hereof), (b) has any interest, directly or indirectly in any assets or properties used or held for use by any Brand Company (other than as a holder of Equity Securities of the Brand Companies), (c) is (or, other than through ownership of less than 5% of the capital stock of any Person, has any direct or indirect interest in) a supplier, customer, competitor, debtor, lessor or creditor of the Brand Companies, (d) is (other than pursuant to the Company Benefit Plans as in effect as of the date hereof) owed any amounts by any Brand Company, or (e) owes any amount to any Brand Company.

5.19 Absence of Unlawful Payments.

- (a) No Brand Company has and no Representative of any Brand Company, acting on their behalf, has (i) violated any applicable provision of the Foreign Corrupt Practices Act of 1977, as amended, (ii) been targeted by any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department, (iii) paid, or offered, promised or authorized payment of, money or any other thing of value to any Governmental Entity or official, political party (or official thereof), candidate for political office, employee of a state-owned enterprise or official of an international organization (each, a <u>Government Person</u>) in violation of Applicable Law for the purpose of influencing, directly or indirectly through another Person, any act, omission or decision of such Government Person in an official capacity so that any of the Company might secure any advantage, obtain or retain business or direct business to any Person, or (iv) accepted or received any contributions, payments, gifts or expenditures that was unlawful.
- (b) The operations of the Brand Companies have been conducted at all times in compliance with the applicable financial recordkeeping and reporting requirements of the U.S. Currency and Foreign Transaction Reporting Act of 1970, as amended, the U.S. Money Laundering Control Act of 1986, as amended, and all money laundering-related laws of other jurisdictions where the Brand Companies conduct business or own assets, and any related or similar Applicable Laws issued, administered or enforced by any Governmental Entity (collectively, the Money Laundering Laws). No Proceeding by or before any Governmental Entity involving any Brand Company with respect to the Money Laundering Laws is pending or threatened.
- **5.20 Brokers**. Other than the Banker, no Brand Company has retained any Person to act as a broker or agreed or become obligated to pay, or taken any action that might result in any Person claiming to be entitled to receive, any brokerage commission, finder s fee or similar commission or fee in connection with any of the Contemplated Transactions.
- **5.21 No Other Operations; Contribution of the Brand Business**. The Company is a holding company that has never had any operations (other than administrative operations as a holding company of the Company Subsidiaries) prior to its acquisition of the Contributed Assets pursuant to the Contribution Agreement. As of the Closing Date, the Brand Business will be the only business operation carried on by the Brand Companies. Seller has duly and validly transferred and assigned to the Company pursuant to the Contribution Agreement, and the Company has assumed and

accepted from Seller, all of the Contributed Assets and the Contributed Liabilities, in each case, in accordance with Applicable Law and free and clear of all Liens (other than Permitted Liens). As of the date hereof, the Company (i) does not have any assets other than the Contributed Assets and any assets

A-31

acquired pursuant to its operation of the Brand Business in the ordinary course consistent with past practice and (ii) does not have any Liabilities other than the Contributed Liabilities and Liabilities incurred pursuant to its operation of the Brand Business in the ordinary course consistent with past practice. Other than this Agreement, the Gaiam-FFL APA, and the Collateral Agreements to which the Company is a party, the Company is not a party to any contract or agreement other than the Contribution Agreement and the Brand Contracts (as defined in the Contribution Agreement). Except as set forth in the Collateral Agreements, the Contributed Assets constitute all of the assets necessary to conduct the Brand Business as currently conducted.

5.22 Environmental Matters.

- (a) The Brand Companies are, and since January 1, 2010 have been in compliance with, and operate, and have operated, the Brand Business in compliance with, all Environmental Laws applicable to the Brand Business or the operations at the facilities currently operated, leased, or previously operated, owned, or leased by them.
- (b) No Hazardous Materials have been produced, sold, used, stored, transported, arranged for transport, handled, released, discharged, or disposed of by any of the Brand Companies or any other Person or any of their respective Representatives at or from any location, including at or from the facilities currently or formerly owned, leased, or operated by any of the Brand Companies, in a manner that violated or would reasonably be expected to give rise to Liability under any applicable Environmental Law.
- (c) There are no facts, circumstances or conditions relating to the past or present business or operations of any Brand Company or any of their respective predecessors (including the disposal of any wastes, Hazardous Materials or other materials at any location), or to any real property or facility at any time owned, leased, or operated by any Brand Company or any of their respective predecessors, that would reasonably be expected to give rise to any Proceeding, or to any Liability, under any Environmental Law.
- **5.23** Company IT Assets. The Company IT Assets constitute the information technology assets necessary to operate the e-commerce business conducted by the Brand Companies as of the date hereof, in the manner in which it is presently conducted. The Brand Companies have implemented backup and security technology, policies and procedures reasonably necessary for the continued operation of the Brand Business. The Company IT Assets have operated in all material respects in accordance with all licenses and all specifications, documentations or warranties therefor, and, to the Knowledge of Seller, have not materially malfunctioned or failed in any manner that adversely impacted the Brand Companies.
- **5.24 Privacy**. Seller and the Brand Companies (a) comply and at all times since January 1, 2010 have complied with all applicable privacy laws and regulations and contractual obligations regarding the collection, processing, disclosure and use of all data, including data consisting of personally identifiable information that is, or is capable of being, associated with specific individuals; (b) comply and at all times since January 1, 2010 have complied with the Brand Companies privacy policies with respect to personally identifiable information; and (c) take and have taken all contractually required measures and all measures a reasonable Person in the same industry as the Brand Business would reasonably be expected to take to protect any personally identifiable information that the Brand Companies have collected or otherwise acquired from unauthorized disclosure. There are no claims, investigations or actions currently pending concerning the data or privacy practices of Seller or the Brand Companies. No Proceedings have been asserted or threatened in writing against Seller or the Brand Companies in the last three (3) years alleging a violation of any of the foregoing. Since January 1, 2010, there has been no loss, damage, or unauthorized access, use, modification, or breach of security of personally identifiable information or other data maintained by or on behalf of Seller or the Brand Companies.

5.25 Inventories. The inventories of the Brand Companies are in the physical possession of the Brand Companies, with third party logistics contractors or in transit to or from a customer or supplier of the Brand Companies, and none of the inventories has been pledged as collateral or otherwise is subject to any Lien (other

A-32

than any Permitted Lien) or is held on consignment from others. The inventories reflected in the Financial Statements was, and the inventory reflected on the Company s books of account have been, determined and valued on a first-in first-out basis, in accordance with GAAP applied, in the case of the inventory reflected on the Company s books and records, on a basis consistent with the Financial Statements. The inventories were acquired or produced in the ordinary course of business consistent with past practice. The inventories are of a quality presently useable and salable in the ordinary course of business.

5.26 Product Liability.

- (a) Since January 1, 2012, there has been no Proceeding by or before any Governmental Entity against or involving any of the Brand Companies or concerning any product manufactured, shipped, sold or delivered by or on behalf of any of the Brand Companies relating to or resulting from a material alleged defect in design, manufacture, materials or workmanship of any product manufactured, shipped, sold or delivered by or on behalf of any of the Brand Companies or any material alleged failure to warn, or any material alleged breach of implied warranties or representations, and none has been threatened.
- (b) Since January 1, 2012, there has not been any Occurrence that has had or would reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect.
- (c) Since January 1, 2012, there has not been any product recall, rework or post sale warning or similar action conducted with respect to any product manufactured (or to be manufactured), or sold by or on behalf of any of the Brand Companies or any investigation or consideration of or decision made by any Person or Governmental Entity concerning whether to undertake or not undertake any material product recall, rework or post sale warning or similar action.
- (d) There have been no material defects in design, manufacturing, materials or workmanship including any failure to warn, or any breach of express or implied warranties or representations, which involve any product manufactured, shipped, sold or delivered by or on behalf of any of the Brand Companies. All material manufacturing standards applied, testing procedures used, and product specifications disclosed to customers by the Brand Companies have complied with all requirements established by Applicable Law.
- **5.27 Suppliers and Customers**. Since the Interim Balance Sheet Date, no customer and no supplier required to be identified in Section 5.11(a)(i) of the Disclosure Schedule and Section 5.11(a)(ii) of the Disclosure Schedule, respectively, has canceled or otherwise terminated, or threatened to cancel or otherwise terminate, its relationship with any of the Brand Companies. Since the Interim Balance Sheet Date, none of Seller or any of the Brand Companies has received any written notice that any such supplier or customer may cancel or otherwise materially and adversely modify its relationship with any of the Brand Companies or limit its services, supplies or materials to any of the Brand Companies, or its usage or purchase of the services and products of any of the Brand Companies either as a result of the transactions contemplated hereby or otherwise.
- **5.28 Accounts Receivable**. All of the accounts receivable of the Brand Companies set forth on the Interim Balance Sheet are (a) properly stated thereon in accordance with GAAP, (b) to the Knowledge of Seller, are valid and enforceable claims, and (c) represent bona fide transactions arising in the ordinary course of business of the Brand Companies. Since the Interim Balance Sheet Date, there have not been any write-offs as uncollectible of any such accounts receivable, except for write-offs in the ordinary course of business consistent with past practice.
- **5.29 Inventory**. The Brand Companies own all of the inventory of the Brand Business free and clear of all Liens, subject only to Permitted Liens. The inventory of the Brand Companies consists of raw materials, components,

finished goods, packaging and capitalized landed costs as reflected on the Interim Balance Sheet, and the changes thereto reflected on the books and records of the Brand Companies since the Interim Balance

A-33

Sheet Date, in the ordinary course of business and consistent with past practice, subject to normal adjustments and reserves. The inventory of the Brand Companies reflected on the Interim Balance Sheet and in the books and records of the Brand Companies since the Interim Balance Sheet Date is of a quality and a quantity useable in the ordinary course of business, and such inventory has been valued at the lesser of cost or market, and all material unmarketable, returned, rejected, damaged, slow moving or obsolete inventory has been written off or written down to net realizable value or adequately reserved against in the books and records of the Brand Companies and in the Interim Balance Sheet.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller as follows:

6.01 Authority; Execution and Delivery; Enforceability.

- (a) Purchaser has the requisite power and authority to execute and deliver this Agreement and the Collateral Agreements to which Purchaser is or will be a party, to perform its obligations hereunder and thereunder and to consummate the Contemplated Transactions.
- (b) The execution, delivery and performance by Purchaser of this Agreement and the Collateral Agreements to which Purchaser is or will be a party and the consummation by Purchaser of the Contemplated Transactions, have been duly authorized by all necessary limited liability company action on the part of Purchaser, and no other action on the part of Purchaser is or will be necessary to authorize the execution, delivery and performance of this Agreement and the Collateral Agreements to which Purchaser is or will be a party and the Contemplated Transactions.
- (c) This Agreement has been, and each of the Collateral Agreements to which Purchaser is a party will be, duly executed and delivered by Purchaser. This Agreement constitutes, and each of the Collateral Agreements to which Purchaser is a party will constitute, the legal, valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with their terms, except to the extent that their enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors rights generally and to general equitable principles.

6.02 Noncontravention.

- (a) Neither the execution and delivery of this Agreement or the Collateral Agreements to which Purchaser is or will be a party, nor the consummation or performance of any of the Contemplated Transactions, will (a) in any material respect contravene, conflict with or result in a violation of any Applicable Law or any Judgment to which Purchaser is subject or by which any of its properties or assets are bound; (b) contravene, conflict with or result in a breach of, or constitute a default under, or result in the acceleration of any obligation or loss of any benefits under, the provisions of any material Contract to which Purchaser is subject or by which any of its properties or assets are bound; or (c) contravene, conflict with or result in a violation of any of the provisions of the Organizational Documents of Purchaser.
- (b) Neither Purchaser nor, to the Knowledge of Purchaser, any of its Affiliates owns an interest in any Person which reasonably could be expected to cause a Governmental Entity to seek to prevent or materially impair or delay the consummation of the Contemplated Transactions pursuant to the HSR Act.

6.03 Solvency. Assuming the representations and warranties of the Company contained in this Agreement are true and correct in all material respects, then as of the Closing and immediately after consummating the

A-34

Contemplated Transactions, Purchaser will not (a) be insolvent (either because its financial condition is such that the sum of its debts is greater than the fair value of its assets or because the present fair salable value of its assets will be less than the amount required to pay its probable liability on its debts as they become absolute and matured), (b) have unreasonably small capital with which to engage in its business or (c) have incurred debts beyond its ability to repay such debts as they become absolute and matured.

6.04 Financing.

- (a) Purchaser has delivered to Seller true and complete copies of (i) a duly executed commitment letter, dated as of the date hereof, between Purchaser and GSO Capital Partners LP and (ii) a duly executed commitment letter, dated as of the date hereof, between Purchaser and Bank of America, N.A., and Fifth Third Bank (collectively, the <u>Debt</u> Commitment Letters) (subject to redaction so long as such redaction does not cover terms that would adversely affect the conditionality or availability of the Debt Financing), pursuant to which the counterparties thereto have committed, subject to the terms and conditions thereof (including, to the extent required in accordance therewith, the exercise of so-called flex provisions in the Fee Letter), to lend the amounts set forth therein (the Debt Financing). Assuming the funding in full of the Debt Financing on the Closing Date, Purchaser on the Closing Date will have, together with cash on hand and available borrowing under existing credit facilities, sufficient funds in immediately available cash, to (A) pay an amount in cash equal to the amounts set forth in Section 3.02(a); (B) pay any and all fees and expenses required to be paid by Purchaser in connection with the transactions contemplated by this Agreement and the Debt Financing; and (C) satisfy all other payment obligations of Purchaser contemplated hereunder (including the payment at Closing of unpaid Transaction Expenses). Purchaser has also delivered to Seller a true and complete copy of any fee letter in connection with the Debt Commitment Letters (any such fee letter, a <u>Fee Letter</u>) (subject to redaction so long as such redaction does not cover terms (other than the actual amount of fees that are the subject of the Fee Letter) that would adversely affect the conditionality or availability of the Debt Financing).
- (b) As of the date hereof, there are no side letters or other Contracts or arrangements related to the Debt Financing other than the Debt Commitment Letters and the Fee Letters or as otherwise expressly set forth therein. None of the Debt Commitment Letters nor the Fee Letters has been amended or modified, no such amendment or modification is contemplated as of the date hereof, and the commitments set forth in the Debt Commitment Letter have not been withdrawn or rescinded in any respect; *provided*, that the exercise of any flex provisions in the Fee Letter shall not be considered an amendment, modification, withdrawal or rescission of the Fee Letter for the purposes of this Agreement.
- (c) The Debt Commitment Letters are in full force and effect and is the valid, binding and enforceable obligations of Purchaser and, to the knowledge of Purchaser, the other parties thereto, except to the extent that its enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors—rights generally and to general equitable principles. There are no conditions precedent or other contingencies related to the funding of the full amount of the Debt Financing, other than as set forth in the Debt Commitment Letters and any related Fee Letter. To the knowledge of Purchaser, no event has occurred, or circumstance exists, which, with or without notice, lapse of time or both, would or would reasonably be expected to constitute a default or breach on the part of Purchaser and, to the knowledge of Purchaser, any other party to the Debt Commitment Letters. As of the date hereof, and assuming the accuracy of the representations and warranties of Seller contained in this Agreement and performance by Seller of its obligations under this Agreement, Purchaser has no reason to believe that any of the conditions to the Debt Financing contemplated in the Debt Commitment Letters and any related Fee Letter will not be satisfied, that the Debt Financing will not be made available at or prior to the time contemplated hereunder for the Closing. Purchaser has fully paid, or caused to be fully paid, any and all commitment or other fees which are due and payable on or prior to the date hereof pursuant to the terms of the Debt Commitment Letters and any related Fee Letter.

6.05 Judgments. There is no outstanding Judgment to which Purchaser is a party or is subject that prohibits or impairs the consummation of the Contemplated Transactions by Purchaser.

A-35

6.06 Non-Reliance of Purchaser.

- (a) PURCHASER HEREBY ACKNOWLEDGES THAT SELLER IS NOT MAKING ANY REPRESENTATIONS OR WARRANTIES, AND PURCHASER IS NOT RELYING ON ANY REPRESENTATIONS OR WARRANTIES OF SELLER, OTHER THAN THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS AGREEMENT. WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, PURCHASER HEREBY ACKNOWLEDGES THAT, EXCEPT TO THE EXTENT EXPRESSLY SET FORTH IN THIS AGREEMENT, SELLER IS NOT MAKING ANY REPRESENTATION OR WARRANTY WITH RESPECT TO (I) ANY FINANCIAL PROJECTION OR FORECAST RELATING TO THE BRAND COMPANIES, (II) MERCHANTABILITY, (III) FITNESS FOR ANY PARTICULAR PURPOSE, (IV) THE VIABILITY OR LIKELIHOOD OF SUCCESS OF THE BRAND BUSINESS, OR (V) ANY OTHER INFORMATION MADE AVAILABLE, WHETHER PURSUANT TO ANY PRESENTATION MADE BY OR ON BEHALF OF ANY BRAND COMPANY, PURSUANT TO ANY ELECTRONIC OR PHYSICAL DELIVERY OF DOCUMENTATION OR OTHER INFORMATION, OR OTHERWISE, TO PURCHASER, ITS AFFILIATES AND THEIR RESPECTIVE REPRESENTATIVES, AND PURCHASER EXPRESSLY DISCLAIMS ANY SUCH REPRESENTATION OR WARRANTY.
- (b) In connection with Purchaser s investigation of Seller, the Brand Companies and Brand Business, Purchaser has received from or on behalf of Seller and/or the Brand Companies certain estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions. Purchaser acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions, that Purchaser is familiar with such uncertainties, that Purchaser is taking full responsibility for making its own evaluation of the adequacy and accuracy of all projections, forecasts, budgets or revenues or earnings predictions so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions), and that Purchaser shall have no claim against Seller or any Brand Company or any of their Representatives or Affiliates with respect thereto; provided, however, that nothing contained in this Section 6.06 shall limit Purchaser s remedies (i) in the event of fraud of Seller or (ii) under this Agreement. Purchaser acknowledges and agrees that Seller makes no representation or warranty with respect to such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions (including any such underlying assumptions). Purchaser confirms that Seller and the Brand Companies have made available to it and its Representatives the opportunity to ask questions of the Representatives of the Brand Companies and Seller and their respective Affiliates and to acquire such additional information about the business, operations and financial condition of the Brand Companies and the Brand Business as requested, and all such information has been received.
- **6.07 Brokers.** Purchaser has not retained any Person to act as a broker or agreed or become obligated to pay, or has taken any action that might result in any Person claiming to be entitled to receive, any brokerage commission, finder s fee or similar commission or fee in connection with any of the Contemplated Transactions.
- **6.08 Investment Intention.** Purchaser is acquiring the Purchased Interests for its own account and not with a view to their distribution within the meaning of Section 2(11) of the Securities Act of 1933, as amended (the <u>Securities Act</u>). Purchaser acknowledges that the Purchased Interests have not been registered for offer or sale under the Securities Act, or other Applicable Law, and that the Purchased Interests may not be transferred, sold or offered for sale except pursuant to the registration provisions of the Securities Act or other Applicable Law, or pursuant to any exemption therefrom. Purchaser is an accredited investor within the meaning of Rule 501(a) of the Securities Act, and has such knowledge and experience in financial and business matters, and is capable of evaluating the merits and risks of the Contemplated Transactions.

A-36

ARTICLE VII

COVENANTS

7.01 Covenants Relating to Conduct of Business.

- (a) Except for matters set forth in Schedule 7.01 or otherwise expressly permitted or required by the terms of this Agreement or except as required by Applicable Law, from the date of this Agreement to the Closing, Seller shall cause the Brand Companies to conduct their business in the ordinary course of business consistent with past practice. Without limiting the generality of the foregoing, except as otherwise set forth in Schedule 7.01, or as otherwise expressly permitted or expressly required by the terms of this Agreement or as required by Applicable Law, Seller shall not (and shall cause the Brand Companies not to) do any of the following without the prior written consent of Purchaser (which consent shall not be unreasonably withheld, conditioned or delayed):
- (i) amend any Organizational Document of the Brand Companies;
- (ii) authorize for issuance, issue, sell or deliver any Equity Securities of the Brand Companies;
- (iii) split, combine, redeem, reclassify, purchase or otherwise acquire, directly or indirectly, any Equity Securities of the Brand Companies, or make any other change in the capital structure of the Brand Companies;
- (iv) allow any Brand Company to:
- (A) incur or assume any Indebtedness for borrowed money or guarantee any such Indebtedness, in excess of \$50,000 other than in the ordinary course of business consistent with past practice;
- (B) permit or allow any Brand Company assets, including the Company Intellectual Property, to become subjected to any Lien which is not released on or before the Closing Date (other than Permitted Liens);
- (C) acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof;
- (D) form any Subsidiary;
- (E) write off as uncollectible any notes or accounts receivable in excess of \$250,000, except write-offs in the ordinary course of business consistent with past practice;
- (F) make or incur capital expenditures outside the ordinary course of business in excess of \$100,000;
- (G) other than pursuant to the Gaiam-FFL APA, or the sale of inventory in the ordinary course of business consistent with past practice, sell, lease, license, transfer, cancel or otherwise dispose of any of its assets, including but not limited to Company Intellectual Property (other than non-exclusive licenses granted in the ordinary course of business consistent with past practice), that are material, individually or in the aggregate, to the Brand Companies as a whole;
- (H) purchase or enter into any lease of real property other than extensions of existing leases in the ordinary course of business;
- (I) except as required by GAAP, make any material change in any method of accounting or auditing practice;

A-37

- (J) (1) make, revoke or amend any Tax election, change in any respect any accounting method for Tax purposes, file any Tax Return outside the ordinary course of business or amend any Tax Return, (2) settle any claim for Taxes or enter into any Tax sharing or similar, contract, (3) surrender any right to claim a refund of Taxes, or (4) consent to any extension or waiver of the limitations period (other than one resulting from obtaining a valid extension of time to file Tax Returns in the ordinary course) applicable to any Tax claim or assessment;
- (K) settle or compromise any dispute regarding a Tax liability outside the ordinary course of business;
- (L) discharge or settle any material Proceeding, other than respect to any of the matters set forth in <u>Section 5.14 of the Disclosure Schedule</u> and other than payments, discharges, settlements or satisfactions in the ordinary course of business and consistent with past practice;
- (M) (1) increase the number of Brand Employees by more than two percent (2%) from that in effect as of the date hereof, (2) increase the base compensation, bonus opportunity, pension, welfare, fringe or other benefits payable to the Brand Employees, (3) pay any bonus or make any incentive awards, to the Brand Employees, (4) create, establish, adopt or enter into, amend, modify or terminate any Company Benefit Plan or any other incentive compensation or benefit plan that would constitute a Company Benefit Plan had it been in effect as of the date of this Agreement, in each case with respect to the Brand Employees, (5) negotiate, assume, amend, extend or renew or enter into any labor or collective bargaining agreement relating to the Brand Business, or (6) take any action to amend, waive or accelerate the vesting or payment of any compensation or benefit under any Company Benefit Plan with respect to the Brand Employees, in each case for clauses (1) through (6) above, except in the ordinary course of business consistent with past practice; *provided*, *however*, that nothing in this Section 7.01(a)(iv)(M) shall limit the ability of Seller to adopt, modify, amend or terminate any employee benefit plan, program, policy, arrangement or practice that is or may become applicable to the employees of Seller who are not Brand Employees;
- (N) plan, announce, implement or effect any reduction in force, lay-off, early retirement program, severance program or other program or effort concerning the termination of employment of employees of the Brand Companies (other than routine employee terminations for cause);
- (O) allow any Brand Insurance Policies to lapse or any changes to be made to any Brand Insurance Policies so as to adversely affect the insurance coverage of the Brand Companies or their properties and assets following the Closing; or
- (P) enter into any Contract, agree, commit or offer (in writing or otherwise) to take any of the foregoing actions.
- (b) Except as otherwise expressly permitted or expressly required by the terms of this Agreement, or as set forth on Schedule 7.01 or as required by Applicable Law, Seller shall, and shall cause the Brand Companies to, use commercially reasonable efforts to maintain in full force and effect all Company Intellectual Property which is material to the operation of the Brand Business.

7.02 Access to Information; Confidentiality.

(a) Seller shall (and shall cause the Brand Companies to) afford to Purchaser, its financing sources and each of their respective Representatives commercially reasonable access during normal business hours throughout the period prior to the Closing Date to the books and records, properties and personnel of the Brand Companies and shall furnish promptly to Purchaser all financial and operating data and other information concerning the Brand Companies businesses, properties, and personnel as Purchaser may request; *provided*, that, Seller and the Brand Companies shall not be required to take any action that would unreasonably disrupt the Brand Companies respective operations. Each

Party agrees that the terms of the Confidentiality Agreement (including the provisions regarding no unauthorized contact and non-solicitation) shall survive and remain in full force and effect following the execution and delivery of this Agreement and Purchaser shall continue to comply

A-38

with the terms thereof through the Closing. Notwithstanding the forgoing, in no event shall the foregoing require Seller or any Brand Company to permit (a) any inspection or to disclose any information that, in its reasonable judgment, might risk or result in (i) the waiver of any attorney-client or other privilege, (ii) the disclosure of any Intellectual Property of any third party, (iii) the violation of any of its or its Affiliates—or Representatives obligations with respect to confidentiality, any other rights of third parties or any Applicable Law, or (iv) the disclosure of any personnel files of any Brand Employee prior to the Closing Date without Purchaser having first obtained a release by such Brand Employee (provided that Purchaser shall indemnify and hold Seller and its Affiliates and Representatives harmless from any Liabilities arising out of or relating to the transfer of such personnel files), or (b) any contact with any customers, vendors, suppliers or any other Person with whom any Brand Company has entered into any Contract.

- (b) For a period of six (6) years after the Closing and subject to <u>Section 7.02(d)</u>, each Party shall, and shall cause their respective Affiliates to, afford the other Party, its Affiliates and their respective Representatives, commercially reasonable access upon reasonable notice during normal business hours, to information and documentation of Seller and the Brand Companies, as applicable, to the extent that such access may be reasonably requested by such Party in connection with such Party s Taxes, reporting obligations and compliance with Applicable Laws.
- (c) Purchaser agrees to hold all the books and records of the Brand Business existing on the Closing Date and not to destroy and dispose of any thereof for a period of six (6) years from the Closing Date or such longer time as may be required by Applicable Law, and thereafter, if it desires to destroy or dispose of such books and records, to offer first in writing at least sixty (60) days prior to such destruction or disposition to surrender them to Seller.
- (d) Seller acknowledges that it is in possession of and may in the future obtain possession, including without limitation pursuant to Section 7.02(b) and Article X, of Confidential Information concerning the Brand Companies and their businesses and operations. Seller shall, and shall use its best efforts to cause its Affiliates and their respective Representatives to, treat confidentially and not disclose all or any portion of such Confidential Information and will use such Confidential Information solely for the purposes of consummating the Contemplated Transactions or in connection with Seller s Taxes, reporting obligations and compliance with Applicable Laws, operating the Gaia Business and for no other purposes; provided, that the Brand Companies may also use the Confidential Information for the purpose of operating the Brand Business in the ordinary course. Seller acknowledges and agrees that such Confidential Information is proprietary and confidential in nature and may be disclosed to its Representatives only to the extent necessary for Seller to consummate the Contemplated Transactions or in connection with Seller s Taxes, reporting obligations and compliance with Applicable Laws and operating the Gaia Business (it being understood that Seller shall be responsible for any disclosure by any such Representative not permitted by this Agreement). If Seller or any of its Affiliates or Representatives are requested or required to disclose (after Seller has used its commercially reasonable efforts to avoid such disclosure and after promptly advising and consulting with Purchaser about Seller s intention to make, and the proposed contents of, such disclosure) any of the Confidential Information (whether by deposition, interrogatory, request for documents, subpoena, civil investigative demand or similar process), Seller shall, or shall use its best efforts to cause such Affiliate or Representative, to provide Purchaser with prompt written notice of such request so that Purchaser may seek an appropriate protective order or other appropriate remedy, at Purchaser s sole cost and expense. At any time that such protective order or remedy has not been obtained, Seller may disclose only that portion of the Confidential Information which such Person is legally required to disclose or of which disclosure is required to avoid sanction for contempt or any similar sanction, and Seller shall exercise its commercially reasonable efforts to obtain assurance that confidential treatment will be accorded to such Confidential Information so disclosed. Seller further agrees that, from and after the Closing Date, Seller will and will use its best efforts to cause its Affiliates and Representatives to, upon the request of Purchaser or the Company, promptly deliver to Purchaser or the Company all documents, or other tangible embodiments, constituting Confidential Information or other information with respect to the Brand Companies; provided, that

A-39

Seller shall have the right to retain copies of all such documents, or other tangible embodiments; *provided*, *further*, that any such retained documents or other tangible embodiments shall be subject to this <u>Section 7.02</u>.

(e) Following the Closing, upon Purchaser s request and at Purchaser s sole cost and expense, Seller will either (i) assign all of its rights and obligations pursuant to the confidentiality agreements entered into with any Person in connection with the sale of the Brand Business or an Alternative Transaction Proposal to the extent permitted by the terms of such confidentiality agreements, or (ii) cooperate with Purchaser in enforcing any such confidentiality agreements.

7.03 Publicity. Purchaser and Seller shall consult in good faith with each other before issuing any press release or making any public statement with respect to this Agreement or the Contemplated Transactions and provide to the other Party for review a copy of any such press release or public statement. The foregoing restrictions shall not apply to any disclosure (a) that may be required by Applicable Law, including the Applicable Laws of any United States or foreign securities exchange in which case the disclosing Party shall inform the non-disclosing Party of the required disclosure, and, to the extent practical provide the non-disclosing Party a copy of such release or statement prior to its release or filing, it being understood that the disclosing Party shall have final say on the wording of any such required disclosure, (b) made in connection with the enforcement of any right or remedy relating to this Agreement or any Collateral Agreement or the Contemplated Transactions, or (c) to a Party s Representatives as necessary in connection with the ordinary course of its business. Each Party agrees that it shall (and shall use commercially reasonable effort to cause its Affiliates and its and their respective Representatives to) hold in confidence all nonpublic information acquired from the other Party or the other Party s Representatives in accordance with the terms of the Confidentiality Agreement, the terms of which are incorporated herein by reference in their entirety.

7.04 Commercially Reasonable Efforts. Between the date of this Agreement and the Closing Date, each Party shall use commercially reasonable efforts to cause the conditions to the obligations of the other Party as set forth in <u>Article VIII</u> to be satisfied and the Contemplated Transactions to be consummated in accordance with their terms.

7.05 HSR Act and Foreign Competition Filings. Without limiting the general nature of Section 7.04, each of Purchaser and Seller shall (and, to the extent required, shall cause their respective Affiliates to): (a) comply promptly, and in any event within ten Business Days after the date of this Agreement, with the notification and reporting requirements of the HSR Act and use commercially reasonable efforts to obtain early termination of the waiting period under the HSR Act; (b) within five Business Days after the date of this Agreement, make such other filings with any similar foreign Antitrust Authorities as may be required under any applicable similar foreign law; and (c) use commercially reasonable efforts to cooperate with the other Parties in connection with making any filing under the HSR Act or any similar foreign law and in connection with any filings, communications or other submissions related to resolving any investigation or other inquiry by any Antitrust Authority under the HSR Act or any similar foreign law.

7.06 Consents and Approvals; Representation and Warranty Policy.

(a) Each of Purchaser and Seller shall (and shall cause each of its Affiliates to), in addition to complying with their respective obligations under Section 7.05 with respect to Antitrust Authorities, including obtaining the termination of the waiting period under the HSR Act, use commercially reasonable efforts to assemble, prepare and file any information (and, as needed, to supplement such information) as may be reasonably necessary in order to obtain, as promptly as practicable, all consents and approvals required to be obtained from any Governmental Entity in order to execute and deliver this Agreement and consummate the Contemplated Transactions in accordance with the terms and conditions of this Agreement.

(b) In addition, Seller shall use its commercially reasonable efforts to obtain, and, Purchaser shall use its commercially reasonable efforts to cooperate with Seller in obtaining, any consents from third parties pursuant

A-40

to the Contracts set forth on Schedule 7.06(b) (the Material Consents). In connection with each Material Consent, Purchaser shall promptly comply with any reasonable requests from third parties in connection with such Material Consents. The Parties agree that nothing under this Section 7.06 shall be interpreted to require Seller, any Brand Company, Purchaser or any of their respective Affiliates to pay any amount, incur any obligation in favor of or offer or grant any accommodation (financial or otherwise, regardless of any provision to the contrary in the underlying Contract, including any requirements for the securing or posting of any bonds, letters of credit or similar instruments, or the furnishing of any guarantees) in order to obtain a Material Consent. If any Material Consent is not obtained prior to the Closing, Seller shall use its commercially reasonable efforts to obtain such Material Consent after the Closing, and Purchaser and Seller shall cooperate in any commercially reasonable arrangement reasonably acceptable to Purchaser to both (i) provide Purchaser with the interests and benefits of such Contract and (ii) cause Purchaser to bear all costs and Liabilities of, or under, any such Contracts, until such time as such Material Consent for such Contract shall have been obtained (provided, that any incremental costs of such arrangement shall be borne by Seller, including any costs incurred by Seller in its performance as agent on behalf of Purchaser if such costs would not have been incurred by Purchaser had such Material Consent been obtained).

- (c) Purchaser s obligations under Section 7.05 and/or this Section 7.06 to use commercially reasonable efforts shall not include (i) proposing, negotiating, committing to or effecting, by consent decree, hold separate order, or otherwise, the sale, transfer, license, divestiture or other disposition of, or any prohibition or limitation on the ownership, operation, effective control or exercise of full rights of ownership, of any of the businesses, product lines or assets of Purchaser or any of its Affiliates or of the Brand Companies or (ii) paying any material amounts or incurring any material obligations or granting any other material financial accommodation in connection with obtaining any consents from third parties.
- (d) Purchaser shall use its reasonable best efforts to take, or cause to be taken, all actions necessary to have fully bound by the Closing Date, the Representation and Warranty Policy, including, as promptly as reasonably practical, furnishing the Representation and Warranty Policy Insurer and its Representatives with all such information as is reasonably requested by such Persons in connection with the Representation and Warranty Policy Insurer s issuance of the Representation and Warranty Policy at the Closing. Seller shall, and shall cause the Brand Companies to, and shall request the Brand Companies Representatives to, provide all cooperation reasonably requested by Purchaser in connection with the arrangement and obtaining of the Representation and Warranty Policy. Purchaser shall keep Seller reasonably informed of the status of its efforts to arrange and obtain the Representation and Warranty Policy and provide to Seller copies of all executed definitive documents related to the Representation and Warranty Policy. From and after the Closing, Seller shall, and shall cause its Representatives to, reasonably cooperate with Purchaser Parties, at such Purchaser Parties expense, to the extent necessary in connection with any claim made by any such Purchaser Party under the Representation and Warranty Policy.
- **7.07 Notification of Certain Matters.** Seller shall promptly notify Purchaser of (a) any material Proceeding in connection with the Contemplated Transactions commenced or, to the Knowledge of Seller, threatened, against Seller, any of the Brand Companies, the Brand Business, or Purchaser, as the case may be; (b) any fact or event which any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the Contemplated Transactions; or (c) the occurrence of any change, event, effect, condition or circumstance which, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect; *provided*, that no such notification, not the obligation to make such notification, shall affect the representations, warranties or covenants, or the conditions to the obligations of, Seller.

7.08 Employees.

(a) No later than four (4) Business Days following the date hereof (the <u>Notice Date</u>), in such form as approved by Seller and Purchaser, which approval shall not be unreasonably withheld, conditioned or delayed,

A-41

Seller shall cause the Brand Companies, as applicable, to deliver to (i) those Brand Employees who work in or are based out of the Brand Companies facility in Colorado (the Colorado Facility) and who are either Designated Employees or Transferred Employees, a joint WARN Notice (as defined below) from Seller and Purchaser notifying such Brand Employees of the Contemplated Transactions and that, as a result of the consummation of the Contemplated Transactions, the employment of such Designated Employees and Transferred Employees will be terminated by the applicable Brand Company effective on the Closing Date (which notice for the Transferred Employees shall reflect that it is anticipated that they shall receive offers of employment from Licensees, with such employment to commence immediately following the Closing Date such that there will be no gap in employment); provided, however, that if the Closing Date occurs before the 60th day following the date such notice was sent, the Brand Companies shall terminate the employment of all such employees on the Closing Date and pay, at Purchaser s sole cost and expense, such employees in lieu for the balance of the applicable WARN notice period, if any, and (ii) those Brand Employees who are either Designated Employees or Transferred Employees that do not work in and are not based out of the Colorado Facility, notice of the Contemplated Transactions and that, as a result of the consummation of the Contemplated Transactions, the employment of such Designated Employees and Transferred Employees will be terminated by the applicable Brand Company effective on the Closing Date (which notice for the Transferred Employees shall reflect that it is anticipated that they shall receive offers of employment from the Licensees, with such employment to commence immediately following the Closing Date such that there will be no gap in employment).

- (b) At least three (3) Business Days prior to the Notice Date, Purchaser shall provide Seller with a list of (i) all Brand Employees designated by the Purchaser (the <u>Designated Employees</u>) that, as a result of the consummation of the Contemplated Transactions, are being terminated pursuant to <u>Section 7.08(a)</u> no later than the Closing Date (but not including any Transferred Employees, even if such Transferred Employees do not accept employment with a Licensee); and (ii) all Brand Employees that, as a result of the consummation of the Contemplated Transactions, will receive offers of employment from Licensees (the <u>Transferred Employees</u>) with such employment commencing immediately following the Closing Date.
- (c) Seller shall provide Purchaser with reasonable assistance and cooperation in communicating any further information to the Brand Employees regarding the Contemplated Transactions and the effects thereof upon the Brand Employees, at such time and in such form mutually agreed upon by Seller and Purchaser following the date of this Agreement. Seller shall give Purchaser at least thirty-six (36) hours to review and provide input on any material written communications Seller intends on delivering to Brand Employees regarding the impact of the transactions contemplated by this Agreement, including regarding the treatment of the Company Benefit Plans following the Closing Date. In that regard, Seller shall ensure that copies of all written communications and/or slides for presentations are provided to Purchaser as soon as practicable in advance of the date of intended communication.
- (d) Seller shall be responsible for maintenance and distribution of benefits accrued under any Company Benefit Plan pursuant to and if required by the provisions of such Company Benefit Plans to Brand Employees for the time period ending on or before the Closing Date.
- (e) The Parties shall cooperate in preparing and distributing any notices (such notices, <u>WARN Notices</u>) to the Brand Employees, and to any required state and local authorities, pursuant to the federal Worker Adjustment and Retraining Notification Act (and any similar state or local law) (collectively, <u>WARN Act</u>), which WARN Notices shall be delivered by the applicable Brand Company.
- (f) Purchaser shall be responsible for any Liabilities relating to the WARN Act and the WARN Notices, and any requirement to provide notice pursuant to the WARN Act, which arises out of or results from any termination of employment or other employment loss of any Brand Employee at the direction of Purchaser; *provided* that Seller has

complied with its obligations under <u>Sections 7.08(a)</u> and <u>7.08(d)</u>. Purchaser agrees to promptly reimburse Seller or its Affiliates for all reasonable, documented out-of-pocket costs incurred in preparing, reviewing or distributing any such WARN Notices.

A-42

- (g) Purchaser and its Affiliates shall be solely liable for all benefits (including, but not limited to, severance in accordance with the severance practice maintained by the Brand Companies as of the date hereof), and other liabilities and obligations arising from or relating to the termination of a Designated Employee and a Transferred Employee on the Closing Date, and Seller shall bear no liability or obligation relating to such terminations. Purchaser shall use commercially reasonable efforts to cause its Licensees, pursuant to the terms of their agreements with Purchaser and/or its Affiliates, to provide any Transferred Employee who is terminated by any such Licensee during the twelve (12) months following the Closing Date, severance in accordance with the severance practice maintained by the Brand Companies as of the date hereof. Seller has previously made available to Purchaser a written description of the severance practice maintained by the Brand Companies as of the date hereof, which at Purchaser's requires in order for severance to payable (i) the execution of a valid release and waiver of claims by employees in favor or their employer and (ii) that an employee has not been offered substantially similar employment (same salary and location) in connection with the sale of the business in which they work (regardless of whether the applicable employee actually accepts such employment). For the avoidance of doubt, neither Purchaser nor the Brand Companies shall have any Liability with respect to severance, benefits, or any other amounts payable by a Licensee to any Transferred Employee following the Closing.
- (h) Seller will provide continued health and medical coverage as may be required under Section 4980B of the Code, Part 6 of Subtitle B of Title I of ERISA or any other Applicable Law or ordinance to all eligible current and former Brand Employees and independent contractors (and the spouses, dependents and beneficiaries of any such current or former Brand Employees and independent contractors) with respect to whom a qualifying event (as such term is defined under Sections 4980B(f)(3) of the Code or 603 of ERISA) or other triggering event described under the Applicable Laws or ordinances occurred on or before the Closing Date.
- (i) Purchaser shall use its commercially reasonable efforts to cause its Licensees to provide, to each Transferred Employee for a period of 12 months immediately following the Closing Date (i) no less than the same wage rate or cash salary level in effect for such Transferred Employee immediately prior to the Closing and (ii) employee benefit plans, cash incentive compensation opportunities and fringe benefits that, when taken as a whole, are substantially the same or similar in value, as those in effect immediately prior to the Closing (other than with respect to the value of any matching contribution under Seller s 401(k) Plan or the value of any additional vacation, sick time or personal time off entitlement such employees may have had prior to the Closing Date).
- (j) Purchaser shall use its commercially reasonable efforts to cause its Licensees to, under any plan providing health benefits in which Transferred Employees, as applicable, will participate after the Closing Date, (i) waive any preexisting condition limitations otherwise applicable to such Transferred Employees and their eligible dependents, as applicable, and (ii) subject to Seller providing Purchaser with the applicable information with respect to each Transferred Employee in a form that Purchaser determines is administratively necessary to take into account under its plans or the plans of its Affiliates or Licensees, provide credit to each Transferred Employee, as applicable, for any co-payments, deductibles and out-of-pocket expenses paid by the Transferred Employee during the portion of the relevant plan year including the Closing Date. For eligibility and vesting purposes under any tax-qualified retirement plan sponsored or maintained by the Licensees, Purchaser shall use its commercially reasonable efforts to cause the Licensee to, provide Transferred Employees with past service credit for such Transferred Employees service to Seller, the Company or any of their respective Affiliates, as applicable. The foregoing provisions of this Section 7.08(i) shall be effective only to the extent: (A) such service would have been credited under a comparable Company Benefit Plan, (B) such service would not result in a duplication of benefits, and (C) such service credit would not violate the qualified status of any tax-qualified retirement plan or violate any provision of any Applicable Law. Furthermore, the provisions of this Section 7.08(j) shall not apply as follows: (1) such service credit shall not be given with respect to benefit accruals under any defined benefit plan or retiree medical savings account plan, and (2) such service credit shall not be given with respect to any newly established plan for which prior service is not taken into account for

employees of Seller.

A-43

- (k) Purchaser agrees that it shall use commercially reasonable efforts to cause its Licensees to agree that each Transferred Employee who receives an eligible rollover distribution (within the meaning of Section 402(c)(4) of the Code) from the Seller 401(k) Plan shall be eligible to rollover such distribution (including an in-kind rollover of outstanding notes associated with participant plan loans) to the applicable Licensee 401(k) Plan, in accordance with reasonable policies and procedures adopted by the plan administrators of such plans. Notwithstanding anything to the contrary herein, the Licensees shall not assume any liabilities arising under or relating to the Seller 401(k) Plan. Purchaser agrees that it shall use commercially reasonable efforts to cause the Licensees to (i) to the extent permitted by Applicable Law, provide Transferred Employees with a 90-day grace period with regard to participant loan repayments with the goal of avoiding any deemed distributions in respect of participant loans, and (ii) cause the trustee and plan administrator of the applicable Licensee 401(k) Plan to accept an in-kind rollover of any such participant loans, subject to being reasonably satisfied that such participant loans are properly documented and compliant with Sections 4975(d)(1) and 72(p)(2) of the Code. Seller agrees to cause the unvested components of any and all accounts of Transferred Employees under the Seller 401(k) Plan to automatically vest on the Closing Date.
- (I) On or prior to the Closing Date (as set forth in <u>Section 8.03(d)</u>), Purchaser shall deliver, or cause to be delivered, to the Payroll Provider the Designated Employee Accrued PTO Amount, the Severance Amount and the WARN Amount, in each case in accordance with Section 8.03(d) hereof.
- (m) On or prior to the Closing Date (as set forth in <u>Section 8.02(e)</u>), Seller shall deliver, or cause to be delivered, to the Payroll Provider (i) the Transferred Employee Accrued PTO Amount in accordance with <u>Section 8.02(e)</u> and (ii) \$500,000 in respect of certain severance obligations of Seller and the Brand Companies.

7.09 D&O Tail Policy & Related Matters. Prior to the Closing Date, Seller may, or may cause the Brand Companies to, purchase a six year tail policy of liability insurance that covers events occurring on or prior to the Closing (the Policy). Notwithstanding anything to the contrary contained herein, the cost of such Tail Policy shall constitute a Company Transaction Expense. Purchaser shall not, and shall not permit any Brand Company to, amend, repeal or modify in a manner adverse to the beneficiary thereof any provision in such Brand Company s Organizational Documents relating to exculpation or indemnification of current (*i.e.*, as of the date hereof and as of the Closing) and former officers and directors of any Brand Company (collectively, the Brand Directors) as in effect as of the date hereof, it being the intent of the Parties that the Brand Directors prior to the Closing shall continue to be entitled to such exculpation and indemnification to the fullest extent permitted under such Organizational Documents. Seller shall not amend, repeal or modify any of its Organizational Documents relating to exculpation or indemnification in a manner adverse to the Brand Directors, it being the intent of the Parties that the Brand Directors shall continue to be entitled to such exculpation and indemnification to the fullest extent permitted thereunder.

7.10 Tax Matters.

(a) In the case of Taxes arising in a taxable period of any Brand Company that begins on or before and ends after the Closing Date (the <u>Straddle Period</u>) (i) the allocation of income Taxes or other Taxes based upon the income or receipts of (or consideration, interest or other compensation paid by) the Brand Companies between the Pre-Closing Period and the Post-Closing Period shall be made on the basis of an interim closing of the books as of the close of business on the Closing Date to the extent that there is no actual closing of the taxable year on such Closing Date; (ii) the amount of ad valorem and other Taxes of the Brand Companies attributable to the Pre-Closing Period shall equal the amount of such Tax for the entire taxable period (excluding extraordinary items attributable to actions taken by Purchaser or a Brand Company after the Closing Date that is outside the ordinary course of business of the applicable Brand Company) multiplied by a fraction, the numerator of which is the number of days in the taxable period; and (iii) the remainder of the ad valorem and of the other Taxes of the Brand Companies shall be attributable to the Post-Closing

Period. To the extent any Brand Company is

A-44

permitted but not required under applicable state, local, or foreign income Tax laws to treat the Closing Date as the last day of a taxable period, then the Parties shall treat the Closing Date as the last day of a taxable period.

- (b) Seller shall prepare and file, or cause to be prepared and filed, all Tax Returns of the Brand Companies for all periods ending on or prior to the Closing Date, including, without limitation, Tax Returns for such periods which are to be filed after the Closing Date. Purchaser shall not amend or revoke any Tax Returns (or any elections relating thereto) filed or to be filed by Seller hereunder without Seller s prior written consent which consent shall not be unreasonably conditioned, delayed or withheld. For any Tax Returns with respect to the Brand Companies that is the responsibility of Seller under this Section 7.10(b), Seller shall (i) prepare and file such Tax Returns in a manner consistent with the past practice of the Brand Companies, as the case may be, unless otherwise required by Applicable Law, and (ii) deliver to Purchaser for its review, comment and consent (which consent shall not be unreasonably withheld, conditioned or delayed) a copy of such proposed Tax Return at least 20 days prior to the due date (giving effect to any validly obtained extension) thereof. Purchaser shall prepare and file, or cause to be prepared and filed, all Tax Returns with respect to the Brand Companies for all Tax periods ending after the Closing Date including any Straddle Periods. For any Tax Return of any of the Brand Companies with respect to a Straddle Period that is the responsibility of Purchaser under this Section 7.10(b), Purchaser shall (i) prepare and file such Tax Returns in a manner consistent with the past practice of the Brand Companies, as the case may be, unless otherwise required by Applicable Law, and (ii) deliver to Seller for its review and comment a copy of such proposed Tax Return (accompanied, where appropriate, by a calculation of the allocation between the Pre-Closing Period and the Post-Closing Period of the Taxes shown to be due on such Tax Return) at least 20 days prior to the due date (giving effect to any validly obtained extension) thereof and Purchaser shall, in good faith, consider any comments from Seller. Seller shall pay (i) to the relevant Governmental Entity any amount due and payable on any Tax Return for all periods ending on or prior to the Closing Date (unless paid to the Purchaser pursuant to clause (ii) or (iii)), (ii) to Purchaser all Taxes attributable to Pre-Closing Periods due with respect to any Tax Return of any of the Brand Companies with respect to a Straddle Period that is the responsibility of Purchaser under this Section 7.10(b), and (iii) to Purchaser any amount due and payable on any Tax Return of any of the Brand Companies for all Pre-Closing Periods which are to be filed after the Closing Date, that Purchaser is required under Applicable Law to file, Payments made pursuant to clause (i) of the preceding sentence shall be paid on or before the due date of the Tax Return on which such Taxes are reported and payments made pursuant to clauses (ii) or (iii) of the preceding sentence shall be paid on or before three (3) Business Days prior to the due date of the Tax Return on which such Taxes are reported.
- (c) Purchaser shall have the right to control any Tax audits, Tax disputes or administrative, judicial or other proceedings (<u>Tax Controversies</u>) with respect to the Brand Companies; *provided, however*, that in the case of any Tax Controversy that relates to any Taxes that could give rise to a liability of Seller under this Agreement and to the extent such Taxes are not covered by the Representation and Warranty Insurance Policy, Seller may participate in such Tax Controversy at Seller s expense, and Purchaser shall not enter into any agreement that represents a final determination of the matter without Seller s prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed; provided, further, that in the case of any Tax Controversy that relates to Taxes that could give rise to Liability of Seller under this Agreement and that could not reasonably be expected to adversely impact the Tax Liability of Purchaser, Seller shall have the right to control such Tax Controversy, in which case, Seller shall not enter into any agreement that represents a final determination of the matter without Purchaser s prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed, and Purchaser may participate in such Tax Controversy at Purchaser s expense. Seller shall execute, or cause to be executed, powers of attorney or other documents reasonably necessary to enable Purchaser to take all actions necessary in connection with any Tax Controversies. Purchaser shall have the authority to represent the interests of the Company with respect to any Tax Controversy related to the Section 338(h)(10) Election; *provided*, *however*, that Purchaser shall keep Seller reasonably informed of such Tax Controversy and shall allow Seller to make comments to Purchaser regarding the conduct of or positions taken by Purchaser in such Tax Controversy; provided, further, that Purchaser shall not consent to any

settlement of any such Tax Controversy without the prior written consent of Seller, such consent not to be unreasonably withheld, conditioned or delayed.

A-45

- (d) If, after the Closing Date, Purchaser, any Brand Company or any Affiliate of Purchaser receives any document with respect to the Tax matters of the Brand Companies, the Purchased Interests, or the assets, operations or activities of the Brand Companies that could affect Seller, or Seller or any Affiliate of Seller receives any document with respect to the Tax matters of the Brand Companies, the Purchased Interests, or the assets, operations or activities of the Brand Companies, the Party receiving such document shall supply a copy of such document to the potentially affected Party within seven days of receipt. For this purpose, such Tax documents shall include requests for information, notices of proposed adjustments, revenue agent s reports or similar reports and notices of deficiency. Any information provided or obtained under this Section 7.10(d) shall be kept confidential, except as may otherwise be necessary in connection with the filing of a Tax Return, refund claims, Tax audits, Tax claims or Tax Controversies or as required by Applicable Law.
- (e) Any refund of Taxes (including any interest thereon) that relates to the Brand Companies and that is attributable to a Post-Closing Period, shall be the property of Purchaser and shall be retained by Purchaser (or promptly paid by Seller to Purchaser if any such refund (or interest thereon) is received by Seller). If after the Closing Date, any Brand Company or any member of Purchaser's consolidated Tax group receives a refund of any Tax that relates to such Brand Company and is attributable to a Pre-Closing Period, then Purchaser shall promptly pay or cause to be paid to Seller the amount of such refund, together with any interest paid thereon. Any refund of Taxes (including any interest thereon) for a Straddle Period shall be allocated between the Pre-Closing Period and the Post-Closing Period in accordance with Section 7.10(a). Seller shall repay to Purchaser the amount paid over pursuant to this Section 7.10(e) (plus any penalties, interest or other charges imposed by the relevant Governmental Entity) in the event that Purchaser is required to repay such refund to such Governmental Entity. Notwithstanding the foregoing, all such refunds shall be reduced by the reasonable out-of-pocket costs and expenses (including reasonable external legal and advisory costs) incurred solely in connection with obtaining such refund.
- (f) After the Closing, Seller and Purchaser shall, each at its own expense, cooperate with each other and with each other s Representative, in connection with the preparation or audit of any Tax Return, refund claim or Tax Controversy matter with respect to the activities of the Brand Companies, and such cooperation shall include the retention and (upon the other Party s reasonable request) the provision of records and information, including work papers (but excluding records and information that are protected by recognized professional privilege) which are reasonably relevant to any Tax Returns, claims for refund, Tax Controversies or other action with respect to the activities of the Brand Companies, and shall include making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Purchaser and Seller agree (i) to retain all books and records with respect to Tax matters pertinent to the activities of the Brand Companies relating to the six year period (or portion thereof) prior to the Closing Date, and (ii) to give the other Party reasonable written notice prior to destroying or discarding any such books and records and, if the other Party so requests, Purchaser or Seller, as the case may be, shall allow the other Party to take possession of such books and records.
- (g) Seller and Purchaser shall join in making an election under Section 338(h)(10) of the Code (and any corresponding equivalent elections under state or local law) (a <u>Section 338(h)(10) Election</u>) with respect to the Contemplated Transaction under this Agreement (i.e. the acquisition of the stock of Gaiam Americas for U.S. federal income Tax purposes) such that Purchaser is treated as acquiring the assets of Gaiam Americas for U.S. federal income, state, local and/or franchise tax purposes. Except as otherwise provided herein, Seller shall include in all relevant Tax calculations and filings any income, gain, loss, deduction, or other Tax item resulting from the Section 338(h)(10) Election, including (i) any Tax imposed under Treasury Regulations Section 1.338(h)(10)-1(d)(2), or (ii) any state or local Tax imposed. In connection with the Section 338(h)(10) Election (or as soon thereafter as reasonably practicable), the Parties shall file all federal, state, local and other forms necessary to make the Section 338(h)(10) Election jointly and shall make any required filings and take any and all other reasonable actions necessary or appropriate to effect the Section 338(h)(10) Election. Seller shall include in Seller s income Tax Returns for the

taxable period, which includes the Closing Date, any forms that are required to be so included on account of the Section 338(h)(10) Election. Seller and Purchaser shall cooperate

A-46

fully, and in good faith, with each other in making the Section 338(h)(10) Election. No later than ninety (90) days after the Closing Date, Purchaser and Seller shall sign and date IRS Form 8023 or a signature schedule attached thereto, shall provide their taxpayer identification numbers, shall sign any corresponding state or local forms required in connection with the Section 338(h)(10) Election, and shall provide any other information required by Form 8023 and any corresponding state or local form. All such forms to be signed shall be prepared by Purchaser. Purchaser shall retain and promptly file the executed IRS Form 8023 and corresponding state and local forms and shall provide to Seller a copy of each such form. Purchaser shall not make an election under Section 338(g) of the Code with respect to Modern Lotus, Ltd. and Gaiam Limited (the <u>338(g) Subsidiaries</u>) without the written consent of Seller, which consent may be withheld in its sole discretion.

- (h) The Parties shall comply with (and not waive) the provisions of any so-called bulk transfer law or similar laws of any jurisdiction in connection with transactions contemplated hereunder.
- 7.11 Allocation. The Purchase Price and the Liabilities of the Company and Gaiam Americas (plus other relevant items, including any amount treated as having been paid for the assets of the Company and Gaiam Americas under Section 338 of the Code) shall be allocated among the assets owned by the Company and Gaiam Americas in accordance with Section 338 of the Code and the Treasury Regulations thereunder. Such allocation (the Allocation) shall be adjusted after the Closing to the extent necessary to reflect any adjustments to the Purchase Price pursuant to Section 2.05 and Section 2.06 in the manner determined by the Parties. Purchaser shall deliver to Seller the Allocation within sixty (60) days after the Closing Date. Purchaser and Seller shall make good faith efforts to agree on the Allocation. If Purchaser and Seller are able to agree upon such Allocation, each of Purchaser, Seller the Company and the Brand Companies shall prepare and file IRS Form 8883 (and any applicable forms required under state or local Applicable Law) and supplementary forms in respect of the Section 338(h)(10) Election in a timely manner consistent with such Allocation and shall file all Tax Returns (including amended returns and claims for refund) and information reports in a manner consistent with such Allocation. Seller and Purchaser shall each provide the other with a copy of the IRS Form 8883 it files and any corresponding forms required under state or local Applicable Law. If Purchaser and Seller are unable to agree upon the Allocation, the Parties shall submit the matter to the Accounting Referee to be resolved in a manner similar to the process described in Section 2.06(c).

7.12 Expenses; Transfer Taxes.

- (a) Whether or not the Closing takes place, except as otherwise specifically provided herein, all Transaction Expenses shall be paid by the Party incurring such expense; *provided, however*, that (i) the fees and expenses of the Accounting Referee, if applicable, shall be paid or reimbursed in accordance with Section 2.06(c)(iii), and (ii) the filing fee payable upon the filing of a Notification and Report Form for Certain Mergers and Acquisitions under the HSR Act and any comparable Antitrust Laws shall be borne by fifty percent (50%) by Seller and fifty percent (50%) by Purchaser.
- (b) Any transfer Taxes, sales Taxes, stamp duties, filing fees, registration fees, recordation expenses, or other similar taxes, fees, charges or expenses incurred by Seller or any other party in connection with the Contemplated Transactions, including any transfer Taxes related to the Section 338(h)(10) Election, shall be borne fifty percent (50%) by Seller and fifty percent (50%) by Purchaser.
- **7.13 Dividends or Distributions.** Prior to Closing, Seller shall be permitted to cause any Brand Company to (a) declare or pay a dividend or distribution to the equityholders of such Brand Company from any Cash on Hand and (b) use any Cash on Hand to pay Company Transaction Expenses.

7.14 Financing.

(a) Subject to the terms and conditions of this Agreement, Purchaser shall use its best efforts to obtain the Debt Financing on the terms and conditions described in the Debt Commitment Letters and any related Fee

A-47

Letters. Purchaser shall not permit any amendment, modification or supplement to be made to, or any waiver of any provision or remedy under, the Debt Commitment Letters or any related Fee Letter, if such amendment, modification or waiver (i) reduces the aggregate amount of the Debt Financing (including by changing the amount of fees to be paid or original issue discount), when taken together with Purchaser s cash on hand, to an amount below the amount required to satisfy Purchaser s obligations under this Agreement, or (ii) imposes new or additional conditions or other terms or otherwise expands, amends or modifies any of the conditions to the receipt of the Debt Financing or other terms in a manner that could reasonably be expected to delay in any material respect or prevent the Closing; provided, however, that Purchaser may, after consultation with Seller, replace or amend the Debt Commitment Letters to add lenders, lead arrangers, bookrunners, syndication agents or similar entities who had not executed the Debt Commitment Letters as of the date hereof so long as such action would not reasonably be expected to materially delay or prevent the Closing; provided, further, that the exercise of any flex provisions in the Fee Letter shall not be considered an expansion, amendment, or modification of any of the conditions to the receipt of the Debt Financing or other terms in a manner that could reasonably be expected to delay in any material respect or prevent the Closing. Any reference in this Agreement to (x) Debt Financing shall include the financing contemplated by the Debt Commitment Letters as amended or modified in compliance with this Section 7.14 and (y) Fee Letter and Debt Commitment Letters shall include such documents as amended or modified in compliance with this Section 7.14.

- (b) Purchaser shall use its best efforts to (i) maintain in effect the Debt Commitment Letters in accordance with the terms and subject to the conditions thereof; (ii) negotiate and enter into all definitive agreements in form and substance reasonably satisfactory to Purchaser with respect to the Debt Financing contemplated by the Debt Commitment Letters on the terms and conditions set forth in the Debt Commitment Letters and any related Fee Letter; (iii) satisfy, to the extent within its control, all conditions to such definitive agreements that are applicable to Purchaser and consummate the Debt Financing at or prior to the Closing; and (iv) comply with its obligations under the Debt Commitment Letters and any related Fee Letter. Upon request (which request may be made telephonically or in electronic mail), Purchaser shall keep the Company and Seller reasonably informed on a reasonably current basis and in reasonable detail of the status of its efforts to arrange the Debt Financing and any material developments relating thereto and provide to Seller copies of all definitive documents related to the Debt Financing, Without limiting the generality of the foregoing, Purchaser shall give the Company and Seller prompt written notice (and in any event within two (2) Business Days) of (A) any breach or default (or any event or circumstance that with or without notice, lapse of time, or both, could reasonably be expected to give rise to any breach or default) or the threatened breach by any party to the Debt Commitment Letters, Fee Letter or any other definitive document related to the Debt Financing of which Purchaser becomes aware, (B) the receipt by Purchaser of any notice or other communication from any Debt Financing source with respect to any (1) breach, default, termination or repudiation by any party to the Debt Commitment Letters, Fee Letter or any other definitive document related to the Debt Financing of any provisions of the Debt Commitment Letters, Fee Letter or any other definitive document related to the Debt Financing or (2) dispute or disagreement between or among any parties to the Debt Commitment Letters, Fee Letter or any other definitive document related to the Debt Financing, and (C) if for any reason Purchaser reasonably believes that it may not be able to obtain on a timely basis all or any portion of the Debt Financing on the terms, in the manner or from the sources contemplated by the Debt Commitment Letters or the definitive documents relating to the Debt Financing. As soon as reasonably practicable, but in any event within two (2) Business Days of the date the Company or Seller delivers to Purchaser a written request, Purchaser shall provide any information reasonably requested by the Company or Seller relating to any circumstance referred to in clause (A), (B) or (C) of the immediately preceding sentence.
- (c) Purchaser shall use its best efforts to cause the lenders and any other Persons providing Debt Financing to fund on the Closing Date the Debt Financing required to consummate the transactions contemplated by this Agreement and the other transactions contemplated by the Debt Commitment Letters, if all conditions to Closing set forth in Article VIII herein are satisfied or waived (other than those conditions that by their terms are to be satisfied or waived at the

Closing, but subject to the fulfillment or waiver of those conditions).

A-48

- (d) Nothing herein shall obligate Purchaser or any of its Affiliates to (i) waive any terms or conditions of this Agreement or of the Debt Commitment Letters, (ii) pay any additional material fees or amounts beyond those expressly set forth in the Debt Commitment Letters to cause the lenders and any other Persons providing Debt Financing to fund the Debt Financing, or (iii) commence any Proceeding against any financing institution or other Person that fails to provide its respective portion of the Debt Financing under the Debt Commitment Letters or New Debt Commitment Letter.
- (e) If any Debt Commitment Letter shall be terminated or modified in a manner adverse to Purchaser (other than as a result of a breach of Seller s obligations hereunder or the exercise of flex provisions in the Fee Letter), or if any portion of the Debt Financing becomes unavailable on the terms and conditions contemplated in the Debt Commitment Letters (other than as a result of obtaining substitute debt financing in accordance with Section 7.14(e)), Purchaser shall use its best efforts to arrange to promptly obtain alternative financing from alternative sources on economic terms not materially less favorable to Purchaser than the terms of the Debt Commitment Letters and on non-economic terms not materially less favorable to Purchaser than the terms of the Debt Commitment Letter, in each case, taken as a whole, and in an amount which, together with its cash on hand, shall be sufficient (subject to the fulfilment of the conditions set forth in this Agreement) to consummate the Contemplated Transactions (the <u>Alternate Debt</u> Financing), and to obtain a new financing commitment letter with respect to such Alternate Debt Financing (the New Debt Commitment Letter), which shall replace the existing Debt Commitment Letter, a true and complete copy of which (together with any related fee or other letter) shall be promptly provided to Seller (subject to redaction so long as such redaction does not cover terms that would adversely affect the conditionality or availability of the Debt Financing). In the event any New Debt Commitment Letter is obtained, (i) any reference in this Agreement to the Debt Financing shall mean the debt financing contemplated by the Debt Commitment Letter as modified pursuant to clause (ii) below, (ii) any reference in this Agreement to the Debt Commitment Letter shall be deemed to include any Debt Commitment Letter that is not superseded by a New Debt Commitment Letter at the time in question and any New Debt Commitment Letter to the extent then in effect and (iii) any reference in this Agreement to Fee Letter shall be deemed to include any fee or other letter relating to the Debt Commitment Letter that are not superseded by a New Debt Commitment Letter at the time in question and any New Debt Commitment Letter to the extent then in effect.

7.15 Financing Cooperation.

- (a) Following the date hereof and prior to the Closing, Seller shall, and shall cause each of the Brand Companies and their respective Representatives and employees to provide to Purchaser such cooperation reasonably requested by Purchaser in connection with the Debt Financing, including:
- (i) causing appropriate officers and employees of Seller and the Brand Companies to be available on a customary basis to meet with prospective lenders, rating agencies and investors in presentations, meetings, road shows and due diligence sessions and assist with the preparation of projections and other pertinent information in form and substance not previously provided to Purchaser as reasonably requested by Purchaser s lender(s) in connection with the Debt Financing;
- (ii) cooperating with prospective lenders due diligence, to the extent customary and reasonable, including, but not limited to, the delivery of Organizational Documents of the Brand Companies and lien searches, and reasonably cooperating with Purchaser to facilitate an appraisal of the Intellectual Property owned by the Brand Companies, each as contemplated by the Debt Commitment Letters;
- (iii) executing and delivering into escrow any credit agreement, customary and reasonable pledge and security documents, guarantee documents, supplemental indenture, currency or interest hedging arrangements, other definitive financing documents, or other certificates, or other documents with respect to the Debt Financing contemplated by the

Debt Commitment Letters as may be reasonably requested by Purchaser or otherwise reasonably facilitating the pledging of collateral;

A-49

- (iv) furnishing Purchaser and its sources of Debt Financing as promptly as practicable financial statements with respect to the period following the date hereof as reasonably required by such sources and not previously provided to Purchaser by Seller;
- (v) furnishing Purchaser, within the time period specified in the Debt Commitment Letters or as otherwise reasonably requested by the sources of the Debt Financing with all documentation and other information required by regulators and authorities under applicable know your customer and anti-money laundering and regulations, including the USA Patriot Act of 2001;
- (vi) taking such reasonable actions as may be required, proper or advisable, as requested by Purchaser, to facilitate the guarantee and pledge of collateral to secure the Debt Financing, including cooperating in connection with the pay-off of existing Indebtedness and the release of Liens related thereto; and
- (vii) using commercially reasonable efforts to obtain customary pay-off letters, Lien releases, instruments of termination, waivers, consents, and approvals from other parties to Contracts and Liens to which the Brand Companies are a party or by which any of them or their assets or properties are bound or subject;

provided, that (1) nothing herein shall require such cooperation to the extent it would require the Company or its Affiliates to waive or amend any terms of this Agreement or to agree to pay any fees, reimburse any expenses or give any indemnities prior to the Closing for which it has not received prior reimbursement by or on behalf of Purchaser or for which it has not otherwise agreed to pay or reimburse under this Agreement, or to give any indemnities or incur any liabilities other than reimbursements and liabilities of the Brand Companies that are effective only after the Closing, (2) nothing herein shall require such cooperation from the Company or its Affiliates to the extent it would unreasonably interfere with or disrupt the ongoing operations of the Company or its Affiliates, (3) no action, liability or obligation of the Company under any certificate, agreement, arrangement, document or instrument relating to the Debt Financing shall be effective until after the Closing and (4) neither the Company nor any of its Affiliates shall be required to issue any offering documents, private placement memoranda, bank information memoranda, prospectuses and similar documents required in relation to the Debt Financing, but such documents may contain disclosure and financial statements reflecting the Company as the obligor following the Closing.

- (b) Purchaser shall promptly, upon request by the Company or Seller, reimburse Seller for all documented and reasonable out-of-pocket costs and expenses (including reasonable attorneys fees) incurred by the Company, Seller or any of their respective Affiliates or Representatives in connection with the cooperation of the Company and any other Brand Company contemplated by this <u>Section 7.15</u>, and the foregoing obligations shall survive termination of this Agreement.
- (c) In no event shall Seller, the Company, any Brand Company or any of their Affiliates be in breach of this Agreement because of the failure by the Company or any Brand Company to deliver any financial or other information that is not currently readily available to the Company or any Brand Company on the date hereof or is not otherwise prepared in the ordinary course of its business at the time requested by Purchaser or for failure to obtain any review of any financial or other information by its accountants.

7.16 Restrictive Covenants of Seller.

(a) Non-Disparagement. Seller agrees not to (and shall use its commercially reasonable efforts to cause, its executive management team and the executive management team of Gaia, Inc. not to) make or publish any derogatory or disparaging written, oral, or electronic statements about (i) the Brand Companies or any of their predecessors, stockholders and the present and former officers, directors, managers, employees, owners, members and agents of the

foregoing, individually and in their official capacities, or (ii) the Brand Business or any products, services, practices, or operations of any of the Brand Companies. Notwithstanding the covenants

A-50

contained in Section 7.16(a), none of Seller or its executive management team or the executive management team of Gaia, Inc. shall be prohibited or restricted from (x) making or publishing any such statements that such Person reasonably believes in good faith to be necessary in responding to or initiating a bona fide claim, demand or dispute involving such Person and the Brand Companies and is otherwise not prohibited by the terms hereof or (y) answering truthfully if compelled to do so in a deposition, lawsuit or similar dispute resolution proceeding.

- (b) Non-Competition; Non-Interference. Seller agrees (i) that for the two-year period commencing on the Closing Date, Seller shall not, and shall each cause its Subsidiaries not to (A) engage in or in any way, directly or indirectly, own, manage, operate, control, consult with, provide products and/or services to or participate in the ownership, management, operation or control of, provide financing to or otherwise advise or assist or be actively connected with, directly or indirectly, any enterprise or business which is of the type or character that engages in, or otherwise carries on, any business activity which in any other manner is in competition with the Brand Business, except that Seller may own, in the aggregate, 5% or less of the stock of any corporation which is listed on a national stock exchange or actively traded in the over-the-counter market; (B) solicit or attempt to solicit (other than on behalf of any Brand Company) any Person that is, or within two (2) years before the Closing Date was, a client, customer, supplier, licensee or business relation of the Brand Business, or who Seller, any Brand Company or any of their respective Subsidiaries solicited to be a client, customer, supplier or licensee of the Brand Business during such two (2)-year period, or induce or attempt to induce any such Person to cease, reduce or not commence doing business with any of the Brand Companies (or assist any other Person in engaging in any such activities); or (C) knowingly and intentionally interfere in any way with the relationship between any Brand Company and any Person that is or was a client, lender, investor, customer, supplier, licensee or other business relation of the Brand Business (or assist any other Person in engaging in any such activities); provided, however, that in no event shall Seller or any of their respective Affiliates be in violation of this Section 7.16(a) as a result of Seller or any of its respective Affiliates or Subsidiaries engaging in and operating the Gaia Business.
- (c) Non-Solicitation of Employees. From and after the Closing Date, Seller shall not, and shall cause its Affiliates not to, for a period of two (2) years after the Closing Date, hire or knowingly solicit for employment any Transferred Employee; *provided*, that this paragraph shall not preclude Seller or its Affiliates from soliciting for employment any such Transferred Employee by a general solicitation through a public medium or general or mass mailing by or on behalf of Seller or any of its Affiliates that is not targeted specifically at employees of the Brand Companies or the Licensees; *provided*, *further*, that Seller shall not during the now current term of the Boulder Road Lease, as the same may be assigned to a Licensee, hire any Transferred Employee who works primarily out of the Boulder Road location notwithstanding the manner of solicitation, without Purchaser s prior written consent. For the avoidance of doubt, this Section 7.16(c) shall not apply to any services that are provided under the Transition Services Agreement.

(d) Acknowledgement.

(i) It is the desire and intent of the Parties that the provisions of Sections 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) and 7.17(c) shall be enforced to the fullest extent permissible under Applicable Law and public policies applied in each jurisdiction in which enforcement is sought. If any particular provisions or portion of any of Section 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) or 7.17(c) shall be adjudicated by a court of competent jurisdiction to be invalid, illegal or unenforceable, such Section 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) or 7.17(c), as applicable, shall be deemed amended to delete therefrom such provision or portion adjudicated to be invalid, illegal or unenforceable, such amendment to apply only with respect to the operation of Section 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) or 7.17(c), as applicable, in the particular jurisdiction in which such adjudication is made. The Parties agree that the scope, area and duration of the covenants set forth in Sections 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) and 7.17(c) are reasonable. Notwithstanding the foregoing, the Parties agree that if the scope, area or duration of any covenant set forth in Sections 7.16(a), 7.16(c), 7.17(a), 7.17(b) and 7.17(c) shall be adjudicated by a court of competent

jurisdiction to be overly broad or otherwise unreasonable, such court may reduce the scope, area and duration thereof to that which it deems the maximum

A-51

scope, area and duration reasonable under the circumstances, such reduced scope, area and duration to apply only with respect to the operation of Section 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) or 7.17(c), as applicable, in the particular jurisdiction in which such adjudication is made.

- (ii) The Parties recognize that the performance of the obligations under Sections 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) and 7.17(c) is special, unique and extraordinary in character, and that in the event of a breach by either Party of the terms and conditions of any of Section 7.16(a), 7.16(b), 7.16(c), 7.17(a), 7.17(b) or 7.17(c) to be performed by such Party, the other Party has no adequate remedy at law for any such breach, and therefore, the Parties agree that, each Party and the Brand Companies shall, in addition to the other remedies that may be available to either of them under this Agreement at law (including to obtain damages for any breach of Sections 7.16(a), 7.16(c), 7.16(c), 7.17(a), 7.17(b) and 7.17(c)), be entitled, if they so elect, to enforce the specific performance thereof by such Party or to enjoin any employees of such Party or any of its Affiliates from performing services for any Person.
- (iii) Seller agrees that the covenants and agreements of Seller set forth in Sections 7.16(a), 7.16(c) and 7.16(c) are independent of all other covenants, representations, warranties and agreements of the Parties set forth in this Agreement, and no default, breach or failure to perform by any Party shall constitute an excuse or other justification for Seller to fail to observe fully its covenants and agreements under Sections 7.16(a), 7.16(c) and 7.16(c). No course of dealing between Purchaser and Seller, and no delay by Purchaser in exercising any right, power or remedy under this Section 7.16(d), in equity or at law, shall constitute a waiver of, or otherwise prejudice, any such right, power or remedy.
- (iv) Purchaser agrees that the covenants and agreements of Purchaser set forth in Sections 7.17(a), 7.17(b) and 7.17(c) are independent of all other covenants, representations, warranties and agreements of the Parties set forth in this Agreement, and no default, breach or failure to perform by any Party shall constitute an excuse or other justification for Purchaser to fail to observe fully its covenants and agreements under Sections 7.17(a), 7.17(b) and 7.17(c). No course of dealing between Purchaser and Seller, and no delay by Seller in exercising any right, power or remedy under this Section 7.16(d), in equity or at law, shall constitute a waiver of, or otherwise prejudice, any such right, power or remedy.

7.17 Restrictive Covenants of Purchaser and Parent.

- (a) Non-Disparagement. Each of Parent and Purchaser agrees not to (and shall direct, and use its commercially reasonable efforts to cause, its respective executive management team not to) make or publish any derogatory or disparaging written, oral, or electronic statements about (i) Seller or any of its predecessors, stockholders and the present and former officers, directors, managers, employees, owners, members and agents of the foregoing, individually and in their official capacities, or (ii) the Gaia Business or any products, services, practices, or operations of Seller or any of its Subsidiaries. Notwithstanding the covenants contained in this Section 7.17(a), none of Purchaser or Parent or their respective executive management teams shall be prohibited or restricted from (x) making or publishing any such statements that such Person reasonably believes in good faith to be necessary in responding to or initiating a bona fide claim, demand or dispute involving such Person and Seller or any of its Subsidiaries and is otherwise not prohibited by the terms hereof or (y) answering truthfully if compelled to do so in a deposition, lawsuit or similar dispute resolution proceeding.
- **(b) Non-Competition; Non-Interference.** Each of Parent and Purchaser hereby agrees that for the two (2)-year period commencing on the Closing Date, Parent and Purchaser shall not, and shall cause their respective Subsidiaries not to, within the continental United States or any other jurisdiction or marketing area in which the Gaia Business operates, (i) engage in or in any way, directly or indirectly, own, manage, operate, control, consult with, provide services to or participate in the ownership, management, operation or control of, provide financing to or otherwise

advise or assist or be actively connected with, directly or indirectly, any enterprise or business which is of the type or character that engages in, or otherwise carries on, any business activity which in any other manner both (x) is in competition with the Gaia Business and (y) utilizes the Gaiam brand and related

A-52

trademarks, except that Parent or Purchaser may own, in the aggregate, 5% or less of the stock of any corporation which is listed on a national stock exchange or actively traded in the over-the-counter market; or (ii) knowingly and intentionally interfere in any way with the relationship between the Gaia Business and any Person that is or was a client, lender, investor, customer, supplier, licensee or other business relation of the Gaia Business (or assist any other Person in engaging in any such activities); provided, however, that in no event shall Purchaser, any Brand Company or any of their respective Affiliates be in violation of this Section 7.17(b) as a result of Purchaser, any Brand Company or any of their respective Affiliates or Subsidiaries engaging in and operating the Brand Business in the manner that such business is conducted by Seller on the Closing Date.

- (c) Non-Solicitation of Employees. From and after the Closing Date, Parent, Purchaser and the Brand Companies shall not, and shall cause their respective Affiliates not to, for a period of two (2) years after the Closing Date, hire or knowingly solicit for employment any employee of the Gaia Business; *provided*, that this paragraph shall not preclude Parent, Purchaser, the Brand Companies or their respective Affiliates from soliciting for employment any such employee by a general solicitation through a public medium or general or mass mailing by or on behalf of Parent, Purchaser, the Brand Companies or any of their respective Affiliates that is not targeted specifically at employees of the Gaia Business; *provided further* that Parent and Purchaser shall not during the now current term of the existing Boulder Road Lease hire any employee of the Gaia Business who works primarily out of the Boulder Road location notwithstanding the manner of solicitation, without Seller s prior written consent. For the avoidance of doubt, this Section 7.17(c) shall not apply to any services that are provided under the Transition Services Agreement.
- **7.18 Intellectual Property Title Matters.** To the extent that any Registered Company Intellectual Property is not in the legal name of the Brand Companies, is subject to a chain of title defect, or is subject to any form of Lien (other than (i) the Permitted Liens, and (ii) non-exclusive licenses granted in the ordinary course of business consistent with past practice), Seller shall, and shall use its reasonable best efforts to cause each of its applicable Subsidiaries to, prior to and following the Closing, at Seller s sole expense, prepare, execute and file for record such instruments and documents necessary to cure such title defects, provide for current ownership of the asset by the Brand Companies and remove any and all such Liens in such Company Intellectual Property following the Closing (all such actions, the <u>IP Title Corrective Actions</u>); provided that, to the extent any IP Title Corrective Actions are not completed prior to the Closing, Seller shall use its reasonable best efforts to complete all IP Title Corrective Actions promptly following the Closing. Seller shall deliver all documentation that is filed concerning the IP Title Corrective Actions to Purchaser promptly following the IP Title Corrective Actions are taken.
- **7.19** Use of Company Intellectual Property. From and after the Closing, unless otherwise agreed to in writing by the Parties, Seller shall not, and shall cause its Subsidiaries not to, use any of the Company Intellectual Property, including any Trademarks confusingly similar to the Trademarks used on or in connection with the Brand Business and/or Brand Companies (the <u>Company Trademarks</u>) except for references in historical, Tax, and similar records, other permissible fair use and as otherwise required by Applicable Law or as set forth in the IP License Agreement; *provided*, that that Seller shall change its name (and amend its certificate of incorporation to reflect such change) to a name that does not use any of the Company Intellectual Property or any Company Trademarks promptly as practicable after the Closing, but in no event later than 15 days after the Closing.
- **7.20 Shareholder Consent; Information Statement.** Seller shall use its best efforts to obtain the Shareholder Consent as promptly as practicable after the execution and delivery of this Agreement, including exercising any rights Seller may have under any voting agreements or similar arrangements that Seller and any of its shareholders is a party to. Within five (5) Business Days following the receipt of the Shareholder Consent, Seller shall prepare and file with the SEC the Information Statement. Seller shall promptly notify Purchaser of the receipt of any comments of the SEC with respect to the Information Statement and of any requests by the SEC for any amendment or supplement thereto or for additional information, and shall promptly provide to Purchaser copies of all correspondence between Seller or any

Representative of Seller and the SEC with respect to the Information Statement. Seller shall (i) give Purchaser and its counsel the opportunity to review and comment on the Information Statement and all responses to requests for additional information by, and replies to

A-53

comments of, the SEC, (ii) take into good faith consideration all comments reasonably proposed by Purchaser and (iii) not file such document with the SEC prior to providing Purchaser and its counsel a reasonable opportunity to review and comment thereon it being acknowledged and agreed that two (2) days constitutes a reasonable opportunity. Seller will use its commercially reasonable efforts to have the Information Statement cleared by the SEC as promptly as reasonably practicable after its filing and thereafter cause the Information Statement to be mailed to its shareholders as promptly as reasonably practicable (but no more than twenty (20) Business Days) after the Information Statement has been declared effective by the SEC. Purchaser shall furnish all information concerning itself and its Subsidiaries as may be reasonably requested in connection with the preparation, filing and distribution of the Information Statement. Purchaser agrees that it shall reasonably cooperate with and assist Seller, including providing Seller upon request (as promptly as reasonably practicable) with the information concerning Purchaser and its Affiliates, directors and officers required to be included in the Information Statement. If at any time prior to the Closing, any information relating to Seller or Purchaser or any of their respective Affiliates, officers or directors should be discovered by Seller or Purchaser that should be set forth in an amendment or supplement to the Information Statement so that the Information Statement shall not contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party which discovers such information shall promptly notify the other parties hereto and an appropriate amendment or supplement describing such information shall be filed with the SEC, and to the extent required by Applicable Law, disseminated to the shareholders of Seller. Seller shall be solely responsible for the costs and expenses incurred in connection with the preparation, mailing and filing of the Information Statement.

7.21 No Solicitation.

- (a) Alternative Transaction. Seller shall not, and shall cause its Subsidiaries (including the Brand Companies) and its and their respective Representatives not to, directly or indirectly, (i) solicit, initiate or knowingly facilitate, induce or encourage the making of any proposal or offer that constitutes or would reasonably be expected to lead to an Alternative Transaction Proposal, or (ii) enter into, continue or otherwise participate in any discussions or negotiations regarding, or furnish to any Person any information with respect to, or otherwise cooperate in any way that would otherwise reasonably be expected to lead to, any Alternative Transaction Proposal. Without limiting the foregoing, it is agreed that any violation of the restrictions set forth in the preceding sentence by Seller, any of its Subsidiaries (including the Brand Companies) or its or their respective Representatives shall be deemed to be a breach of this Section 7.21(a) by Seller. Seller will, and will cause each of the Brand Companies and each of its and their respective Representatives to, immediately cease and cause to be terminated any and all existing activities, discussions or negotiations with any Person conducted heretofore with respect to any Alternative Transaction Proposal. Seller agrees that it will take all commercially reasonable steps to promptly inform its Representatives of the obligations undertaken in this Section 7.21(a).
- (b) <u>Superior Proposal</u>. Notwithstanding anything to the contrary contained in <u>Section 7.21(a)</u>, in the event that Seller or any Brand Company receives after the date of this Agreement and prior to the termination of this Agreement in accordance with its terms, an unsolicited, bona fide written Alternative Transaction Proposal which the Board determines to be, or to be reasonably likely to result in, a Superior Proposal, Seller may then take the following actions (but only if (x) (A) the Board concludes in good faith, after consultation with its outside legal counsel, that the failure to take the following actions would be inconsistent with its fiduciary duties under Applicable Law, and (B) Seller has given Purchaser prior written notice of its intention to take any of the following actions and of the identity of the Person or group making such Alternative Transaction Proposal and the material terms and conditions of such Alternative Transaction Proposal and (y) Seller shall not have breached any of, and continues to comply with all of, the provisions of this Section 7.21):

(i) furnish nonpublic information with respect to Seller and its Subsidiaries and the Brand Business to the Person or group making such Alternative Transaction Proposal; <u>provided</u>, that prior to furnishing any such nonpublic information, it receives from such Person or group an executed confidentiality agreement containing confidentiality terms at least as restrictive as the terms contained in the Confidentiality Agreement,

A-54

and which shall not contain any exclusivity provision or other term that would restrict, in any manner, Seller s ability to consummate the transactions contemplated by this Agreement; and

- (ii) engage in discussions or negotiations with such Person or group with respect to such Alternative Transaction Proposal.
- (c) Notification. In addition to the obligations of Seller set forth in Sections 7.21(a), 7.21(b), and 7.21(d), as promptly as practicable (and in any event within two (2) Business Days) after receipt of any Alternative Transaction Proposal, Seller shall provide Purchaser with the identity of the Person or group making any such Alternative Transaction Proposal and oral and written notice of the material terms and conditions of such Alternative Transaction Proposal. Seller shall provide Purchaser as promptly as practicable (and in any event within two (2) Business Days) with such information as is reasonably necessary to keep Purchaser informed in all material respects regarding, and the status and terms of, and changes in any such Alternative Transaction Proposal. Seller shall provide Purchaser with two (2) Business Days prior notice (or such lesser prior notice as is provided to the Board) of any meeting of the Board at which the Board is reasonably expected to consider any Alternative Transaction Proposal. Any material amendment (including the form, amount and timing of payment of consideration) to any Alternative Transaction Proposal will be deemed to be a new Alternative Transaction Proposal for the purposes of this Section 7.21.
- (d) Changes of Recommendation. Except as provided in Section 7.21(e) below, neither the Board nor any committee thereof shall (i) (A) withdraw or qualify (or amend or modify in a manner adverse to Purchaser), or fail to make the approval, recommendation or declaration of advisability by the Board or any committee thereof of this Agreement, or the transactions contemplated by this Agreement, (B) recommend, adopt or approve any Alternative Transaction Proposal, (C) publicly make any recommendation in connection with an Alternative Transaction Proposal other than a recommendation against such proposal, or (D) if an Alternative Transaction Proposal shall have been publicly announced or disclosed, if so requested by Purchaser, fail to recommend against such Alternative Transaction Proposal or fail to reaffirm the approval, recommendation and declaration of advisability of this Agreement and the transactions contemplated by this Agreement, on or prior to the fourth (4th) Business Day after the Alternative Transaction Proposal shall have been publicly announced or disclosed (any action described in this clause (i) being referred to as an Adverse Recommendation Change) or (ii) approve or recommend or allow Seller or any of its Affiliates to execute or enter into, any letter of intent, memorandum of understanding, agreement in principle, merger agreement, acquisition agreement, option agreement, joint venture agreement, partnership agreement or other similar agreement, arrangement or understanding (A) relating to any Alternative Transaction Proposal or any offer or proposal that would reasonably be expected to lead to an Alternative Transaction Proposal or (B) requiring it (or that would require it) to abandon, terminate or fail to consummate the transactions contemplated by this Agreement.
- (e) Notwithstanding anything to the contrary set forth in this Agreement, solely in response to a Superior Proposal, the Board may make an Adverse Recommendation Change or terminate this Agreement to enter into a definitive agreement with respect to such Superior Proposal in accordance with Section 9.01(h), only if all of the following conditions in clauses (i) through (iv) are met:
- (i) a Superior Proposal has been made and has not been withdrawn and continues to be a Superior Proposal;
- (ii) Seller shall have (A) provided to Purchaser four (4) Business Days prior written notice, which notice shall state expressly (1) that it has received a Superior Proposal, (2) the material terms and conditions of the Superior Proposal (including the consideration offered therein (and the portion thereof attributable to the Brand Business) and the identity of the Person or group of Persons making the Superior Proposal), and shall have contemporaneously provided copies of the relevant proposed transaction agreements with the Person or group of Persons making such Superior Proposal (the <u>Alternative Acquisition Agreement</u>) (it being understood and agreed that any material amendment to the

financial terms or any other material term of such Superior Proposal shall require a new notice and a new four (4) Business Day period) and (3) that it intends to effect an Adverse

A-55

Recommendation Change and the manner in which it intends to do so or that it intends to terminate this Agreement in accordance with Section 9.01(h) in order to enter into the Alternative Acquisition Agreement, as applicable, and (B) prior to making such Adverse Recommendation Change or terminating this Agreement in accordance with Section 9.01(h), as applicable, to the extent requested by Purchaser, engaged in good faith negotiations with Purchaser during such four (4) Business Day period to amend this Agreement in such a manner that the Alternative Acquisition Agreement ceases to constitute a Superior Proposal;

- (iii) the Board shall have determined in good faith, after consultation with its outside legal counsel, that, in light of such Superior Proposal and taking into account any revised terms offered by Purchaser, the failure to make an Adverse Recommendation Change or terminate this Agreement in accordance with Section 9.01(h), as applicable, is inconsistent with its fiduciary duties under Applicable Law; and
- (iv) Seller shall have complied with <u>Section 7.21(a)</u> and shall not have breached any of the other provisions set forth in this Section 7.21.
- 7.22 Gaiam-FFL APA. Seller shall not, without the prior written consent of Purchaser (which consent may be withheld in its reasonable discretion), terminate, amend, modify or supplement the Gaiam-FFL APA in any material respect. Purchaser and Seller acknowledge and agree that in the event FFL shall fail to close the transactions contemplated by the Gaiam-FFL APA, or otherwise fail to fund the Net Closing Cash Consideration, as defined in the Gaiam-FFL APA on the Closing Date, (a) Seller shall have the right to (i) terminate the Gaiam-FFL APA, and (ii) to cause the Brand Companies to transfer and assign the Acquired Assets, as defined in the Gaiam-FFL APA, to Seller immediately prior to the sale and transfer of the Purchased Interests to Purchaser hereunder, and such transfer and assignment shall not result in any breach of any representation, warranty or covenant of Seller or the Brand Companies hereunder; (b) during the sixty day period following the Closing, Seller and Purchaser will work together in good faith to allow Seller to sell and/or collect such Acquired Assets in the ordinary course of business, and such sale shall not be subject to any restrictive covenants set forth herein; and (c) following the Closing, the Brand Companies shall grant to Seller a limited royalty-free, sublicensable, worldwide license to use, reproduce and display the Material Registered Company Intellectual Property solely as necessary to market and sell and/or collect the Acquired Assets and such license shall be sublicensable to any subsequent purchaser or assignee of the Acquired Assets solely as necessary to market and sell any Acquired Assets that are inventory bearing any Material Registered Company Intellectual Property. For the avoidance of doubt, this limited license is granted conditionally and solely to allow Seller (i) to market and sell Acquired Assets that are inventory and (ii) to sell the entirety of the Acquired Assets to a purchaser thereof, in each case, until such time as Purchaser, Seller and the Brand Companies identify a third party to fill FFL s role as a Licensee and purchase the Acquired Assets.

ARTICLE VIII

CONDITIONS PRECEDENT

- **8.01** Conditions to Obligations of Each Party to Effect the Contemplated Transactions. The respective obligations of each Party to effect the Contemplated Transactions shall be subject to the waiver by such Party or the satisfaction at or prior to the Closing Date of the following conditions:
- (a) All waiting periods (and all extensions thereof) applicable to the Contemplated Transactions under Antitrust Laws set forth on <u>Schedule 8.01(a)</u> shall have terminated or expired and all consents, waivers, approvals, authorizations and orders required under the Antitrust Laws set forth on <u>Schedule 8.01(a)</u> shall have been received.

(b) No temporary restraining order, preliminary, or permanent injunction or other order issued by any Governmental Entity of competent jurisdiction prohibiting or challenging, or seeking to prohibit or challenge, the consummation of any of the Contemplated Transactions shall be in effect.

A-56

- (c) Seller shall have obtained the Shareholder Consent, which Shareholder Consent shall not have been revoked, modified or amended, and remains in full force and effect.
- **8.02** Conditions to Obligation of Purchaser. The obligation of Purchaser to consummate the transactions that are to be consummated at the Closing is subject to the satisfaction, as of the Closing Date, of the following further conditions (any of which may be waived by Purchaser in whole or in part):
- (a) Each of the (i) representations and warranties of Seller set forth in Article IV and Article V (other than the Fundamental Representations) shall be true and correct (without giving effect to any material, materially, materiality, Company Material Adverse Effect, material adverse effect, material adverse change or similar qualifiers contained in any of such representations and warranties other than those contained in <u>Section 5.05</u> or <u>Section 5.07</u>) in all respects, except where the failure to be so true and correct, individually or in the aggregate, does not and would not reasonably be expected to have a Company Material Adverse Effect, in each case, as of the date of this Agreement, the FFL Closing Date (to the extent such date occurs) and as of the Closing Date as though made on and as of the FFL Closing Date and the Closing Date, as applicable (except to the extent that any such representations and warranties are made as of another date, which such representations and warranties shall be true and correct in all respects as of such other date), (ii) representations and warranties of Seller set forth in Sections 4.05 (Judgments), 4.07 (Brokers), 5.02(a) (Capitalization), <u>5.10(b)</u> (Intellectual Property), <u>5.14(b)</u> (Proceedings; Judgments), and <u>5.20</u> (Brokers) shall be true and correct in all material respects as of the date of this Agreement and as of the FFL Closing Date (to the extent such date occurs) and the Closing Date as though made on and as of the FFL Closing Date and the Closing Date, as applicable (except to the extent that any such representations and warranties are made as of another date, which such representations and warranties shall be true and correct in all respects as of such other date) and (iii) representations and warranties of Seller set forth in Sections 4.01 (Organization), 4.02(a) (Authority), 4.02(b) (Execution and Delivery), 4.04 (Title to Purchased Interests), 5.02(b) (Capitalization), 5.02(c) (Capitalization), 5.02(d) (Capitalization), and 5.03(a) (Authority) of Seller shall be true and correct in all respects as of the date of this Agreement and as of the FFL Closing Date (to the extent such date occurs) and the Closing Date as though made on and as of the FFL Closing Date and Closing Date, as applicable (except to the extent that any such representations and warranties are made as of another date, which such representations and warranties shall be true and correct in all respects as of such other date).
- (b) Seller shall have performed or complied in all material respects with all covenants required by this Agreement to be performed or complied with by Seller at or prior to the Closing Date.
- (c) Since the date of this Agreement, no Company Material Adverse Effect shall have occurred and no other events shall have occurred that would, in the aggregate, reasonably be expected to have a Company Material Adverse Effect.
- (d) Seller shall have delivered, or cause to be delivered, to Purchaser the following:
- (i) membership interest certificates representing the Purchased Interests, duly endorsed in blank or accompanied by duly executed membership interest powers, in each case, with all necessary transfer tax and other revenue stamps (the cost of which shall be allocated as set forth in Section 7.12(b)), affixed and canceled;
- (ii) a certificate signed by Seller certifying that each of the conditions specified in subsections (a), (b) and (c) of this Section 8.02 have been satisfied;
- (iii) resignations, effective as of the Closing Date, of each manager (if applicable), director (if applicable) and officer of the Brand Companies identified on Schedule 8.02(d)(iii);

- (iv) the IP License Agreement, executed by Gaiam Americas and Gaia, Inc.;
- (v) the Transition Services Agreement, executed by Seller;

A-57

- (vi) the SVOD Rights Assignment Agreement, executed by Gaiam Americas and Gaia, Inc.;
- (vii) the SVOD Sub-License Agreement, executed by Gaiam Americas and Gaia, Inc.;
- (viii) the Escrow Agreement, executed by Seller and the Escrow Agent;
- (ix) a certificate of good standing or existence of Seller and each domestic Brand Company issued as of a date not more than 15 days prior to the Closing Date by the appropriate Governmental Entity (*e.g.*, Secretary of State) of its jurisdiction of formation or incorporation;
- (x) a certificate of the Secretary of Seller certifying, (A) as complete, accurate and in effect as of the Closing, (1) attached copies of Seller s and each Brand Company s Organizational Documents, and (2) all requisite resolutions or actions of the Board approving the execution and delivery of this Agreement, the other Collateral Agreements and the consummation of the Contemplated Transactions, and (B) as to the incumbency and signatures of the officers of Seller executing this Agreement and any other Collateral Agreement or other document, certificate or instrument relating to the Contemplated Transactions;
- (xi) a certificate pursuant to U.S. Treasury Regulations §1.1445-2(b) that Seller is not a foreign person within the meaning of Section 1445 of the Code;
- (xii) at least two (2) Business Days prior to the Closing, pay-off letters, in form and substance reasonably satisfactory to Purchaser, duly executed by each holder of Funded Indebtedness then outstanding, which include evidence that all Liens arising under such Funded Indebtedness have been or will be released upon payment in full of all outstanding amounts thereunder (the <u>Pay-Off Letters</u>); and
- (xiii) at least two (2) Business Days prior to the Closing, final invoices, in form and substance reasonably satisfactory to Purchaser, duly executed by each payee of Company Transaction Expenses, which invoices include the amount of Company Transaction Expenses incurred and owing to such Person as of the Closing and releasing the Brand Companies and its Affiliates from all obligations and Liabilities in connection with the repayment in full of such Company Transaction Expenses.
- (e) Seller shall have delivered, or cause to be delivered, to the Payroll Provider (i) the Transferred Employee Accrued PTO Amount and (ii) \$500,000 in respect of certain severance obligations of Seller and the Brand Companies, in each case, to be paid in accordance with the instructions provided by Seller to the Payroll Provider.
- (f) Either, the transactions contemplated by the Gaiam-FFL APA shall have been consummated, or Seller shall have caused the Brand Companies to have transferred and assigned the Acquired Assets, as defined in the Gaiam FFL APA, to Seller in accordance with <u>Section 7.22</u> hereof.
- **8.03** Conditions to Obligation of Seller. The obligation of Seller to consummate the transactions that are to be consummated at the Closing is subject to the satisfaction, as of the Closing Date, of the following further conditions (any of which may be waived in writing by Seller in whole or in part):
- (a) Each of the (i) representations and warranties of Purchaser set forth in $\underline{\text{Article VI}}$ shall be true and correct (without giving effect to any material, materially, materiality, material adverse effect, material adverse change or similar qualifiers contained in any of such representations and warranties) in all respects, except where the failure to be so true and correct, individually or in the aggregate, does not and would not reasonably be expected to have a material adverse effect on Purchaser's ability to perform its obligations under this Agreement and to consummate the

Contemplated Transactions, in each case as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date (except to the extent that any such representations and warranties speak as of another date, which such representations and warranties shall be true

A-58

and correct in all respects as of such other date), and (ii) Fundamental Representations of Purchaser shall be true and correct in all respects as of the date of this Agreement and as of the Closing Date as though made on and as of the Closing Date (except to the extent that any such Fundamental Representations are made as of another date, which such Fundamental Representations shall be true and correct in all respects as of such other date).

- (b) Purchaser shall have performed or complied in all material respects with all covenants required by this Agreement to be performed or complied with by Purchaser at or prior to the Closing Date.
- (c) Purchaser shall have delivered, or cause to be delivered, to Seller the following:
- (i) a certificate signed by an officer of Purchaser certifying that each of the conditions specified in subsections (a) and (b) of this Section 8.03 have been satisfied;
- (ii) evidence reasonably satisfactory to Seller that the requirements of Purchaser to be completed on the Closing Date set forth in <u>Section 7.08</u> will be satisfied immediately following the Closing;
- (iii) the Third Amendment and Assignment of Lease, executed by FFL;
- (iv) the Transition Services Agreement, executed by FFL;
- (v) the Reception Services Agreement, executed by FFL;
- (vi) the Escrow Agreement, executed by Purchaser and the Escrow Agent;
- (vii) a certificate of good standing for Purchaser issued as of a date not more than 15 days prior to the Closing Date by the appropriate Governmental Entity (e.g., Secretary of State) of its jurisdiction of formation; and
- (viii) a certificate of the Secretary of Purchaser certifying, (A) as complete, accurate and in effect as of the Closing, (1) attached copies of Purchaser's Organizational Documents, and (2) all requisite resolutions or actions of Purchaser's board of managers approving the execution and delivery of this Agreement, the other Collateral Agreements and the consummation of the Contemplated Transactions, and (B) as to the incumbency and signatures of the officers of Purchaser executing this Agreement and any other Collateral Agreement or other document, certificate or instrument relating to the Contemplated Transactions.
- (d) In addition to the Purchase Price paid in accordance with <u>Section 3.02(a)</u>, Purchaser shall have delivered, or cause to be delivered, to the Payroll Provider the following amounts:
- (i) the Designated Employee Accrued PTO Amount, to be paid in accordance with the instructions provided by Seller to the Payroll Provider;
- (ii) the Severance Amount, to be paid in accordance with the instructions provided by Seller to the Payroll Provider; and
- (iii) the WARN Amount, if any, to be paid in accordance with the instructions provided by Seller to the Payroll Provider.
- (e) The transactions contemplated by the Gaiam-FFL APA shall have been consummated.

8.04 Frustration of Closing Conditions. Neither Purchaser nor Seller may rely on the failure of any condition set forth in this <u>Article VIII</u> to be satisfied if such failure was caused by such Party s failure to act in good faith or to comply with <u>Section 7.04</u> or <u>Section 7.05</u>.

A-59

ARTICLE IX

TERMINATION

- **9.01 Termination.** This Agreement may be terminated at any time prior to the Closing Date:
- (a) by mutual written consent of Purchaser and Seller;
- (b) by either Purchaser or Seller if the Contemplated Transactions shall not have been consummated within ninety (90) days following the date hereof (the <u>Outside Date</u>); *provided, however*, that the Parties may mutually consent in writing to extend the Outside Date for any reason and, if any Person shall initiate a Proceeding seeking to enjoin the Contemplated Transactions and succeed in obtaining a temporary restraining order or injunction, the Outside Date shall be automatically extended for an additional thirty (30) days and thereafter, at Seller s or Purchaser s option, for up to two (2) additional thirty (30)-day periods (such extended date shall then be the <u>Outside Date</u>). The right to terminate this Agreement under this <u>Section 9.01(b)</u> shall not be available to any Party whose willful breach of any material obligation under this Agreement has been the principal cause of or resulted in the failure of the Closing to occur on or before the Outside Date:
- (c) by either Purchaser or Seller, if a Governmental Entity of competent jurisdiction shall have issued a nonappealable final Judgment or taken any other action having the effect of permanently restraining, enjoining, or otherwise prohibiting the Contemplated Transactions; *provided*, *however*, that the Party seeking to terminate pursuant to this Section 9.01(c) shall have complied with its obligations under Section 7.04;
- (d) by Purchaser, if Seller shall have breached or failed to perform any of its representations, warranties, covenants, or other agreements contained in this Agreement (other than Section 7.21(a)), which breach or failure to perform would cause the conditions to the obligations of Purchaser at the Closing to not be satisfied by the Outside Date and which breach or failure, if capable of being cured, shall not have been cured prior to the earlier of (i) the Outside Date and (ii) 10 calendar days following receipt by Seller of written notice of such breach or failure from Purchaser, with specific reference to the alleged facts underlying such breach or failure and the conclusion it would lead to such conditions not being met;
- (e) by Seller, if Purchaser shall have breached or failed to perform any of its representations, warranties, covenants, or other agreements contained in this Agreement, which breach or failure to perform would cause the conditions to the obligations of Seller at the Closing to not be satisfied by the Outside Date and which breach or failure, if capable of being cured, shall not have been cured prior to the earlier of (i) the Outside Date or (ii) 10 calendar days following receipt by Purchaser of written notice of such breach or failure from Seller, with specific reference to the alleged facts underlying such breach or failure and the conclusion it would lead to such conditions not being met;
- (f) by Purchaser, if Seller shall have breached in any material respect any of its obligations under Section 7.21(a);
- (g) by Purchaser, if Seller, the Board or any committee thereof, for any reason, shall have (i) effected an Adverse Recommendation Change, or (ii) approved or recommended any Alternative Transaction Proposal; or
- (h) by Seller, if: (i) Seller shall have received a Superior Proposal; (ii) Seller shall have complied in all material respects with its obligations under <u>Section 7.21</u>; (iii) the Board approves, and Seller, concurrently with the termination of this Agreement, enters into, such Alternative Acquisition Agreement with respect to such Superior Proposal; and (iv) concurrently with such termination, Seller pays to Purchaser the Termination Fee.

9.02 Termination Procedures. If Purchaser or Seller exercises their right to terminate this Agreement pursuant to Section 9.01, it shall deliver to the other Party a written notice specifying a reasonably detailed basis for termination; *provided, however*, that (i) Purchaser s right to terminate this Agreement pursuant to

A-60

<u>Section 9.01(d)</u> shall be subject to the cure period set forth in <u>Section 9.01(d)</u> and (ii) Seller s right to terminate this Agreement pursuant to <u>Section 9.01(e)</u> shall be subject to the cure period set forth in such Section <u>9.01(e)</u>.

9.03 Effect of Termination. If this Agreement is terminated and the Contemplated Transactions are abandoned as described in Section 9.01, this Agreement shall become null and void and of no further force and effect and there shall be no Liability on the part of any Party or any of its Affiliates, directors, officers, shareholders, partners, members or Representatives, except (i) that the provisions of Section 4.07, Section 5.20, Section 6.07, Section 7.03, Section 7.12, this Section 9.03, Section 9.04 and Article XI shall survive termination, and (ii) nothing herein shall relieve any Party from Liability for any intentional and material breach of this Agreement. The Confidentiality Agreement shall survive termination of this Agreement as provided therein. Purchaser shall promptly cause to be returned to Seller, the Company or their designated Representatives or certified destroyed by Purchaser, all documents and information obtained in connection with this Agreement and the Contemplated Transactions and all documents and information obtained in connection with Purchaser s investigation of the Company s business, operations, and legal affairs, including any copies made by Purchaser or its Representatives of any such documents or information and take all other such action as required by the Confidentiality Agreement.

9.04 Termination Fee.

- (a) In the event that:
- (i) Purchaser terminates this Agreement pursuant to Section 9.01(f) or 9.01(g);
- (ii) (1) this Agreement is terminated by (x) Seller or Purchaser pursuant to <u>Section 9.01(b)</u>, and Seller did not have the right to terminate this Agreement pursuant to <u>Section 9.01(e)</u>, or (y) by Purchaser pursuant to <u>Section 9.01(d)</u> and prior to such termination, an Alternative Transaction Proposal has been made and (2) within nine (9) months of such termination, Seller or any of its Subsidiaries (including any Brand Company) enters into a definitive agreement with any third party to consummate, or consummates, an Alternative Transaction;
- (iii) (1) this Agreement is terminated by (x) Seller or Purchaser pursuant to <u>Section 9.01(b)</u>, and Seller did not have the right to terminate this Agreement pursuant to <u>Section 9.01(e)</u>, or (y) by Purchaser pursuant to <u>Section 9.01(d)</u> and prior to such termination, no Alternative Transaction Proposal has been made and (2) within six (6) months of such termination, Seller or any of its Subsidiaries (including any Brand Company) enters into a definitive agreement with any third party to consummate, or consummates, an Alternative Transaction; or
- (iv) Seller terminates this Agreement pursuant to Section 9.01(h);

then, Seller shall pay Purchaser a one-time fee equal to \$5,010,000 (the <u>Termination Fee</u>), which Termination Fee shall be the sole and exclusive remedy available at law or in equity to Purchaser for a termination of this Agreement which results in the payment of the Termination Fee. Any amounts due under this <u>Section 9.04(a)</u> shall be paid by wire transfer of same-day funds on the date of such termination or entry into such definitive agreement, as applicable, to an account provided in writing by Purchaser to Seller. The Parties agree that the liquidated damages described in this <u>Section 9.04</u> are based in part on the damages which the Purchaser would suffer on account of the failure of the Closing to occur under the circumstances specified (which damages the parties acknowledge and agree are incapable of exact determination of amount). It is therefore understood and agreed that such liquidated damages amount represents the Parties reasonable estimate of actual damages and do not constitute a penalty.

(b) Interest and Costs; Other Remedies. Each of Seller and Purchaser acknowledges that the agreements contained in this <u>Section 9.04</u> are an integral part of the transactions contemplated by this Agreement

A-61

and that, without these agreements, neither Purchaser nor Seller would have entered into this Agreement. Accordingly, in the event that Seller shall fail to pay the Termination Fee when due, and in order to obtain such payment, Purchaser commences a suit which results in a judgment against Seller for such Termination Fee, then Seller shall pay to Purchaser its costs and expenses (including attorneys fees and expenses of enforcement) in connection with such suit, together with interest on the amounts owed at the prime lending rate prevailing at such time, as published in the Wall Street Journal.

ARTICLE X

INDEMNIFICATION AND RELATED MATTERS

10.01 Indemnification by Seller. Subject to the limitations set forth in this <u>Article X</u>, from and after the Closing, Seller shall indemnify and hold harmless Parent, Purchaser, each Brand Company and their respective Affiliates, Representatives, successors and assigns (each a, <u>Purchaser Party</u> and collectively, the <u>Purchaser Parties</u>) from and against any and all Damages that any Purchaser Party suffers, incurs or pays, directly or indirectly, resulting from, arising out of or related to:

- (a) any breach by Seller of a covenant contained in this Agreement;
- (b) any Indebtedness of the Brand Companies that is not paid and discharged in full on or prior to the Closing Date and has not been taken into account pursuant to Section 2.06;
- (c) any Company Transaction Expenses that are not paid and discharged on or prior to the Closing Date and have not been taken into account pursuant to Section 2.06;
- (d) any Transferred Liabilities;
- (e) any of the following Taxes and liabilities:
- (i) any Taxes relating to a Pre-Closing Period imposed on Seller, Purchaser or any of the Brand Companies resulting from, arising out of, or incurred with respect to the Section 338(h)(10) Election;
- (ii) any Taxes relating to a Pre-Closing Period arising from the Contribution or the contribution of the Company Subsidiaries to the Company;
- (iii) any liability of any of the Brand Companies under Treasury Regulations Section 1.1502-6 or the analogous provisions of any state, local or foreign law for any Taxes relating to a Pre-Closing Period;
- (iv) any Taxes arising from any breach of the representations and warranties in Section 5.13(p);
- (v) any Taxes arising from any violation of the covenants contained in <u>Section 7.10</u> or <u>Section 7.12(b)</u>;
- (vi) any Taxes relating to the matters disclosed in <u>Section 5.13(c)</u>, <u>Section 5.13(g)</u>, and <u>Section 5.13(h)</u> of the Disclosure Schedule; and
- (vii) any Taxes or penalties in excess of \$250,000 relating to the matters disclosed in <u>Section 5.13(f) of the Disclosure Schedule</u>;

(f) all Taxes imposed on, asserted against or attributable to the properties, income or operations of Seller or the Brand Companies or any Taxes for which Seller or any Brand Company is otherwise liable, with respect to all Pre-Closing Periods, other than those described in <u>Section 10.01(e)</u>;

A-62

- (g) any pre-Closing Liabilities of the Brand Companies relating to or arising from any events, circumstances or matters unrelated to the operation of the Brand Business; and
- (h) any Proceeding commenced or brought by any shareholder of Seller in connection with the Contemplated Transactions.
- **10.02 Indemnification by Purchaser.** Subject to the limitations set forth in this <u>Article X</u>, from and after the Closing, Purchaser shall indemnify and hold harmless Seller and its Affiliates, Representatives, successors and assigns (each, a <u>Seller Party</u> and collectively, the <u>Seller Parties</u>) from and against any and all Damages that any Seller Party suffers, incurs or pays, directly or indirectly, resulting from, arising out of or related to:
- (a) any failure of any representation or warranty of Purchaser contained in this Agreement to be true and correct (without giving effect to any material, materially, materiality, material adverse effect, material adverse change of similar qualification contained in any such representation or warranty) in all respects on and as of the date of this Agreement and on and as of the Closing Date as if made on such date (other than those made on a specified date, which shall be true and correct in all respects as of such specified date);
- (b) any breach by Purchaser of any covenant in this Agreement; and
- (c) any claim made by FFL against a Seller Party pursuant to the Gaiam-FFL APA (excluding any actual breach by a Seller Party thereunder), and subject to the Parties compliance with Section 7.22 after giving effect to any sale or collection of the Acquired Assets thereunder, any breach by FFL of the Gaiam-FFL APA; and
- (d) any Damages imposed on, sustained, incurred or suffered by, or asserted against, any of them, whether in respect of third-party claims, direct claims or otherwise, directly or indirectly relating to, arising out of or resulting from the arrangement of the Debt Financing, any other financing and/or the provision of information utilized in connection therewith.

10.03 Survival and Expiration of Representations and Warranties and Covenants.

- (a) Except as set forth in Section 10.03(b) and Section 10.03(c), all of the representations and warranties of the Parties set forth in this Agreement terminate and expire, and shall cease to be of any force or effect, at 5:00 p.m. (Mountain Time) upon the date that is the 14-month anniversary following the Closing Date (the Survival Date), and all liability of the Parties with respect to such representations and warranties (including for indemnification for breach of such representations and warranties under this Article X), shall thereupon be extinguished. The covenants and agreements of the Parties shall survive the Closing until the Survival Date that is the 12-month anniversary following the Closing Date except (i) with respect to those covenants and agreements of the Parties that by their terms contemplate performance in whole or in part after the Closing shall remain in full force and effect until such covenant or agreement has been fully performed and (ii) with respect to the indemnification obligations of Seller pursuant to Section 10.01(e) and the related covenants contained in Section 7.10, which shall survive the Closing until sixty (60) days following the expiration of the applicable statute of limitations.
- (b) Except as set forth in Section 10.03(c), the representations and warranties set forth in Section 4.04 (*Title to Purchased Interests*), Sections 5.02(a) (*Capitalization*), 5.02(b) (*Capitalization*), 5.02(c) (*Capitalization*) and 5.02(d) (*Capitalization*) shall survive until the expiration of the applicable statute of limitations, and the representations and warranties set forth in Section 5.13 (*Taxes*) shall survive until 60 days after the expiration of the applicable statute of limitations.

(c) Notwithstanding anything in this Section 10.03 to the contrary, if, prior to the relevant survival period set forth in Section 10.03(a) and Section 10.03(b) (the Claim Expiration Date), a Purchaser Party or a

A-63

Seller Party, as applicable, has duly delivered in good faith a conforming Claim Notice to Purchaser or Seller, as applicable, then the specific indemnification claim set forth in such Claim Notice (to the extent of the matter specified in the Claim Notice) shall survive the Claim Expiration Date and shall not be extinguished thereby until resolution of the matter specified in the Claim Notice in accordance with this Agreement.

10.04 Indemnification Limitations.

- (a) Subject to the other provisions set forth in this Agreement, no Seller Party shall be entitled to any indemnification payment under this Agreement pursuant to Section 10.02(a) for any individual item or group of items arising out of the same or related events, sets of facts or circumstances, where the amount of Damages relating thereto is less than \$50,000 (the Sub-Basket) and in respect of each individual item, or group of items arising out of the same or related events, sets of facts or circumstances, where the Damages relating thereto is equal to or greater than the Sub-Basket, unless and until the amount of Damages actually incurred by such Seller Parties as a result thereof exceeds \$1,000,000 (the Basket Amount), at which time the Indemnified Party shall be entitled to recover the amount of the Damages incurred as a direct result of all such breaches of such representations and warranties that exceeds the Basket Amount.
- (b) Subject to the other provisions set forth in this Agreement, the aggregate Liability of Seller to any and all Purchaser Parties under this Agreement and the aggregate Liability of Purchaser to any and all Seller Parties under this Agreement shall, in each case, be limited to \$11,690,000 (the <u>Cap Amount</u>).
- (c) Except as set forth in Section 10.01(e)(iv), the Purchaser Parties—sole recourse in respect of (i) any breach or inaccuracy by Seller of the representations and warranties contained in Article IV and Article V and (ii) indemnification claims made by a Purchaser Party pursuant to Section 10.01(f) shall be against the Representation and Warranty Policy Insurer pursuant to the terms of the Representation and Warranty Policy and no Seller Party shall be liable for any Damages with respect to (A) any breach or inaccuracy by Seller of the representations and warranties contained in Article IV and Article V and (B) any indemnification claim made by a Purchaser Party pursuant to Section 10.01(f).
- (d) Neither the Sub-Basket nor the Basket Amount set forth in Section 10.04(a) shall apply to (i) any breach of any Fundamental Representations made by Purchaser; (ii) any indemnification claim made by a Purchaser Party pursuant to Sections 10.01(a) through 10.01(h); (iii) any indemnification claim made by a Seller Party pursuant to Section 10.02(b); or (iv) fraud in connection with any breach of a representation or warranty set forth in this Agreement.
- (e) From and after the date that is the 12-month anniversary of the Closing Date, no Purchaser Party or Seller Party shall be entitled to indemnification, to sue for Damages or to assert any other right or remedy under this Agreement (whether under this Article X or otherwise) against any Seller Party or Purchaser Party, as applicable, with respect to any Damages, cause of action or other claim (other than, for the avoidance of doubt, Damages incurred pursuant to Section 10.01(f) or, except as set forth in Section10.01(e)(iv), as a result of any breach or inaccuracy by Seller of the representations and warranties contained in Article IV and Article V, for which the Purchaser Party s sole recourse shall be against the Representation and Warranty Policy Insurer pursuant to the terms of the Representation and Warranty Policy in accordance with Section 10.04(c)), to the extent the Purchaser Party or Seller Party, as applicable, failed to give written notice of such Damages, cause of action or claim within ninety (90) days following the date on which such Purchaser Party or Seller Party, as applicable, learned of such Damages, cause of action or claim.
- (f) The Cap Amount set forth in Section 10.04(b) shall not apply to (i) any indemnification claim made by a Purchaser Party pursuant to Section 10.01(d), Section 10.01(e), Section 10.01(g) and Section 10.01(h), (ii) fraud in connection with any breach of a representation or warranty set forth in this Agreement and (iii) any indemnification claim made

by a Seller Party pursuant to Section 10.02(c).

A-64

(g) No Purchaser Party or Seller Party shall be entitled to recover any Damages or other amounts relating to any matter arising under one provision of this Agreement to the extent that such Purchaser Party or Seller Party (or other Purchaser Parties in the event of a Purchaser Party, or other Seller Parties in the event of a Seller Party), as applicable, has already recovered such Damages or amounts with respect to such matter pursuant to other provisions of this Agreement. Without limiting the generality of the foregoing, the operation of Section 2.05 and Section 2.06 is an exclusive remedy in respect of the assets and liabilities and related items taken into account in connection with the determination of Final Net Closing Cash Consideration, and to the extent any breach of a representation or warranty and the Damages resulting therefrom were specifically taken into account in such determination, Purchaser shall not be entitled to any additional recourse in respect thereof under this Article X.

10.05 No Liability of Representatives, Lenders, or Financing Sources. No Party shall have any recourse whatsoever against (a) any Representatives of the other Party, or (b) any of the assets of any of the Persons identified in subsection (a), in connection with any indemnification claim or any other claim of any nature in connection with this Agreement or the Contemplated Transactions. Notwithstanding anything to the contrary contained in this Agreement, neither Seller or Affiliates nor either of their respective Representatives shall have any rights or claims against any lender or financing source with respect to the Debt Financing (including without limitation any lender or financing source that is a party to the Debt Commitment Letters or Fee Letter) in connection with this Agreement, the Debt Financing or the transactions contemplated hereby or thereby, and no such lender or financing source shall have any rights or claims against Seller, its Affiliates or either of their respective Representatives in connection with this Agreement, the Debt Financing or the transactions contemplated hereby or thereby, in each case whether at law or in equity, in contract, in tort or otherwise; *provided*, that, following the consummation of the transactions contemplated by this Agreement, the foregoing shall not limit the rights of the parties to the Debt Financing under the Debt Commitment Letters.

10.06 Indemnification Claims.

- (a) If Purchaser or Seller (the Party seeking indemnification, the <u>Indemnified Party</u>) wishes to assert an indemnification claim in accordance with this <u>Article X</u>, then the Indemnified Party shall deliver to the other Party (the <u>Indemnifying Party</u>) a written notice (a <u>Claim Notice</u>) setting forth:
- (i) the specific representation, warranty, or covenant alleged to have been breached by the Indemnifying Party or other indemnifiable matter described in <u>Section 10.01</u> or <u>Section 10.02</u>, as applicable;
- (ii) a reasonably detailed description of the facts and circumstances giving rise to the alleged breach of such representation, warranty or covenant or other indemnifiable matter described in <u>Section 10.01</u> or <u>Section 10.02</u>, as applicable; and
- (iii) a reasonably detailed description of, and a reasonable estimate of the total amount of, the Damages incurred or expected to be incurred by the Indemnified Party as a result of such alleged breach or other indemnifiable matter described in <u>Section 10.01</u> or <u>Section 10.02</u>, as applicable.
- (b) In the event that the Indemnifying Party shall object to the indemnification of an Indemnified Party in respect of any claim or claims specified in any Claim Notice (other than a Third Party Claim, which is addressed in Section 10.07), the Indemnifying Party shall, within twenty (20) days after receipt by the Indemnifying Party of such Claim Notice, deliver to the Indemnified Party a notice to such effect, specifying in reasonable detail the basis for such objection, and the Indemnifying Party and the Indemnified Party shall, within the 20 day period beginning on the date of receipt by the Indemnified Party of such objection and prior to submitting such dispute in accordance with Section 11.11, attempt in good faith to agree upon the rights of the respective parties with respect to each of such

claims to which the Indemnifying Party shall have so objected. If the Indemnified Party and the Indemnifying Party shall succeed in reaching agreement on their respective rights with respect to any of such claims, the Indemnified Party and the Indemnifying Party shall promptly prepare and

A-65

sign a memorandum of agreement setting forth such agreement. Should the Indemnified Party and the Indemnifying Party be unable to agree as to any particular item or items or amount or amounts within such time period, then the Indemnified Party shall be permitted to submit such dispute in accordance with <u>Section 11.11</u>.

(c) Claims for Damages specified in any Claim Notice to which an Indemnifying Party shall not object in writing within twenty (20) days of receipt of such Claim Notice (other than a Third Party Claim, which is addressed in Section 10.07), claims for Damages covered by a memorandum of agreement of the nature described in Section 10.06(b), and claims for Damages the validity and amount of which have been the subject of judicial determination as described in Sections 10.06(b) and 11.11 or shall have been settled or determined as described in Section 10.07, are hereinafter referred to, collectively, as Agreed Claims. Within 10 Business Days of the determination of the amount of any Agreed Claim (or at such other time as the Indemnified Party and the Indemnifying Party shall agree), the Indemnifying Party shall pay to the Indemnified Party an amount equal to the Agreed Claim by wire transfer in immediately available funds to the bank account or accounts designated by the Indemnified Party in a notice to the Indemnifying Party not less than two Business Days prior to such payment.

10.07 Defense of Third Party Claims.

- (a) If an Indemnified Party receives notice or otherwise obtains knowledge of any Proceeding commenced or threatened by a third party, other than any claim relating to Taxes (such Tax-related matters being governed by Section 7.10) (each, a Third Party Claim) against the Indemnified Party that the Indemnified Party actually knows may give rise to an indemnification claim against the Indemnifying Party, then the Indemnified Party shall reasonably promptly after such actual knowledge deliver to the Indemnifying Party a written notice (the Third Party Claim Notice) describing the Third Party Claim in reasonable detail. The delivery of such written notice by the Indemnified Party to the Indemnifying Party shall be a condition precedent to any Liability on the part of the Indemnifying Party under this Article X with respect to such Third Party Claim; provided, that any failure to timely notify shall not relieve the Indemnifying Party of its obligations hereunder, except to the extent that the Indemnifying Party is actually and materially prejudiced thereby. The Indemnifying Party shall have the right, at its option, to assume the defense of any such Third Party Claim with counsel reasonably acceptable to the Indemnified Party; provided, that (i) the Indemnifying Party shall be entitled to assume the defense of such Third Party Claim only to the extent the Indemnifying Party acknowledges in writing its indemnity obligation and assumes and holds the Indemnified Party harmless from and against the full amount of any Damages resulting therefrom; and (ii) the Indemnifying Party shall not be entitled to assume control of such defense and shall pay the fees and expenses of counsel retained by the Indemnified Party if (A) such Third Party Claim is reasonably foreseeable to result in Damages which are more than the remaining Cap Amount at the time such claim is submitted by the Indemnified Party; (B) such Third Party Claim relates to or arises in connection with any criminal proceeding, action, indictment, allegation or investigation; (C) such Third Party Claim seeks an injunction or equitable relief against the Indemnified Party; (D) the Indemnified Party has been advised in writing by counsel that a reasonable likelihood exists of a conflict of interest between the Indemnifying Party and the Indemnified Party; (E) the Indemnified Party reasonably believes an adverse determination with respect to the action, lawsuit, investigation, proceeding or other claim giving rise to such claim for indemnification would be detrimental to or injure the Indemnified Party's reputation or future business prospects; or (F) upon petition by the Indemnified Party, the appropriate court rules that the Indemnifying Party failed or is failing to vigorously prosecute or defend such Third Party Claim.
- **(b)** If the Indemnifying Party is entitled to and elects in writing (which writing shall also contain the acknowledgement referenced in <u>Section 10.07(a)(i)</u>) within 20 days of receipt of the Third Party Claim Notice to assume the defense of any such Third Party Claim, then:

(i) the Indemnifying Party shall permit the Indemnified Party to participate in such defense through counsel chosen by such Indemnified Party and reasonably satisfactory to the Indemnifying Party and the Indemnifying Party shall not be required to pay or otherwise indemnify the Indemnified Party against any fees or

A-66

expenses of such counsel incurred on behalf of the Indemnified Party in connection with such Third Party Claim following the Indemnifying Party s election to assume the defense of such Third Party Claim, *provided*, that the Indemnifying Party shall pay the fees and expenses of such counsel if (A) the Indemnifying Party subsequently ceases to control the defense of such Third Party Claim; *provided further* that in such event, the Indemnifying Party shall only be responsible for such fees and expenses from and after the date on which the Indemnifying Party ceases to control the defense of such Third Party Claim, or (B) the named parties to any such action (including any impleaded parties) include both such Indemnified Party and the Indemnifying Party and such Indemnified Party shall have been advised in writing by such counsel that there may be one or more legal defenses available to the Indemnified Party which are not available to the Indemnifying Party, or available to the Indemnifying Party the assertion of which would be adverse to the interests of the Indemnified Party;

- (ii) subject to the entry into a confidentiality agreement acceptable to the Indemnified Party, the Indemnified Party shall make reasonably available to the Indemnifying Party all nonprivileged books, records, and other documents and materials that are under the direct or indirect control of the Indemnified Party or any of the Indemnified Party s Representatives and that the Indemnifying Party considers necessary or desirable for the defense of such Third Party Claim;
- (iii) the Indemnified Party shall reasonably cooperate as reasonably requested by the Indemnifying Party in the defense of such Third Party Claim;
- (iv) the Indemnified Party shall not admit any liability with respect to or pay or settle such Third Party Claim so long as the Indemnifying Party is reasonably contesting any such Third Party Claim in good faith; provided, that notwithstanding the foregoing, the Indemnified Party shall have the right to pay or settle any such Third Party Claim if (A) it shall waive any right to indemnity therefor by the Indemnifying Party or (B) the Indemnifying Party shall have consented in writing to such payment or settlement; and
- (v) the Indemnifying Party shall have the right to settle, adjust, or compromise such Third Party Claim, on such terms as it may deem appropriate, without the consent or approval of the Indemnified Party or any other Person, unless such settlement, adjustment or compromise (i) does not include a full, unconditional and complete release by all of the Persons asserting such Third Party Claim of Purchaser or other applicable Purchaser Party from all Liability with respect to such Third Party Claim, (ii) is not entirely indemnifiable by the Indemnifying Party pursuant to this Article X, or, and (iii) imposes any injunctive relief or other restrictions of any kind or nature on the Indemnified Party (or any Purchaser Party or Seller Party, as applicable), in which case the Indemnifying Party shall not enter into such settlement, adjustment, or compromise of such Third Party Claim without Purchaser s or other applicable Purchaser Party s prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed).
- (c) If the Indemnifying Party elects not to assume the defense of such Third Party Claim, then (i) the Indemnified Party shall defend such Third Party Claim in a commercially reasonable manner; *provided, however*, that the Indemnified Party shall not settle, adjust, or compromise such Third Party Claim, or admit any Liability with respect to such Third Party Claim, without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld, conditioned or delayed, and (ii) the Indemnifying Party shall reasonably cooperate as reasonably requested by the Indemnified Party in the defense of such Third Party Claim and make reasonably available to the Indemnified Party all nonprivileged books, records, and other documents and materials that are under the direct or indirect control of the Indemnifying Party or any of the Indemnifying Party s Representatives and that the Indemnified Party reasonably considers necessary or desirable for the defense of such Third Party Claim.
- (d) In all events, the Representation and Warranty Policy Insurer and its agents and advisors shall be permitted to associate effectively with any Party in the defense of any matter which might reasonably involve Loss (as defined in

the Representation and Warranty Policy).

A-67

- (e) Without limiting the other rights of Seller hereunder, the Representation and Warranty Policy Insurer shall be entitled to participate, at its sole cost and expense, in any Third Party Claim covered by the Representation and Warranty Policy to the extent set forth in the Representation and Warranty Policy, but only to the extent such participation is expressly required by the Representation and Warranty Policy. Notwithstanding anything to the contrary in this <u>Section 10.07</u>, the provisions of this <u>Section 10.07</u> shall be subject to the rights of the Representation and Warranty Policy Insurer under the Representation and Warranty Policy with respect to any Third Party Claims covered by the Representation and Warranty Policy.
- **(f)** Notwithstanding anything in this <u>Section 10.07</u> to the contrary, <u>Section 7.10</u> shall govern for purposes of any Third Party Claim that is a Tax Controversy
- **10.08 Exclusive Remedy.** Except (a) in the case of (i) fraud by a member of the Seller Knowledge Group with respect to a representation or warranty in <u>Article V</u> or (ii) fraud by a member of the Purchaser Knowledge Group with respect to a representation or warranty in <u>Article VI</u>, (b) the dispute resolution provisions set forth in <u>Section 2.06</u> relating to the calculation of the Final Net Closing Cash Consideration, (c) the right to the Termination Fee pursuant to <u>Section 9.04</u>, and (d) the right to specific performance in accordance with <u>Section 11.14</u>, in the event of a breach by any Party of a covenant under this Agreement, the right of each Party to assert indemnification claims and receive indemnification payments pursuant to <u>Section 7.10</u> (with respect to Tax matters) and this <u>Article X</u> shall, from and after the Closing Date, be the sole and exclusive right and remedy exercisable by such Party with respect to any breach by any other Party of any covenant, representation, warranty, or otherwise under this Agreement, relating to this Agreement or relating to the Contemplated Transactions.

10.09 Characterization of Indemnification Payment. Any payment made pursuant to or in connection with this Article X shall be deemed to be an adjustment to the Purchase Price to the extent permitted by Applicable Law.

10.10 Seller s Waiver. Notwithstanding anything to the contrary contained herein, Seller hereby waives and acknowledges that it shall not exercise or assert, any right of contribution or right to indemnity or any other right or remedy against any Brand Company in connection with any indemnification obligation or any other Liability to which Seller may become subject under this Agreement or otherwise in connection with any of Contemplated Transactions; *provided* that this waiver shall not apply to or limit any of Seller s rights under or pursuant to any Collateral Agreement.

ARTICLE XI

GENERAL PROVISIONS

11.01 Disclosure Schedule. Each section of the Disclosure Schedule qualifies the correspondingly numbered representation and warranty or covenant, and each disclosure set forth therein shall qualify any other representation or warranty to the extent that the applicability of such disclosure to such other representation or warranty is reasonably apparent on the face of such disclosure as set forth in the Disclosure Schedule. The Disclosure Schedule is qualified in its entirety by reference to specific provisions of the Agreement, and is not intended to constitute, and shall not be construed as constituting, any representation or warranty or covenant of Seller, except as and to the extent expressly provided in the Agreement. Inclusion of information in the Disclosure Schedule shall not be construed as an admission that such information is material to Seller or any Brand Company or its assets, liabilities, financial condition, results, business and/or operations. The fact that any item of information is contained in the Disclosure Schedule shall not be used as a basis for interpreting the term material, materially or materiality in the Agreement. References to any document in the Disclosure Schedule do not purport to be complete and are qualified in their

entirety by the document itself. Capitalized terms used but not defined in the Disclosure Schedule shall have the same meanings given them in this Agreement.

A-68

11.02 Exclusivity of Agreement. The Parties have voluntarily agreed to define their rights, liabilities, and obligations respecting the subject matter of this Agreement exclusively in contract pursuant to the express terms and provisions of this Agreement; and the Parties expressly disclaim that they are owed any duties or are entitled to any remedies not expressly set forth in this Agreement. Furthermore, the Parties each hereby acknowledge that this Agreement embodies the justifiable expectations of sophisticated parties derived from arm s-length negotiations; all Parties specifically acknowledge that no Party has any special relationship with another party that would justify any expectation beyond that of an ordinary acquiror and an ordinary target in an arm s-length transaction. The sole and exclusive remedies for any breach of the terms and provisions of this Agreement (including any representations and warranties set forth herein) shall be those remedies set forth herein.

11.03 No Third Party Liability. This Agreement may only be enforced against the named Parties. All claims or cause of action (whether in contract or tort) that may be based upon, arise out of or related to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), may be made only against the Persons that are expressly identified as Parties; and no officer, director, shareholder, employee or affiliate of any Party (including any Person negotiating or executing this Agreement on behalf of a Party) shall have any liability or obligation with respect to this Agreement or with respect to any claim or cause of action (whether in contract or tort) that may arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including a representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement).

11.04 Assignment.

- (a) This Agreement shall be binding upon: Seller and its respective successors and permitted assigns (if any) and Purchaser and its successors and permitted assigns (if any). This Agreement shall inure to the benefit of Seller and Purchaser and their respective successors and permitted assigns (if any).
- (b) Neither Purchaser, Parent nor Seller shall be permitted to assign any of its rights or delegate any of its obligations under this Agreement without the other Party's prior written consent; provided, that notwithstanding the foregoing: (i) Seller may assign its rights under this Agreement to any wholly-owned Subsidiary of Seller upon written notice to Purchaser; and (ii) Purchaser may assign or delegate (A) its rights under this Agreement to any wholly-owned Subsidiary of Purchaser or Sequential Brands Group, Inc., (B) its rights, interests and obligations under this Agreement in connection with a merger, exchange, consolidation or sale of all or substantially of its stock or assets or other similar transaction, and (C) its rights under this Agreement as collateral to its lenders and financing sources, in each case, without the prior written consent of Seller; *provided further* that no assignment or delegation by Seller or Purchaser pursuant to clauses (i) or (ii), respectively, shall limit or release Seller or Purchaser of its obligations hereunder, including, without limitation, Purchaser's obligations with respect to the payment of the Net Closing Cash Consideration and the Final Net Closing Cash Consideration hereunder.
- **11.05** No Third-Party Beneficiaries. This Agreement is for the sole benefit of the Parties and their successors and assigns and nothing herein expressed or implied shall give or be construed to give to any Person (other than the Parties and such Party s successors and permitted assigns) any legal or equitable rights hereunder, other than (i) the persons intended to benefit from the provisions of Article X (Indemnification), and (ii) the persons intended to benefit from the provisions of Sections 6.04, 7.14, 9.04(a), 10.05, 11.04, 11.09, 11.11(c), 11.13 and this Section 11.05 (including without limitation the lenders and financing sources of Purchaser), who, in each case, shall be intended beneficiaries under this Agreement with respect to such provisions and have the right to rely on and enforce such provisions directly in the event of a breach thereof.

11.06 Notices. All notices and other communications pursuant to this Agreement shall be in writing and shall be deemed given if delivered personally, sent by electronic mail, sent by nationally-recognized overnight

A-69

courier or mailed by registered or certified mail (return receipt requested), postage prepaid, to the Parties at the addresses set forth below or to such other address as the Party to whom notice is to be given may have furnished to the other Parties in writing in accordance herewith. Any such notice or communication shall be deemed to have been delivered and received (a) in the case of personal delivery, on the date of such delivery, (b) in the case of electronic mail, on the date sent if either (i) confirmation of receipt is received or (ii) such notice is promptly mailed by registered or certified mail (return receipt requested) or a nationally recognized overnight courier, (c) in the case of a nationally recognized overnight courier in circumstances under which such courier guarantees next Business Day delivery, on the next Business Day after the date sent, and (d) in the case of mailing, on the third Business Day following that on which the piece of mail containing such communication is posted.

(i) if to Seller, to:

Gaiam Americas, Inc.

833 West South Boulder Rd.

Louisville, CO 80027-2452

Email: john.jackson@gaiam.com

Attention: General Counsel

with a copy to:

Gaiam, Inc.

833 West South Boulder Rd.

Louisville, CO 80027-2452

Email: jirka.rysavy@gaiam.com

Attention: Jirka Rysavy

Brownstein Hyatt Farber Schreck, LLP

410 Seventeenth Street, Suite 2200

Denver, Colorado 80202-4432

Email: kmacdonald@bhfs.com

Attention: Kristin Macdonald

(ii) if to Purchaser or Parent, to:

c/o Sequential Brands Group, Inc.

5 Bryant Park, 30th Floor

New York, New York 10018

Email: cwagenheim@sbg-ny.com

Attention: Chad Wagenheim

with a copy to:

White & Case LLP

1155 Avenue of the Americas

New York, NY 10036

Email: nzilkha@whitecase.com

dorzechowski@whitecase.com

Attention: Nazim Zilkha

Daren Orzechowski

11.07 Counterparts. This Agreement may be executed in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or other electronic transmission (including in portable document format or PDF) shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.

A-70

- **11.08 Entire Agreement**. This Agreement (including the documents and instruments referred to herein, including the Confidentiality Agreement and the Collateral Agreements and all Exhibits and Schedules hereto and thereto) constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the Parties, or any of them, with respect to the subject matter hereof.
- **11.09 Amendments**. This Agreement may not be amended except pursuant to a written agreement executed by each of Purchaser and Seller; *provided*, that Sections 6.04, 7.14, 9.04(a), 11.04, 11.05, 11.11(c), and this Section 11.09 shall also not be amended without the written consent of the lenders or financing sources of Purchaser in connection with the Debt Financing.
- **11.10 Severability**. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision will be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision, and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances will not be affected by such invalidity or unenforceability, nor will such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

11.11 Governing Law; Venue.

- (a) This Agreement, and all claims of causes of action (whether in contract or tort) that may be based upon, arise out of, or relate to this Agreement or the negotiation, execution or performance of this Agreement (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed solely by the internal laws of the State of New York, without regard to the conflict-of-law principles thereof.
- (b) Each Party acknowledges that any Proceeding or other legal action relating to this Agreement or the enforcement of any provision of this Agreement (including any Proceeding relating to a claim for indemnification in accordance with Article X or for specific performance in accordance with Section 11.14) may only be brought or otherwise commenced in any state or federal court located in the Borough of Manhattan of the City of New York, New York, or Boulder, Colorado. Each Party:
- (i) expressly and irrevocably consents and submits to the jurisdiction of each state and federal court located in the Borough of Manhattan of the City of New York, New York and Boulder, Colorado, in connection with any such Proceeding;
- (ii) agrees that each state and federal court located in the Borough of Manhattan of the City of New York, New York and Boulder, Colorado, shall be deemed to be a convenient forum; and
- (iii) agrees not to assert (by way of motion, as a defense or otherwise), in any such Proceeding commenced in a state or federal court located in the Borough of Manhattan of the City of New York, New York or Boulder, Colorado, any claim that such Party is not subject personally to the jurisdiction of such court, that such Proceeding has been brought in an inconvenient forum, that the venue of such Proceeding is improper or that this Agreement or the subject matter of this Agreement may not be enforced in or by such court.
- (c) Notwithstanding anything in this <u>Section 11.11</u> to the contrary, each of the parties hereto agrees that it will not bring or support any Proceeding (whether at law, in equity, in contract, in tort or otherwise) against the lenders or

financing sources to Purchaser or any other Persons that have committed to provide or otherwise entered into agreements in connection with the Debt Financing or other financings in connection with the transactions contemplated hereby in any way relating to this Agreement or any of the transactions contemplated

A-71

by this Agreement, including any dispute arising out of or relating in any way to the Debt Commitment Letters or the Fee Letter or the performance of either of them, in any forum other than the Supreme Court of the State of New York, County of New York, or, if under Applicable Law exclusive jurisdiction is vested in the Federal courts, the United States District Court for the Southern District of New York (and appellate courts thereof).

11.12 Attorney s Fees. If any Proceeding relating to this Agreement or any of the Contemplated Transactions or the enforcement thereof is brought against any Party, the prevailing Party, if any, shall be entitled to recover reasonable attorneys fees, costs and disbursements (in addition to any other relief to which the prevailing Party may be entitled). In the event that both Parties prevail in part, the applicable trier of fact shall determine the allocation of such attorneys fees, costs and disbursements in proportion to each Party s success and failure of claims brought in connection with such Proceeding.

11.13 Waiver of Jury Trial. EACH PARTY TO THIS AGREEMENT WAIVES TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY ANY OF THEM AGAINST THE OTHER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS AGREEMENT, OR ANY OTHER AGREEMENTS EXECUTED IN CONNECTION HEREWITH OR THE ADMINISTRATION THEREOF OR THE CONTEMPLATED TRANSACTIONS. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 11.13. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS SECTION 11.13 WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.

11.14 Remedies. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party shall be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy shall not preclude the exercise of any other remedy. Each Party acknowledges and agrees that irreparable damage would occur in the event that any of the provisions of this Agreement (including failing to take such actions as are required of it hereunder to consummate this Agreement) were not performed in accordance with its specific terms or were otherwise breached. Accordingly, each Party acknowledges and agrees that, except to the extent prohibited in the event the Termination Fee becomes due and payable to the Purchaser pursuant to Section 9.04, the other Party shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which they are entitled at law or in equity, without having to post a bond. Each Party agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other Party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity.

11.15 Waiver.

(a) No failure on the part of any Person to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any Person in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy.

(b) No Person shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or

A-72

remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Person; and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.

11.16 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neutral genders; the feminine gender shall include the masculine and neutral genders; and the neutral gender shall include the masculine and feminine genders.
- (b) Each Party acknowledges that it has participated in the drafting of this Agreement, and, as a result, the Parties agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words include and including, and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words without limitation.
- (d) Except as otherwise indicated, all references in this Agreement to subsections, Sections, Schedules and Exhibits are intended to refer to subsections and Sections of this Agreement, Schedules of this Agreement and Exhibits to this Agreement.
- (e) The words this Agreement, hereby, hereof, herein, hereunder, and comparable words refer to all of this Agree including the Appendices, Exhibits, Schedules, and Disclosure Schedule to this Agreement, and not to any particular Article, Section, preamble, recital, or other subdivision of this Agreement or Appendix, Exhibit, Schedule, or Disclosure Schedule to this Agreement.
- (f) The headings contained in this Agreement, any Exhibit or Schedule hereto, the Disclosure Schedule and in the table of contents to this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. The Disclosure Schedule and all Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Exhibit, Schedule or in the Disclosure Schedule but not otherwise defined therein, shall have the meaning as defined in this Agreement.
- (g) The term—fraud—shall not include constructive fraud or other claims based on constructive knowledge, negligent misrepresentation or similar theories.
- 11.17 Waivers; Terminations. Recognizing that Brownstein Hyatt Farber Schreck, LLP has acted as legal counsel to Seller and certain of its Affiliates, and the Company prior to the Closing, and that Brownstein Hyatt Farber Schreck, LLP intends to act as legal counsel to Seller and certain of their respective Affiliates (which will no longer include the Company) after the Closing, each of Purchaser, the Company and the Company s Subsidiaries hereby waives, on its own behalf and agrees to cause its Affiliates to waive, any conflicts that may arise in connection with Brownstein Hyatt Farber Schreck, LLP representing Seller and/or any of Seller s Affiliates after the Closing as such representation may relate to Purchaser, or any of the Brand Companies or the Contemplated Transactions (including in respect of litigation). In addition, all communications involving attorney-client confidences between Seller, its Affiliates, the Brand Companies, on the one hand, and Brownstein Hyatt Farber Schreck, LLP on the other hand, in the course of the negotiation, documentation and consummation of the transactions contemplated hereby shall be deemed to be attorney-client confidences that belong solely to Seller and its respective Affiliates (and not the Brand Companies). Accordingly, Purchaser agrees that the Brand Companies shall not have access to any such communications, or to the files of Brownstein Hyatt Farber Schreck, LLP and relating to their engagement in connection with the Contemplated

Transactions, whether or not the Closing shall have occurred. Without limiting the generality of the foregoing, upon and after the Closing, (a) Seller and its Affiliates (and not the Brand Companies) shall be the sole holders of the attorney-client

A-73

privilege with respect to such engagement, and none of the Brand Companies shall be a holder thereof, (b) to the extent that files of Brownstein Hyatt Farber Schreck, LLP in respect of such engagement constitute property of the client, only Seller and its Affiliates (and not the Brand Companies) shall hold such property rights and (c) Brownstein Hyatt Farber Schreck, LLP shall not have any duty whatsoever to reveal or disclose any such attorney-client communications or files to any of the Brand Companies by reason of any attorney-client relationship between Brownstein Hyatt Farber Schreck, LLP and the Brand Companies or otherwise. This Section 11.17 is for the benefit of Seller and Brownstein Hyatt Farber Schreck, LLP, and Brownstein Hyatt Farber Schreck, LLP is an intended third-party beneficiary of this Section 11.17. This Section 11.17 shall be irrevocable, and no term of this Section 11.17 may be amended, waived or modified, without the prior written consent of Seller.

11.18 Non-Recourse. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, financial advisor, accountant, advisor, agent, attorney or other representative of Seller, Purchaser or any of their respective Affiliates or Representatives shall have any liability for any obligations or liabilities of Seller or Purchaser (as applicable) under this Agreement or the Collateral Agreements or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby, except that nothing in this Section 11.18 shall limit the express obligations of a Party under this Agreement or the Collateral Agreements.

11.19 Time of Essence. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.

11.20 Further Assurances. From and after the Closing, upon the request of a Party, each other Party shall do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the Contemplated Transactions, including using commercially reasonable efforts to provide (at Purchaser s sole cost) any information requested by Purchaser pertaining to the Company Intellectual Property, to the extent such information is readily available and within Seller s possession, in order for Purchaser (and not Seller) to prosecute, maintain and enforce the Company Intellectual and to register the same in any appropriate Governmental Entity; *provided, however*, that Purchaser shall use its best efforts to aggregate all such requests for information or action from Seller and to avoid making isolated requested for the same.

11.21 Parent Guaranty.

- (a) Parent irrevocably and unconditionally guarantees the prompt, complete and punctual payment of all payment obligations of Purchaser under this Agreement (<u>Guaranteed Obligations</u>). In furtherance of the foregoing and without limiting the generality thereof, the parties hereto agree that (i) Parent s guaranty under this <u>Section 11.2</u>1 (this <u>Guaranty</u>) is a guaranty of payment when due and not collectability; (ii) this Guaranty is a primary obligation of Parent and not merely a contract of surety; and (iii) payment by Parent of a portion, but not all, of the Guaranteed Obligations shall in no way limit, affect, modify or abridge any liability of Parent for any portion of the Guaranteed Obligations which have not been paid or performed.
- (b) Seller may recover from Parent the full amount of any Guaranteed Obligation on demand, but only after Purchaser has breached or defaulted on such Guaranteed Obligation. Nothing herein shall require Seller to first seek or exhaust any remedy against Purchaser, its successors and assigns, or any other person obligated with respect to the Guaranteed Obligations. It is agreed that Seller may, upon any breach or default of Purchaser of a Guaranteed Obligation, or at any time thereafter, make demand upon Parent and receive payment and performance of such Guaranteed Obligation, with or without notice or demand for payment or performance by Purchaser, its successors or assigns, or any other person. Suit may be brought and maintained against Parent, at Seller s election, without joinder of Purchaser or any other person as parties thereto.

(c) Parent agrees that its obligations to Seller under this Guaranty shall be primary, absolute, continuing and unconditional (other than a discharge of Parent with respect to any Guaranteed Obligation as a

A-74

result of performance of such Guaranteed Obligation in accordance with its terms or as a result of defenses to the performance of such Guaranteed Obligation that would be available to Purchaser hereunder), irrespective of and unaffected by any of the following actions or circumstances (regardless of any notice to or consent of Parent): (i) the genuineness, validity, regularity and enforceability of this Agreement or any other document executed in connection herewith; (ii) any extension, renewal, amendment, change, waiver or other modification of this Agreement or any other document executed in connection herewith; (iii) Purchaser s voluntary or involuntary bankruptcy, assignment for the benefit of creditors, reorganization, or similar proceedings affecting Purchaser or any of its assets; or (iv) any other action or circumstances which might otherwise constitute a legal or equitable discharge or defense of a surety or guarantor (other than as otherwise provided in this Section 11.21(c)).

- (d) To the extent the waiver of Parent s rights of subrogation, reimbursement and contribution as set forth herein is found by a court of competent jurisdiction to be void or voidable for any reason, Parent s rights of subrogation and reimbursement against Purchaser, shall be junior and subordinate to any rights Seller may have against Purchaser.
- (e) Any modification, limitation or discharge of all or any part of the Guaranteed Obligations arising out of or by virtue of any bankruptcy, reorganization or similar proceeding for relief of Purchaser under Applicable Law initiated by or against Purchaser shall not modify, limit, lessen, reduce, impair, discharge or otherwise affect Parent s obligations hereunder in any manner whatsoever. If at any time any payment or portion thereof of any Guaranteed Obligation, whether or not made by or for the account of Parent, is set aside by any court or trustee having jurisdiction as a voidable preference or fraudulent conveyance or must otherwise be restored or returned by Seller to Purchaser under any insolvency, bankruptcy or other Applicable Laws or as a result of any dissolution, liquidation or reorganization of Purchaser or upon or as a result of the appointment of any receiver, intervenor or conservator of, or trustee or similar officer of Purchaser or any substantial part of Purchaser s properties or assets, or in connection with any compromise or settlement relating to any of the above, Parent hereby agrees that this Guaranty shall continue and remain in full force and effect or be reinstated, as the case may be, all as though such payment(s) had not been made.

11.22 Release. Effective upon the Closing, Boulder Road, LLC, the landlord under the Boulder Road Lease, on behalf of itself and its Affiliates hereby waives, releases and discharges Purchaser and the Brand Companies from any and all claims, causes of action, liabilities, losses, costs, damages, penalties, charges, expenses and all other forms of liability or obligation whatsoever, in law or equity, arising prior to the Closing Date under the Boulder Road Lease.

[Signature Page Follows]

A-75

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first written above.

PURCHASER:

STRETCH & BEND HOLDINGS, LLC

By: /s/ Yehuda Shmidman Name: Yehuda Shmidman

Title: CEO

PARENT:

SEQUENTIAL BRANDS GROUP, INC.,

solely for the purposes of Section 7.17 and Article XI

By: /s/ Yehuda Shmidman Name: Yehuda Shmidman

Title: CEO

[Signature Page to Membership Interest Purchase Agreement]

SELLER:

GAIAM, INC.

By: /s/ Jirka Rysavy Name: Jirka Rysavy Title: Chairman

[Signature Page to Membership Interest Purchase Agreement]

LANDLORD:

BOULDER ROAD, LLC, solely for the purposes of Section 11.22.

By: /s/ John Jackson
Name: John Jackson

Title: VP

[Signature Page to Membership Interest Purchase Agreement]

Annex A

Brand Business

The Brand Business includes the Gaiam, Gaiam Restore, and SPRI branded products. The Brand Business develops, markets and sells fitness and yoga accessories, apparel, and media. These products are sold primarily through major national retailers in the United States, Canada, Europe, and other countries, but also through websites and catalogs. The products are targeted to all levels of yoga and fitness enthusiasts, including professionals. Accessories include yoga mats, bags, straps and blocks, media content including digital media and apps, restorative and massage accessories such as rollers, resistance cords and balance balls, and various other offerings. Apparel products include pants, shorts, tops and jackets designed around yoga.

The Brand Business has a broad distribution network including brick-and-mortar retailers, third party online retailers, third party digital partners as well as the Brand Business own e-commerce websites (gaiam.com and spri.com websites). The Brand Business also operates internationally with both the Gaiam and SPRI brands in Canada, the UK, Europe, and Korea. The Brand Business is vertically integrated from product design and content creation, to product development and sourcing, and ultimately to customer service and distribution.

A-79

Annex B

Gaia Business

Gaia, Inc. (<u>Gaia</u>) operates a global video service (known as gaia) which currently features approximately 7,000 video titles specially curated for its subscriber base. Gaia s video content is available to its subscribers via online digital streaming on virtually any Internet-connected device, anytime, anywhere, on a commercial-free basis. In addition to streaming, Gaia s subscribers can download Gaia s video content to their devices, so they can view our content without being actively connected to the Internet. Through the Gaia service, Gaia s subscribers have unlimited access to a vast library of inspiring films, personal growth-related content, cutting edge documentaries, interviews, yoga classes, fitness and more.

Gaia s core strategy is to grow our business on a worldwide basis by continuing to expand our unique and exclusive content library (including by producing and acquiring new content, and forging into new areas such as longevity), forging strong distribution relationships and growing our conscious community.

In addition to the online Gaia service, Gaia has also created a fitness and yoga-focused version of its video service (known as GaiamTV Fit & Yoga) which is marketed and sold to subscribers through third party partners such as Comcast and Verizon FiOS.

Gaia also operates a legacy DVD subscription club that provides members with a curated selection of video content on DVD, on a monthly basis. Certain Gaia content is also available for purchase by the general public via download or on DVD on third party platforms such as Amazon and iTunes.

A-80

LIST OF OMITTED SCHEDULES AND EXHIBITS

<u>Schedules</u>	
Schedule 1.01(a)	Funded Indebtedness
Schedule 1.01(b)	Knowledge Groups
Schedule 1.01(c)	Gaia Agreements
Schedule 1.01(d)	Affiliate Agreements
Schedule 1.01(e)	Material Registered Company Intellectual Property
Schedule 2.05(b)	Pro Forma Closing Balance Sheet
Schedule 7.01	Conduct of Business
Schedule 7.06(b)	Material Consents
Schedule 8.01(a)	Antitrust Laws
Schedule 8.02(d)(iii)	Resigning Persons

Exhibits

<u>Exilibits</u>	
Exhibit A	Contribution Agreement
Exhibit B	Form of Escrow Agreement
Exhibit C	Form of IP License Agreement
Exhibit D	Form of Transition Services Agreement
Exhibit E	Form of SVOD Rights Assignment Agreement
Exhibit F	Form of SVOD Rights Sub-License Agreement
Exhibit G	Form of Reception Services Agreement
Exhibit H	Form of Third Amendment and Assignment of Lease

A-81

ANNEX B

FFL PURCHASE AGREEMENT

[Remainder of Page Intentionally Blank]

ANNEX B

ASSET PURCHASE AGREEMENT

by and between

FIT FOR LIFE LLC,

a Delaware limited liability company,

as Purchaser,

and

GAIAM, INC.,

a Colorado corporation,

as Seller

Dated as of May 10, 2016

TABLE OF CONTENTS

ARTIC	CLE I DEFINITIONS	Page B-1
1.01	Definitions	B-1
ARTICLE II PURCHASE AND SALE		В-3
2.01 2.02 2.03 2.04 2.05 2.06	Sale of Assets; Excluded Assets Assumption of Liabilities; Excluded Liabilities Purchase Price Purchase Price Adjustment Working Capital Adjustment Procedures Inventory Reserve Adjustment	B-3 B-4 B-4 B-5 B-6 B-7
ARTIC	CLE III CLOSING	B-7
3.01 3.02	Closing Date Deliveries at Closing	B-7 B-8
ARTIC	CLE IV REPRESENTATIONS AND WARRANTIES RELATING TO SELLER	B-8
4.01 4.02 4.03 4.04	Organization Authority; Execution and Delivery; Enforceability Noncontravention Brokers	B-8 B-8 B-9 B-9
ARTIC	CLE V REPRESENTATIONS AND WARRANTIES OF PURCHASER	B-9
5.01 5.02 5.03 5.04 5.05 5.06	Authority; Execution and Delivery; Enforceability Noncontravention Brokers Solvency Financing Non-Reliance of Purchaser	B-9 B-10 B-10 B-10 B-11
	CLE VI TERMINATION	B-11
6.01 6.02	Termination Effect of Termination	B-11 B-12
ARTIC	CLE VII FINANCING COVENANTS	B-12
7.01	Financing	B-12
ARTIC	CLE VIII GENERAL PROVISIONS AND CERTAIN COVENANTS	B-13
8.01 8.02 8.03 8.04 8.05 8.06	Restrictive Covenants Assignment No Third-Party Beneficiaries Notices Amendments Remedies	B-13 B-15 B-15 B-16 B-16
0.00	Remedies	D- 10

8.07	Construction	B-17
8.08	Survival and Expiration of Representations and Warranties and Covenants	B-17
8.09	Further Assurances	B-17
8.10	Exclusivity of Agreement	B-17
8.11	No Third Party Liability	B-18
8.12	Counterparts	B-18

-i-

TABLE OF CONTENTS

(continued)

		Page
8.13	Entire Agreement	B-18
8.14	Severability	B-18
8.15	Governing Law; Venue	B-18
8.16	Attorney s Fees	B-19
8.17	Waiver of Jury Trial	B-19
8.18	Waiver	B-19
8.19	Transfer Taxes	B-19
8.20	Non-Recourse	B-19
8.21	Time of Essence	B-20

-ii-

SCHEDULES AND EXHIBITS

Schedules

Schedule 1.01(c) Working Capital Target Calculation

Schedule 2.04(b) Closing Working Capital Amount Example

Exhibits

Exhibit A Form of Reception Services Agreement

Exhibit B Form of Third Amendment and Assignment of Lease

Exhibit C Form of Transition Services Agreement

Exhibit D Form of Working Capital Assignment and Bill of Sale

-111-

ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement, dated as of May 10, 2016 (this <u>Agreement</u>), is by and between (i) FIT FOR LIFE LLC, a Delaware limited liability company (<u>Purchaser</u>); and (ii) GAIAM, INC., a Colorado corporation (<u>Seller</u>). Certain terms used in this Agreement are defined in <u>Section 1.01</u>.

RECITALS

- A. Seller is a party to that certain Membership Interest Purchase Agreement, dated as of the date hereof (the <u>MIPA</u>), by and among Stretch & Bend Holdings, LLC, a Delaware limited liability company (<u>SBH</u>), Sequential Brands Group, Inc., a Delaware corporation (<u>Parent</u>), and Seller, a copy of which is attached hereto <u>as Annex</u> A.
- B. It is contemplated that, pursuant to the MIPA, SBH would acquire the Brand Business (as defined in the MIPA), including, but not limited to, the Company Trademarks (as defined in the MIPA), which includes the Gaiam trademark, through the purchase of all of the equity interests of Gaiam Brand Holdco LLC, a Delaware limited liability company (<u>HoldCo</u>, and together with HoldCo s Subsidiaries, the <u>Brand Companies</u>).
- C. The MIPA provides that, immediately prior to the acquisition by SBH of all of the equity interests of HoldCo,
 (i) Seller shall cause the Brand Companies to sell, to Purchaser certain assets as specified herein and
 (ii) Purchaser shall assume certain liabilities of the Brand Companies as specified herein, in each case on the terms and conditions and as more specifically provided in this Agreement.
- D. Concurrently with the execution and delivery of this Agreement, SBH and Purchaser shall enter into that certain Assignment and Assumption Agreement, dated as of the date hereof (the <u>SBH-Purchaser AAA</u>), pursuant to which, after the closing of the transactions contemplated by the MIPA, (i) SBH will assign, or cause the Brand Companies to assign, to Purchaser certain contracts as provided therein and (ii) Purchaser will assume from the Brand Companies certain liabilities as provided therein, a copy of which is attached hereto as <u>Annex B</u>.

AGREEMENT

In consideration of the covenants and agreements contained herein and the other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Seller and Purchaser agree as follows:

ARTICLE I

DEFINITIONS

1.01 Definitions. Capitalized terms and other terms used in this Agreement have the following respective meanings:

Accounting Expert has the meaning set forth in Section 2.05(c)(i).

<u>Accrued Liabilities Credit Amount</u> means an amount equal to \$500,000.

Adjustment Escrow Account has the meaning ascribed to such term in the MIPA,

Adjustment Escrow Amount has the meaning ascribed to such term in the MIPA.

Affiliate has the meaning ascribed to such term in the MIPA.

Agreement has the meaning set forth in the opening paragraph of this Agreement.

Applicable Law has the meaning ascribed to such term in the MIPA.

B-1

Board means the board of directors of Seller.

Boulder Road Lease has the meaning ascribed to such term in the MIPA.

<u>Brand Companies</u> has the meaning set forth in the Recitals.

Business Day has the meaning ascribed to such term in the MIPA.

<u>Closing</u> has the meaning set forth in Section 3.01.

<u>Closing Date</u> has the meaning set forth <u>in Section 3</u>.01.

<u>Closing Statement</u> has the meaning set forth <u>in Section 2.05(a)</u>.

<u>Closing Working Capital Amount</u> means the amount of the Working Capital of the Brand Companies as of the Closing Date and immediately prior to the Closing, as determined in accordance with <u>Section 2.05</u>.

<u>Collateral Agreements</u> means the documents, instruments, certificates and agreements executed in connection with this Agreement, including the Reception Services Agreement, Third Amendment and Assignment of Lease, Transition Services Agreement and the Working Capital Assignment and Bill of Sale.

<u>Company</u> has the meaning set forth in the Recitals.

<u>Contemplated Transactions</u> means Purchaser s acquisition of the Acquired Assets, assumption of the Assumed Liabilities and all other transactions contemplated by this Agreement.

<u>Current Assets</u> means only the Brand Companies current assets identified as being included in the calculation of Working Capital on <u>Schedule 2.04(b)</u>, which, to the extent identified as being included in the calculation of Working Capital, shall be determined in accordance with the Brand Companies historical practices actually used to prepare Schedule 2.04(b), subject to Section 2.06.

<u>Current Liabilities</u> means only the Brand Companies current liabilities identified as being included in the calculation of Working Capital on <u>Schedule 2.04(b)</u>, which, to the extent identified as being included in the calculation of Working Capital, shall be determined in accordance with the Brand Companies historical practices actually used to prepare <u>Schedule 2.04(b)</u>.

<u>Disputed Items</u> has the meaning set forth in Section 2.05(b).

Escrow Agent means UMB Bank, N.A.

<u>Escrow Agreement</u> has the meaning ascribed to such term in the MIPA.

<u>Estimated Closing Statement</u> has the meaning set forth in Section 2.04(b).

Estimated Closing Working Capital Amount has the meaning set forth in Section 2.04(b).

<u>Final Net Closing Cash Consideration</u> means an amount equal to (a) the Purchase Price, minus (b) the Working Capital Target, plus (c) the Closing Working Capital Amount (as finally determined pursuant to <u>Section 2.05</u>) minus

(d) the Accrued Liabilities Credit Amount.

<u>Liens</u> has the meaning ascribed to such term in the MIPA.

<u>Net Closing Cash Consideration</u> means an amount equal to (a) the Purchase Price, minus (b) the Working Capital Target, plus (c) the Estimated Closing Working Capital Amount, minus (d) the Accrued Liabilities Credit Amount.

B-2

- Net Deficit Amount has the meaning set forth in Section 2.04(c)(ii).
- Net Overage Amount has the meaning set forth in Section 2.04(c)(i).
- Objection Notice has the meaning set forth in Section 2.05(b).
- Party or Parties means Purchaser or Seller.
- <u>Permitted Liens</u> has the meaning ascribed to such term in the MIPA.
- <u>Person</u> has the meaning ascribed to such term in the MIPA.
- <u>Purchase Price</u> has the meaning set forth in Section 2.03.
- <u>Purchaser</u> has the meaning set forth in the opening paragraph of this Agreement.
- <u>Reception Services Agreement</u> means that certain Reception Services Agreement in substantially the form attached hereto as <u>Exhibit A</u>.
- <u>Representative</u> has the meaning ascribed to such term in the MIPA.
- <u>Seller</u> has the meaning set forth in the opening paragraph of this Agreement.
- <u>Subsidiary</u> has the meaning ascribed to such term in the MIPA.
- <u>Third Amendment and Assignment of Lease</u> means that certain Third Amendment to and Assignment and Assumption of Lease in substantially the form attached hereto as <u>Exhibit B</u>.
- <u>Transferred Employees</u> has the meaning ascribed to such term in the MIPA.
- <u>Transition Services Agreement</u> means that certain Transition Service Agreement in substantially the form attached hereto as <u>Exhibit C</u>.
- Working Capital means Current Assets minus Current Liabilities.
- <u>Working Capital Assignment and Bill of Sale</u> means an assignment and assumption agreement and bill of sale, substantially in the form attached hereto as <u>Exhibit D</u>.
- Working Capital Target means an amount equal to \$26,840,308 calculated in accordance with Schedule 1.01(c).

ARTICLE II

PURCHASE AND SALE

2.01 Sale of Assets; Excluded Assets.

(a) Acquired Assets. Subject to the terms and conditions of this Agreement, Purchaser agrees to purchase from the applicable Brand Company, and Seller agrees to cause the applicable Brand Company to sell, convey, transfer, assign

and deliver to Purchaser, on the Closing Date, the Acquired Assets (as defined below) and all of the applicable Brand Company s right, title and interest in and to the Acquired Assets, free and clear of any Liens of any kind whatsoever except Permitted Liens. The <u>Acquired Assets</u> shall mean all the right, title

B-3

and interest of the applicable Brand Company as of the Closing Date in and to the following assets relating to the business of the Brand Companies: (i) inventory, (ii) accounts receivable, and (iii) all other physical assets including, without limitation, equipment, furniture and other operating assets.

- (b) Excluded Assets. Notwithstanding anything herein to the contrary, the Acquired Assets shall not include (i) any assets of the Brand Companies other than the Acquired Assets, (ii) any cash or cash equivalents of the Brand Companies, or (iii) any Intellectual Property (as defined in the MIPA) owned by any of the Brand Companies.
- (c) AS IS PURCHASE. NEITHER SELLER NOR ANY OF ITS AFFILIATES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES IS MAKING ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, OF ANY NATURE WHATSOEVER WITH RESPECT TO ANY OF THE ACQUIRED ASSETS, AND THE CONDITION OF THE ACQUIRED ASSETS SHALL BE AS IS AND WHERE IS. SELLER DISCLAIMS ANY OTHER REPRESENTATIONS OR WARRANTIES, WHETHER MADE BY SELLER OR ANY OF ITS AFFILIATES, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES MADE BY SELLER CONTAINED IN ARTICLE IV AND HEREBY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, PROJECTION, FORECAST, STATEMENT, OR INFORMATION MADE, COMMUNICATED, OR FURNISHED (ORALLY OR IN WRITING) TO PURCHASER OR ANY OF ITS AFFILIATES OR REPRESENTATIVES.

2.02 Assumption of Liabilities; Excluded Liabilities.

- (a) Assumed Liabilities. Subject to the terms and conditions of this Agreement, Purchaser agrees, effective at the Closing, to assume and to pay, perform and discharge when due, only the following (and no other) liabilities and obligations: (i) the trade payables of the Brand Companies reflected on the Closing Statement (as finally determined pursuant to Section 2.05), (ii) any and all other liabilities and obligations (whether known or unknown, fixed, absolute, matured, unmatured, accrued or contingent) arising from or related to the Acquired Assets in respect of the period on or after the Closing Date, and (iii) all obligations and liabilities set forth on the Closing Statement (as finally determined pursuant to Section 2.05) (collectively, the Assumed Liabilities).
- (b) Excluded Liabilities. Notwithstanding anything contained herein to the contrary, Purchaser shall not assume, or cause to be assumed, or be deemed to have assumed or caused to have assumed or be liable or responsible for any liabilities or obligations (whether known or unknown, fixed, absolute, matured, unmatured, accrued or contingent, now existing or arising after the date hereof), other than the Assumed Liabilities (such liabilities and obligations not assumed, the Excluded Liabilities). For the avoidance of doubt, Excluded Liabilities shall include but not be limited to (i) any claims by Transferred Employees (as defined in the MIPA) arising from their employment with any Brand Company prior to the Closing Date, (ii) any liabilities and obligations of any Brand Company that do not arise from and are not related to any Acquired Asset, (iii) any liabilities and obligations of any Brand Company to the extent based upon a theory of successor liability, including any successor liability claims with respect to employees of the Brand Companies (provided, that this clause (iii) shall not apply in any respect to Purchaser's assumption of the Assumed Liabilities), (iv) any liabilities and obligations under any agreement, contract or license relating to any event occurring prior to Closing and (v) any claims by, liabilities or obligations to any employee or former employee of any of the Brand Companies, other than Transferred Employees, under the federal Worker Adjustment and Retraining Notification Act of 1988, as amended, or under any state or local plant closing or mass layoff law. All Excluded Liabilities will remain liabilities of the Brand Companies following the Closing.
- **2.03 Purchase Price**. In consideration for the purchase of the Acquired Assets and the assumption of the Assumed Liabilities, Purchaser shall pay at Closing to Seller, in accordance with <u>Section 3.02(a)</u>, an aggregate purchase price equal to US \$21,300,000 (the <u>Purchase Price</u>), subject to adjustment in accordance with <u>Sections 2.04 and 2.05</u>.

B-4

2.04 Purchase Price Adjustment.

- (a) General. The Purchase Price shall be subject to adjustment at the Closing, and the Net Closing Cash Consideration shall be subject to adjustment following the Closing, in each case, in accordance with this Section 2.04 and Section 2.05.
- (b) Closing Adjustment. As soon as practicable before the Closing, but in no event later than 5:00 p.m. (Mountain Time) on the date that is two (2) Business Days prior to the Closing Date, Seller shall deliver to Purchaser a statement (the <u>Estimated Closing Statement</u>) setting forth in reasonable detail Seller s (A) estimate of the Closing Working Capital Amount, which estimate shall not exceed an amount equal to the Working Capital Target, plus \$5,000,000 (such estimate, the <u>Estimated Closing Working Capital Amount</u>), and (B) calculation of the Net Closing Cash Consideration based on the foregoing, in each case, prepared by Seller in good faith in accordance with the Brand Companies historical practices and <u>Section 2.06</u> and presented in a manner consistent with the illustrative example attached hereto as <u>Schedule 2.04(b)</u>. In preparing the Estimated Closing Statement, Seller shall employ the same judgments, accounting methods, policies, practices, procedures, classifications or estimation methodologies employed by Seller in preparing <u>Schedule 1.01(c)</u> and <u>Schedule 2.04(b)</u>. Together with the Estimated Closing Statement, Seller shall use good faith efforts to deliver to Purchaser an estimate of the breakdown between non-apparel and apparel assets and liabilities included in the Estimated Closing Working Capital Amount, it being acknowledged and agreed by Seller and Purchaser that Seller is not making any representation or warranty hereunder with respect to such estimate.
- (c) **Post-Closing Adjustment**. After the final calculation of Final Net Closing Cash Consideration (as determined pursuant to <u>Section 2.05</u> below):
- (i) If the Final Net Closing Cash Consideration minus the Net Closing Cash Consideration is greater than \$0 (such difference, the Net Overage Amount), then, within three (3) Business Days following the determination of the Final Net Closing Cash Consideration in accordance with Section 2.05, Purchaser shall deliver to Seller the Net Overage Amount by wire transfer of immediately available funds to the account or accounts designated by Seller; and
- (ii) If the Final Net Closing Cash Consideration minus the Net Closing Cash Consideration is less than \$0 (such difference a Net Deficit Amount), then, within three (3) Business Days following the determination of the Final Net Closing Cash Consideration in accordance with Section 2.05, Seller and SBH shall deliver a joint instruction to the Escrow Agent to distribute from the Adjustment Escrow Account (A) to Purchaser the Net Deficit Amount by wire transfer of immediately available funds to the account or accounts designated by Purchaser, and (B) to Seller any funds remaining in the Adjustment Escrow Account by wire transfer of immediately available funds to the account or accounts designated by Seller; provided, however, that in no event shall the amount payable by Seller under this Section 2.04(c)(ii), if any, exceed an amount equal to (1) the Adjustment Escrow Amount less (2) any amounts payable to SBH under Section 2.05(c)(ii) of the MIPA less (3) any amounts payable from the Adjustment Escrow Amount to the Accounting Referee under Section 2.06(c)(iii) of the MIPA or the Accounting Expert under Section 2.05(c)(iii).
- (d) Acknowledgement. The Parties agree that the purpose of determining the Closing Working Capital Amount and the Final Net Closing Cash Consideration pursuant to Section 2.05 is to correct for any differences between the amounts of the foregoing as set forth on the Estimated Closing Statement and the actual amounts thereof as of the Closing Date, and the adjustment procedures set forth in this Section 2.04 and Section 2.05 are not intended to permit the introduction of new or different judgments, accounting methods, policies, practices, procedures, classifications, or estimation methodologies for the purpose of determining the Final Net Closing Cash Consideration or the components thereof. Purchaser further acknowledges and agrees that that the calculation of Final Net Closing Consideration shall

be made in accordance with this <u>Section 2.04</u> and <u>Section 2.05</u>, and shall not take into account any breach of Seller s representations, warranties or covenants under the MIPA which, for the avoidance of doubt, shall be addressed as claims for indemnification under Article X of the MIPA. For purposes of this Agreement, in the event of any uncertainty or disagreement with respect to the

B-5

judgments, accounting methods, policies, practices, procedures, classifications or estimation methodologies employed in the preparation of <u>Schedule 1.01(c)</u>, to the extent any adjustment is made to any line item in the Closing Balance Sheet to reflect or address such uncertainty or disagreement, the same corresponding adjustment shall be made to the line item on the Working Capital Target calculation set forth on <u>Schedule 1.01(c)</u>, with the amount of the Working Capital Target adjusted accordingly for purposes of determining the Closing Working Capital Amount.

2.05 Working Capital Adjustment Procedures.

- (a) Preparation of Closing Balance Sheet. As promptly as practicable, but no later than 5:00 p.m. (Mountain Time) on the 75th day after the Closing Date or such later date as Purchaser and Seller agree in writing, Purchaser will prepare or cause to be prepared a statement (the <u>Closing Statement</u>) setting forth in reasonable detail Purchaser s calculation of (i) the Closing Working Capital Amount prepared by Purchaser in good faith in accordance with the Brand Companies historical practices and <u>Section 2.05</u> and presented in a manner consistent with the pro forma example attached hereto as <u>Schedule 2.04(b)</u>, and (ii) the Final Net Closing Cash Consideration based on the foregoing. In the event that Purchaser does not deliver the Closing Statement within such 75-day period, Purchaser shall be conclusively deemed to have accepted the Estimated Closing Working Capital Amount as the Closing Working Capital Amount. Purchaser acknowledges and agrees that, for the purposes of Seller s review of the Closing Statement delivered by Purchaser pursuant to this <u>Section 2.05(a)</u>, Purchaser shall afford Seller commercially reasonable access during normal business hours to the books and records (other than privileged documents) of Purchaser and its Representatives.
- (b) Disagreement by Seller. If Seller disagrees with Purchaser s calculation of the Closing Working Capital Amount, Seller may, within 20 days after receipt of the Closing Statement, deliver a notice to Purchaser disagreeing with such calculation(s) and setting forth in reasonable detail Seller s (i) objection(s) to such calculation(s) of the Closing Working Capital Amount as set forth on the Closing Statement, and (ii) calculation(s) thereof and of the Final Net Closing Cash Consideration (an <u>Objection Notice</u>). Any Objection Notice shall specify those items or amounts as to which Seller disagrees (the <u>Disputed Items</u>). If Seller fails to deliver an Objection Notice during such 20-day period, the Closing Working Capital Amount determined by Purchaser and delivered to Seller in accordance with Section 2.05(a) shall be deemed final, binding, and conclusive for purposes of this Agreement.

(c) Dispute Resolution.

(i) If an Objection Notice shall have been timely delivered by Seller to Purchaser pursuant to Section 2.05(b), Seller and Purchaser shall, during the 10 days following such delivery, use commercially reasonable efforts to reach agreement on the Disputed Items in order to determine the Closing Working Capital Amount (to the extent disputed in the Objection Notice); provided, that the amount of any agreed upon Disputed Item shall fall between (or be equal to either) Purchaser s calculation thereof delivered pursuant to Section 2.05(a) and Seller s calculation thereof delivered pursuant to Section 2.05(b). If, during such 10-day period, Seller and Purchaser agree as to any Disputed Item, then Seller and Purchaser shall execute a written acknowledgement of the amount of such Disputed Item and any Disputed Item so agreed upon shall be deemed final, binding, and conclusive for purposes of this Agreement. If, during such period, Seller and Purchaser are unable to reach agreement on all Disputed Items, then the Disputed Items that remain in dispute shall be resolved by KPMG LLP or, if such firm is conflicted or declines to act in such capacity by the office of such other nationally recognized major firm of independent accountants having no material relationship with any Party and reasonably acceptable to both Seller and Purchaser (the Accounting Expert): provided, that if an Accounting Referee (as defined in the MIPA) is retained pursuant to Section 2.06(c) of the MIPA, then the Parties hereto shall select the Accounting Referee under the MIPA to be the Accounting Expert hereunder.

(ii) Within 15 days of the retention of the Accounting Expert, Purchaser and Seller shall jointly submit the matter to the Accounting Expert and instruct the Accounting Expert that it (A) shall act as an expert in accounting, and not as an arbitrator, to resolve, in accordance with the Brand Companies historical practices and

B-6

Section 2.06 and presented in a manner consistent with Schedule 2.04(b), only the Disputed Items that remain in dispute; (B) shall adjust the calculation(s) of the Closing Working Capital Amount based thereon to reflect such resolution; (C) shall determine the amount of any Disputed Item to be an amount that falls between (or is equal to either) Purchaser s calculation thereof delivered pursuant to Section 2.05(a) and Seller s calculation thereof delivered pursuant to Section 2.05(b); and (D) shall deliver to Purchaser and Seller a written decision as promptly as practicable and in any event within 30 days following the submission of the remaining Disputed Items to the Accounting Expert for resolution. The Accounting Expert shall deliver to Purchaser and Seller, as promptly as practicable, a report setting forth such calculation. The Parties acknowledge and agree that, if any Disputed Item is submitted to the Accounting Expert pursuant to this Section 2.05(c), the Closing Working Capital Amount as determined by the Accounting Expert shall be, absent manifest error, final, binding, and conclusive for purposes of this Agreement.

- (iii) If requested by the Accounting Expert, Purchaser and Seller agree to execute a reasonable engagement letter. The cost of such review and report (including any retainer) shall be borne (A) by Purchaser if Seller s calculation of the Final Net Closing Cash Consideration (based on Seller s position with respect to the Disputed Items submitted to the Accounting Expert) is closer to the Final Net Closing Cash Consideration calculated by the Accounting Expert (based on the Accounting Expert s determination of such Disputed Items), (B) from the Adjustment Escrow Account, if Purchaser s calculation of the Final Net Closing Cash Consideration (based on Purchaser s position with respect to the Disputed Items submitted to the Accounting Expert) is closer to the Final Net Closing Cash Consideration calculated by the Accounting Expert (based on the Accounting Expert s determination of such Disputed Items), and (C) otherwise equally by Purchaser and from the Adjustment Escrow Account. However, initially, any retainer charged by the Accounting Expert shall be paid 50% by Purchaser and 50% from the Adjustment Escrow Account.
- (d) Cooperation. Seller and Purchaser will reasonably cooperate and assist as requested in the preparation of the calculations of the Closing Working Capital Amount and in the conduct of the audits and reviews referred to in this Section 2.05, including the making available to the extent necessary of books, records, work papers, and personnel.
- **2.06 Inventory Reserve Adjustment**. Solely for purposes of calculating the Closing Working Capital Amount, any inventory in the Brand Companies possession which (a) is included in Current Assets and (b) is in excess of a three (3) year supply remaining on hand with the Brand Companies as of the Closing Date, shall be reserved for using the Agreed Upon Reserve Percentage for the applicable categories as set forth on Schedule 2.06, as opposed to in accordance with the Brand Companies historical practices under Section 2.04(b) hereof.

ARTICLE III

CLOSING

3.01 Closing Date. If Seller delivers a written notice to Purchaser containing (a) a statement that the conditions precedent to the consummation of the transactions contemplated by the MIPA have been satisfied or waived in accordance therewith and that Seller and SBH are ready and prepared to close the transactions contemplated by the MIPA (the _MIPA Closing), subject only to the occurrence of the Closing under this Agreement, and (b) the date on which the MIPA Closing is to occur (which date shall be at least three (3) Business Days after the delivery of such notice), then the Parties shall cause the purchase and sale of the Acquired Assets and the assumption of the Assumed Liabilities to be consummated by the exchange of signatures by facsimile or other electronic transmission or, if such exchange is not practicable, at a closing (the _Closing) to be held at the offices of Brownstein Hyatt Farber Schreck, LLP, 410 Seventeenth Street, Suite 220, Denver, Colorado 80202 at 10:00 a.m. (Mountain Time) immediately prior to the MIPA Closing on the date set forth in such notice; provided, however, that in no event shall the Closing occur prior to the 45th day following the date hereof (such date on which the Closing occurs, the _Closing Date), subject only to the subsequent occurrence of the MIPA Closing.

- **3.02 Deliveries at Closing**. Subject to the terms and conditions of this Agreement, at the Closing, the following Persons shall deliver or cause to be delivered the following:
- (a) Purchaser shall deliver to Seller the Net Closing Cash Consideration by wire transfer of immediately available funds to an account or accounts designated by Seller (such designation to occur no less than two Business Days prior to the Closing).
- **(b)** Purchaser shall deliver, or cause to be delivered, to Seller:
- (i) the Third Amendment and Assignment of Lease, executed by Purchaser or an Affiliate of Purchaser designated by Purchaser;
- (ii) the Transition Services Agreement, executed by Purchaser;
- (iii) the Reception Services Agreement, executed by Purchaser; and
- (iv) the Working Capital Assignment and Bill of Sale, executed by Purchaser.
- (c) Seller shall deliver, or cause to be delivered, to Purchaser:
- (i) the Third Amendment and Assignment of Lease, executed by Seller;
- (ii) the Transition Services Agreement, executed by Seller;
- (iii) the Reception Services Agreement, executed by Seller; and
- (iv) the Working Capital Assignment and Bill of Sale, executed by Seller.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES RELATING TO SELLER

Seller hereby represents and warrants to Purchaser, except as set forth in the disclosure schedule delivered by Seller to Purchaser on the date hereof, as follows:

- **4.01 Organization**. Seller is a corporation duly incorporated and is validly existing and in good standing under the laws of the State of Colorado.
- 4.02 Authority; Execution and Delivery; Enforceability.
- (a) Seller has the requisite power and authority to execute and deliver this Agreement and the Collateral Agreements to which Seller is or will be a party, to perform its obligations hereunder and thereunder and, subject to obtaining the Shareholder Consent, to consummate Contemplated Transactions.
- (b) The execution, delivery and performance by Seller of this Agreement and the Collateral Agreements to which Seller is or will be a party, and the consummation by Seller of the Contemplated Transactions, have been duly authorized by all necessary action on the part of Seller, and, subject to obtaining the Shareholder Consent, no other action on the part of Seller is or will be necessary to authorize the execution, delivery and performance of this

Agreement and the Collateral Agreements to which Seller is or will be a party and the Contemplated Transactions. This Agreement has been, and each of the Collateral Agreements to which Seller is a party will be, duly executed and delivered by Seller.

B-8

- (c) This Agreement constitutes, and each of the Collateral Agreements to which Seller is a party will constitute, the legal, valid and binding obligations of Seller, enforceable against Seller in accordance with their terms, except to the extent that their enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors rights generally and to general equitable principles.
- (d) The Board, by resolutions duly adopted by vote at a meeting of the Board duly called and held at which a quorum was present, has (i) determined that this Agreement and the Contemplated Transactions are advisable and in the best interests of Seller and its shareholders, (ii) authorized, adopted and approved the execution, delivery and performance of this Agreement and the Contemplated Transactions, (iii) approved and declared advisable this Agreement and the Contemplated Transactions, and (iv) resolved to recommend that the shareholders of Seller adopt and approve the consummation of the Contemplated Transactions, and such resolutions have not been rescinded, withdrawn or modified in any way as of the date hereof.
- **4.03** Noncontravention. Neither the execution and delivery of any of this Agreement or the Collateral Agreements to which Seller is or will be a party, nor the consummation or performance of any of the Contemplated Transactions, will, subject to obtaining the Shareholder Consent, (i) in any material respect contravene, conflict with or result in a violation of any Applicable Law or any Judgment to which Seller is subject or by which any of its properties or assets are bound, assuming that all filings and notifications in accordance with the HSR Act have been made and the applicable waiting period has expired or been terminated; (ii) in any material respect contravene, conflict with or result in a breach of, or constitute a default under, or result in the acceleration of any obligation or loss of any benefits under, the provisions of any material Contract to which Gaiam is subject or by which any of its properties or assets are bound; or (iii) contravene, conflict with or result in a violation of any of the provisions of the Organizational Documents of Seller.
- **4.04 Brokers**. Other than the Banker, Seller has not retained any Person to act as a broker or agreed or become obligated to pay, or has taken any action that might result in any Person claiming to be entitled to receive, any brokerage commission, finder s fee or similar commission or fee in connection with any of the Contemplated Transactions.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller as follows:

5.01 Authority; Execution and Delivery; Enforceability.

- (a) Purchaser has the requisite power and authority to execute and deliver this Agreement and the Collateral Agreements to which Purchaser is or will be a party, to perform its obligations hereunder and thereunder and to consummate the Contemplated Transactions.
- (b) The execution, delivery and performance by Purchaser of this Agreement and the Collateral Agreements to which Purchaser is or will be a party and the consummation by Purchaser of the Contemplated Transactions, have been duly authorized by all necessary limited liability company action on the part of Purchaser, and no other action on the part of Purchaser is or will be necessary to authorize the execution, delivery and performance of this Agreement and the Collateral Agreements to which Purchaser is or will be a party and the Contemplated Transactions.

(c) This Agreement has been, and each of the Collateral Agreements to which Purchaser is a party will be, duly executed and delivered by Purchaser. This Agreement constitutes, and each of the Collateral Agreements to which Purchaser is a party will constitute, the legal, valid and binding obligation of Purchaser, enforceable

B-9

against Purchaser in accordance with their terms, except to the extent that their enforceability may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors—rights generally and to general equitable principles.

- **5.02** Noncontravention. Neither the execution and delivery of this Agreement or the Collateral Agreements to which Purchaser is or will be a party, nor the consummation or performance of any of the Contemplated Transactions, will (a) in any material respect contravene, conflict with or result in a violation of any Applicable Law or any Judgment to which Purchaser is subject or by which any of its properties or assets are bound; (b) in any material respect, contravene, conflict with or result in a breach of, or constitute a default under, or result in the acceleration of any obligation or loss of any benefits under, the provisions of any material Contract to which Purchaser is subject or by which any of its properties or assets are bound; or (c) contravene, conflict with or result in a violation of any of the provisions of the Organizational Documents of Purchaser.
- **5.03 Brokers**. Purchaser has not retained any Person to act as a broker or agreed or become obligated to pay, or has taken any action that might result in any Person claiming to be entitled to receive, any brokerage commission, finder s fee or similar commission or fee in connection with any of the Contemplated Transactions.
- **5.04 Solvency**. As of the Closing Purchaser will not (a) be insolvent (either because its financial condition is such that the sum of its debts is greater than the fair value of its assets or because the present fair salable value of its assets will be less than the amount required to pay its probable liability on its debts as they become absolute and matured), (b) have unreasonably small capital with which to engage in its business or (c) have incurred debts beyond its ability to repay such debts as they become absolute and matured.

5.05 Financing.

- (a) Purchaser has delivered to Seller true and complete copies of a duly executed commitment letter, dated as of the date hereof, between Purchaser and Israel Discount Bank of New York (the <u>Debt Commitment Letter</u>) (subject to redaction so long as such redaction does not cover terms that would adversely affect the conditionality or availability of the Debt Financing), pursuant to which the counterparties thereto have committed, subject to the terms and conditions thereof, to lend the amounts set forth therein (the <u>Debt Financing</u>). Assuming the funding in full of the Debt Financing on the Closing Date, Purchaser on the Closing Date will have, together with cash on hand and available borrowing under existing credit facilities, sufficient funds in immediately available cash, to (A) pay an amount in cash equal to the amounts set forth in <u>Section 3.02(a)</u>; (B) pay any and all fees and expenses required to be paid by Purchaser in connection with the transactions contemplated by this Agreement and the Debt Financing; and (C) satisfy all other payment obligations of Purchaser contemplated hereunder (including the payment at Closing of unpaid Transaction Expenses). Purchaser has also delivered to Seller a true and complete copy of any fee letter in connection with the Debt Commitment Letter (any such fee letter, a <u>Fee Letter</u>) (subject to redaction so long as such redaction does not cover terms (other than the actual amount of fees that are the subject of the Fee Letter) that would adversely affect the conditionality or availability of the Debt Financing).
- (b) As of the date hereof, there are no side letters or other contracts or arrangements related to the Debt Financing other than the Debt Commitment Letter and the Fee Letter or as otherwise expressly set forth therein. Neither the Debt Commitment Letter nor the Fee Letter has been amended or modified, no such amendment or modification is contemplated as of the date hereof, and the commitments set forth in the Debt Commitment Letter have not been withdrawn or rescinded in any respect.
- (c) The Debt Commitment Letter is in full force and effect and is the valid, binding and enforceable obligations of Purchaser and, to the knowledge of Purchaser, the other parties thereto, except to the extent that its enforceability may

be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors—rights generally and to general equitable principles. There are no conditions precedent or other contingencies related to the funding of the full amount of the Debt Financing, other

B-10

than as set forth in the Debt Commitment Letter and any related Fee Letter. To the knowledge of Purchaser, no event has occurred, or circumstance exists, which, with or without notice, lapse of time or both, would or would reasonably be expected to constitute a default or breach on the part of Purchaser and, to the knowledge of Purchaser, any other party to the Debt Commitment Letter. Purchaser has no reason to believe that any of the conditions to the Debt Financing contemplated in the Debt Commitment Letter and any related Fee Letter will not be satisfied, that the Debt Financing will not be made available at or prior to the time contemplated hereunder for the Closing. Purchaser has fully paid, or caused to be fully paid, any and all commitment or other fees which are due and payable on or prior to the date hereof pursuant to the terms of the Debt Commitment Letter and any related Fee Letter.

5.06 Non-Reliance of Purchaser.

- (a) PURCHASER HEREBY ACKNOWLEDGES THAT SELLER IS NOT MAKING ANY REPRESENTATIONS OR WARRANTIES, AND PURCHASER IS NOT RELYING ON ANY REPRESENTATIONS OR WARRANTIES OF SELLER, OTHER THAN THE REPRESENTATIONS AND WARRANTIES SET FORTH IN THIS AGREEMENT. WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, PURCHASER HEREBY ACKNOWLEDGES THAT, EXCEPT TO THE EXTENT EXPRESSLY SET FORTH IN THIS AGREEMENT, SELLER IS NOT MAKING ANY REPRESENTATION OR WARRANTY WITH RESPECT TO (I) ANY FINANCIAL PROJECTION OR FORECAST RELATING TO THE BRAND COMPANIES OR ACQUIRED ASSETS, (II) MERCHANTABILITY, (III) FITNESS FOR ANY PARTICULAR PURPOSE, (IV) THE VIABILITY OR LIKELIHOOD OF SUCCESS OF THE BRAND BUSINESS, OR (V) ANY OTHER INFORMATION MADE AVAILABLE, WHETHER PURSUANT TO ANY PRESENTATION MADE BY OR ON BEHALF OF ANY BRAND COMPANY, PURSUANT TO ANY ELECTRONIC OR PHYSICAL DELIVERY OF DOCUMENTATION OR OTHER INFORMATION, OR OTHERWISE, TO PURCHASER, ITS AFFILIATES AND THEIR RESPECTIVE REPRESENTATIVES, AND PURCHASER EXPRESSLY DISCLAIMS ANY SUCH REPRESENTATION OR WARRANTY.
- (b) In connection with Purchaser s investigation of Seller, the Brand Companies, and the Acquired Assets, Purchaser has received from or on behalf of Seller and/or the Brand Companies certain estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions. Purchaser acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions, that Purchaser is familiar with such uncertainties, that Purchaser is taking full responsibility for making its own evaluation of the adequacy and accuracy of all projections, forecasts, budgets or revenues or earnings predictions so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions), and that Purchaser shall have no claim against Seller or any Brand Company or any of their Representatives or Affiliates with respect thereto. Purchaser acknowledges and agrees that Seller makes no representation or warranty with respect to such estimates, projections, forecasts, pro formas, plans, budgets or revenues or earnings predictions (including any such underlying assumptions). Purchaser confirms that Seller and the Brand Companies have made available to it and its Representatives the opportunity to ask questions of the Representatives of the Brand Companies and Seller and their respective Affiliates and to acquire such additional information about the business, operations and financial condition of the Brand Companies and the Brand Business as requested, and all such information has been received.

ARTICLE VI

TERMINATION

6.01 Termination. Either Seller or Purchaser may terminate this Agreement by delivery of written notice to the other Party only if the MIPA has been validly terminated in accordance with its terms. In addition, Seller may, at Seller s

election, terminate this Agreement upon written notice to Purchaser in the event that Purchaser fails to fund the Net Closing Consideration on the Closing Date as provided in <u>Section 3.02(a)</u>.

B-11

6.02 Effect of Termination. If this Agreement is terminated and the Contemplated Transactions are abandoned as described in <u>Section 6.01</u>, this Agreement shall become null and void and of no further force and effect and there shall be no Liability on the part of any Party or any of its Affiliates, directors, officers, shareholders, partners, members or Representatives, except (a) that <u>Section 6.02</u> and Article VIII shall survive termination, and (b) nothing herein shall relieve any Party from Liability for any intentional and material breach of this Agreement.

ARTICLE VII

FINANCING COVENANTS

7.01 Financing.

- (a) Subject to the terms and conditions of this Agreement, Purchaser shall use its best efforts to obtain the Debt Financing on the terms and conditions described in the Debt Commitment Letter and any related Fee Letter. Purchaser shall not permit any amendment, modification or supplement to be made to, or any waiver of any provision or remedy under, the Debt Commitment Letter or any related Fee Letter, if such amendment, modification or waiver (i) reduces the aggregate amount of the Debt Financing (including by changing the amount of fees to be paid or original issue discount), when taken together with Purchaser's cash on hand, to an amount below the amount required to satisfy Purchaser's obligations under this Agreement, or (ii) imposes new or additional conditions or other terms or otherwise expands, amends or modifies any of the conditions to the receipt of the Debt Financing or other terms in a manner that could reasonably be expected to delay in any material respect or prevent the Closing; *provided*, *however*, that Purchaser may, after consultation with Seller, replace or amend the Debt Commitment Letter to add lenders, lead arrangers, bookrunners, syndication agents or similar entities who had not executed the Debt Commitment Letter as of the date hereof so long as such action would not reasonably be expected to materially delay or prevent the Closing. Any reference in this Agreement to (x) Debt Financing shall include the financing contemplated by the Debt Commitment Letter as amended or modified in compliance with this Section 7.01 and (y) Fee Letter and Debt Commitment Letter shall include such documents as amended or modified in compliance with this Section 7.01.
- (b) Purchaser shall use its best efforts to (i) maintain in effect the Debt Commitment Letter in accordance with the terms and subject to the conditions thereof; (ii) negotiate and enter into all definitive agreements in form and substance reasonably satisfactory to Purchaser with respect to the Debt Financing contemplated by the Debt Commitment Letter on the terms and conditions set forth in the Debt Commitment Letter and any related Fee Letter; (iii) satisfy, to the extent within its control, all conditions to such definitive agreements that are applicable to Purchaser and consummate the Debt Financing at or prior to the Closing; and (iv) comply with its obligations under the Debt Commitment Letter and any related Fee Letter. Upon request (which request may be made telephonically or in electronic mail), Purchaser shall keep Seller reasonably informed on a reasonably current basis and in reasonable detail of the status of its efforts to arrange the Debt Financing and any material developments relating thereto and provide to Seller copies of all definitive documents related to the Debt Financing. Without limiting the generality of the foregoing, Purchaser shall give Seller prompt written notice (and in any event within two (2) Business Days) of (A) any breach or default (or any event or circumstance that with or without notice, lapse of time, or both, could reasonably be expected to give rise to any breach or default) or the threatened breach by any party to the Debt Commitment Letter, Fee Letter or any other definitive document related to the Debt Financing of which Purchaser becomes aware, (B) the receipt by Purchaser of any notice or other communication from any Debt Financing source with respect to any (1) breach, default, termination or repudiation by any party to the Debt Commitment Letter, Fee Letter or any other definitive document related to the Debt Financing of any provisions of the Debt Commitment Letter, Fee Letter or any other definitive document related to the Debt Financing or (2) dispute or disagreement between or among any parties to the Debt Commitment Letter, Fee Letter or any other definitive document related to the Debt Financing, and (C) if for any reason Purchaser reasonably believes that it may not be able to obtain on a

timely basis all or any portion of the Debt Financing on the terms, in the manner or from the sources contemplated by

B-12

the Debt Commitment Letter or the definitive documents relating to the Debt Financing. As soon as reasonably practicable, but in any event within two (2) Business Days of the date the Seller delivers to Purchaser a written request, Purchaser shall provide any information reasonably requested by the Seller relating to any circumstance referred to in clause (A), (B) or (C) of the immediately preceding sentence.

- (c) Purchaser shall use its best efforts to cause the lenders and any other Persons providing Debt Financing to fund on the Closing Date the Debt Financing required to consummate the transactions contemplated by this Agreement and the other transactions contemplated by the Debt Commitment Letter, if all conditions to Closing set forth in Article III herein are satisfied or waived (other than those conditions that by their terms are to be satisfied or waived at the Closing, but subject to the fulfillment or waiver of those conditions).
- (d) Nothing herein shall obligate Purchaser or any of its Affiliates to (i) waive any terms or conditions of this Agreement or of the Debt Commitment Letter, (ii) pay any additional material fees or amounts beyond those expressly set forth in the Debt Commitment Letter to cause the lenders and any other Persons providing Debt Financing to fund the Debt Financing, or (iii) commence any Proceeding against any financing institution or other Person that fails to provide its respective portion of the Debt Financing under the Debt Commitment Letter or New Debt Commitment Letter.
- (e) If any Debt Commitment Letter shall be terminated or modified in a manner adverse to Purchaser, or if any portion of the Debt Financing becomes unavailable on the terms and conditions contemplated in the Debt Commitment Letter (other than as a result of obtaining substitute debt financing in accordance with Section 7.01(e)), Purchaser shall use its best efforts to arrange to promptly obtain alternative financing from alternative sources on economic terms not materially less favorable to Purchaser than the terms of the Debt Commitment Letter and on non-economic terms not materially less favorable to Purchaser than the terms of the Debt Commitment Letter, in each case, taken as a whole, and in an amount which, together with its cash on hand, shall be sufficient (subject to the fulfilment of the conditions set forth in this Agreement) to consummate the Contemplated Transactions (the <u>Alternate Debt Financing</u>), and to obtain a new financing commitment letter with respect to such Alternate Debt Financing (the <u>New Debt Commitment</u> <u>Letter</u>), which shall replace the existing Debt Commitment Letter, a true and complete copy of which (together with any related fee or other letter) shall be promptly provided to Seller (subject to redaction so long as such redaction does not cover terms that would adversely affect the conditionality or availability of the Debt Financing). In the event any New Debt Commitment Letter is obtained, (i) any reference in this Agreement to the Debt Financing shall mean the debt financing contemplated by the Debt Commitment Letter as modified pursuant to clause (ii) below, (ii) any reference in this Agreement to the Debt Commitment Letter shall be deemed to include any Debt Commitment Letter that is not superseded by a New Debt Commitment Letter at the time in question and any New Debt Commitment Letter to the extent then in effect and (iii) any reference in this Agreement to Fee Letter shall be deemed to include any fee or other letter relating to the Debt Commitment Letter that are not superseded by a New Debt Commitment Letter at the time in question and any New Debt Commitment Letter to the extent then in effect.

ARTICLE VIII

GENERAL PROVISIONS AND CERTAIN COVENANTS

8.01 Restrictive Covenants.

(a) Seller Non-Solicitation of Employees. From and after the Closing Date, Seller shall not, and shall cause its Affiliates not to, for a period of two (2) years after the Closing Date, hire or knowingly solicit for employment any Transferred Employee; *provided*, that this paragraph shall not preclude Seller or its Affiliates from soliciting for employment any such Transferred Employee by a general solicitation through a public medium or general or mass

mailing by or on behalf of Seller or any of its Affiliates that is not targeted specifically at employees of Purchaser; *provided, further*, that Seller shall not during the now current term of the Boulder Road Lease, hire any Transferred Employee who works primarily out of the Boulder Road location

B-13

notwithstanding the manner of solicitation, without Purchaser s prior written consent. For the avoidance of doubt, this restriction shall not apply to any services that are provided under the Transition Services Agreement.

- (b) Purchaser Non-Solicitation of Employees. From and after the Closing Date, Purchaser shall not, and shall cause their respective Affiliates not to, for a period of two (2) years after the Closing Date, hire or knowingly solicit for employment any employee of Seller or its Affiliates; *provided*, that this paragraph shall not preclude Purchaser, or its Affiliates from soliciting for employment any such employee by a general solicitation through a public medium or general or mass mailing by or on behalf of Purchaser or its Affiliates that is not targeted specifically at employees of the Seller or its Affiliates; *provided further* that Purchaser shall not during the now current term of the existing Boulder Road Lease hire any employee of the Seller who works primarily out of the Boulder Road location notwithstanding the manner of solicitation, without Seller s prior written consent. For the avoidance of doubt, this Section 8.01(b) shall not apply to any services that are provided under the Transition Services Agreement or any Transferred Employees.
- (c) Seller Non-Competition; Non-Interference. Seller agrees that for the two (2)-year period commencing on the Closing Date, Seller shall not, and shall each cause its Subsidiaries not to (i) engage in or in any way, directly or indirectly, own, manage, operate, control, consult with, provide products and/or services to or participate in the ownership, management, operation or control of, provide financing to or otherwise advise or assist or be actively connected with, directly or indirectly, any enterprise or business which is of the type or character that engages in, or otherwise carries on, any business activity which in any other manner is in competition with Purchaser s operation of the Brand Business (as defined in the MIPA), except that Seller may own, in the aggregate, 5% or less of the stock of any corporation which is listed on a national stock exchange or actively traded in the over-the-counter market; (ii) solicit or attempt to solicit any Person that is, or within two (2) years before the Closing Date was, a client, customer, supplier, licensee or business relation of the Brand Business, or who Seller or any of their respective Subsidiaries solicited to be a client, customer, supplier or licensee of the Brand Business during such two (2)-year period, or induce or attempt to induce any such Person to cease, reduce or not commence doing business with Purchaser (or assist any other Person in engaging in any such activities); or (iii) knowingly and intentionally interfere in any way with the relationship between any Purchaser and any Person that is or was a client, lender, investor, customer, supplier, licensee or other business relation of the Brand Business (or assist any other Person in engaging in any such activities); provided, however, that in no event shall Seller or any of their respective Affiliates be in violation of this Section 8.01(c) as a result of Seller or any of its respective Affiliates or Subsidiaries engaging in and operating Seller s business which were not otherwise sold or transferred in connection with the sale of the Brand Business.
- (d) Purchaser Non-Competition; Non-Interference. Purchaser hereby agrees that for the two (2)-year period commencing on the Closing Date, Purchaser shall not, and shall cause its Subsidiaries not to, within the continental United States or any other jurisdiction or marketing area in which the Gaia Business (as defined in the MIPA) operates, (i) engage in or in any way, directly or indirectly, own, manage, operate, control, consult with, provide services to or participate in the ownership, management, operation or control of, provide financing to or otherwise advise or assist or be actively connected with, directly or indirectly, any enterprise or business which is of the type or character that engages in, or otherwise carries on, any business activity which in any other manner in competition with the Gaia Business, except that Purchaser may own, in the aggregate, 5% or less of the stock of any corporation which is listed on a national stock exchange or actively traded in the over-the-counter market; or (ii) knowingly and intentionally interfere in any way with the relationship between the Gaia Business and any Person that is or was a client, lender, investor, customer, supplier, licensee or other business relation of the Gaia Business (or assist any other Person in engaging in any such activities); provided, however, that in no event shall Purchaser, or its Affiliates be in violation of this Section 8.01(d) as a result of Purchaser, or its Affiliates or Subsidiaries engaging in and operating the Brand Business, as defined in the MIPA, in the manner that such business is conducted by Seller on the Closing Date.

(e) Acknowledgement. It is the desire and intent of the Parties that the provisions of this <u>Section 8.01</u> shall be enforced to the fullest extent permissible under Applicable Law and public policies applied in each jurisdiction in which enforcement is sought. If any particular provisions or portion of any of this <u>Section 8.01</u>

B-14

shall be adjudicated by a court of competent jurisdiction to be invalid, illegal or unenforceable, such provision as applicable, shall be deemed amended to delete therefrom such provision or portion adjudicated to be invalid, illegal or unenforceable, such amendment to apply only with respect to the operation of such provisions, as applicable, in the particular jurisdiction in which such adjudication is made. The Parties agree that the scope, area and duration of the covenants set forth in this Section 8.01 are reasonable. Notwithstanding the foregoing, the Parties agree that if the scope, area or duration of any covenant set forth in this Section 8.01 shall be adjudicated by a court of competent jurisdiction to be overly broad or otherwise unreasonable, such court may reduce the scope, area and duration thereof to that which it deems the maximum scope, area and duration reasonable under the circumstances, such reduced scope, area and duration to apply only with respect to the operation of such provision, as applicable, in the particular jurisdiction in which such adjudication is made. The Parties recognize that the performance of the obligations under this Section 8.01 is special, unique and extraordinary in character, and that in the event of a breach by either Party of the terms and conditions of any of this Section 8.01 to be performed by such Party, the other Party has no adequate remedy at law for any such breach, and therefore, the Parties agree that, each Party and shall, in addition to the other remedies that may be available to either of them under this Agreement at law (including to obtain damages for any breach of the provisions of this Section 8.01), be entitled, if they so elect, to enforce the specific performance thereof by such Party or to enjoin any employees of such Party or any of its Affiliates from performing services for any Person. Seller agrees that the covenants and agreements of Seller set forth in with respect to Seller in this Section 8.01 are independent of all other covenants, representations, warranties and agreements of the Parties set forth in this Agreement, and no default, breach or failure to perform by any Party shall constitute an excuse or other justification for Seller to fail to observe fully its covenants and agreements under this Section 8.01. No course of dealing between Purchaser and Seller, and no delay by Purchaser in exercising any right, power or remedy under this Section 8.01, in equity or at law, shall constitute a waiver of, or otherwise prejudice, any such right, power or remedy. Purchaser agrees that the covenants and agreements of Purchaser set forth in with respect to Purchaser in this Section 8.01 are independent of all other covenants, representations, warranties and agreements of the Parties set forth in this Agreement, and no default, breach or failure to perform by any Party shall constitute an excuse or other justification for Purchaser to fail to observe fully its covenants and agreements under this Section 8.01. No course of dealing between Purchaser and Seller, and no delay by Seller in exercising any right, power or remedy under this Section 8.01 in equity or at law, shall constitute a waiver of, or otherwise prejudice, any such right, power or remedy.

8.02 Assignment.

- (a) This Agreement shall be binding upon Seller and its respective successors and permitted assigns (if any), and Purchaser and its successors and permitted assigns (if any). This Agreement shall inure to the benefit of Seller and Purchaser and their respective successors and permitted assigns (if any).
- (b) Neither Purchaser nor Seller shall be permitted to assign any of its rights or delegate any of its obligations under this Agreement without the other Party's prior written consent: provided, that notwithstanding the foregoing: (i) Seller may assign its rights under this Agreement to any wholly-owned Subsidiary of Seller upon written notice to Purchaser; and (ii) Purchaser may assign or delegate (A) its rights under this Agreement to any wholly-owned Subsidiary of Purchaser, (B) its rights, interests and obligations under this Agreement in connection with a merger, exchange, consolidation or sale of all or substantially of its stock or assets or other similar transaction, and (C) its rights under this Agreement as collateral to its lenders and financing sources, in each case, without the prior written consent of Seller; provided further that no assignment or delegation by Seller or Purchaser pursuant to clauses (i) or (ii), respectively, shall limit or release Seller or Purchaser of its obligations hereunder, including, without limitation, Purchaser s obligations with respect to the payment of the Net Closing Cash Consideration and the Final Net Closing Cash Consideration hereunder.

8.03 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the Parties and their successors and assigns and nothing herein expressed or implied shall give or be construed to give to any Person (other than the Parties and such Party s successors and permitted assigns) any legal or equitable rights hereunder.

B-15

8.04 Notices. All notices and other communications pursuant to this Agreement shall be in writing and shall be deemed given if delivered personally, sent by electronic mail, sent by nationally-recognized overnight courier or mailed by registered or certified mail (return receipt requested), postage prepaid, to the Parties at the addresses set forth below or to such other address as the Party to whom notice is to be given may have furnished to the other Parties in writing in accordance herewith. Any such notice or communication shall be deemed to have been delivered and received (a) in the case of personal delivery, on the date of such delivery, (b) in the case of electronic mail, on the date sent if either (i) confirmation of receipt is received or (ii) such notice is promptly mailed by registered or certified mail (return receipt requested) or a nationally recognized overnight courier, (c) in the case of a nationally recognized overnight courier in circumstances under which such courier guarantees next Business Day delivery, on the next Business Day after the date sent, and (d) in the case of mailing, on the third Business Day following that on which the piece of mail containing such communication is posted.

(i) if to Seller, to:

Gaiam, Inc.

833 West South Boulder Rd.

Louisville, Colorado 80027-2452

Email: jirka.rysavy@gaiam.com

Attention: Jirka Rysavy

Brownstein Hyatt Farber Schreck, LLP

410 Seventeenth Street, Suite 2200

Denver, Colorado 80202-4432

Email: kmacdonald@bhfs.com

Attention: Kristin Macdonald

(ii) if to Purchaser, to:

Fit For Life LLC

31 West 34th St.

11th Floor

New York, New York 10001

Email: joeys@ffl-group.com

Attention: Joey Shamah

with a copy to:

Sills Cummis & Gross P.C.

101 Park Avenue, 28th Floor

New York, New York 10178

Email: mgoldsmith@sillscummis.com

rslotkin@sillscummis.com

Attention: Michael B. Goldsmith

Richard M. Slotkin

8.05 Amendments. This Agreement may not be amended except pursuant to a written agreement executed by each of Purchaser and Seller.

8.06 Remedies. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party shall be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy shall not preclude the exercise of any other remedy. Each Party acknowledges and agrees that irreparable damage would occur in the event that any of the provisions of this Agreement (including failing to take such actions as are required of it hereunder to consummate this Agreement) were not performed in accordance with its specific terms or were otherwise

B-16

breached. Accordingly, each Party acknowledges and agrees that the other Party shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which they are entitled at law or in equity, without having to post a bond. Each Party agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other Party has an adequate remedy at law or that any award of specific performance is not an appropriate remedy for any reason at law or in equity.

8.07 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neutral genders; the feminine gender shall include the masculine and neutral genders; and the neutral gender shall include the masculine and feminine genders.
- (b) Each Party acknowledges that it has participated in the drafting of this Agreement, and, as a result, the Parties agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words include and including, and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words without limitation.
- (d) Except as otherwise indicated, all references in this Agreement to subsections, Sections, Schedules and Exhibits are intended to refer to subsections and Sections of this Agreement, Schedules of this Agreement and Exhibits to this Agreement.
- (e) The words this Agreement, hereby, hereof, herein, hereunder, and comparable words refer to all of this Agreement including the Appendices, Exhibits and Schedules to this Agreement, and not to any particular Article, Section, preamble, recital, or other subdivision of this Agreement or Appendix, Exhibit or Schedule to this Agreement.
- (f) The headings contained in this Agreement, any Exhibit or Schedule hereto, and in the table of contents to this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Exhibit or Schedule but not otherwise defined therein, shall have the meaning as defined in this Agreement.
- **8.08** Survival and Expiration of Representations and Warranties and Covenants. All of the representations and warranties of the Parties set forth in this Agreement terminate and expire on the Closing Date upon the consummation of the Contemplated Transactions and all liability of the Parties with respect to such representations and warranties shall thereupon be extinguished. The covenants and agreements of the Parties shall survive the Closing the 12-month anniversary following the Closing Date except with respect to those covenants and agreements of the Parties that by their terms contemplate performance in whole or in part after the Closing shall remain in full force and effect until such covenant or agreement has been fully performed.
- **8.09 Further Assurances**. From and after the Closing, upon the request of a Party, each other Party shall do, execute, acknowledge and deliver all such further acts, assurances, deeds, assignments, transfers, conveyances and other instruments and papers as may be reasonably required or appropriate to carry out the Contemplated Transactions.
- **8.10 Exclusivity of Agreement**. The Parties have voluntarily agreed to define their rights, liabilities, and obligations respecting the subject matter of this Agreement exclusively in contract pursuant to the express terms and provisions of

this Agreement; and the Parties expressly disclaim that they are owed any duties or are entitled to any remedies not expressly set forth in this Agreement. Furthermore, the Parties each hereby acknowledge that

B-17

this Agreement embodies the justifiable expectations of sophisticated parties derived from arm s-length negotiations; all Parties specifically acknowledge that no Party has any special relationship with another party that would justify any expectation beyond that of an ordinary acquirer and an ordinary target in an arm s-length transaction. The sole and exclusive remedies for any breach of the terms and provisions of this Agreement (including any representations and warranties set forth herein) shall be those remedies set forth herein.

- **8.11 No Third Party Liability**. This Agreement may only be enforced against the named Parties. All claims or cause of action (whether in contract or tort) that may be based upon, arise out of or related to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), may be made only against the Persons that are expressly identified as Parties; and no officer, director, shareholder, employee or affiliate of any Party (including any Person negotiating or executing this Agreement on behalf of a Party) shall have any liability or obligation with respect to this Agreement or with respect to any claim or cause of action (whether in contract or tort) that may arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including a representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement).
- **8.12** Counterparts. This Agreement may be executed in one or more counterparts, each of which when executed and delivered shall be an original, and all of which when executed shall constitute one and the same instrument. The exchange of copies of this Agreement and of signature pages by facsimile or other electronic transmission (including in portable document format or PDF) shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.
- **8.13 Entire Agreement**. This Agreement (including the documents and instruments referred to herein, including Collateral Agreements and all Exhibits and Schedules hereto and thereto) constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the Parties, or any of them, with respect to the subject matter hereof.
- **8.14 Severability**. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision will be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision, and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances will not be affected by such invalidity or unenforceability, nor will such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

8.15 Governing Law; Venue.

- (a) This Agreement, and all claims of causes of action (whether in contract or tort) that may be based upon, arise out of, or relate to this Agreement or the negotiation, execution or performance of this Agreement (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed solely by the internal laws of the State of New York, without regard to the conflict-of-law principles thereof.
- (b) Each Party acknowledges that any Proceeding or other legal action relating to this Agreement or the enforcement of any provision of this Agreement (including any Proceeding relating specific performance in accordance with Section 8.06) may only be brought or otherwise commenced in any state or federal court located in the Borough of

Manhattan of the City of New York, New York, or Boulder, Colorado. Each Party:

(i) expressly and irrevocably consents and submits to the jurisdiction of each state and federal court located in the Borough of Manhattan of the City of New York, New York and Boulder, Colorado, in connection with any such Proceeding;

B-18

- (ii) agrees that each state and federal court located in the Borough of Manhattan of the City of New York, New York and Boulder, Colorado, shall be deemed to be a convenient forum; and
- (iii) agrees not to assert (by way of motion, as a defense or otherwise), in any such Proceeding commenced in a state or federal court located in the Borough of Manhattan of the City of New York, New York or Boulder, Colorado, any claim that such Party is not subject personally to the jurisdiction of such court, that such Proceeding has been brought in an inconvenient forum, that the venue of such Proceeding is improper or that this Agreement or the subject matter of this Agreement may not be enforced in or by such court.
- **8.16 Attorney s Fees**. If any Proceeding relating to this Agreement or any of the Contemplated Transactions or the enforcement thereof is brought against any Party, the prevailing Party, if any, shall be entitled to recover reasonable attorneys fees, costs and disbursements (in addition to any other relief to which the prevailing Party may be entitled). In the event that both Parties prevail in part, the applicable trier of fact shall determine the allocation of such attorneys fees, costs and disbursements in proportion to each Party s success and failure of claims brought in connection with such Proceeding.
- 8.17 Waiver of Jury Trial. EACH PARTY TO THIS AGREEMENT WAIVES TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY ANY OF THEM AGAINST THE OTHER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS AGREEMENT, OR ANY OTHER AGREEMENTS EXECUTED IN CONNECTION HEREWITH OR THE ADMINISTRATION THEREOF OR THE CONTEMPLATED TRANSACTIONS. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 8.17. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS SECTION 8.17 WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.
- **8.18 Waiver.** No failure on the part of any Person to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any Person in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy.
- (b) No Person shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Person; and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.
- **8.19 Transfer Taxes**. Any transfer taxes, sales taxes, stamp duties, filing fees, registration fees, recordation expenses, or other similar taxes, fees, charges or expenses incurred by Seller or any other party in connection with the Contemplated Transactions, shall be borne fifty percent (50%) by Seller and fifty percent (50%) by Purchaser.
- **8.20 Non-Recourse**. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, financial advisor, accountant, advisor, agent, attorney or other representative of Seller,

Purchaser or any of their respective Affiliates or Representatives shall have any liability for any obligations or liabilities of Seller or Purchaser (as applicable) under this Agreement or the Collateral Agreements or for any

B-19

claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby, except that nothing in this <u>Section 8.20</u> shall limit the express obligations of a Party under this Agreement or the Collateral Agreements.

8.21 Time of Essence. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.

8.22 Release. Effective upon the Closing, Boulder Road, LLC, the landlord under the Boulder Road Lease, on behalf of itself and its Affiliates hereby waives, releases and discharges Purchaser from any and all claims, causes of action, liabilities, losses, costs, damages, penalties, charges, expenses and all other forms of liability or obligation whatsoever, in law or equity, arising prior to the Closing Date under the Boulder Road Lease, as assigned to Purchaser on the Closing Date, other than with respect to any current liabilities related to the Boulder Road Lease shown on the Closing Statement.

[Signature Page Follows]

B-20

IN WITNESS WHEREOF, the Parties have duly executed this Agreement as of the date first written above.

PURCHASER:

FIT FOR LIFE LLC

By: /s/ Joseph Shamah Name: Joseph Shamah

Title: CEO

[Signature Page to Asset Purchase Agreement]

SELLER:

GAIAM, INC.

By: /s/Lynn Powers
Name: Lynn Powers

Title: CEO

[Signature Page to Asset Purchase Agreement]

SOLELY WITH RESPECT TO PARAGRAPH 8.22,

LANDLORD:

BOULDER ROAD, LLC

By: /s/ John Jackson Name: John Jackson

Title: VP

[Signature Page to Asset Purchase Agreement]

LIST OF OMITTED SCHEDULES AND EXHIBITS

Schedules

Schedule 1.01(c) Working Capital Target Calculation Schedule 2.04(b) Closing Working Capital Amount Example

Exhibits

Exhibit A	Form of Reception Services Agreement
Exhibit B	Form of Third Amendment and Assignment of Lease
Exhibit C	Form of Transition Services Agreement
Exhibit D	Form of Working Capital Assignment and Bill of Sale

B-24

ANNEX C

OPINION OF FINANCIAL ADVISOR

[Remainder of Page Intentionally Blank]

ANNEX C

May 10, 2016

Board of Directors

Gaiam, Inc.

833 W. South Boulder Rd.

Louisville, CO 80027

Members of the Board:

Stifel, Nicolaus & Company, Incorporated (Stifel or we) has been advised that Gaiam, Inc. (the Seller) is considering entering into a Membership Interest Purchase Agreement (the MIPA) with Stretch & Bend Holdings, LLC (SBH) and Sequential Brands Group, Inc. (Sequential). As provided in the MIPA, immediately prior to the date thereof, the Seller will have contributed to Gaiam Brand Holdco, LLC (the Company), a wholly owned subsidiary of the Seller, all of the assets of the Brand Business (as defined in the MIPA) as a result of which the Company, together with the Company Subsidiaries (as defined in the MIPA) (collectively with the Company, the Brand Companies), own and operate the Brand Business. Concurrently with the execution and delivery of the MIPA, Seller will enter into an Asset Purchase Agreement (the APA, together with the MIPA, the Agreements) with Fit For Life LLC (FFL, together with SBH, the Buyers). Pursuant to the Agreements, (i) the Seller will sell and transfer to SBH, and SBH will acquire from the Seller, the Brand Business (other than the FFL Acquired Assets and Assumed Liabilities as defined below) and all of the issued and outstanding Equity Securities (as defined in the MIPA) of the Company (the MIPA Transaction), and (ii) the Seller will cause the Brand Companies to sell and transfer to FFL, and FFL will purchase and assume from the Brand Companies, certain assets and liabilities relating to the Brand Business (the FFL Acquired Assets and Assumed Liabilities; such transaction, the APA Transaction, and together with the MIPA Transaction, the Transactions), for an aggregate value of \$167,000,000 in cash (the Consideration) subject to adjustment and on terms and conditions more fully set forth in the Agreements.

The Board of Directors of the Seller (the Board) has requested Stifel s opinion, as investment bankers, as to the fairness, from a financial point of view, to the Seller of the Consideration to be received by the Seller from the Buyers in the Transactions pursuant to the Agreements (the Opinion).

In rendering our Opinion, we have, among other things:

(i) discussed the Transactions and related matters with the Seller s counsel and reviewed a draft copy of the MIPA, dated May 9, 2016, and a draft copy of the APA, dated May 9, 2016, such drafts being the latest drafts of the Agreements provided to us;

(ii)

reviewed unaudited consolidated financial results of the Brand Companies and the Brand Business for the fiscal years ended on December 31, 2013, December 31, 2014 and December 31, 2015, respectively, prepared by the management of the Seller;

- (iii) reviewed information regarding estimated revenue and expenses of the Brand Companies and the Brand Business for the fiscal year ending December 31, 2016 prepared by the management of the Seller;
- (iv) reviewed and analyzed certain publicly available information concerning the terms of selected transactions that we considered relevant to our analysis;
- (v) reviewed and analyzed certain publicly available financial and stock market data relating to selected public companies that we deemed relevant to our analysis;

C-2

- (vi) participated in certain discussions and negotiations between representatives of the Seller and the Buyers;
- (vii) considered the results of our efforts, at the direction of the Seller, to solicit indications of interest from selected third parties with respect to a sale transaction relating to the Brand Companies and the Brand Business;
- (viii) conducted such other financial studies, analyses and investigations and considered such other information as we deemed necessary or appropriate for purposes of the Opinion; and
- (ix) taken into account our assessment of general economic, market and financial conditions and our experience in other transactions, as well as our experience in securities valuations and our knowledge of the Brand Companies industry and the Brand Business generally.

In rendering our Opinion, we have relied upon and assumed, without independent verification, the accuracy and completeness of all of the financial and other information that was provided to Stifel by or on behalf of the Seller, or that was otherwise reviewed by Stifel, and have not assumed any responsibility for independently verifying any of such information. With respect to the revenue and expense forecasts supplied to us by the Seller, we have assumed, at the direction of the Seller, that they were reasonably prepared on the basis reflecting the best currently available estimates and judgments of the management of the Seller as to the revenues and expenses of the Brand Companies and the Brand Business for the periods indicated, and that they provided a reasonable basis upon which we could form our opinion. Such forecasts were not prepared with the expectation of public disclosure. Such revenue and expense forecasts are based on numerous variables and assumptions that are inherently uncertain, including, without limitation, factors related to general economic and competitive conditions. Accordingly, actual results could vary significantly from those set forth in such forecasts. Stifel has relied on this forecasted information without independent verification or analyses and does not in any respect assume any responsibility for the accuracy or completeness thereof. We have not received any internal financial analyses, financial forecasts, reports or other information concerning the Brand Companies or the Brand Business prepared by the management of the Seller of a nature that would have enabled us to perform a discounted cash flow analysis of the Brand Companies and the Brand Business s future cash flows.

We also assumed that there were no material changes in the assets, liabilities, financial condition, results of operations, business or prospects of the Brand Companies or the Brand Business since the date of the last financial results of the Brand Companies and the Brand Business made available to us. We did not make or obtain any independent evaluation, appraisal or physical inspection of the Brand Companies or the Brand Business s assets or liabilities, including without limitation the FFL Acquired Assets and Assumed Liabilities, nor have we been furnished with any such evaluation or appraisal. Estimates of values of companies and assets do not purport to be appraisals or necessarily reflect the prices at which companies or assets may actually be sold. Because such estimates are inherently subject to uncertainty, Stifel assumes no responsibility for their accuracy.

We have assumed, with your consent, that there are no factors that would delay or subject to any adverse conditions any necessary regulatory or governmental approval and that all conditions to the Transactions will be satisfied and not waived. In addition, we have assumed that the definitive Agreements will not differ materially from the drafts we reviewed. We have also assumed that the Transactions will be consummated simultaneously substantially on the terms and conditions described in the Agreements, without any waiver of material terms or conditions by the Seller or any other party and without any anti-dilution or other adjustment to the Consideration, and that obtaining any necessary regulatory approvals or satisfying any other conditions for consummation of the Transactions will not have an adverse effect on the Seller, the Brand Companies, the Brand Business or the Transactions. We have assumed that the

Transactions will be consummated in a manner that complies with the applicable provisions of the Securities Act of 1933, as amended, the Securities Exchange Act of 1934, as amended, and all other applicable federal and state statutes, rules and regulations. We have further assumed that the Seller has relied upon the advice of its counsel, independent accountants and other advisors (other than Stifel) as to all legal, financial reporting, tax, accounting and regulatory matters with respect to the Seller, the Brand Companies, the Brand Business, the Transactions and the Agreements.

C-3

Our Opinion is limited to whether the Consideration is fair to the Seller, from a financial point of view, and does not address any other terms, aspects or implications of the Transactions, including, without limitation, the form or structure of the Transactions, any consequences of the Transactions on the Seller, the Brand Companies, the Brand Business or the Seller s and the Brand Companies stockholders, creditors or otherwise, or any terms, aspects or implications of any voting, support, stockholder or other agreements, arrangements or understandings contemplated or entered into in connection with the Transactions or otherwise. Our Opinion also does not consider, address or include: (i) any other strategic alternatives currently (or which have been or may be) contemplated by the Board or the Seller; (ii) the legal, tax or accounting consequences of the Transactions on, the Seller, the Brand Companies, the Brand Business or the respective stockholders of the Seller and the Brand Companies; (iii) the fairness of the amount or nature of any compensation to any of the Seller s or the Brand Companies officers, directors or employees, or class of such persons, relative to the compensation to the holders of the Seller s securities; and (iv) the effect of the Transactions on, or the fairness of the consideration to be received by, holders of any class of securities of the Seller or the Brand Companies other than the Equity Securities, or any class of securities of any other party to any transactions contemplated by the Agreements. Furthermore, we are not expressing any opinion herein as to the prices, trading range or volume at which the Seller s securities will trade following public announcement or consummation of the Transactions.

Our Opinion is necessarily based on economic, market, financial and other conditions as they exist on, and on the information made available to us by or on behalf of the Seller or its advisors, or information otherwise reviewed by Stifel, as of the date of this Opinion. It is understood that subsequent developments may affect the conclusion reached in this Opinion and that Stifel does not have any obligation to update, revise or reaffirm this Opinion. Our Opinion is for the information of, and directed to, the Board for its information and assistance in connection with its consideration of the financial terms of the Transactions. Our Opinion does not constitute a recommendation to the Board as to how the Board should vote on the Transactions or to any stockholder of the Seller or the Brand Companies as to how any such stockholder should vote at any stockholders meeting at which the Transactions are considered, or whether or not any stockholder of the Seller should enter into a voting, stockholders or affiliates agreement with respect to the Transactions, or exercise any dissenters or appraisal rights that may be available to such stockholder. In addition, the Opinion does not compare the relative merits of the Transactions with any other alternative transactions or business strategies which may have been available to the Seller, the Brand Companies or the Brand Business, and does not address the underlying business decision of the Board or the Seller to proceed with or effect the Transactions.

Stifel, as part of its investment banking services, is regularly engaged in the independent valuation of businesses and securities in connection with mergers, acquisitions, underwritings, sales and distributions of listed and unlisted securities, private placements and valuations for estate, corporate and other purposes. We have acted as financial advisor to the Seller in connection with the Transactions and will receive a fee for our services, a substantial portion of which is contingent upon the completion of the Transactions (the Advisory Fee). We have also acted as financial advisor to the Board and will receive a fee upon the delivery of this Opinion that is not contingent upon consummation of the Transactions (the Opinion Fee), provided that such Opinion Fee is creditable against any Advisory Fee. We were also engaged as Seller s financial advisor in connection with the sale of the Seller s interest in Natural Habitat Adventures, Inc. and received a fee upon consummation of such transaction. Upon the consummation of the Transactions, we may be entitled to receive an additional fee relating to the Transactions and the sale of the Seller s interest in Natural Habitat Adventures, Inc. We will not receive any other significant payment or compensation contingent upon the successful consummation of the Transactions. In addition, the Company has agreed to indemnify us for certain liabilities arising out of our engagement. There are no other material relationships that existed during the two years prior to the date of this Opinion or that are mutually understood to be contemplated in which any compensation was received or is intended to be received as a result of the relationship between Stifel and any party to the Transactions. Stifel may seek to provide investment banking services to the Seller, the Buyers or their respective affiliates in the future, for which we would seek customary compensation. In the ordinary course of business, Stifel

and our clients may transact in the securities of each of the Seller and Sequential and may at any time hold a long or short position in such securities.

C-4

Stifel s Fairness Opinion Committee has approved the issuance of this Opinion. Our Opinion may not be published or otherwise used or referred to, nor shall any public reference to Stifel be made, without our prior written consent, except in accordance with the terms and conditions of Stifel s engagement letter agreement with the Seller.

Based upon and subject to the foregoing, we are of the opinion that, as of the date hereof, the Consideration to be received by the Seller from the Buyers in the Transactions pursuant to the Agreements is fair to the Seller, from a financial point of view.

Very truly yours,

/s/ Stifel Nicolaus & Company Incorporated

C-5

ANNEX D

CONSOLIDATED FINANCIAL STATEMENTS

[Remainder of Page Intentionally Blank]

Index to Financial Statements

Unaudited Interim Condensed Consolidated Financial Statements for the Quarterly Period Ended

March 31, 2016

Condensed Consolidated Balance Sheets at March 31, 2016 and December 31, 2015	D-3
Condensed Consolidated Statements of Operations for the Three Months Ended March 31, 2016 and 2015	D-4
Condensed Consolidated Statements of Comprehensive Loss for the Three Months Ended March 31, 2016 and 2015	D-5
Condensed Consolidated Statements of Cash Flows for the Three Months Ended March 31, 2016 and 2015	D-6
Notes to Interim Condensed Consolidated Financial Statements Consolidated Financial Statements for the Years Ended December 31, 2015 and 2014	D-7
Report of Independent Registered Public Accounting Firm	D-17
Gaiam, Inc. Consolidated Financial Statements:	
Consolidated Balance Sheets	D-18
Consolidated Statements of Operations	D-19
Consolidated Statements of Comprehensive Income (Loss)	D-20
Consolidated Statement of Changes in Equity	D-21
Consolidated Statements of Cash Flows	D-22
Notes to Consolidated Financial Statements	D-23
Financial Statement Schedule: Schedule II Consolidated Valuation and Qualifying Accounts	D-45

D-2

GAIAM, INC.

Condensed consolidated balance sheets

(in thousands, except share and per share data) ASSETS		(arch 31, 2016 naudited)	Dec	ember 31, 2015
Current assets:				
Cash	\$	11,228	\$	13,772
Accounts receivable, net	φ	13,002	Ψ	26,999
Inventory, less allowances		16,847		17,303
Other current assets		14,351		14,843
Other current assets		14,551		14,043
Total current assets		55,428		72,917
Property and equipment, net		25,590		25,046
Media library, net		9,348		9,117
Goodwill		15,448		15,448
Other intangibles, net		782		866
Other assets		4,160		5,148
Total assets	\$	110,756	\$	128,542
LIABILITIES AND EQUITY				
Current liabilities:				
Accounts payable	\$	10,437	\$	19,384
Accrued liabilities		20,779		20,365
Total current liabilities		31,216		39,749
Commitments and contingencies				
Equity:				
Gaiam, Inc. shareholders equity:				
Class A common stock, \$.0001 par value, 150,000,000 shares authorized, 19,141,888 and 19,130,681 shares issued and outstanding at March 31,				
2016 and December 31, 2015, respectively		2		2
Class B common stock, \$.0001 par value, 50,000,000 shares authorized,				
5,400,000 issued and outstanding at March 31, 2016 and December 31, 2015		1		1
Additional paid-in capital		172,672		172,371
Accumulated other comprehensive loss		(273)		(399)
Accumulated deficit		(95,659)		(88,035)
Total Gaiam, Inc. shareholders equity		76,743		83,940
Noncontrolling interest		2,797		4,853
Total equity		79,540		88,793

Total liabilities and equity

\$ 110,756

128,542

\$

See accompanying notes to the interim condensed consolidated financial statements.

D-3

GAIAM, INC.

Condensed consolidated statements of operations

	For	the Three Marc		,
(in thousands, except per share data)		2016		2015
L.	Φ.	(unau		/
Net revenue	\$	35,327	\$	37,638
Cost of goods sold		18,937		20,391
Gross profit		16,390		17,247
Expenses:				
Selling and operating		20,223		16,720
Corporate, general and administration		4,370		3,412
Total expenses		24,593		20,132
Loss from operations		(8,203)		(2,885)
Interest and other income (expense), net		509		(484)
Loss before income taxes and noncontrolling interest		(7,694)		(3,369)
Income tax (benefit) expense		(9)		49
Net loss from continuing operations		(7,685)		(3,418)
Loss from discontinued operations, net of tax				(466)
AV . 4		(5 60 5)		(2.00.1)
Net loss		(7,685)		(3,884)
Net loss (income) attributable to noncontrolling interest		61		(8)
Net loss attributable to Gaiam, Inc.	\$	(7,624)	\$	(3,892)
Net loss attributable to Galani, nic.	Ψ	(7,024)	φ	(3,092)
Net loss per share attributable to Gaiam, Inc. common shareholders basic and diluted:				
From continuing operations	\$	(0.31)	\$	(0.14)
From discontinued operations	\$		\$	(0.02)
Net loss per share attributable to Gaiam, Inc.	\$	(0.31)	\$	(0.16)
The 1055 per share authoritable to Garani, file.	Ф	(0.31)	Ф	(0.10)
Weighted-average shares outstanding:				
Basic and Diluted		24,531		24,490

See accompanying notes to the interim condensed consolidated financial statements.

D-4

GAIAM, INC.

Condensed consolidated statements of comprehensive loss

	For the Three Months En			
	March 31,			
(in thousands, except per share data)		2016		2015
		(unau	dited	
Net loss	\$	(7,685)	\$	(3,884)
Accumulated other comprehensive income (loss):				
Foreign currency translation gain (loss), net of tax		75		(67)
Comprehensive loss		(7,610)		(3,951)
Less: comprehensive loss attributable to the noncontrolling interest		112		14
Comprehensive loss attributable to Gaiam, Inc.	\$	(7,498)	\$	(3,937)

See accompanying notes to the interim condensed consolidated financial statements

GAIAM, INC.

Condensed consolidated statements of cash flows

(in thousands)	For the Thr Ended M 2016 (unauc	arch 31, 2015
Operating activities		
Net loss	\$ (7,685)	\$ (3,884)
Loss from discontinued operations	Ì	466
Net loss from continuing operations	(7,685)	(3,418)
Adjustments to reconcile net loss from continuing operations to net cash provided by		
(used in) operating activities continuing operations:		
Depreciation	594	590
Amortization	764	543
Share-based compensation expense	301	283
Loss (gain) on equity method investments	413	(24)
Unrealized (gain) loss on translation of foreign currency contracts and balances	(945)	544
Changes in operating assets and liabilities:		
Accounts receivable, net	14,076	14,657
Inventory, less allowances	488	(12)
Other current and long term assets	974	(3,318)
Accounts payable	(8,110)	(4,924)
Accrued liabilities	523	(81)
Net cash provided by operating activities continuing operations	1,393	4,840
Net cash used in operating activities discontinued operations		(184)
Net cash provided by operating activities	1,393	4,656
Investing activities		
Purchase of property, equipment and media rights	(2,041)	(2,317)
Purchase of businesses, net of acquired cash		(223)
Net cash used in investing activities	(2,041)	(2,540)
Financing activities		
Proceeds from issuance of stock		133
Drawdowns on line of credit	2,000	100
Repayments on line of credit	(2,000)	
Dividends paid to noncontrolling interest	(1,944)	(486)
	(1,7)	(100)
Net cash used in financing activities	(1,944)	(353)

Edgar Filing: TRUPANION INC. - Form 3

Effect of exchange rates on cash		48		(641)
Net change in cash	((2,544)		1,122
Cash at beginning of period	1	3,772	1	5,772
Cash at end of period	\$ 1	1,228	1	6,894
Supplemental cash flow information				
Income taxes paid	\$	260	\$	285
Interest paid	\$	15	\$	3

See accompanying notes to the interim condensed consolidated financial statements

Notes to interim condensed consolidated financial statements

References in this report to we, us, our or Gaiam refer to Gaiam, Inc. and its consolidated subsidiaries, unless we indicate otherwise.

1. Organization, Nature of Operations, and Principles of Consolidation

Gaiam provides a broad selection of yoga, fitness, and wellness products, content, and eco-travel services. Our products are sold through major retailers in the United States, Canada, Europe and other countries. We also sell our products through digital partners, websites and ecommerce channels. Gaia, our global subscription video streaming service, provides our members with access to conscious media on most internet-connected devices anytime, anywhere commercial free. We were incorporated under the laws of the State of Colorado on July 7, 1988.

We have prepared the accompanying unaudited interim condensed consolidated financial statements in accordance with accounting principles generally accepted in the United States (GAAP) and they include our accounts and those of our subsidiaries. Intercompany transactions and balances have been eliminated.

The unaudited condensed consolidated financial position, results of operations and cash flows for the interim periods disclosed in this report are not necessarily indicative of future financial results.

Use of Estimates and Reclassifications

The preparation of financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the amounts reported in the accompanying financial statements and disclosures. Although we base these estimates on our best knowledge of current events and actions that we may undertake in the future, actual results may be different from the estimates. We have made certain reclassifications to prior period amounts to conform to the current period presentations.

Recent Accounting Pronouncements

The Financial Accounting Standards Board issued Accounting Standards Update (ASU) No. 2015-17, Income Taxes - Balance Sheet Classification of Deferred Taxes (Topic 740). The amendments under the new guidance require that deferred tax liabilities and assets be classified as noncurrent in a classified statement of financial position. The guidance is effective for consolidated financial statements issued for annual periods beginning after December 15, 2016, and interim periods within those annual periods. Earlier application is permitted for all entities as of the beginning of an interim or annual reporting period. The new ASU is not expected to have a material impact on our reported financial position or results of operations.

2. Spin-off of Gaia

On February 20, 2015, our wholly-owned subsidiary Gaia, Inc. (Gaia) filed a registration statement on Form 10 with the Securities and Exchange Commission in connection with the previously announced proposed separation of the Gaia segment from the Gaiam Brand segment into two separate publicly traded companies. Gaia filed an amended Form 10 with the Securities and Exchange Commission on May 29, 2015, September 9, 2015 and February 17, 2016. Gaia has filed a listing application with NASDAQ and is in the process of listing on NASDAQ in anticipation of the spin-off. As of March 31, 2016, we expected that the proposed tax-free spin-off would occur through a distribution to Gaiam, Inc. s shareholders of all the stock of Gaia. Gaia would hold all of the assets and liabilities of the Gaia segment. The Gaiam Brand segment would remain with Gaiam, Inc. after the distribution. The completion of the separation is

subject to satisfaction of several conditions. Furthermore, our board of directors has the right and ability, in its sole discretion, to abandon the proposed separation at any time before the distribution date. As a result, there can be no assurance that the separation will occur.

D-7

In connection with the proposed spin-off, Gaia anticipated entering into a reorganization agreement with Gaiam, Inc. to provide for, among other things, the principal corporate transactions required to effect the spin-off, certain conditions to the spin-off and provisions governing the relationship between Gaia and Gaiam, Inc. with respect to and resulting from the spin-off. The reorganization agreement would also provide that the holders of options to purchase Gaiam, Inc. Class A common stock who are employees or non-employee directors of Gaiam, Inc. on the record date for the distribution would receive options to purchase shares of Gaia s Class A common stock in the same ratio as shareholders. Additionally there would be a corresponding adjustment to the existing Gaiam, Inc. options held by such holder. The spin-off would not constitute a change in control for purposes of Gaiam, Inc. s equity plans, and therefore no vesting of awards would occur as a result of the spin-off. In addition, the reorganization agreement would address the treatment of the various insurance policies held by Gaiam, Inc. and Gaia after the spin-off. In anticipation of the spin-off, Gaia has entered into multiple license agreements with Gaiam, Inc. including a license agreement for the use of the Gaiam trade name, and related trademarks and service marks following the spin-off.

Providing the spin-off is completed, Gaia anticipates entering into a transition services agreement with Gaiam, Inc. in connection with the separation. Under the transition services agreement, Gaiam, Inc. and Gaia would agree to provide certain services to the other for a period of up to 24 months following the spin-off, or such other shorter period as may be provided in the transition services agreement. The services to be provided may include certain corporate services including, but not limited to, management, financial, accounting, tax, human resources, payroll, technical, fulfillment, software quality control, and certain office services as required from time to time in the ordinary course of our business. Charges for these services would be based on the actual cost of such services without premium or mark-up, although applicable administrative and other overhead costs associated with the services would be allocated and included in the service charge. The employees of Gaia would remain eligible employees under certain employee benefit plans currently maintained by Gaiam, Inc., which would be managed under the transition services agreement.

Effective January 1, 2015, Gaiam, Inc. contributed to Gaia its 100% membership interest in Boulder Road LLC, a Colorado limited liability company. Boulder Road LLC is the sole owner of the property located at 833 West South Boulder Road in Louisville, Colorado, which is the location for our operations and the principal executive offices of Gaiam, Inc., Gaia and various other companies. The Gaiam, Inc. business unit and the Gaia business unit have entered into lease agreements with Boulder Road LLC effective with the contribution. The intercompany transactions have been eliminated in the accompanying condensed consolidated financial statements.

3. Joint Venture

In 2015 the Company formed a partnership with a third party to jointly market a fitness-based infomercial and its associated products. Under the arrangement, the Company provides rights to an infomercial which it has developed, and the third party provides the necessary working capital to advertise and market the infomercial worldwide. We hold a 49% interest in the partnership, and are accounting for it as an equity method investment. During the three months ended March 31, 2016, the Company recognized a non-cash loss of \$0.4 million associated with the venture which is included in Interest and other income (expense), net in the accompanying condensed consolidated statements of operations.

4. Credit Facility

In 2015 Boulder Road LLC, a subsidiary of Gaia, entered into a revolving line of credit agreement with a bank in the amount of \$5.5 million. The note bears interest at the prime rate plus 3.25%, is guaranteed by Gaiam, Inc. and Gaia, and is secured by a Deed of Trust filed against the real property on which the principal offices of the Company are located. The value of our principal offices is in excess of the amount of the line of credit. During the three months ended March 31, 2016, Gaia drew and repaid \$2.0 million on the line of credit. No amounts were outstanding under

the line of credit as of March 31, 2016.

D-8

5. Derivatives

Our operations in foreign countries expose us to market risk associated with foreign currency exchange rate fluctuations between the U.S. dollar and various foreign currencies. We use derivative instruments to manage a portion of our exposure to changes in currency exchange rates due to payments made by our eco-travel subsidiary to tour operators in Canada, South Africa, Australia, Europe, Denmark, and New Zealand. Our primary objective for entering into currency derivatives is to reduce the volatility that changes in currency exchange rates have on our earnings. In general, the market risk related to these contracts is offset by corresponding gains and losses on the hedged transactions. By working only with major banks and closely monitoring current market conditions, we seek to limit the risk that counterparties to these contracts may be unable to perform. We have presented amounts on a gross basis as we do not have any master netting agreements with counterparties. We do have the legal right to net settle transactions of the same currency with a single net amount payable by one party or the other. We do not enter into derivative contracts for trading purposes.

The cash flow effects of our derivative contracts for the three months ended March 31, 2016 are included within Net cash provided by operating activities in the Statements of Cash Flows. Realized and unrealized gains and losses on currency derivatives without hedge accounting designation are included in Interest and other (expense) income, net in the accompanying statements of operations. For the period ended March 31, 2016 the gain recognized was \$0.9 million. The asset related to the fair value of the hedging instruments has been included in Other current assets in the accompanying balance sheet.

Total notional amounts and gross fair values for derivative instruments without hedge accounting designation were as follows as of March 31, 2016:

	March	31, 201	16	December 31, 2015				
(in thousands)	Notional Amount in USD	of C	Value urrent ssets	Notional Amount in USD	Cu	Value of rrent bilities		
Currency forward contracts								
Canadian dollars	\$ 10,701	\$	309	\$ 4,200	\$	(246)		
Other currencies	1,435		44	1,372		(83)		
	\$ 12,136	\$	353	5,572		(329)		
Currency option contracts								
Canadian dollars				8,514		(252)		
Total	\$ 12,136	\$	353	\$ 14,086	\$	(581)		

As of March 31, 2016, \$9.0 million of our outstanding contracts will mature in the next 12 months, and \$3.1 million will mature the following year.

6. Equity

During the first three months of 2016, we issued 11,000 shares of our Class A common stock under our 2009 Long-Term Incentive Plan to our independent directors, in lieu of cash compensation, for services rendered in 2016. We valued the shares issued to our independent directors at estimated fair value based on the closing price of our

shares on the date the shares were issued, which by policy is the last trading day of each quarter in which the services were rendered.

D-9

During the first three months of 2016, we did not issue any stock in connection with option exercises. The following is a reconciliation from December 31, 2015 to March 31, 2016 of the carrying amount of total equity, equity attributable to Gaiam, Inc., and equity attributable to the noncontrolling interest.

Gaiam, Inc. Shareholders Accumulate Class A Other and Class B

	C	omp	orehensi A	ecumula@	lmj	orehenst	ve m	mon	Paid-in N	Nonc	ontrolling
(in thousands)	Total		Loss	Deficit		Loss	Sto	ck	Capital	Iı	nterest
Balance at December 31, 2015	\$88,793			\$ (88,035)	\$	(399)	\$	3	\$172,371	\$	4,853
Issuance of Gaiam, Inc. common											
stock for stock option exercises and											
share-based compensation	301								301		
Dividends paid to noncontrolling											
interest	(1,944)										(1,944)
Comprehensive loss:											
Net (loss) income	(7,685)		(7,685)	(7,624)							(61)
Foreign currency translation											
adjustment, net of taxes of \$32	75		75			126					(51)
Comprehensive loss		\$	(7,610)								
Balance at March 31, 2016	\$79,540			\$ (95,659)	\$	(273)	\$	3	\$172,672	\$	2,797

7. Share-Based Payments

During the first three months of 2016 and 2015, we extended the term of certain options granted under our 2009 Long-Term Incentive Plan to members of our executive team for an additional three months, and recognized \$40,000 and \$50,000 of associated stock compensation expense. Total share-based compensation expense is reported in corporate, general and administration expenses on our condensed consolidated statements of operations.

8. Net Loss per Share Attributable To Gaiam, Inc. Common Shareholders

Basic net loss per share attributable to Gaiam, Inc. common shareholders excludes any dilutive effects of options. We compute basic net loss per share attributable to Gaiam, Inc. common shareholders using the weighted average number of shares of common stock outstanding during the period. We compute diluted net loss per share attributable to Gaiam, Inc. common shareholders using the weighted average number of shares of common stock and common stock equivalents outstanding during the period. We excluded common stock equivalents of 1,071,000 and 1,088,000 from the computation of diluted net loss per share attributable to Gaiam, Inc. common shareholders for the three months ended March 31, 2016 and 2015, respectively, because their effect was antidilutive.

The following table sets forth the computation of basic and diluted net loss per share attributable to Gaiam, Inc. common shareholders:

	Three Months End March 31,			
(in thousands, except per share data)	2016	2015		
Net loss attributable to Gaiam, Inc. common shareholders:				
Loss from continuing operations	\$ (7,624)	\$ (3,426)		
Loss from discontinued operations		(466)		
Net loss attributable to Gaiam, Inc.	\$ (7,624)	\$ (3,892)		
Weighted average shares for basic and diluted net loss per				
share	24,531	24,490		
Net loss per share attributable to Gaiam, Inc. common				
shareholders basic and diluted:				
Loss from continuing operations	\$ (0.31)	\$ (0.14)		
Loss from discontinued operations	\$	\$ (0.02)		
Basic and diluted net loss per share attributable to Gaiam,				
Inc.	\$ (0.31)	\$ (0.16)		
Weighted average shares for basic and diluted net loss per share Net loss per share attributable to Gaiam, Inc. common shareholders basic and diluted: Loss from continuing operations Loss from discontinued operations Basic and diluted net loss per share attributable to Gaiam,	\$ (0.31) \$	\$ (0.14) \$ (0.02)		

9. Income Taxes

During 2013, we determined that a full valuation allowance against our deferred tax assets was necessary due to the cumulative loss incurred over the three-year period ended December 31, 2013. Since that time, we have continued to provide a full valuation allowance against deferred tax assets. As income is generated in future periods, the Company expects to reverse the valuation allowance as utilization of the deferred tax assets occurs. As of March 31, 2016, our gross net operating losses were \$86.3 million and \$16.4 million for federal and state, respectively.

10. Segment Information

We manage our company and aggregate our operational and financial information in accordance with two reportable segments, which are aligned based on their products or services:

Gaiam Brand: This segment includes all our branded yoga, fitness, and wellness products. It also includes our

eco-travel subsidiary.

Gaia: This segment includes our digital video streaming service. It also includes the results of

Boulder Road LLC. We previously announced that we are pursuing the potential spin off of this

segment into a separate company.

D-11

Amounts shown as Other unallocated corporate expenses in the table below are primarily expenses of being a public company or general corporate expenses which we do not allocate to our segments. Although we are able to track sales by channel, the management, allocation of resources, and analysis and reporting of expenses are presented on a combined basis, at the reportable segment level. Contribution margin is defined as net revenue less cost of goods sold and total operating expenses. Financial information for our segments is as follows:

	Three Mon Marc	
(in thousands)	2016	2015
Net revenue:		
Gaiam Brand	\$31,497	\$ 34,532
Gaia	3,830	3,106
Consolidated net revenue	35,327	37,638
Contribution loss:		
Gaiam Brand	(3,468)	(53)
Gaia	(3,449)	(2,359)
Segment contribution loss	(6,917)	(2,412)
Other unallocated corporate expenses	(1,286)	(473)
Consolidated contribution loss	(8,203)	(2,885)
Reconciliation of contribution loss to net loss attributable to Gaiam, Inc.:		
Interest and other income (expense), net	509	(484)
Income tax benefit (expense)	9	(49)
Loss from discontinued operations		(466)
Net loss (income) attributable to noncontrolling interest	61	(8)
Net loss attributable to Gaiam, Inc.	\$ (7,624)	\$ (3,892)

As discussed in Note 2, Gaia filed amendments to its registration statement on Form 10 with the SEC on September 9, 2015 and February 17, 2016. The segment amounts presented here and discussed elsewhere in this quarterly report on Form 10-Q vary insignificantly from the amounts reported by Gaia in the Form 10, as amended, as the Form 10 requires that certain items be recast for stand-alone presentation.

11. Commitments and Contingencies

During 2015, we were involved in various legal disputes with Cinedigm Corp. and Cinedigm Entertainment Holdings, LLC (together, Cinedigm) in connection with our sale of GVE Newco, LLC (GVE) to Cinedigm. The disputes were fully settled at the end of 2015.

During the three months ended March 31, 2015, the Company recorded charges of \$0.5 million for legal costs associated with the disputes. The expenses associated with the disputes have all been included in Loss from discontinued operations in the accompanying condensed consolidated statements of operations.

From time to time, we are involved in legal proceedings that we consider to be in the normal course of business. Claimed amounts against us may be substantial but may not bear any reasonable relationship to the merits of the claim or the extent of any real risk of court or arbitral awards. We record accruals for losses related to those matters against us that we consider to be probable and that can be reasonably estimated. Although it is not feasible to predict the outcome of these matters with certainty, it is reasonably possible that some legal proceedings may be disposed of or decided unfavorably to us and in excess of the amounts currently accrued. Based on available information, in the opinion of management, settlements, arbitration awards and final

D-12

judgments, if any, which are considered probable of being rendered against us in litigation or arbitration in existence at March 31, 2016 and can be reasonably estimated are reserved against or would not have a material adverse effect on our financial condition, results of operations or cash flows.

12. Discontinued Operations

During 2013, we consummated the sale of GVE, a wholly-owned subsidiary to Cinedigm. As discussed in Note 11, Commitments and Contingencies, we were involved in legal disputes with Cinedigm associated with the sale, which were settled during 2015. The results of discontinued operations reported in the accompanying financial statements relate solely to legal costs associated with the disputes.

13. Exit Activity Costs

During the fourth quarter of 2013, as a result of the sale of GVE and the discontinuation of DRTV, we recorded impairment charges on certain media and assets; and recorded accruals for termination benefits. Those accruals are included in accrued expenses in our condensed consolidated balance sheets. The balance in the accrual for termination benefits remained at \$1.7 million during the three months ended March 31, 2016.

14. Subsequent Events

Pending Sale of the Brand Business

On May 10, 2016 we entered into agreements to sell the assets and liabilities of our Gaiam Brand Business in exchange for \$167.0 million. The Brand Business consists of all the assets and liabilities of our Gaiam Brand segment, except for our interest in our eco-travel subsidiary, which was sold on May 4, 2016 as explained below. Our Gaiam Brand Business constitutes the vast majority of our consolidated revenues and expenses, and consists of our yoga, fitness and wellness consumer products, and content. If the sale of the Brand Business is consummated, our remaining business will consist of our Gaia video subscription service business and on-line community and will operate in our Gaia segment. Upon closing the sale of the Brand Business, we expect to conduct a tender offer for up to 12 million shares of our Class A common stock or vested stock options. The sale of our Brand Business is planned to occur after the period covered by this report. Therefore, the information in this report should be read in light of the pending sale of our Brand Business. There can be no assurance that we will be able to successfully consummate the sale of our Brand Business.

Sale of Natural Habitat

On May 4, 2016 we sold all our interest in Natural Habitat, Inc. (Natural Habitat), our eco-travel subsidiary, in exchange for \$12.85 million in cash.

The following pro forma condensed consolidated financial information is based on our historical financial statements, including certain pro forma adjustments, and has been prepared to illustrate the pro forma effect of the sale of Natural Habitat. The unaudited pro forma condensed consolidated statements of operations for the quarter ending March 31, 2016 and for the year ending December 31, 2015 assume that the sale occurred as of January 1, 2015. The unaudited pro forma condensed consolidated balance sheet as of March 31, 2016 is presented as if the sale had occurred on March 31, 2016.

The unaudited pro forma condensed consolidated financial information has been prepared based upon available information and management estimates; actual amounts may differ from these estimated amounts. The unaudited pro

forma condensed consolidated financial information is not necessarily indicative of the financial position or results of operations that might have occurred had the sale occurred as of the dates stated above. The pro forma adjustments are described in the notes.

The unaudited pro forma condensed consolidated financial information should be read in conjunction with the audited financial statements and notes and related Management s Discussion and Analysis of Financial Condition and Results of Operations (MD&A) included in the Company s Annual Report on Form 10-K for the fiscal year ended December 31, 2015.

D-13

Unaudited Pro Forma Condensed Consolidated Financial Information

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Statement of Operations

Quarter Ending March 31, 2016

(in thousands, except per share data)

	Gaiam Historical Financial Statements	Less: Natural Habitat Historical Financial Statements (a)(c)	Pro Forma Adjustments (Disposition of Natural Habitat)	Notes	I (Ad Disp N	o Forma Results justed for position of Natural Iabitat)
Net revenues	\$ 35,327	\$ 9,585	\$		\$	25,742
Cost of goods sold	18,937	6,312				12,625
Gross profit	16,390	3,273				13,117
Selling, general, and administrative	24,593	4,209	3	(b)		20,387
Loss from operations	(8,203)	(936)	(3)			(7,270)
Interest and other income (expense)	509	944				(435)
(Loss) income before income taxes	(7,694)	8	(3)			(7,705)
Income tax (benefit) expense	(9)	1				(10)
Net (loss) income from continuing operations	(7,685)	7	(3)			(7,695)
Net loss attributable to noncontrolling						
interest	61	47				14
Net (loss) income from continuing						
operations attributable to Gaiam, Inc.	\$ (7,624)	\$ 54	\$ (3)		\$	(7,681)
Net loss from continuing operations per share:						
Basic and Diluted	\$ (0.31)				\$	(0.31)
Weighted-average shares outstanding:						
Basic and Diluted	24,531					24,531

- (a) Represents the revenues and expenses excluding certain fixed overhead costs of Natural Habitat included in Gaiam s historical financial statements.
- (b) Operating expenses are increased to reverse the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (c) Results are consolidated then adjusted for noncontrolling interests. Gaiam owned 51.4% of Natural Habitat.

D-14

Unaudited Pro Forma Condensed Consolidated Financial Information

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Statement of Operations

Year Ending December 31, 2015

(in thousands, except per share data)

Net revenues \$ 188,018 \$ 43,514 \$ 144,504 Cost of goods sold 103,249 27,544 75,705 Gross profit 84,769 15,970 68,799 Selling, general, and administrative 80,588 12,126 10 (b) 68,472 Income (loss) from operations 4,181 3,844 (10) 327		Hi Fi	Gaiam storical nancial tements	N Hi Hi Fi Sta	Less: Vatural Vabitat Vastorical Value of the contents Value of th	Pro Forma Adjustments (Disposition of Natural Habitat)	Notes	(Ad Di of	ro Forma Results ljusted for sposition 'Natural Iabitat)
Cost of goods sold 103,249 27,544 75,705 Gross profit 84,769 15,970 68,799 Selling, general, and administrative 80,588 12,126 10 (b) 68,472 Income (loss) from operations 4,181 3,844 (10) 327	Net revenues					· · · · · · · · · · · · · · · · · · ·	1,000		
Selling, general, and administrative 80,588 12,126 10 (b) 68,472 Income (loss) from operations 4,181 3,844 (10) 327						1		Ť	
Income (loss) from operations 4,181 3,844 (10) 327	Gross profit		84,769		15,970				68,799
	Selling, general, and administrative		80,588		12,126	10	(b)		68,472
	Income (loss) from operations		4,181		3,844	(10)			327
Interest and other income (expense) $(1,406)$ $(1,168)$ (238)	Interest and other income (expense)		(1,406)		(1,168)				(238)
Loss from equity method investments (465)	Loss from equity method investments		(465)						(465)
Income (loss) before income taxes 2,310 2,676 (10) (376) Income tax expense (benefit) 1,219 1,154 65						(10)			` ′
Net income (loss) from continuing operations 1,091 1,522 (10)	Net income (loss) from continuing operations		1 091		1 522	(10)			(441)
Net income attributable to noncontrolling interest (694) (672) (22)	Net income attributable to noncontrolling		ĺ		ĺ	(10)			Ì
Net income (loss) from continuing operations attributable to Gaiam, Inc. \$ 397 \$ 850 \$ (10) \$ (463)		\$	397	\$	850	\$ (10)		\$	(463)
Net income (loss) from continuing operations per share:									
Basic and Diluted \$ 0.02 \$ (0.02)	Basic and Diluted	\$	0.02					\$	(0.02)
Weighted-average shares outstanding:	Weighted-average shares outstanding:								
Basic 24,510 24,510			24,510						24,510
Diluted 24,612 24,612	Diluted		24,612						24,612

- (a) Represents the revenues and expenses excluding certain fixed overhead costs of Natural Habitat included in Gaiam s historical financial statements.
- (b) Operating expenses are increased to reverse the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (c) Results are consolidated then adjusted for noncontrolling interests. Gaiam owned 51.4% of Natural Habitat.

D-15

Unaudited Pro Forma Condensed Consolidated Financial Information

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Balance Sheet

As of March 31, 2016

(in thousands)

	H Fi	Gaiam istorical inancial atements	N Hi Hi	Less: Natural Iabitat istorical nancial ntements (a)	Adj (Dis	o Forma ustments sposition Natural (abitat)	Notes) (A Di	Pro Forma Results Adjusted for sposition Natural
Cash	\$	11,228	\$	3,951	\$	11,050	(b)	\$	18,327
Accounts receivable, net		13,002		389					12,613
Inventory, less allowances		16,847							16,847
Other current assets		14,351		8,714					5,637
Total current assets		55,428		13,054		11,050			53,424
Property and equipment, net		25,590		1,418					24,172
Media library, net		9,348							9,348
Goodwill		15,448		3,384					12,064
Other intangibles, net		782		75					707
Other assets		4,160		423					3,737
Total assets	\$	110,756	\$	18,354	\$	11,050		\$	103,452
Accounts payable	\$	10,437	\$	2,615	\$			\$	7,822
Accrued liabilities		20,779		13,310		544	(c)		8,013
Total current liabilities		31,216		15,925		544			15,835
Total Gaiam, Inc. shareholders equity		76,743		1,501		10,506	(d)		85,748
Noncontrolling interest		2,797		928		10,500	(u)		1,869
0		,							,
Total equity		79,540		2,429		10,506			87,617
Total liabilities and equity	\$	110,756	\$	18,354	\$	11,050		\$	103,452

- (a) Represents the assets, liabilities, and equity attributable to Natural Habitat included in Gaiam s historical financial statements.
- (b) Cash is adjusted to reflect the \$11.0 million estimated net proceeds from the sale.
- (c) Accrued liabilities have been adjusted to reflect the accrual of estimated nonrecurring costs directly related to the disposition (not paid in close).
- (d) Equity is adjusted to reflect the estimated net gain of \$9.0 million on the sale of Natural Habitat. We expect that there will be minimal income taxes on the gain due to utilization of our net operating losses.

D-16

Report of Independent Registered Public Accounting Firm

To the Board of Directors and Shareholders of

Gaiam, Inc.

Louisville, Colorado

We have audited the accompanying consolidated balance sheets of Gaiam, Inc. and subsidiaries (the Company) as of December 31, 2015 and 2014, and the related consolidated statements of operations, comprehensive loss, changes in equity, and cash flows for each of the years in the three-year period ended December 31, 2015. Our audits also included the consolidated financial statement schedule II for the years ended December 31, 2015, 2014, and 2013. We have also audited the Company s internal control over financial reporting as of December 31, 2015, based on criteria established in Internal Control-Integrated Framework (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). The Company s management is responsible for these consolidated financial statements, for maintaining effective internal control over financial reporting, and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying management s report on internal control over financial reporting. Our responsibility is to express an opinion on these consolidated financial statements and an opinion on the effectiveness of the Company s internal control over financial reporting based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement and whether effective internal control over financial reporting was maintained in all material respects. Our audits of the consolidated financial statements included examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall consolidated financial statement presentation. Our audits of internal control over financial reporting included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audits also included performing such other procedures as we considered necessary in the circumstances. We believe that our audits provide a reasonable basis for our opinions.

A company s internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company s internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company s assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Gaiam, Inc. and subsidiaries as of December 31, 2015 and 2014, and the results of their operations and their cash flows for each of the years in the three-year period ended December 31, 2015, in conformity with accounting principles generally accepted in the United States of America. In our opinion, the related consolidated financial statement schedule II for the years ended December 31, 2015, 2014, and 2013, when considered in relation to the consolidated financial statements taken as a whole, presents fairly in all material respects the information set forth therein. Also in our opinion, Gaiam, Inc. and subsidiaries maintained, in all material respects, effective internal control over financial reporting as of December 31, 2015, based on the criteria established in Internal Control Integrated Framework (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO).

EKS&H LLLP

March 15, 2016

Boulder, Colorado

D-17

GAIAM, INC.

Consolidated Balance Sheets

	December 31,	
(in thousands, except share and per share data)	2015	2014
ASSETS		
Current assets:	¢ 12.772	¢ 15.770
Cash	\$ 13,772	\$ 15,772
Accounts receivable, net	26,999	30,266
Inventory, less allowances	17,303	20,154
Other current assets	14,843	11,998
Current assets of discontinued operations		582
Total current assets	72,917	78,772
Property and equipment, net	25,046	23,231
Media library, net	9,117	7,691
Goodwill	15,448	15,448
Other intangibles, net	866	823
Other assets	5,148	12,667
Total assets	\$ 128,542	\$ 138,632
	Ψ 120,3 12	ψ 130,03 2
LIABILITIES AND EQUITY		
Current liabilities:		
Accounts payable	\$ 19,384	\$ 18,837
Accrued liabilities	20,365	20,236
Total current liabilities	39,749	39,073
Commitments and contingencies		
Equity:		
Gaiam, Inc. shareholders equity:		
Class A common stock, \$.0001 par value, 150,000,000 shares authorized, 19,130,681 and		
19,084,958 shares issued and outstanding at December 31, 2015 and 2014, respectively	2	2
Class B common stock, \$.0001 par value, 50,000,000 shares authorized, 5,400,000 shares		
issued and outstanding at December 31, 2015 and 2014	1	1
Additional paid-in capital	172,371	171,315
Accumulated other comprehensive loss	(399)	(200)
Accumulated deficit	(88,035)	(76,329)
Total Gaiam, Inc. shareholders equity	83,940	94,789
Noncontrolling interest	4,853	4,770
Total equity	88,793	99,559
Total liabilities and equity	\$ 128,542	\$ 138,632
20mm macmato and equity	Ψ120,012	ψ 100,002

See accompanying notes to consolidated financial statements.

D-18

GAIAM, INC.

Consolidated Statements of Operations

(in thousands, except per share data) Consolidated Statements of Operations Data:	2015	2014	2013
Net revenues	\$ 188,018	\$ 166,694	\$ 155,463
Cost of goods sold	103,249	91,189	90,155
Cost of goods sold	103,249	91,109	90,133
Gross profit	84,769	75,505	65,308
Expenses:			
Selling and operating	64,160	68,470	64,657
Corporate, general and administration	16,428	12,121	11,325
Other general income and expense			10,967
Total expenses	80,588	80,591	86,949
Income (loss) from operations	4,181	(5,086)	(21,641)
Interest and other (expense) income	(1,406)	(600)	2,421
Gain on sale of investments		1,480	25,096
Loss from equity method investment	(465)	(55)	
Income (loss) before income taxes	2,310	(4,261)	5,876
	1,219	1,369	25,974
Income tax expense	1,219	1,309	23,914
Net income (loss) from continuing operations	1,091	(5,630)	(20,098)
Loss from discontinued operations, net of tax	(12,103)	(3,327)	(1,995)
AV . 1	(11.010)	(0.055)	(22.002)
Net loss	(11,012)	(8,957)	(22,093)
Net income attributable to noncontrolling interest	(694)	(959)	(659)
Net loss attributable to Gaiam, Inc.	\$ (11,706)	\$ (9,916)	\$ (22,752)
Net income (loss) per share attributable to Gaiam, Inc. common shareholders basic and diluted:			
From continuing operations	\$ 0.01	\$ (0.27)	\$ (0.90)
From discontinued operations	(0.49)	(0.14)	(0.09)
Basic and diluted net loss per share attributable to Gaiam, Inc.	\$ (0.48)	\$ (0.41)	\$ (0.99)
Weighted-average shares outstanding:			
Basic	24,510	24,228	22,972
Diluted	24,612	24,228	22,972

See Accompanying Notes to Consolidated Financial Statements.

D-19

GAIAM, INC.

Consolidated Statements of Comprehensive Loss

	Years Ended December 31,		
(in thousands)	2015	2014	2013
Net loss	\$ (11,012)	\$ (8,957)	\$ (22,093)
Other comprehensive (loss) income:			
Foreign currency translation loss	(324)	(62)	(292)
Unrealized gain on equity securities		202	116
Reclassification of gain on equity securities to net income		(319)	
Other comprehensive loss	(324)	(179)	(176)
Comprehensive loss	(11,336)	(9,136)	(22,269)
Less: comprehensive income attributable to the noncontrolling interest	569	947	634
Comprehensive loss attributable to Gaiam, Inc.	\$ (11,905)	\$ (10,083)	\$ (22,903)

See Accompanying Notes to Consolidated Financial Statements.

GAIAM, INC.

Consolidated Statement of Changes in Equity

	Gaiam, Inc. Shareholders Accumulated							
	Total	Total Gaiam <i>A</i>	Accumu late	OtherCo	mm	oandditional Callonider Paid-in	Common Stock No	ncontrollin
(in thousands, except shares)	Equity	Equity			mou	nt Capital	Shares	Interest
Balance at December 31, 2012	\$ 118,872	\$ 116,074	\$ (43,661)	\$ 118	\$3	\$ 159,614	22,730,464	\$2,798
Issuance of Gaiam, Inc. common stock in conjunction with an acquisition								
and share-based compensation	8,261	8,261				8,261	1,264,657	
Noncontrolling interest portion of subsidiary s business								
combinations	426							426
Comprehensive (loss) income	(22,269)	(22,903)	(22,752)	(151)				634
Balance at December 31, 2013	\$ 105,290	\$ 101,432	\$ (66,413)	\$ (33)	\$3	\$ 167,875	23,995,121	\$3,858
Issuance of Gaiam, Inc. common stock in conjunction with an acquisition								
and share-based compensation	3,440	3,440				3,440	489,837	
Noncontrolling interest portion of subsidiary s business	Í	,				·	ŕ	
combinations	115							115
Subsidiary s dividend to								
noncontrolling interest	(150)							(150)
Comprehensive (loss) income	(9,136)	(10,083)	(9,916)	(167)				947
Balance at December 31, 2014	\$ 99,559	\$ 94,789	\$ (76,329)	\$ (200)	\$3	\$ 171,315	24,484,958	\$4,770
Issuance of Gaiam, Inc. common stock and share-based								
compensation	1,056	1,056				1,056	45,723	
Subsidiary s dividend to		,						
noncontrolling interest	(486)							(486)
Comprehensive (loss) income	(11,336)	(11,905)	(11,706)	(199)				569
Balance at December 31, 2015	\$ 88,793	\$ 83,940	\$ (88,035)	\$ (399)	\$3	\$ 172,371	24,530,681	\$4,853

See Accompanying Notes to Consolidated Financial Statements.

GAIAM, INC.

Consolidated Statements of Cash Flows

(in thousands)	Years ended December 31, 2015 2014 2013 (
Operating activities:			_ = = = (65)
Net loss	\$(11,012)	\$ (8,957)	\$ (22,093)
Loss from discontinued operations	12,103	3,327	1,995
	,	- /-	,
Net income (loss) from continuing operations	1,091	(5,630)	(20,098)
Adjustments to reconcile net income (loss) from continuing operations to net			
cash (used in) provided by operating activities:			
Depreciation	2,393	1,989	2,301
Amortization	2,782	2,223	1,659
Share-based compensation expense	895	839	809
Deferred income tax expense (benefit)		28	23,861
Loss on translation of foreign currency	408	564	42
Gain on sale of investments		(1,480)	(25,096)
Loss on equity method investment	465	55	
Gain from collection of note receivable			(2,300)
Impairment and other losses			9,194
Changes in operating assets and liabilities, net of effects from acquisitions and			
dispositions:			
Accounts receivable, net	3,247	(3,872)	(2,072)
Inventory, net	2,780	56	1,097
Deferred advertising costs		(250)	508
Other current and long-term assets	(2,586)	(3,939)	(2,887)
Accounts payable	(2,458)	5,535	(4,407)
Accrued liabilities	2,946	(4,945)	1,834
	11.062	(0, 007)	(15.555)
Net cash provided by (used in) operating activities continuing operations	11,963	(8,827)	(15,555)
Net cash used in operating activities discontinued operations	(3,807)	(142)	(7,569)
Net cash provided by (used in) operating activities	8,156	(8,969)	(23,124)
ive cash provided by (ased in) operating activities	3,200	(0,2 02)	(==,== -)
Investing activities:			
Net proceeds from the sale of investments		2,646	25,096
Collection from related party			2,100
Purchase of property, equipment and media rights	(9,335)	(5,590)	(3,386)
Proceeds from sale of business			47,500
Purchase of businesses and equity-method investments, net of cash acquired		(5,493)	(6,333)
Net cash (used in) provided by investing activities	(9,335)	(8,437)	64,977

Edgar Filing: TRUPANION INC. - Form 3

Financing activities:

Net proceeds from issuance of stock	160	1,806	777
Payment of dividends to non-controlling interest	(486)	(150)	
Net cash (used in) provided by financing activities continuing operations	(326)	1,656	777
Net cash used in financing activities discontinued operations			(19,967)
Net cash (used in) provided by financing activities	(326)	1,656	(19,190)
Effect of exchange rates on cash	(495)	(707)	(292)
Net (decrease) increase in cash	(2,000)	(16,457)	22,371
Cash at beginning of year	15,772	32,229	9,858
Cash at end of year	\$ 13,772	\$ 15,772	\$ 32,229
Supplemental cash flow information:			
Interest paid	\$ 4	\$ 14	\$ 442
Income taxes paid	1,322	1,114	577
Liabilities and debt assumed from acquisitions		466	337
Stock issued in connection with business acquisitions	\$	\$ 840	\$ 7,303

⁽a) Cash flows in 2013 include the \$25.0 million gain from the sale of RGSE stock, the sale of GVE and the discontinuation of the DRTV Business Unit.

See Accompanying Notes to Consolidated Financial Statements.

Notes to Consolidated Financial Statements

References in this report to we, us, our, Company or Gaiam refer to Gaiam, Inc. and its consolidated subsidiari unless we indicate otherwise.

1. Organization, Nature of Operations, and Principles of Consolidation

Gaiam, Inc. and its consolidated subsidiaries (the Company) provide a broad selection of yoga, fitness, and wellness products, with media, subscription and travel services to customers who value personal development, wellness, ecological lifestyles, responsible media, and conscious community. We were incorporated under the laws of the State of Colorado on July 7, 1988.

We have prepared the accompanying consolidated financial statements in accordance with accounting principles generally accepted in the United States, or GAAP, and they include our accounts and those of our subsidiaries, over which we exercise control. Intercompany transactions and balances have been eliminated.

Spin-off of Gaia

On February 20, 2015, our wholly-owned subsidiary Gaia, Inc. (Gaia formerly referred to as Gaiam TV) filed a registration statement on Form 10 with the Securities and Exchange Commission in connection with the previously announced proposed separation of the Gaia segment from the Gaiam Brand segment into two separate publicly traded companies. Gaia filed amendments to the Form 10 with the Securities and Exchange Commission on May 29, 2015, September 9, 2015, and February 17, 2016. Gaia has filed a listing application with NASDAQ and is in the process of listing on NASDAQ in anticipation of the spin-off. We currently expect that the proposed tax-free spin-off will occur through a distribution to Gaiam, Inc. s shareholders of all the stock of Gaia. Gaia would hold all of the assets and liabilities of the Gaia segment. The Gaiam Brand segment will remain with Gaiam, Inc. after the distribution. The completion of the separation is subject to satisfaction of several conditions. Furthermore, our board of directors has the right and ability, in its sole discretion, to abandon the proposed separation at any time before the distribution date. As a result, there can be no assurance that the separation will occur.

If the proposed spin-off occurs, Gaia anticipates entering into a reorganization agreement with Gaiam, Inc. to provide for, among other things, the principal corporate transactions required to effect the spin-off, certain conditions to the spin-off and provisions governing the relationship between Gaia and Gaiam, Inc. with respect to and resulting from the spin-off. The reorganization agreement would also provide that the holders of options to purchase Gaiam, Inc. Class A common stock who are employees or non-employee directors of Gaiam, Inc. on the record date for the distribution will receive options to purchase shares of Gaia s Class A common stock in the same ratio as shareholders. Additionally there will be a corresponding adjustment to the existing Gaiam, Inc. options held by such holder. The spin-off would not constitute a change in control for purposes of Gaiam, Inc. s equity plans, and therefore no vesting of awards will occur as a result of the spin-off. In addition, the reorganization agreement would address the treatment of the various insurance policies held by Gaiam, Inc. and Gaia after the spin-off. In anticipation of the spin-off, Gaia has entered into multiple license agreements with Gaiam, Inc. including a license agreement for the use of the Gaia trade name, and related trademarks and service marks following the spin-off.

Providing the spin-off is completed, Gaia anticipates entering into a transition services agreement with Gaiam, Inc. in connection with the separation. Under the transition services agreement, Gaiam, Inc. and Gaia would agree to provide certain services to the other for a period of up to 24 months following the spin-off, or such other shorter period as may be provided in the transition services agreement. The services to be provided may include certain corporate services including, but not limited to, management, financial, accounting, tax, human resources, payroll, technical, fulfillment,

software quality control, and certain office services as required from time to time in the ordinary course of our business. Charges for these services would be based on the actual cost of such

D-23

services without premium or mark-up, although applicable administrative and other overhead costs associated with the services will be allocated and included in the service charge. The employees of Gaia would remain eligible employees under certain employee benefit plans currently maintained by Gaiam, Inc., which will be managed under the transition services agreement.

Effective January 1, 2015, Gaiam, Inc. contributed to Gaia its 100% membership interest in Boulder Road LLC, a Colorado limited liability company. Boulder Road LLC is the sole owner of the property located at 833 West South Boulder Road in Louisville, Colorado, which is the location for our operations and the principal executive offices of Gaiam, Inc., Gaia and various other companies. The Gaiam, Inc. business unit and the Gaia business unit have entered into lease agreements with Boulder Road LLC effective with the contribution. The intercompany transactions have been eliminated in the accompanying condensed consolidated financial statements.

Discontinued Operations

During 2013, we sold our non-Gaiam-branded entertainment media distribution operation and discontinued our DRTV operations. Accordingly, the assets and liabilities, operating results, and cash flows for these businesses are presented as discontinued operations, separate from our continuing operations, for all periods presented in these consolidated financial statements and footnotes, unless indicated otherwise. See Note 17. Discontinued Operations.

2. Significant Accounting Policies

Cash

Cash represents on-demand accounts with financial institutions that are denominated in U.S. dollars and foreign currencies. At each balance sheet date, cash on hand that is denominated in a foreign currency is adjusted to reflect the exchange rate that existed at the balance sheet date. The difference is reported as a gain or loss in our statement of operations each period. Historically, such gains or losses have been immaterial.

Concentration of Risk

We have a concentration of credit risk in our accounts receivable with our top customers as shown in the table below. Target, Kohl s, and Amazon are major retailers in the United States to which we make significant sales during the year-end holiday season. A significant downturn in sales with or loss of any of these customers, would have a material adverse impact on our financial statements.

Customer Receivables as a Percent of Total Gross Receivables	2015	2014
Target	34%	50%
Kohl s	23%	11%
Amazon	17%	9%

Allowances for Doubtful Accounts and Other Accounts Receivable Reserves

We maintain allowances for doubtful accounts for estimated losses resulting from the inability of our customers to make required payments. We make estimates of the collectability of our accounts receivable by analyzing historical bad debts, specific customer creditworthiness, and current economic trends. It is our policy to write off receivables after all avenues of collection have been pursued and the probability of collection is considered remote. If the financial condition of our customers were to deteriorate such that their ability to make payments to us was impaired, additional

allowances could be required.

We record allowances for product returns to be received in future periods at the time we recognize the original sale. We record a reduction in revenue and a corresponding reduction in cost of sales. We estimate the amount of returns based on historical experience and future expectations.

D-24

We also maintain reserves for other deductions taken by our customers - such as co-op advertising and markdowns. These are estimated based on historical experience and future expectations. We estimate the amount of these deductions at the time revenue is recognized and record a reduction of revenue for markdowns, or an expense for co-op advertising.

As of December 31, 2015 and 2014, we had bad debt reserves of \$1.0 and \$0.4 million, respectively, and we had accounts receivable reserves for returns and other deductions totaling \$6.1 million and \$5.9 million, respectively.

Inventory

Inventory consists primarily of finished goods held for sale and is stated at the lower of cost (first-in, first-out method) or market. We identify the inventory items to be written down for obsolescence based on the item s current sales status and condition. We write down discontinued or slow moving inventories based on an estimate of the markdown required to sell off the inventory. As of December 31, 2015 and 2014, we estimated obsolete or slow-moving inventory to be \$0.7 million and \$1.2 million, respectively.

Advertising Costs

We expense other advertising and promotional costs as incurred. Amounts recorded as advertising expense were \$15.5 million, \$16.4 million, and \$15.3 million for the years ended December 31, 2015, 2014, and 2013, respectively, and we include these amounts in selling and operating expense.

We record sales discounts or other sales incentives as a reduction to revenue. We identify and record any cooperative advertising expenses we pay, which are for advertisements meeting the separable benefit and fair value tests, as part of selling and operating expense.

Prior to December 2014, we deferred advertising costs related to the preparation, printing, advertising and distribution of catalogs. Those costs were deferred for financial reporting purposes until the catalogs were distributed, and then they were amortized over succeeding periods based on forecasted sales, typically within 6 months.

Property and Equipment

We state property and equipment at cost less accumulated depreciation and amortization. We include in property and equipment the cost of internal-use software, including software used in connection with our websites. We expense all costs related to the development of internal-use software other than those incurred during the application development stage. We capitalize the costs we incur during the application development stage and amortize them over the estimated useful life of the software, which is typically three years. We compute depreciation of property and equipment on the straight-line method over estimated useful lives, generally three to forty-five years. We amortize leasehold and building improvements over the shorter of the estimated useful lives of the assets or the remaining term of the lease or remaining life of the building, respectively. Depreciation expense is included in Selling and operating expense, and Corporate, general and administration expense in the accompanying statements of operations.

Investments

We account for investments in equity securities that have readily determinable fair values that are not trading securities as available-for-sale securities. Unrealized changes in the fair value of an available-for-sale security are reported in accumulated other comprehensive income, net of tax, until disposed of or determined to be other-than-temporarily impaired, at which time the realized changes are reported in our statement of operations.

D-25

Purchase Accounting

We account for the attainment of a controlling interest in a business using the acquisition method. In determining the estimated fair value of certain acquired assets and liabilities, we make assumptions based upon many different factors, such as historical and other relevant information and analyses performed by independent parties. Assumptions may be incomplete, and unanticipated events and circumstances may occur that could affect the validity of such assumptions, estimates, or actual results.

Media Library

Our media library asset represents the fair value of libraries of media acquired through business combinations, the purchase price of media rights to both video and audio titles, and the capitalized cost to produce media products, all of which we market to retailers and to e-commerce and subscription customers. Our media library is shown in the accompanying balance sheets net of accumulated amortization of \$15.8 million and \$14.5 million at December 31, 2015 and 2014, respectively, and is amortized over the estimated useful lives of the titles, which range from five to fifteen years.

Capitalized media library production costs consist of costs incurred to produce the media content, net of accumulated amortization. We recognize these costs, as well as participation costs, as expenses on an individual title basis equal to the ratio that the current year s gross revenues bear to our estimate of total ultimate gross revenues from all sources to be earned over a maximum seven-year period. We state capitalized production costs at the lower of unamortized cost or estimated fair value on an individual title basis. We continually review revenue forecasts, based primarily on historical sales statistics, and revise these forecasts when warranted by changing conditions. When estimates of total revenues and other events or changes in circumstances indicate that a title has an estimated fair value that is less than its unamortized cost, we recognize an impairment loss in the current period for the amount by which the unamortized cost exceeds the title s estimated fair value.

Amortization expense for capitalized produced media content and acquired media rights is shown in the table below.

	For the Y	ears Ended De	ecember 31,
(in thousands)	2015	2014	2013
Total media amortization expense	\$ 2.071	\$ 1.021	\$ 1341

Based on total media library costs at December 31, 2015 and assuming no subsequent impairment of the underlying assets or a material increase in the video productions or media acquired, we expect the amortization expense for 2016, 2017, 2018, 2019 and 2020 to be \$2.2 million, \$1.5 million, \$1.2 million \$1.0 million and \$0.9 million, respectively.

Goodwill and Other Intangibles

Goodwill represents the excess of the purchase consideration over the estimated fair value of assets acquired less liabilities assumed in a business acquisition. Our other intangibles consist of customer related assets. We review goodwill for impairment annually or more frequently if impairment indicators arise on a goodwill reporting unit level. We have the option of first assessing qualitative factors to determine whether events and circumstances indicate that it is more likely than not that the fair value of a goodwill reporting unit is less than its carrying amount. If it is determined that the fair value for a goodwill reporting unit is more likely than not greater than the carrying amount for that goodwill reporting unit, then the two-step impairment test is unnecessary. If it is determined that the two-step

impairment test is necessary, then for step one, we compare the estimated fair value of a goodwill reporting unit with its carrying amount, including goodwill. If the estimated fair value of a goodwill reporting unit exceeds its carrying amount, we consider the goodwill of the reporting unit not impaired. If the carrying amount of a goodwill reporting unit exceeds its estimated fair value, we perform the second step

D-26

of the goodwill impairment test to measure the amount of impairment loss. We use either a comparable market approach or a traditional present value method to test for potential impairment. The process of evaluating the potential impairment of goodwill is highly subjective and requires significant judgment at many points during the analysis. Application of alternative assumptions and definitions could yield significantly different results.

Long-Lived Assets

We evaluate the carrying value of long-lived assets held and used, other than goodwill, when events or changes in circumstances indicate the carrying value may not be recoverable. We consider the carrying value of a long-lived asset impaired when the total projected undiscounted cash flows from such asset are separately identifiable and are less than the carrying value. We recognize a loss based on the amount by which the carrying value exceeds the estimated fair value of the long-lived asset. We determine the estimated fair value primarily using the projected cash flows from the asset discounted at a rate commensurate with the risk involved.

Income Taxes

We provide for income taxes pursuant to the liability method. The liability method requires recognition of deferred income taxes based on temporary differences between financial reporting and income tax bases of assets and liabilities, using current enacted income tax rates and regulations. These differences will result in taxable income or deductions in future years when the reported amount of the asset or liability is recovered or settled, respectively. Considerable judgment is required in determining when these events may occur and whether recovery of an asset, including the utilization of a net operating loss or other carryforward prior to its expiration, is more likely than not. Due to historical losses, we established a full valuation allowance on our deferred tax assets at the end of 2013.

Revenue

We recognize revenue in our Gaiam Brand segment when the goods are shipped to the customer and collection is either probable or has occurred. The amount of revenue recognized is net of estimated returns and other chargebacks (or channel credits), which are estimated using historical return and credit rates. If the actual amount of returns and chargebacks were to vary significantly from our estimates, it could materially impact our results of operations in subsequent periods. We recognize amounts billed to customers for postage and handling as revenue at the same time we recognize the revenue arising from the product sale. Travel revenues are recognized in the period which the trip begins. We recognize revenue in our Gaia segment ratably over the subscription period after collection has occurred. We present revenue net of taxes collected from customers.

Share-Based Compensation

We recognize compensation cost for share-based awards based on the estimated fair value of the award on date of grant. We measure compensation cost at the grant date based on the estimated fair value of the award and recognize compensation cost upon the probable attainment of a specified performance condition over the estimated performance period or for time based awards over the service period. We use the Black-Scholes option valuation model to estimate the fair value of the award. In estimating this fair value, we use certain assumptions, as disclosed in Note 13. Share-Based Compensation, consisting of the expected life of the option, risk-free interest rate, dividend yield, and volatility. The use of a different estimate for any one of these components could have a material impact on the amount of calculated compensation expense.

Defined Contribution Plan

We have adopted a defined contribution retirement plan under Section 401(k) of the Internal Revenue Code of 1986, as amended (Internal Revenue Code), which covers substantially all employees. Eligible employees may contribute amounts to the plan, via payroll withholding, subject to certain limitations. The 401(k) plan permits,

D-27

but does not require, us to make additional matching contributions to the 401(k) plan on behalf of all participants in the 401(k) plan. We match 50% of an employee s contribution, up to an annual maximum matching contribution of \$1,500. We made matching contributions to the 401(k) plan of \$0.2 million, \$0.2 million, and \$0.3 million in each of the years ended December 31, 2015, 2014, and 2013, respectively.

Foreign Currency Translation

Our foreign subsidiaries use their local currency as their functional currency. We translate assets and liabilities into U.S. dollars at exchange rates in effect at the balance sheet date. We translate income and expense accounts at the average monthly exchange rates during the year. We record resulting translation adjustments, net of income taxes, as a separate component of accumulated other comprehensive income.

Fair Value Measurements

Fair value represents the price that would be received to sell an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. Assets and liabilities are valued based upon observable and non-observable inputs. Valuations using Level 1 inputs are based on unadjusted quoted prices that are available in active markets for the identical assets or liabilities at the measurement date. Level 2 inputs utilize significant other observable inputs available at the measurement date, other than quoted prices included in Level 1, either directly or indirectly; and valuations using Level 3 inputs are based on significant unobservable inputs that cannot be corroborated by observable market data and reflect the use of significant management judgment.

Derivative and Hedging Instruments

We use derivative instruments to manage a portion of our exposure to changes in currency exchange rates. Our eco-travel subsidiary makes a significant amount of payments to tour operators in Canadian dollars, and to a lesser extent in South African Rand, Australian dollars, Euros, Danish Krone, and New Zealand dollars. Our derivatives have consisted of forward and option contracts. We classify the cash flows from these instruments in the same category as the cash flows from the hedged items. We do not use derivative instruments for trading or speculative purposes. Derivative instruments are measured at their fair values, using Level 2 inputs, and recognized as either assets or liabilities. The accounting for changes in the fair value of derivative instruments is based on the intended use of the derivative and the resulting designation. Our derivative instruments have not been designated as hedges for accounting purposes, therefore, gains or losses from changes in fair values are recognized in other expense, net in the accompanying statements of operations.

Comprehensive Loss

Our comprehensive loss is comprised of our net loss, noncontrolling interest net income, foreign currency translation adjustments, net of tax, and unrealized changes in the fair value of equity securities, net of tax.

The tax effects allocated to our accumulated other comprehensive loss components were as follows:

	For the Y	For the Years Ended December 31				
(in thousands)	2015	2	014	2	2013	
Before-tax amount	\$ (463)	\$	(268)	\$	(262)	

Edgar Filing	TRUPANION	INC Form 3
--------------	-----------	------------

Tax benefit	(139)	(89)	(86)
Net-of-tax amount	\$ (324)	\$ (179)	\$ (176)

Net Income (Loss) Per Share Attributable To Gaiam, Inc. Common Shareholders

Basic net income (loss) per share attributable to Gaiam, Inc. common shareholders excludes any dilutive effects of options. We compute basic net income (loss) per share attributable to Gaiam, Inc. common shareholders using

D-28

the weighted average number of common shares outstanding during the period. We compute diluted net income (loss) per share attributable to Gaiam, Inc. common shareholders using the weighted average number of common shares and common stock equivalents outstanding during the period. We excluded weighted average common stock equivalents of 954,000, 725,000, and 1,440,000 from the computation of diluted net income (loss) per share attributable to Gaiam, Inc. common shareholders for 2015, 2014 and 2013, respectively, because their effect was antidilutive.

The following table sets forth the computation of basic and diluted net (loss) income per share attributable to Gaiam, Inc. common shareholders:

	For the Years Ended December 31			ber 31,		
(in thousands, except per share data)		2015		2014	2	2013
Net income (loss) attributable to Gaiam, Inc. common shareholders:						
Income (loss) from continuing operations	\$	397	\$	(6,589)	\$(20,757)
Loss from discontinued operations	((12,103)		(3,327)		(1,995)
Net loss attributable to Gaiam, Inc.	\$ ((11,706)	\$	(9,916)	\$(22,752)
Weighted average shares basic		24,510	,	24,228		22,972
Effect of dilutive securities: weighted average stock options		102				
Weighted average shares diluted		24,612		24,228		22,972
Net income (loss) per share attributable to Gaiam, Inc. common shareholders - basic and diluted:						
Income (loss) from continuing operations	\$	0.01	\$	(0.27)	\$	(0.90)
Loss from discontinued operations		(0.49)		(0.14)		(0.09)
Basic and diluted net loss per share attributable to Gaiam, Inc.	\$	(0.48)	\$	(0.41)	\$	(0.99)
Fair Value of Financial Instruments						

The carrying amounts of our cash and cash equivalents, accounts receivable, accounts payable and other current liabilities approximate their fair values.

Recently Issued Accounting Pronouncements

In February of 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update No. 2016-02, *Leases* (Topic 842). The new standard establishes a right-of-use (ROU) model that requires a lessee to record a ROU asset and a lease liability on the balance sheet for all leases with terms longer than 12 months. Leases will be classified as either finance or operating, with classification affecting the pattern of expense recognition in the income statement. The new standard is effective for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. A modified retrospective transition approach is required for lessees for capital and operating leases existing at, or entered into after, the beginning of the earliest comparative period presented in the financial statements, with certain practical expedients available. We are currently evaluating the impact of our pending adoption of the new standard on our consolidated financial statements.

In May of 2014, the FASB issued ASU No. 2014-09, *Revenue from Contracts with Customers (Topic 606)*. The new standard supersedes most previously existing revenue recognition rules, and will become effective for us in the first quarter of 2017. The core principle of the guidance is that an entity should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. Our revenue transactions typically consist of one, distinct, fixed-price performance obligation which is delivered to the customer at a single point in time, or

D-29

over a subscription period. We are monitoring the evolving interpretations and implementation guidance. Our preliminary assessment is that we do not expect the new standard to have a material impact on our reported financial position or results of operations.

Use of Estimates and Reclassifications

The preparation of financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the amounts reported in the accompanying financial statements and disclosures. Although we base these estimates on our best knowledge of current events and actions that we may undertake in the future, actual results may be different from the estimates. We have made certain reclassifications to prior period amounts to conform to the current period presentations.

3. Related Party Transactions

Joint Venture

On July 24, 2015, the Company formed a partnership with a third party to jointly market a fitness-based infomercial and its associated products. Under the arrangement, the Company has provided rights to an infomercial which it has developed, and the third party will provide the necessary working capital to advertise and market the infomercial worldwide. We hold a 49% interest in the partnership, and are accounting for it as an equity method investment. During 2015 we recognized a loss of \$0.4 million associated with the venture which is included in Loss from equity method investment in the accompanying condensed consolidated statements of operations. As of December 31, 2015 our investment in the joint venture totaled \$1.8 million and is included in Other assets in the accompanying balance sheet.

Real Goods Solar

Real Goods Solar, Inc. (RGSE) was a division of our company, until it was spun off in an initial public offering in 2008.

At the beginning of 2013, we still held some shares in RGSE and were reporting it as an equity method investment; and we also had loans receivable from RGSE totaling \$2.7 million. The loans had zero carrying value. On April 23, 2013, we agreed to convert \$0.1 of the loan balance into 62,111 shares of RGSE s Class A common stock. On November 5, 2013, we collected \$2.1 million in cash from RGSE and \$0.2 million of tenant improvements in settlement of the two outstanding loans made. The \$2.3 million gain resulting from the collection of these loans is reported in Interest and other income on our consolidated statement of operations for the year ended December 31, 2013.

During 2013, we also sold the majority of our investment in RGSE for total net proceeds of approximately \$25 million. Following the sale of the majority of our position in May 2013, our voting ownership percentage declined to below 20% and our Chairman resigned as Chairman of RGSE s board of directors and, thus, we no longer had significant influence over RGSE. Therefore, we changed our accounting for our investment in RGSE from the equity method to the cost method. From that time forward, we have not reported any portion of RGSE s net income or loss in our results.

4. Other Current Assets

Other current assets consist of the following as of December 31:

Edgar Filing: TRUPANION INC. - Form 3

(in thousands)	2015	2014
Prepaid travel deposits	\$ 5,494	\$ 5,216
Advances	4,603	2,592
Other current assets	4,746	4,190
	\$ 14,843	\$ 11,998

5. Property and Equipment and Other Assets

Property and equipment, stated at lower of cost or estimated fair value, consists of the following as of December 31:

(in thousands)	2015	2014
Land	\$ 5,603	\$ 5,603
Buildings	17,115	16,809
Furniture, fixtures and equipment	8,727	7,327
Leasehold improvements	1,141	1,622
Website development costs and other software	13,987	11,678
Studios, computer and telephone equipment	9,475	9,293
Warehouse and distribution equipment	1,765	1,765
	57,813	54,097
Accumulated depreciation and amortization	(32,767)	(30,866)
	\$ 25,046	\$ 23,231

Other Assets consists of the following as of December 31:

(in thousands)	2015	2014
Working capital and related receivables, net (See Note 9)	\$	\$ 7,250
Other assets	5,148	5,417
	\$ 5,148	\$ 12,667

6. Acquisitions

Yoga Studio

In September 2014, our Gaiam Brand segment acquired all the outstanding stock of Modern Lotus Limited (Yoga Studio), the number-one-ranked paid yoga app on the U.S. App Store. The aggregate consideration paid by us exceeded the aggregate estimated fair value of the assets acquired and the liabilities assumed by \$1.5 million which we have recognized as goodwill. We attribute the goodwill to the recognized market position of the app.

My Yoga Online

In October 2013, our Gaia segment acquired all of the outstanding common stock of My Yoga Online, ULC (My Yoga Online), an on-line yoga video streaming subscription business in Canada. The aggregate consideration paid by us exceeded the aggregate estimated fair value of the assets acquired and liabilities assumed by \$10.6 million, which we have recognized as goodwill. This goodwill is attributable to the domain expertise of the assembled workforce that has demonstrated an ability to generate new content, subscriber growth and revenues from future subscribers.

The following table sets forth the changes in goodwill for the period December 31, 2013 through December 31, 2015 by segment.

(in thousands) Balance at December 31, 2013 Acquisitions	Gaiam Brand Segment 3,390 1,449	Gaia Segment 10,609	Total 13,999 1,449
Balance at December 31, 2014 Acquisitions	\$ 4,839	\$ 10,609	\$ 15,448
Balance at December 31, 2015	\$ 4,839	\$ 10,609	\$ 15,448

D-31

The following table represents our intangibles assets by major class as of December 31, 2015 and 2014.

(in thousands)	As of December 31, 2015 2014		
Indefinite Lived Intangibles	\$ 340	\$ 340	
Intangibles Subject to Amortization			
Customer related:			
Gross carrying amount	\$ 1,417	\$ 1,741	
Accumulated amortization	(1,171)	(1,399)	
	\$ 246	\$ 342	
Marketing related:			
Gross carrying amount	\$ 1,029	\$ 729	
Accumulated amortization	(749)	(588)	
	\$ 280	\$ 141	

The amortization periods range from 24 to 60 months. Amortization expense for the years ended December 31, 2015, 2014, and 2013 was \$0.4 million, \$0.7 million, and \$0.3 million, respectively.

7. Accrued Liabilities

Accrued liabilities consist of the following as of December 31:

(in thousands)	2015	2014
Accrued compensation	\$ 3,896	\$ 3,323
Customer travel deposits	12,747	11,072
Accrued legal expense and related reserves		3,000
Other accrued liabilities	3,722	2,841
	\$ 20,365	\$ 20,236

Accrued compensation at December 31, 2015 and 2014 includes severance and termination benefits of \$1.7 million.

8. Credit Facility

On July 23, 2015, Boulder Road LLC, a subsidiary of Gaia, entered into a revolving line of credit agreement with a bank in the amount of \$5.5 million. The note bears interest at the prime rate plus 3.25%, is guaranteed by Gaiam, Inc. and Gaia, and is secured by a Deed of Trust filed against the real property on which the principal offices of the Company are located. The value of our principal offices is far in excess of the amount of the line of credit. No amounts were drawn or outstanding under the line of credit during 2015.

9. Commitments and Contingencies

Working Capital Arbitration

Since 2014, Cinedigm Corp. (Cinedigm) and Gaiam have been engaged in various legal disputes relating to Gaiam s sale of its entertainment media distribution business to Cinedigm. In response to the disputes, the Company accrued a litigation-related reserve of \$3.0 million in 2014.

In a settlement agreement made effective as of September 30, 2015, Cinedigm and Gaiam agreed to the following: (1) a mutual release of all claims, with only one exception (described immediately below), that the

D-32

parties held against each other; (2) the commencement of a further arbitration to resolve Cinedigm s single preserved claim that it did not receive all of the cash collected by Gaiam on Cinedigm s behalf during the transition period following the sale (the Cash Reconciliation Claim); and (3) Gaiam would pay Cinedigm \$2.3 million to Cinedigm.

In a further settlement agreement made effective as of December 31, 2015, Cinedigm and Gaiam agreed to resolve the Cash Reconciliation Claim in exchange for a further payment by Gaiam to Cinedigm in the amount of \$1.6 million.

As a result, all legal disputes between the parties have now been finally and fully settled. The parties settlements do not constitute an admission by either party of any liability or wrongdoing whatsoever. The Company entered into the Settlement in order to (i) eliminate the uncertainty and risk inherent in any litigation, (ii) significantly reduce the professional fees and costs that the litigation would require, and (iii) significantly reduce the distraction imposed by the process on management.

During 2015, the Company recorded charges of \$12.1 million which include a non-cash charge of \$7.3 million for the working capital receivable due from Cinedigm. The expenses associated with the settlements and arbitration have all been included in Loss from discontinued operations in the accompanying condensed consolidated statements of operations.

Risks and Uncertainties

From time to time, we are involved in legal proceedings that we consider to be in the normal course of business. Claimed amounts against us may be substantial but may not bear any reasonable relationship to the merits of the claim or the extent of any real risk of court or arbitral awards. We record accruals for losses related to those matters against us that we consider to be probable and that can be reasonably estimated. Although it is not feasible to predict the outcome of these matters with certainty, it is reasonably possible that some legal proceedings may be disposed of or decided unfavorably to us and in excess of the amounts currently accrued. Based on available information, in the opinion of management, settlements, arbitration awards and final judgments, if any, which are considered probable of being rendered against us in litigation or arbitration in existence at December 31, 2015 and can be reasonably estimated are reserved against or would not have a material adverse effect on our financial condition, results of operations or cash flows.

Operating Leases

We lease office and warehouse space through operating leases. Some of the leases have renewal clauses, which range from 3 to 6 years.

The following schedule represents the annual future minimum payments under these commitments, as of December 31, 2015:

(in thousands)	_	rating eases
2016	\$	829
2017		754
2018		827
2019		432

Total minimum lease payments

\$ 2,842

We incurred rent expense of \$1.7 million, \$1.1 million and \$1.0 million for the years ended December 31, 2015, 2014 and 2013, respectively.

D-33

Media Distribution and Other Minimum Royalty Payments

From time to time, we enter into media distribution or royalty agreements which require that we make periodic minimum payments against future liabilities. The following schedule shows the annual future minimum payments under these agreements, as of December 31, 2015:

	Distribution
(in thousands)	Payments
2016	\$ 2,698
2017	528
2018	96
2019	25
Total minimum payments	\$ 3,347

Spin-off of Gaia

The potential spin-off of Gaia will have a significant impact on our financial statements if it occurs. See further discussion in Notes 1 and 16 to our consolidated financial statements.

10. Derivative Contracts

Our operations in foreign countries expose us to market risk associated with foreign currency exchange rate fluctuations between the U.S. dollar and various foreign currencies. We use derivative instruments to manage a portion of our exposure to changes in currency exchange rates due to payments made by our eco-travel subsidiary to tour operators in Canada, South Africa, Australia, Europe, Denmark, and New Zealand. Our primary objective for entering into currency derivatives is to reduce the volatility that changes in currency exchange rates have on our earnings. In general, the market risk related to these contracts is offset by corresponding gains and losses on the hedged transactions. By working only with major banks and closely monitoring current market conditions, we seek to limit the risk that counterparties to these contracts may be unable to perform. We have presented amounts on a gross basis as we do not have any master netting agreements with counterparties. We do have the legal right to net settle transactions of the same currency with a single net amount payable by one party or the other. We do not enter into derivative contracts for trading purposes.

The cash flow effects of our derivative contracts for the year ended December 31, 2015 are included within net cash provided by operating activities in the Statements of Cash Flows. Realized and unrealized gains and losses on currency derivatives without hedge accounting designation are included in Interest and other (expense) income in the accompanying statements of operations. For the period ended December 31, 2015 the losses recognized were \$1.3 million. Prior to 2015, outstanding contracts and associated gains or losses were immaterial. The liability related to the fair value of the hedging instruments has been included in Accounts payable in the accompanying balance sheet.

Total notional amounts and gross fair values for derivative instruments without hedge accounting designation were as follows as of December 31, 2015:

Edgar Filing: TRUPANION INC. - Form 3

(in thousands)	Notional Amount in USD	Fair Value of Current Liabilities
Currency forward contracts		
Canadian dollars	\$ 4,200	\$ (246)
Other currencies	1,372	(83)
	5,572	(329)
Currency option contracts		
Canadian dollars	8,514	(252)
Total	\$ 14,086	\$ (581)

11. Equity

Our common stock has two classes, Class A and Class B. Each holder of our Class A common shares is entitled to one vote for each share held on all matters submitted to a vote of shareholders. Each of our Class B common shares is entitled to ten votes on all matters submitted to a vote of shareholders. There are no cumulative voting rights. All holders of our Class A common shares and our Class B common shares vote as a single class on all matters that are submitted to the shareholders for a vote. Shareholders may consent to an action in writing and without a meeting under certain circumstances. Jirka Rysavy, our chairman, holds 100% of our 5,400,000 outstanding shares of class B common stock and also owns 348,682 shares of Class A common stock. Consequently, our chairman holds approximately 74% of our voting stock and thus is able to exert substantial influence over us and to control matters requiring approval by our shareholders, including the election of directors, increasing our authorized capital stock, or a merger or sale of substantially all of our assets. As a result of Mr. Rysavy s control of us, no change of control can occur without Mr. Rysavy s consent.

Our Class A common shares and our Class B common shares are entitled to receive dividends, if any, as may be declared by the board of directors out of legally available funds. In the event of a liquidation, dissolution or winding up of our Company, our Class A common shares and our Class B common shares are entitled to share ratably in our assets remaining after the payment of all of our debts and other liabilities. Holders of our Class A common shares and our Class B common shares have no preemptive, subscription or redemption rights, and there are no redemption or sinking fund provisions applicable to our Class A common shares or our Class B common shares.

Our Class B common shares may not be transferred unless converted into our Class A common shares, other than certain transfers to affiliates, family members, and charitable organizations. Our Class B common shares are convertible one-for-one into our Class A common shares, at the option of the holder of the Class B common shares.

During 2015, 2014 and 2013, we issued shares of our Class A common stock as shown in the table below under our 2009 Long-Term Incentive Plan. We recorded the shares issued to our directors at their estimated fair value based on the market s closing price of our stock on the date the shares were issued, which by policy is the last trading day of each quarter in which the services were rendered.

	For the Years Ended December 31,			
	2015 2014 201			
Restricted shares issued to independent directors for services rendered, in lieu				
of cash compensation	16,887	19,542	49,187	
Shares issued to employees upon exercise of stock options	33,825	354,926	160,470	
As of December 31, 2015, we had the following Class A common shares reser	ved for future	iccuance.		

Conversion of Class B common shares	5,400,000
Awards under the 2009 and 1999 Long-Term Incentive Plans:	
Stock options outstanding	1,478,724
Total shares reserved for future issuance	6,878,724

12. Non-Controlling Interests

We own 51.4% of our eco-travel subsidiary, Natural Habitat Adventures (Natural Habitat). The balance is owned by its founder. In addition, some of Natural Habitat subsidiaries also have minority shareholders. We own 50.01% of our Australian subsidiary, Gaiam Pty. The amount of these non-controlling interests is reflected separately in our consolidated financial statements, and all intercompany transactions have been eliminated.

D-35

During 2015 and 2014, Natural Habitat paid its shareholders dividends of \$1.0 million and \$0.3 million, respectively, and, as a result, the noncontrolling interests decreased by \$0.5 million and \$0.2 million, respectively. No dividends were declared or paid by Natural Habitat during 2013.

13. Share-Based Compensation

We are currently issuing options under the 2009 Long-Term Incentive Plan (the Plan). We previously issued options under the 1999 Long-Term Incentive Plan, which was replaced with the Plan. The purpose of the Plan is to advance our interests and those of our shareholders by providing incentives to certain persons who contribute significantly to our strategic and long-term performance objectives and growth. An aggregate of not more than 3 million of our Class A common shares, subject to certain adjustments, may be issued under the Plan, and the Plan terminates no later than April 23, 2019. The exercise price for our options is generally equal to the closing market price of our stock at the date of the grant, and the options normally vest at 2% per month for the 50 months beginning in the eleventh month after the grant date. Follow on option grants begin vesting in the first month after grant. We recognize the compensation expense related to share-based payment awards on a straight-line basis over the requisite service periods of the awards, which are generally five years for employees, and two years for board members. Commencing with options granted during 2012, we extended the exercise period from seven to ten years.

The determination of the estimated fair value of share-based payment awards on the date of grant using the Black-Scholes option-pricing model is affected by our stock price as well as assumptions regarding a number of complex and subjective variables. We derive the expected terms from the historical behavior of participant groupings. We base expected volatilities on the historically realized volatility of our stock over the expected term. Our use of historically realized volatilities is based upon the expectation that future volatility over the expected term is not likely to differ significantly from historical results. We base the risk-free interest rate used in the option valuation model on U.S. Treasury zero-coupon issues with remaining terms similar to the expected term on the options. We estimate forfeitures at the time of grant and revise those estimates in subsequent periods if actual forfeitures differ from those estimates. We primarily use historical data by participant groupings to estimate option forfeitures and record share-based compensation expense only for those awards that are expected to vest.

The following are the variables we used in the Black-Scholes option pricing model to determine the estimated grant date fair value for options granted under the Plan for each of the years presented:

	2015	2014	2013
Expected volatility	34% - 50%	48% - 59%	57% - 61%
Weighted-average volatility	43%	55%	58%
Expected dividends	%	%	%
Expected term (in years)	0.5 - 5.9	1.1 - 7.8	5.1 - 7.8
Risk-free rate	0.26% - 1.73%	0.14% - 2.37%	1.33% - 2.32%

The table below presents a summary of option activity under both of our plans as of December 31, 2015, and changes during the year then ended:

Shares	Weighted-	Weighted-	Aggregate
	Average	Average	Intrinsic
	Exercise	Remaining	Value

Edgar Filing: TRUPANION INC. - Form 3

		P	rice	Contractual Term	
Outstanding at January 1, 2015	1,448,684	\$	6.17		
Granted	514,610		6.09		
Exercised	(28,836)		5.54		
Cancelled or forfeited	(455,734)		6.06		
Outstanding at December 31, 2015	1,478,724	\$	6.19	5.8	\$ 622,420
Exercisable at December 31, 2015	804,484	\$	5.90	3.7	\$ 504,629

During 2015, 2014 and 2013, we extended the exercise period on the options of certain former board members and key employees and recognized an additional immediate expense of \$0.1 million, \$0.1 million and \$0.1 million, respectively. On October 11, 2013, we issued 15,759 shares of our Class A common stock under restricted stock award agreements to certain former board members. The estimated fair value of these restricted stock awards was \$0.1 million and was based on the closing market price of our stock on October 11, 2013. These restricted stock awards vested 100% on April 10, 2014.

We issue new shares upon the exercise of options. We received approximately \$0.2 million, \$1.8 million and \$0.8 million in cash from stock options exercised during 2015, 2014 and 2013, respectively. The weighted-average grant-date fair value of options granted during the years 2015, 2014, and 2013 was \$1.86, \$2.98, and \$3.14, respectively. The total intrinsic value of options exercised during 2015 and 2014 was \$0.1 million and \$0.9 million, respectively. The total fair value of shares vested was \$0.9 million, \$0.6 million, and \$0.6 million during 2015, 2014, and 2013.

Our share-based compensation cost charged against income was \$0.9 million, \$0.8 million, and \$0.8 million during 2015, 2014, and 2013, respectively, and is included in corporate, general and administration expenses. The total income tax benefit recognized for share-based compensation was \$0.3 million, \$0.3 million, and \$0.3 million for 2015, 2014, and 2013, respectively. As of December 31, 2015, there was \$2.4 million of unrecognized cost related to non-vested shared-based compensation arrangements granted under our 2009 and 1999 Long-Term Incentive Plans. We expect that cost to be recognized over a weighted-average period of 3.0 years.

Long-term deferred equity plan

On February 26, 2015, our board of directors approved the Long-Term Deferred Equity Plan (the deferred equity plan) as an incentive plan for the management of our Gaia segment. In anticipation of the spin-off, our board of directors granted restricted stock units (RSUs) for 173,976 shares of our subsidiary Gaia s Class A common stock under the deferred equity plan to certain of our officers and employees involved in the Gaia segment. In connection with the grants, each recipient entered into an individual restricted stock unit award agreement with Gaia with the following terms: (i) the recipient is entitled to receive one share of our subsidiary Gaia s Class A common stock for each RSU upon vesting, and (ii) the RSUs will vest on March 16, 2020, provided that (a) the spin-off has occurred by that date, and (b) the recipient is still an employee or director of the Company on such date. The RSUs will be automatically forfeited and of no further force and effect if either of the vesting conditions are not met. Given the performance vesting provisions of these awards, the fair value of these awards will be determined upon the completion of the spin-off and recognized over the remaining vesting period.

14. Asset Impairments and Exit Activity Costs

During 2013, as a result of the reorganization and re-focus of our continuing businesses following the discontinuation of our non-Gaiam-branded entertainment media distribution and direct response television marketing operations, we impaired \$4.4 million of media libraries and capitalized production costs, \$1.5 million of advances, and \$1.3 million of property, plant, and equipment, net of accumulated depreciation, and other investments. These noncash impairments reduced the carrying value of assets for our Gaiam Brand segment by \$7.2 million. We estimated the fair value of each impaired asset category using a traditional present value technique, relying upon various sources of information for our assumptions, such as estimated future sales, internal budgets and projections, and judgment about the related product—s future earnings potential (level 3 of the fair value hierarchy). We also recorded termination benefits of \$2.5 million related to the termination of certain employees associated with our restructuring and future retirement benefits for one of our executive officers. These asset impairment and termination benefit charges were recorded in other general expense on our consolidated statement of operations. Also included in other general

expenses on our 2013 consolidated statement of operations are \$1.3 million of expenses related to a brand study, recruiting for a new CEO, and other operating expenses that management believes are not ongoing expenses related to the operations of the Company.

D-37

Changes in the accrual liability associated with termination benefits were as follows:

Charges	\$ 2,472
Payments, net	(298)
Balance December 31, 2013	2,174
Payments, net	(308)
Reversals and Adjustments	(101)
Balance December 31, 2014	1,765
Payments, net	(112)
Balance December 31, 2015	\$ 1,653

The \$1.7 million accrual balance at December 31, 2015 is expected to be paid \$0.9 million in 2016, \$0.4 million in 2017 and \$0.4 million in 2018.

15. Income Taxes

Our provision for income tax expense is comprised of the following:

	For the Years Ended December 31,				
(in thousands)	2015	,	2014	2	2013
Current:					
Federal	\$1,148	\$	887	\$	536
State	14		157		(68)
International	57		175		223
	1,219		1,219		691
Deferred:					
Federal					22,418
State					1,538
International					73
					24,029
Total income tax expense	\$1,219	\$	1,219	\$	24,720

Because our eco-travel subsidiary is not part of the consolidated tax group, the provision includes Federal taxes associated with its income. The state provision consists of the taxes due for subsidiaries of Gaiam which file taxes on a separate basis, and are not able to utilize combined group net operating losses.

The components of our income taxes consisted of the following:

(in thousands)	2015	2014	2013
Income tax expense from continuing operations	\$1,219	\$ 1,369	\$ 25,974
Income tax (benefit) expense from discontinued operations		(150)	209
Income tax benefit from loss on disposal of discontinued operations			(1,463)
Total income tax expense	\$1,219	\$ 1,219	24,720

Variations from the federal statutory rate are as follows:

(in thousands)	2015	2014	2013
Expected federal income tax expense (benefit) at statutory rate of			
34%	\$ 785	\$ (2,631)	\$ 898
Effect of 2008 State NOL s and option forfeitures			49
Effect of permanent other differences	103	37	213
Effect of repatriation and nondeductible foreign losses	422		
State income tax (benefit) expense, net of federal benefit		(150)	24
Establishment of valuation allowance on net deferred tax assets	(103)	4,071	23,153
Other	14	(70)	278
Effect of differences between U.S. taxation and foreign taxation	(2)	(38)	105
Total income tax expense	\$ 1,219	\$ 1,219	24,720

Deferred income taxes reflect net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes. The components of the net accumulated deferred income tax assets as of December 31, 2015 and 2014 are as follows:

Deferred tax assets (liabilities): Current: Provision for doubtful accounts \$ 318 \$ 1,211 Inventory-related expense 573 483 Accrued liabilities 714 1,420 Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803		December 31,	
Current: Provision for doubtful accounts \$ 318 \$ 1,211 Inventory-related expense 573 483 Accrued liabilities 714 1,420 Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803	(in thousands)	2015	2014
Provision for doubtful accounts \$ 318 \$ 1,211 Inventory-related expense 573 483 Accrued liabilities 714 1,420 Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803	Deferred tax assets (liabilities):		
Inventory-related expense 573 483 Accrued liabilities 714 1,420 Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803	Current:		
Accrued liabilities 714 1,420 Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803	Provision for doubtful accounts	\$ 318	\$ 1,211
Nonqualified stock options 1,723 1,221 Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803 *** Total current deferred tax assets 4,935 4,804 Valuation allowance (4,935) (4,804) Non-current: *** *** Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Inventory-related expense	573	483
Worthless stock deduction 207 206 Other 597 263 Exit activity accruals 803 Total current deferred tax assets 4,935 4,804 Valuation allowance (4,935) (4,804) Total current deferred tax assets, net of valuation allowance \$ Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Accrued liabilities	714	1,420
Other597263Exit activity accruals803Total current deferred tax assets4,9354,804Valuation allowance(4,935)(4,804)Total current deferred tax assets, net of valuation allowance\$Non-current:Depreciation and amortization\$ (1,896)\$ (1,506)Section 181 qualified production expense(3,563)(2,770)Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)	Nonqualified stock options	1,723	1,221
Exit activity accruals Total current deferred tax assets Valuation allowance 4,935 4,804 Valuation allowance (4,935) Total current deferred tax assets, net of valuation allowance Non-current: Depreciation and amortization Section 181 qualified production expense (3,563) Section 181 qualified production expense (3,563) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Worthless stock deduction	207	206
Total current deferred tax assets Valuation allowance Total current deferred tax assets, net of valuation allowance S Non-current: Depreciation and amortization Section 181 qualified production expense Net operating loss carryforward Charitable carryforward Signature of investment, net Gain from foreign business acquisition 4,804 4,804 4,804 4,804 4,804 5 (1,896) \$ (1,506) \$ (1,506) \$ (2,770) 1,899 1,414 48 48 63in from foreign business acquisition (103) (347)	Other	597	263
Valuation allowance (4,935) (4,804) Total current deferred tax assets, net of valuation allowance \$ Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Exit activity accruals	803	
Valuation allowance (4,935) (4,804) Total current deferred tax assets, net of valuation allowance \$ Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)			
Total current deferred tax assets, net of valuation allowance \$ Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Total current deferred tax assets	4,935	4,804
Total current deferred tax assets, net of valuation allowance \$ Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	Valuation allowance	(4,935)	(4,804)
Non-current: Depreciation and amortization \$ (1,896) \$ (1,506) Section 181 qualified production expense (3,563) (2,770) Net operating loss carryforward 30,727 26,966 Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)			
Depreciation and amortization\$ (1,896)\$ (1,506)Section 181 qualified production expense(3,563)(2,770)Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)	Total current deferred tax assets, net of valuation allowance	\$	\$
Depreciation and amortization\$ (1,896)\$ (1,506)Section 181 qualified production expense(3,563)(2,770)Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)			
Section 181 qualified production expense(3,563)(2,770)Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)	Non-current:		
Section 181 qualified production expense(3,563)(2,770)Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)	Depreciation and amortization	\$ (1,896)	\$ (1,506)
Net operating loss carryforward30,72726,966Charitable carryforward1,8991,414Loss from change in financial statement carrying value of investment, net4848Gain from foreign business acquisition(103)(347)	^	(3,563)	
Charitable carryforward 1,899 1,414 Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)	* *		
Loss from change in financial statement carrying value of investment, net 48 48 Gain from foreign business acquisition (103) (347)		1,899	1,414
Gain from foreign business acquisition (103) (347)			
•		(103)	(347)
	Impairment of intangibles	• • •	, ,

Edgar Filing: TRUPANION INC. - Form 3

Spin-off and capitalized legal costs	587	
Tax credits	922	921
Other	(3)	(3)
Total non-current deferred tax assets	28,985	25,090
Valuation allowance	(28,985)	(25,090)
Total non-current deferred tax assets, net of valuation allowance		
Total net deferred tax assets	\$	\$

As of December 31, 2015, our gross net operating losses were \$79.3 million and \$16.6 million for federal and state, respectively. The sources of income (loss) before income taxes are as follows:

(in thousands)	2015	2014	2013
Domestic	\$ 2,109	\$ (4,947)	\$5,503
International	201	686	373
	\$ 2,310	\$ (4,261)	\$5,876

Certain of our subsidiaries, namely those for which we own less than 80% of their shares and voting rights and/or are foreign entities, file tax returns separately from Gaiam s consolidated tax group. At December 31, 2015, we had made a provision for U.S. federal and state income taxes on approximately \$0.2 million of undistributed foreign earnings, which are not expected to remain outside of the U.S. indefinitely. Deferred tax liabilities have been established for future taxes on distribution of foreign earnings in the form of dividends or otherwise, in order to derive, for financial statement purposes, the U.S. income taxes (net of tax on foreign tax credits), state income taxes, and withholding taxes payable to the various foreign countries.

Periodically, we perform assessments of the realization of our net deferred tax assets considering all available evidence, both positive and negative. On the basis of this assessment, we recorded a charge of \$23.2 million to income tax expense to record a full valuation allowance against our deferred tax assets as of December 31, 2013. A significant piece of evidence evaluated was the cumulative loss incurred over the three-year period ended December 31, 2013. Because the valuation allowance will remain in place until we return to profitability, we did not record any tax benefit in 2015 associated with our net loss or other deferred tax assets.

We realized \$0.1 million and \$1.3 million in tax benefits recorded to additional paid-in capital as a result of the exercise of stock options during 2015 and 2014, respectively. We recognize the tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. We measure the tax benefits recognized in the consolidated financial statements from such a position based on the largest benefit that has a greater than 50% likelihood of being realized upon ultimate resolution. The application of income tax law is inherently complex. Laws and regulations in this area are voluminous and are often ambiguous. As such, we are required to make many subjective assumptions and judgments regarding our income tax exposures. Interpretations of and guidance surrounding income tax law and regulations change over time and may result in changes to our subjective assumptions and judgments which can materially affect amounts recognized in our consolidated balance sheets and consolidated statements of operations. The result of our assessment of our uncertain tax positions did not have a material impact on our consolidated financial statements. Our federal and state tax returns for all years after 2010 are subject to future examination by tax authorities for all our tax jurisdictions. We recognize interest and penalties related to income tax matters in interest and other income (expense) and corporate, general and administration expenses, respectively.

As specified by our Tax Sharing Agreement with RGSE, to the extent RGSE becomes entitled to utilize certain loss carryforwards relating to periods prior to its initial public offering, it will distribute to us the tax effect (estimated to be 34% for federal income tax purposes) of the amount of such tax loss carryforwards so utilized. These net operating loss carryforwards expire beginning in 2018 if not utilized, and are subject to IRS limitations. These tax assets have a full valuation allowance based on RGSE s current financial performance. As of December 31, 2015, \$4.4 million of these net operating loss carryforwards remained available for current and future utilization, meaning that RGSE s potential future payments to us, which would be made over a period of several years, could therefore aggregate to

Edgar Filing: TRUPANION INC. - Form 3 $\,$

approximately \$1.6 million based on current tax rates.

D-40

16. Segment and Geographic Information

Segment Information

We report two business segments which are aligned based on their products or services:

Gaiam Brand: This segment includes all our branded yoga, fitness, and well-being products. It also includes our

eco-travel subsidiary.

Gaia: This segment includes our digital video streaming service and was formerly referred to as Gaiam TV.

This segment also includes the results of Boulder Road LLC from January 1, 2015 forward. We previously announced that we are pursuing the potential spin-off of this segment into a separate

company.

Amounts shown as Other unallocated corporate in the table below represents a portion of our revenues, expenses and assets that we do not allocate to our segments. Portions of the unallocated corporate amounts may be included in the spin-off with Gaia, if and when that occurs. Certain prior year amounts have been reclassified to conform to the current year presentation.

Although we are able to track sales by channel, the management, allocation of resources, and analysis and reporting of expenses are presented on a combined basis, at the reportable segment level. Segment contribution margin is defined as net revenue less cost of goods sold and total operating expenses. Financial information for our segments is as follows:

Year Ended December 31,					
2015	2014	2013			
\$ 174,559	\$ 156,784	\$ 149,812			
13,459(a)	9,910(a)	5,651(a)			
188,018	166,694	155,463			
13,147	6,395	(9,394)(b)			
(5,125)	(8,718)	(10,144)(c)			
8,022	(2,323)	(19,538)			
(3,841)	(2,763)	(2,103)			
4,181	(5,086)	(21,641)			
(1,406)	(600)	2,421			
	1,480	25,096			
(465)	(55)				
(1,219)	(1,369)	(25,974)			
	2015 \$ 174,559 13,459(a) 188,018 13,147 (5,125) 8,022 (3,841) 4,181 (1,406) (465)	2015 2014 \$ 174,559 \$ 156,784 13,459(a) 9,910(a) 188,018 166,694 13,147 6,395 (5,125) (8,718) 8,022 (2,323) (3,841) (2,763) 4,181 (5,086) (1,406) (600) 1,480 (465) (55)			

Edgar Filing: TRUPANION INC. - Form 3

(Loss) income from discontinued operations, net of tax	(12,103)	(3,327)	(1,995)
Net income attributable to noncontrolling interest	(694)	(959)	(659)
Net loss attributable to Gaiam, Inc.	\$ (11,706)	\$ (9,916)	\$ (22,752)

- (a) Gaia filed its stand-alone audited financial statements on Form 10 with the SEC on February 20, 2015 and February 17, 2016. The segment amounts presented here and discussed elsewhere in this Form 10-K vary insignificantly from the amounts in the Form 10, as the Form 10 required that certain items be recast for stand-alone presentation. As reported in Form 10, revenues were \$14.2 million, \$10.1 million, and \$5.5 million for 2015, 2014, and 2013 respectively. Contribution loss was \$5.1 million, \$8.5 million and \$10.0 million for 2015, 2014, and 2013 respectively.
- (b) During 2013 we recognized impairment and severance charges of \$9.2 million.
- (c) During 2013 we recognized impairment charges of \$1.8 million.

D-41

The following is a reconciliation of reportable segments—assets to our consolidated total assets. Other unallocated corporate amounts are comprised of cash, current and deferred income taxes, and property and equipment.

	As of December 31,						
(in thousands)	2015	2014	2013				
Total assets Continuing Operations:							
Gaiam Brand	\$ 84,329	\$ 114,388	\$ 120,604				
Gaia	44,213	23,662	19,193				
	\$ 128,542	\$ 138,050	\$ 139,797				
Total assets Discontinued Operations:							
Gaiam Brand	\$	\$ 582	\$ 1,889				
	\$ 128,542	\$ 138,632	\$ 141,686				

Major Customers

The following Gaiam Brand segment customers make up 10% or more of our total revenues. No other customer accounted for 10% or more of our total net revenue. The loss of either of these customers would have a material adverse effect on our business.

	2015	2014	2013
Target	21.8%	29.3%	32.1%
Kohl s	18.1%	4.0%	
Consumation Information			

Geographic Information

We sell and distribute essentially the same products in the United States and several foreign countries. The major geographic territories are the U.S., Canada, Australia and the U.K., and are based on the location of the customer. The following represents geographical data for our operations as of and for the years ended December 31, 2015, 2014 and 2013:

(in thousands)	2015	2014	2013
Revenue:			
United States	\$ 179,956	\$ 156,284	\$ 147,527
International	8,062	10,410	7,936
	\$ 188,018	\$ 166,694	\$ 155,463
Long-Lived Assets:			
United States	\$ 38,119	\$ 34,123	\$ 29,072
International	146	243	246

Edgar Filing: TRUPANION INC. - Form 3

\$ 38,265 \$ 34,366 \$ 29,318

	As of December 31,							
(in thousands)	2015	2014	2013					
Components of Long-Lived Assets (a):								
Property and equipment, net	\$ 25,046	\$ 23,231	\$ 22,540					
Media Library, net	9,117	7,691	5,211					
Other Intangibles, net	866	823	1,155					
Other assets	3,236	2,621	412					
	\$ 38,265	\$ 34,366	\$ 29,318					

(a) Excludes other non-current assets (goodwill, long-term receivables, and investments) of \$17.4 million, \$25.5 million, and \$22.3 million for 2015, 2014, and 2013, respectively.

D-42

17. Discontinued Operations

As of December 31, 2015, we no longer have assets or liabilities associated with discontinued operations. As of December 31, 2014 we had inventory of \$0.3 million and other current assets of \$0.3 million.

The income from discontinued operations amounts as reported on our consolidated statements of operations were comprised of the following amounts:

	Years E	Ended Decemb	ber 31,
(in thousands)	2015	2014	2013
Net revenue	\$	\$ 2,516	\$ 53,539
(Loss) income from operations before income taxes (a)	(12,103)	(3,477)	2,386
Exit activity and asset impairment charges before income taxes			
(b)			(1,776)
Income tax benefit (expense)		150	(209)
Income from operations of discontinued operations	(12,103)	(3,327)	401
Gain (loss) on disposal of discontinued operations:			
Gain on sale of GVE before income taxes (c)			5,622
Impairment of DRTV before income taxes (c)			(9,481)
Income tax benefit			1,463
Loss from disposal of discontinued operations			(2,396)
(Loss) income from discontinued operations	\$ (12,103)	\$ (3,327)	\$ (1,995)

- (a) We recorded non-cash and cash charges of \$12.1 million and \$3.3 million during 2015 and 2014, respectively, associated with the legal dispute and settlement with Cinedigm.
- (b) In direct conjunction with the discontinuing of our GVE and DRTV operations, during 2013 we recognized exit activity charges of \$0.8 million for employee termination benefits and \$1.0 million for non-cancellable facility leases, of which \$0.3 million had been paid as of December 31, 2013, and the balance of these amounts was paid in 2014.
- (c) As a direct result of the discontinuance of our GVE and DRTV operations, we recognized impairment charges of \$2.5 million for inventory, \$3.8 million for deferred advertising costs, \$0.8 million for advances, \$0.4 million for property and equipment, \$2.1 million for media library, \$6.7 million for goodwill, and \$3.5 million for other intangibles.

GVE

On October 21, 2013, we consummated the sale of GVE Newco, LLC (GVE), a wholly-owned subsidiary representing our non-Gaiam-branded entertainment media business, to Cinedigm for \$51.7 million, comprised of cash, stock and other assets and liabilities. The sale was subject to customary adjustments, including a post-closing working capital adjustment. After the sale was consummated, we continued providing extensive administrative and accounting

Edgar Filing: TRUPANION INC. - Form 3

services to the buyer through May 2014. We have not provided any significant services since that time. As discussed in Note 9, Commitments and Contingencies, we have been involved in legal disputes with Cinedigm associated with the sale, which were settled during 2015. All the associated legal and settlement costs have been reported in discontinued operations.

During 2014, the Class A shares of Cinedigm's common stock which we received in the GVE sale increased in value and were sold. The unrealized gains were reflected in accumulated other comprehensive income prior to the sale, and were reclassified into gain on sale of investments in the accompanying consolidated statements of operations after the sale.

In conjunction with the sale of GVE, we paid in full the outstanding balance owed under a Revolving Credit and Security Agreement (the PNC Credit Agreement) with PNC Bank, N.A. (PNC), of \$19.6 million (inclusive

D-43

of principal and interest and other fees), and terminated the underlying PNC Credit Agreement. We also paid an early termination fee of \$0.4 million. Upon termination, PNC released all liens granted in its favor on the collateral pledged under the PNC Credit Agreement. All interest charges under the PNC Credit Agreement have been allocated to discontinued operations.

DRTV

During the fourth quarter of 2013, we also discontinued our DRTV operations. In connection with these discontinued operations, we recognized certain exit activity and asset impairment charges. Accordingly, the assets and liabilities, operating results, and cash flows for these businesses, and their related exit activity and asset impairment charges, are presented as discontinued operations in our financial statements and footnotes presented herein.

18. Quarterly Results of Operations (Unaudited)

The following tables set forth our unaudited results of operations for each of the quarters in 2015 and 2014.

	Year 2015 Quarters Ended					
(in thousands, except per share data)	March 31	June 30	Sep	tember 30	Dec	ember 31
Net revenue	\$ 37,638	\$41,146	\$	51,300	\$	57,934
Gross profit	17,247	18,207		21,853		27,462
(Loss) income from continuing operations	(3,418)	12		2,431		2,066
(Loss) income from discontinued operations	(466)	(1,121)		(10,695)		179
Net (loss) income	(3,884)	(1,109)		(8,264)		2,245
Net (loss) income attributable to Gaiam, Inc.	(3,892)	(1,117)		(8,814)		2,117
Diluted net (loss) income per share attributable to Gaiam,						
Inc.	\$ (0.16)	\$ (0.05)	\$	(0.36)	\$	0.09
Weighted average shares outstanding-diluted	24,490	24,610		24,604		24,626

		Year 2014 Quarters Ended						
(in thousands, except per share data)	March 31	June 30	Sep	tember 30	Dec	ember 31		
Net revenue	\$37,611	\$ 32,451	\$	41,256	\$	55,376		
Gross profit	17,020	15,468		18,018		24,999		
Gain on sale of investment (a)	438	1,042						
(Loss) income from continuing operations	(2,098)	(2,216)		(2,559)		1,243		
Income (loss) from discontinued operations	26	2		(82)		(3,273)		
Net loss	(2,072)	(2,214)		(2,641)		(2,030)		
Net loss attributable to Gaiam, Inc.	(2,134)	(2,388)		(3,026)		(2,368)		
Diluted net loss per share attributable to Gaiam, Inc.	\$ (0.09)	\$ (0.10)	\$	(0.12)	\$	(0.10)		
Weighted average shares outstanding-diluted	24,006	24,090		24,340		24,470		

Edgar Filing: TRUPANION INC. - Form 3

(a) We reported gains on the sale of our RGSE stock during 2014, the carrying value for which had previously been reduced to zero through the recognition of our portion of RGSE s net losses. We also recognized a gain of \$0.5 million in connection with the sale of Cinedigm Corp. Class A common stock which we received in connection with the sale of GVE to Cinedigm in October 2013.

D-44

GAIAM, INC.

Financial Statement Schedule II

Consolidated Valuation and Qualifying Accounts

(in thousands)	Begi	lance at inning of Year	Cha Co	ditions arged to ests and enses, net	to nd (Recoveries)		` '		E	lance at End of Year
Allowance for Doubtful										
Accounts:										
2015	\$	412	\$	525	\$	(15)	\$	952		
2014	\$	556	\$	184	\$	328	\$	412		
2013	\$	611	\$	63	\$	118	\$	556		
Allowance for Product Returns:										
2015	\$	817	\$	2,709	\$	2,950	\$	576		
2014	\$	1,576	\$	4,484	\$	5,243	\$	817		
2013	\$	2,579	\$	6,359	\$	7,362	\$	1,576		

D-45

ANNEX E

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

[Remainder of Page Intentionally Blank]

Unaudited Pro Forma Condensed Consolidated Financial Information

On May 4, 2016, Gaiam and its wholly owned subsidiary Gaiam Travel, Inc. (Gaiam Travel) entered into a Stock Purchase Agreement (the Travel Purchase Agreement) with Ben Bressler (together with Gaiam Travel, collectively, Sellers), Lindblad Expeditions Holdings, Inc. and Lindblad Expeditions, LLC (the Travel Purchaser) and closed the transactions contemplated by the Travel Purchase Agreement. Pursuant to the Travel Purchase Agreement, Travel Purchaser purchased from Gaiam Travel, and Gaiam Travel sold to Travel Purchaser, 51.4% of the equity securities of Natural Habitat, Inc. (Natural Habitat), representing all of Gaiam Travel s interest in Natural Habitat (the Travel Business Sale). Natural Habitat owns all of the assets and liabilities primarily related to, or used in, Gaiam s conservation adventure travel and ecotourism business, referred to in this filing as the Travel Business. At the closing, Travel Purchaser paid \$12,850,000 in cash to Gaiam Travel.

On May 10, 2016, Gaiam entered into a Membership Interest Purchase Agreement (the Brand Purchase Agreement) with Stretch & Bend Holdings, LLC (Brand Purchaser) and its direct parent, Sequential Brands Group, Inc. (Brand Purchaser Parent) and an Asset Purchase Agreement with Fit For Life, LLC (FFL Purchaser). Pursuant to the Asset Purchase Agreement, subject to the satisfaction of closing conditions described below, Gaiam will sell certain assets and FFL Purchaser will assume certain liabilities of the Brand Business for \$21,300,000. Pursuant to the Brand Purchase Agreement, subject to the satisfaction of closing conditions described below, Gaiam will sell all of the equity interests in Gaiam Brand Holdco, LLC (Brand Holdco, and, together with its subsidiaries, the Brand Companies) to Brand Purchaser (the Brand Business Sale) for \$145,700,000. At the time of the closing of the Brand Business Sale, Brand Holdco will hold, directly or indirectly through its subsidiaries, all of Gaiam s assets and liabilities primarily related to, or used in, Gaiam s fitness, yoga and wellness consumer products business (the Brand Business) other than the assets and liabilities transferred and assigned to FFL Purchaser.

The following pro forma condensed consolidated financial information is based on our historical financial statements, including certain pro forma adjustments, and has been prepared to illustrate the pro forma effect of the sale of Natural Habitat and the Brand Business. The unaudited pro forma condensed consolidated statements of operations for the three months ending March 31, 2016 and for the year ending December 31, 2015 assume that the sale occurred as of January 1, 2016 and January 1, 2015, respectively. The unaudited pro forma condensed consolidated balance sheet as of March 31, 2016 is presented as if the sale had occurred on March 31, 2016.

The unaudited pro forma condensed consolidated financial information has been prepared based upon available information and management estimates; actual amounts may differ from these estimated amounts. The unaudited pro forma condensed consolidated financial information is not necessarily indicative of the financial position or results of operations that might have occurred had the sale occurred as of the dates stated above. The pro forma adjustments are described in the notes.

The unaudited pro forma condensed consolidated financial information should be read in conjunction with the audited financial statements and notes and related Management s Discussion and Analysis of Financial Condition and Results of Operations (MD&A) included in the Company s Quarterly Report on Form 10-Q for the quarterly period ended March 31, 2016 and Annual Report on Form 10-K for the fiscal year ended December 31, 2015.

E-2

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Statement of Operations

Three Months Ending March 31, 2016

(in thousands, except per share data)

							Pro					
							Forma					
			Less:	Pr	0		Results	Less:	Pro			
			Natural	For	ma		(Adjusted	Brand	Forma			
			Habitat	•			for		Adjustment			
	Gaiam		Historica	_		n	-	n Historical(_	l		Pro
			Financia				of	Financial	of		I	Forma
	Financi	al	Statement	tsNatu	ıral		Natural	Statements			F	Results
	Statemen	nts	(a)(e)	Habi	tat)	Notes	Habitat)	(a)	Business)	Notes	(Co	mbined)
Net revenues	\$ 35,32	7	\$ 9,585	\$			\$ 25,742	\$ 21,913	\$ 193	(d)	\$	4,022
Cost of goods sold	18,93	7	6,312				12,625	12,024				601
Gross profit	16,39	0	3,273				13,117	9,889	193			3,421
Selling, general, and												
administrative	24,59	3	4,209		3	(b)	20,387	12,371	190	(c)(d)		8,206
Loss from operations	(8,20	3)	(936))	(3)		(7,270	(2,482)	3			(4,785)
Interest and other												
income (expense)	50	9	944				(435)	(399)				(36)
(Loss) income before												
income taxes	(7,69	4)	8		(3)		(7,705	2,881	3			(4,821)
Income tax (benefit)												
expense	(9)	1				(10) (13)				3
Net (loss) income												
from continuing												
operations	(7,68	5)	7		(3)		(7,695	(2,868)	3			(4,824)
Net loss attributable to		,	•		(0)		(.,0>0	(=,000)				(1,021)
noncontrolling interest		1	47				14	15				(1)
Net (loss) income												
from continuing												
operations attributable												
to Gaiam, Inc.	\$ (7,62	4)	\$ 54	\$	(3)		\$ (7,681)	\$ (2,853)	\$ 3		\$	(4,825)

Net loss from continuing operations

Edgar Filing: TRUPANION INC. - Form 3

per share: Basic and Diluted	\$ (0.31)	\$ (0.31)	\$ (0.20)
Weighted-average shares outstanding:			
Basic and Diluted	24.531	24.531	24,531

- (a) Represents the revenues and expenses excluding certain fixed overhead costs of Natural Habitat and the Brand Business included in Gaiam s historical financial statements.
- (b) Operating expenses are increased to reverse the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (c) Operating expenses are decreased to include the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (d) Revenue and expenses are increased to reflect rental income from the Brand Business.
- (e) Results are consolidated then adjusted for noncontrolling interests. Gaiam owned 51.4% of Natural Habitat.

E-3

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Statement of Operations

For the Year Ending December 31, 2015

(in thousands, except per share data)

					Pro Forma				
		Less:	Pro Forma		Results	Less: Brand	Pro Forma		
		Natural HabitatA	djustments		(Adjusted for	BusinessA		S	
	Gaiam		Disposition		Disposition		•		Pro
	Historical	Financial	of		of	Financial	of		Forma
		Statements			Natural	Statements			Results
	Statements	() ()	Habitat) No	otes	· · · · · · · · · · · · · · · · · · ·	` '	· · · · · · · · · · · · · · · · · · ·		(Combined)
Net revenues	\$ 188,018	\$ 43,514	\$		\$ 144,504	\$ 131,045	\$ 758	(d)	\$ 14,217
Cost of goods sold	103,249	27,544			75,705	73,684			2,021
Gross profit	84,769	15,970			68,799	57,361	758		12,196
Selling, general, and	00 =00	10.104	4.0	<i>(</i> 1 \)	<0.4 7.	40.050	= 40		•••
administrative	80,588	12,126	10	(b)	68,472	48,970	748	(c)(d)	20,250
Income (loss) from			(4.0)						(0.0-4)
operations	4,181	3,844	(10)		327	8,391	10		(8,054)
Interest and other	(1.406)	(1.160)			(220)	(10)			(220)
income (expense) Loss from equity	(1,406)	(1,168)			(238)	(10)			(228)
method investments	(465)				(465)	(382)			(83)
method myestments	(403)				(403)	(362)			(03)
Income (loss) before									
income taxes	2,310	2,676	(10)		(376)	7,999	10		(8,365)
Income tax expense									
(benefit)	1,219	1,154			65	68			(3)
Net income (loss) from continuing									
operations	1,091	1,522	(10)		(441)	7,931	10		(8,362)
Net income	1,071	1,522	(10)		(111)	7,751	-10		(0,002)
attributable to									
noncontrolling									
interest	(694)	(672)			(22)	(22)			
Net income (loss) from continuing	\$ 397	\$ 850	\$ (10)		\$ (463)	\$ 7,909	\$ 10		\$ (8,362)

operations attributable to Gaiam, Inc.

Net income (loss) from continuing operations per share: Basic and Diluted	\$ 0.02	\$ (0.02)	\$ (0.34)
Weighted-average shares outstanding:			
Basic	24,510	24,510	24,510
Diluted	24,612	24,612	24,612

- (a) Represents the revenues and expenses excluding certain fixed overhead costs of Natural Habitat and the Brand Business included in Gaiam s historical financial statements.
- (b) Operating expenses are increased to reverse the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (c) Operating expenses are decreased to include the impact of a human resources service allocation to Natural Habitat from Gaiam.
- (d) Revenue and expenses are increased to reflect rental income from the Brand Business.
- (e) Results are consolidated then adjusted for noncontrolling interests. Gaiam owned 51.4% of Natural Habitat.

E-4

Gaiam, Inc.

Unaudited Pro Forma Condensed Consolidated Balance Sheet

As of March 31, 2016

(in thousands)

									Pro									
									Forma									
				Less:		Pro			Results		Less:							
				latural		'orma		((Adjusted		Brand		Pro					
			H	Iabitat A	djı	ustmen	ts		for	В	Business		Forma					
	(Gaiam	Hi	istorical(Dis	positio	n	Ι	Disposition	H	istorical A	4d	justment	S			Pro	
	H	istorical	Fi	nancial		of			of	F	inancial(l	Dis	position (of		F	orma	
	Fi	inancial	Sta	tements	N	atural			Natural	St	atements		Brand			R	esults	
	Sta	atements		(a)	H	abitat)	Note	S	Habitat)		(a)	B	usiness)	Not	es (Coı	nbined)
Cash	\$	11,228	\$	3,951	\$	11,050	(b)	1	\$ 18,327	\$	6,715	\$	155,927	(b)	\$ 1	67,539	
Accounts																		
receivable, net		13,002		389					12,613		11,885						728	
Inventory, less																		
allowances		16,847							16,847		16,830						17	
Other current																		
assets		14,351		8,714					5,637		2,993						2,644	
Total current																		
assets		55,428		13,054		11,050			53,424		38,423		155,927			1	70,928	
Property and																		
equipment, net		25,590		1,418					24,172		2,301						21,871	
Media library, net		9,348							9,348		2,448						6,900	
Goodwill		15,448		3,384					12,064		1,455						10,609	
Other intangibles,																		
net		782		75					707		400						307	
Other assets		4,160		423					3,737		2,294						1,443	
Total assets	\$	110,756	\$	18,354	\$	11,050		1	\$ 103,452	\$	47,321	\$	155,927			\$ 2	12,058	
									ĺ									
Accounts payable	\$	10,437	\$	2,615	\$			1	\$ 7,822	\$	5,957	\$				\$	1,865	
Accrued																		
liabilities		20,779		13,310		544	(c)		8,013		2,110		8,337	(c)		14,240	
Total current																		
liabilities		31,216		15,925		544			15,835		8,067		8,337				16,105	
Total Gaiam, Inc.									,		·		,					
shareholders																		
equity		76,743		1,501		10,506	(d)		85,748		37,385		147,590	(d	.)	1	95,953	
									,				,				,	

Edgar Filing: TRUPANION INC. - Form 3

Noncontrolling interest	2,797	928		1,869	1,869		
Total equity	79,540	2,429	10,506	87,617	39,354	147,590	195,953
Total liabilities and equity	\$ 110,756	\$ 18,354	\$ 11,050	\$ 103,452	\$ 47,321	\$ 155,927	\$ 212,058

- (a) Represents the assets, liabilities, and equity attributable to Natural Habitat and the Brand Business included in Gaiam s historical financial statements.
- (b) Cash is adjusted to reflect the \$11.0 million estimated net proceeds from the sale of Natural Habitat and the \$155.9 million estimated net proceeds from the sale of the Brand Business.
- (c) Accrued liabilities have been adjusted to reflect the accrual of estimated nonrecurring costs directly related to the disposition (not paid in close).
- (d) Equity is adjusted to reflect the estimated net gain of \$9.0 million on the sale of Natural Habitat and to reflect the estimated net gain of \$110.2 million on the sale of the Brand Business.

E-5

ANNEX F

UNAUDITED CONSOLIDATED CARVE-OUT FINANCIAL STATEMENTS FOR THE BRAND BUSINESS

[Remainder of Page Intentionally Blank]

GAIAM Brand Company (without Travel),

a Division of Gaiam, Inc.

Unaudited Interim Condensed Consolidated Carve-Out Financial Statements

As of and for the Quarter Ended

March 31, 2016

F-2

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Interim Condensed Consolidated Carve-out Balance Sheets

(in thousands)

(in thousands)	March 31, 2016 (un	Dec audite	ember 31, 2015 d)
ASSETS	Ì		ĺ
Current assets:			
Cash	\$ 6,715	\$	3,668
Accounts receivable, net	11,885		26,057
Inventory, less allowances	16,830		17,301
Other current assets	2,993		3,979
Total current assets	38,423		51,005
Property and equipment, net	2,301		2,214
Media library, net	2,448		2,555
Goodwill	1,455		1,455
Other intangibles, net	400		404
Other assets	2,294		3,074
Total assets	\$ 47,321	\$	60,707
LIABILITIES AND EQUITY			
Current liabilities:			
Accounts payable	\$ 5,957	\$	12,122
Accrued liabilities	2,110		1,902
Total current liabilities	8,067		14,024
Commitments and contingencies			
Equity:			
Accumulated other comprehensive loss	(273)		(382)
Parent s equity in division	37,658		45,150
Total parent equity	37,385		44,768
Noncontrolling interest	1,869		1,915
Total equity	39,254		46,683
Total liabilities and equity	\$47,321	\$	60,707

See accompanying notes to the unaudited interim condensed consolidated carve-out financial statements.

Edgar Filing: TRUPANION INC. - Form 3

F-3

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Interim Condensed Consolidated Carve-out Statements of Operations

(in thousands)

	For the Three Months End March 31,				
(in thousands, except per share data)		2016		2015	
		(unaudited)			
Net revenue	\$	21,913	\$	26,254	
Cost of goods sold		12,024		14,596	
Gross profit		9,889		11,658	
Selling, general and administrative expense		12,371		12,164	
Loss from operations		(2,482)		(506)	
Other (expense) income, net		(399)		8	
Loss before income taxes and noncontrolling interest		(2,881)		(498)	
Income tax (benefit) expense		(13)		19	
Net loss		(2,868)		(517)	
Net loss (income) attributable to noncontrolling interest		15		(22)	
Net loss attributable to Gaiam, Inc.	\$	(2,853)	\$	(539)	

See accompanying notes to the unaudited interim condensed consolidated carve-out financial statements.

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Interim Condensed Consolidated Carve-out Statements of Comprehensive Income

(in thousands)

	For	the Three I	 Ended
(in thousands, except per share data)		2016 (unau	 2015
Net loss	\$	(2,868)	\$ (517)
Accumulated other comprehensive income (loss):			
Foreign currency translation gain (loss), net of tax		78	(103)
Comprehensive loss		(2,790)	(620)
Less: comprehensive loss attributable to the noncontrolling interest		46	31
Comprehensive loss attributable to Gaiam, Inc.	\$	(2,744)	\$ (589)

See accompanying notes to the unaudited interim condensed consolidated carve-out financial statements.

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Interim Condensed Consolidated Carve-out Statements of Cash Flows

(in thousands)

(in thousands)	For the Thi Ended M 2016 (unau	larch 31, 2015
Operating activities	(0/22002	
Net loss	\$ (2,868)	\$ (517)
Adjustments to reconcile net loss to net cash provided by operating activities:		
Depreciation	211	244
Amortization	107	115
Loss on equity method investments	378	
Other	(31)	
Changes in operating assets and liabilities:		
Accounts receivable, net	14,195	15,773
Inventory, less allowances	504	218
Other current and long term assets	1,295	(1,610)
Accounts payable	(6,261)	(8,622)
Accrued liabilities	378	459
Net cash provided by operating activities	7,908	6,060
Investing activities		
Purchase of property, equipment and media rights	(287)	(347)
Net cash used in investing activities	(287)	(347)
Financing activities		
Net capital paid to Parent	(4,639)	(2,120)
Net cash used in financing activities	(4,639)	(2,120)
Effect of exchange rates on cash	65	(90)
Net change in cash	3,047	3,503
Cash at beginning of period	3,668	5,612
Cash at end of period	\$ 6,715	9,115
Supplemental cash flow information		
Income taxes paid See accompanying notes to the unaudited interim condensed consolidated carve-	\$ 158 out financial states	\$ 68 ments.

Notes to unaudited interim condensed consolidated carve-out financial statements

1. Organization, Nature of Operations, and Principles of Carve-Out

The accompanying unaudited interim condensed consolidated carve-out financial statements represent the Gaiam branded yoga, fitness, and wellness products, GAIAM Brand Company (without Travel) (we , us , our or the Company), of Gaiam, Inc. and its subsidiaries (Gaiam or our Parent) as though we had been operating as a separate, stand-alone business since December 31, 2014. Gaiam was incorporated under the laws of the State of Colorado on July 7, 1988.

We were not operating as a separate legal entity within Gaiam. Accordingly, our financial statements have been prepared on a carve-out basis. The accompanying statements include allocations of certain expenses for human resources, facilities, accounting, and other services, plus share-based compensation allocated from Parent. The expense allocations have been determined on bases that we and our Parent consider to be reasonable reflections of the utilization of services or the benefits provided. In addition, the assets and liabilities include only those assigned to the carve-out business. The allocations and related estimates and assumptions are described more fully in Note 2, Related Party Transactions.

These unaudited interim condensed consolidated carve-out financial statements are prepared in accordance with accounting principles generally accepted in the United States, or GAAP, and they include our accounts and those of our subsidiaries, over which we exercise control. Intercompany transactions and balances have been eliminated.

Certain information and note disclosures normally included in annual financial statements prepared in accordance with accounting principles generally accepted in the United States have been condensed or omitted, although we believe that the disclosures made are adequate to make the information not misleading. In our opinion, the unaudited interim condensed consolidated carve-out financial statements contain all adjustments (consisting of only normal recurring adjustments) necessary to present fairly, in all material respects, our consolidated financial position as of March 31, 2016, the interim results of operations for the three months ended March 31, 2016 and 2015, and cash flows for the three months ended March 31, 2016 and 2015. These interim statements have not been audited. The balance sheet as of December 31, 2015 was derived from our audited consolidated financial statements included in our annual report on Form 10-K to the Securities and Exchange Commission. The unaudited interim condensed consolidated carve-out financial statements contained herein should be read in conjunction with our audited financial statements, including the notes thereto, for the year ended December 31, 2015.

The financial position, results of operations and cash flows for the interim periods disclosed in this report are not necessarily indicative of future financial results.

Use of Estimates

The preparation of financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the amounts reported in the accompanying financial statements and disclosures. Although we base these estimates on our best knowledge of current events and actions that we may undertake in the future, actual results may be different from the estimates.

Recent Accounting Pronouncements

The Financial Accounting Standards Board issued Accounting Standards Update (ASU) No. 2015-17, Income Taxes Balance Sheet Classification of Deferred Taxes (Topic 740). The amendments under the new guidance require

Edgar Filing: TRUPANION INC. - Form 3

that deferred tax liabilities and assets be classified as noncurrent in a classified statement of financial position. The guidance is effective for consolidated financial statements issued for annual periods beginning after December 15, 2016, and interim periods within those annual periods. Earlier application is permitted for all entities as of the beginning of an interim or annual reporting period. The new ASU is not expected to have a material impact on our reported financial position or results of operations due to our full valuation allowances.

F-7

2. Related Party Transactions

We provide human resources, facility, accounting, and other services to our Parent. Certain of our employees have been granted options under Gaiam s 1999 and 2009 Long-Term Incentive Plans. Our employees participate in Gaiam s medical, dental, 401(k) and other benefit programs. We obtain our business insurance under Gaiam s policies. The accompanying financial statements include allocations of all of these expenses. The allocation method varies based on the type of expense and is generally based on either proportionate revenue, efforts expanded, number of employees, space occupied, or benefits realized. We believe the allocation methodologies used are reasonable, have been consistently applied, and result in an appropriate allocation of costs incurred. However, these allocations may not be indicative of the cost had we been a stand-alone entity or of future services. Gaiam allocated \$0.7 million and \$0.7 million of costs to us, and we allocated \$0.2 million and \$0.2 million of costs to Gaiam during the quarters ended March 31, 2016 and 2015, respectively.

We lease our corporate headquarters from our Parent under a triple net operating lease ending in December of 2024, respectively. We incurred rent expense of \$0.1 million and \$0.1 million and allocated operating costs of \$0.1 million and \$0.1 million under these leases for the quarters ended March 31, 2016 and 2015, respectively.

As a public company, Gaiam incurs corporate expenses and certain affiliate charges for directors, reporting, stock exchange listing, certain legal costs, and other related costs. None of these costs have been allocated to us in the accompanying financial statements.

One of our executive officers has an outstanding loan with the company totaling \$0.1 million at March 31, 2016. The loan is due December 31, 2016 and is forgiven in annual installments of \$0.1 million each year on December 31, provided the executive continues employment with the Company.

3. Joint Venture

In 2015 the Company formed a partnership with a third party to jointly market a fitness-based infomercial and its associated products. Under the arrangement, the Company provides rights to an infomercial which it has developed, and the third party provides the necessary working capital to advertise and market the infomercial worldwide. We hold a 49% interest in the partnership, and are accounting for it as an equity method investment. During the three months ended March 31, 2016, the Company recognized a non-cash loss of \$0.4 million associated with the venture which is included in Interest and other income (expense), net in the accompanying statements of operations.

4. Equity

The following is a reconciliation from December 31, 2015 to March 31, 2016 of the carrying amount of total equity, equity attributable to our Parent, and equity attributable to the noncontrolling interest.

Noncontrolling		
erest		
1,915		

44 91 4 111 4 D

Edgar Filing: TRUPANION INC. - Form 3

Comprehensive loss:				
Net income	(2,868)	(2,853)		(15)
Foreign currency translation adjustment, net of				
taxes of \$34	78		109	(31)
Balance at March 31, 2016	\$ 39,254	\$ 37,658	\$ (273)	\$ 1,869

F-8

5. Share-Based Compensation

Share-based compensation expenses are included in selling, general, and administration expenses in the accompanying statement of operations. We recognize compensation expenses allocated to us by our Parent for options granted to our employees under Gaiam equity incentive plans. Our share-based compensation cost charged against income was \$0.2 million and \$0.2 million during the three months ended March 31, 2016 and 2015, respectively.

6. Income Taxes

During 2013, we determined that a full valuation allowance against our deferred tax assets was necessary due to the cumulative loss incurred over the three-year period ended December 31, 2013. Since that time, we have continued to provide a full valuation allowance against deferred tax assets. As income is generated in future periods, the Company expects to reverse the valuation allowance as utilization of the deferred tax assets occurs. As of March 31, 2016, our gross net operating losses were \$86.3 million and \$16.4 million for federal and state, respectively.

7. Commitments and Contingencies

From time to time, we are involved in legal proceedings that we consider to be in the normal course of business. Claimed amounts against us may be substantial but may not bear any reasonable relationship to the merits of the claim or the extent of any real risk of court or arbitral awards. We record accruals for losses related to those matters against us that we consider to be probable and that can be reasonably estimated. Although it is not feasible to predict the outcome of these matters with certainty, it is reasonably possible that some legal proceedings may be disposed of or decided unfavorably to us and in excess of the amounts currently accrued. Based on available information, in the opinion of management, settlements, arbitration awards and final judgments, if any, which are considered probable of being rendered against us in litigation or arbitration in existence at March 31, 2016 and can be reasonably estimated are reserved against or would not have a material adverse effect on our financial condition, results of operations or cash flows.

8. Subsequent Events

The Company evaluated subsequent events through May 13, 2016, the date these financial statements were issued. There were no material subsequent events that required recognition or additional disclosure in these financial statements, except as disclosed below

Pending Sale of the Brand Business

On May 10, 2016 we entered into agreements to sell all our assets and liabilities in exchange for \$167.0 million. The sale is planned to occur after the period covered by this report. Therefore, the information in this report should be read in light of the pending sale. There can be no assurance that we will be able to successfully consummate the sale.

F-9

GAIAM Brand Company (without Travel),

a Division of Gaiam, Inc.

Unaudited Consolidated Carve-Out Financial Statements

As of and For the Years Ended

December 31, 2015 and 2014

F-10

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Consolidated Carve-out Balance Sheets

December 31, 2015 and 2014

(in thousands)

	2015	2014
ASSETS		
Current assets:		
Cash	\$ 3,668	\$ 5,612
Accounts receivable, net	26,057	31,940
Inventory	17,301	20,265
Other current assets	3,979	3,118
Total current assets	51,005	60,935
Property and equipment, net	2,214	1,262
Media library, net	2,555	3,045
Goodwill	1,455	1,455
Other intangibles, net	404	419
Other assets	3,074	3,227
Total assets	\$ 60,707	\$70,343
LIABILITIES AND EQUITY		
Current liabilities:		
Accounts payable	\$ 12,122	\$ 17,116
Accrued liabilities	1,902	3,164
Total current liabilities	14,024	20,280
Commitments and contingencies		
Equity:		
Accumulated other comprehensive loss	(382)	(222)
Parent s equity in division:	45,150	48,328
Total parent equity	44,768	48,106
Noncontrolling interest	1,915	1,957
Total equity	46,683	50,063
Total liabilities and equity	\$ 60,707	\$ 70,343

See Accompanying Notes to Unaudited Consolidated Carve-out Financial Statements.

Edgar Filing: TRUPANION INC. - Form 3

F-11

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Consolidated Carve-out Statement of Operations

For the Year Ended December 31, 2015 and 2014

(in thousands)

	2015	2015		2014	
Net revenues	\$ 131,0)45	\$11	19,189	
Cost of goods sold	73,6	584	e	55,365	
Gross profit	57,3	861	5	53,824	
Selling, general and administration expense	48,9	970	5	50,356	
Income from operations	8,3	891		3,468	
Other expense, net	(3	392)		(55)	
Income before income taxes	7,9	99		3,413	
Income tax expense		68		478	
Net income	7,9	931		2,935	
Net income attributable to noncontrolling interest		(22)		(125)	
Net income attributable to Gaiam Brand	\$ 7,9	909	\$	2,810	

See Accompanying Notes to Unaudited Consolidated Carve-out Financial Statements.

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Consolidated Carve-out Statement of Comprehensive Income

For the Year Ended December 31, 2015 and 2014

(in thousands)

	2015	2014
Net income	\$7,931	\$ 2,935
Other comprehensive loss:		
Foreign currency translation loss	(224)	(148)
Comprehensive income	7,707	2,787
Comprehensive loss (income) attributable to the noncontrolling interest	42	(66)
Comprehensive income attributable to Gaiam Brand	\$7,749	\$ 2,721

See Accompanying Notes to Unaudited Consolidated Carve-out Financial Statements.

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Consolidated Carve-out Statement of Changes in Parent s Equity

		Attributable to Parent Accumulated				
		Parent s Equity Other				
		in	_	rehensive		ontrolling
(in thousands)	Total	Division		Loss	In	terest
Balance at December 31, 2013	\$ 60,803	\$ 59,045	\$	(133)	\$	1,891
Capital contribution by Parent of investment in						
acquisition	840	840				
Net capital withdrawal by Parent	(14,367)	(14,367)				
Comprehensive loss:						
Net income	2,935	2,810				125
Foreign currency translation adjustment, net of						
taxes of \$61	(148)			(89)		(59)
Balance at December 31, 2014	\$ 50,063	\$ 48,328	\$	(222)	\$	1,957
Net capital withdrawal by Parent	(11,087)	(11,087)				
Comprehensive loss:						
Net income	7,931	7,909				22
Foreign currency translation adjustment, net of						
taxes of \$96	(224)			(160)		(64)
Balance at December 31, 2015	\$ 46,683	\$ 45,150	\$	(382)	\$	1,915

See Accompanying Notes to Unaudited Consolidated Carve-out Financial Statements.

GAIAM Brand Company (without Travel), a Division of GAIAM, INC.

Unaudited Consolidated Carve-out Statement of Cash Flows

For the Year Ended December 31, 2015 and 2014

(in thousands)

	2015	2014
Operating activities:		
Net income	\$ 7,931	\$ 2,935
Adjustments to reconcile net income to net cash provided by (used in) operating		
activities:		
Depreciation	843	768
Amortization	754	341
Loss on equity method investment	382	
Other		(2)
Changes in operating assets and liabilities, net of effects from acquisitions:		
Accounts receivable, net	5,772	(3,145)
Inventory	2,890	(315)
Other assets	133	(1,167)
Accounts payable	(5,059)	4,301
Accrued liabilities	(1,192)	(4,389)
Net cash provided by (used in) operating activities	12,454	(673)
Investing activities:	(2.221)	(1.022)
Purchase of property, equipment and media rights	(3,221)	(1,023)
Purchase of businesses and equity-method investments, net of cash acquired		(4,155)
Net cash used in by investing activities	(3,221)	(5,178)
Financing activities:		
Net capital paid to Parent	(11,087)	(14,367)
Net cash used in financing activities	(11,087)	(14,367)
Effect of exchange rates on cash	(90)	(149)
Net change in cash	(1,944)	(20,367)
Cash at beginning of year	5,612	25,979
Cash at end of year	\$ 3,668	\$ 5,612

Supplemental cash flow information:

Income taxes paid	\$ 89	\$ 224
Liabilities and debt assumed from acquisitions		135
Stock issued by Parent in connection with business acquisitions	\$	\$ 840

See Accompanying Notes to Unaudited Consolidated Carve-out Financial Statements.

Notes to Unaudited Consolidated Carve-out Financial Statements

1. Organization, Nature of Operations, and Principles of Carve-Out

The accompanying unaudited carve-out financial statements represent the Gaiam branded yoga, fitness, and wellness products, GAIAM Brand Company (without Travel) (we , us , our or the Company), of Gaiam, Inc. and its subsidiar (Gaiam or our Parent) as though we had been operating as a separate, stand-alone business since December 31, 2013. Gaiam was incorporated under the laws of the State of Colorado on July 7, 1988.

We were not operating as a separate legal entity within Gaiam. Accordingly, our financial statements have been prepared on a carve-out basis. The accompanying statements include allocations of certain expenses for human resources, facilities, accounting, and other services, plus share-based compensation allocated from Parent. The expense allocations have been determined on bases that we and our Parent consider to be reasonable reflections of the utilization of services or the benefits provided. In addition, the assets and liabilities include only those assigned to the carve-out business. The allocations and related estimates and assumptions are described more fully in Note 3, Related Party Transactions.

The unaudited carve-out financial statements are prepared in accordance with accounting principles generally accepted in the United States, or GAAP, and they include our accounts and those of our subsidiaries, over which we exercise control. Intercompany transactions and balances have been eliminated.

2. Significant Accounting Policies

Subsequent Events

The Company evaluated subsequent events through March 3, 2015, the date these financial statements were issued. There were no material subsequent events that required recognition or additional disclosure in these financial statements.

Cash

Cash represents on-demand accounts with financial institutions that are denominated in U.S. dollars and foreign currencies. At each balance sheet date, cash on hand that is denominated in a foreign currency is adjusted to reflect the exchange rate that existed at the balance sheet date. The difference is reported as a gain or loss in our statement of operations each period. Historically, such gains or losses have been immaterial.

Concentration of Risk

We have a concentration of credit risk in our accounts receivable with our top customers as shown in the table below. Target, Kohl s, and Amazon are major retailers in the United States to which we make significant sales during the year-end holiday season. A significant downturn in sales with or loss of any of these customers would have a material adverse impact on our financial statements.

Customer Receivables as a Percent of Total Gross Receivables	2015	2014
Target	35%	48%
Kohl s	24%	10%

Edgar Filing: TRUPANION INC. - Form 3

Amazon

8%

18%

Customer Revenue as a Percent of Total Net Revenues	2015	2014
Target	31.3%	40.9%
Kohl s	25.9%	5.6%
Amazon	13.3%	11.4%

F-16

Allowances for Doubtful Accounts and Other Accounts Receivable Reserves

We maintain allowances for doubtful accounts for estimated losses resulting from the inability of our customers to make required payments. We make estimates of the collectability of our accounts receivable by analyzing historical bad debts, specific customer creditworthiness, and current economic trends. It is our policy to write off receivables after all avenues of collection have been pursued and the probability of collection is considered remote. If the financial condition of our customers were to deteriorate such that their ability to make payments to us was impaired, additional allowances could be required.

We record allowances for product returns to be received in future periods at the time we recognize the original sale. We record a reduction in revenue and a corresponding reduction in cost of sales. We estimate the amount of returns based on historical experience and future expectations.

We also maintain reserves for other deductions taken by our customers—such as co-op advertising and markdowns. These are estimated based on historical experience and future expectations. We estimate the amount of these deductions at the time revenue is recognized and record a reduction of revenue for markdowns, or an expense for co-op advertising. As of December 31, 2015 and 2014, we had bad debt reserves of \$1.0 and \$0.4 million, respectively, and we had accounts receivable reserves for returns and other deductions totaling \$6.1 million and \$5.9 million, respectively.

Inventory

Inventory consists primarily of finished goods held for sale and is stated at the lower of cost (first-in, first-out method) or market. Costs associated with in-bound freight and duties are capitalized in inventory and included in cost of goods sold. We identify the inventory items to be written down for obsolescence based on the item s current sales status and condition. We write down discontinued or slow moving inventories based on an estimate of the markdown required to sell off the inventory. As of December 31, 2015 and 2014, we estimated obsolete or slow-moving inventory to be \$0.7 million and \$1.1 million, respectively.

Advertising Costs

Deferred advertising costs relate to the preparation, printing, advertising and distribution of catalogs and are included in other current assets in the accompanying balance sheet. We defer such costs for financial reporting purposes until the catalogs are distributed, and then we amortize these costs over succeeding periods on the basis of estimated direct relationship sales. We amortize our seasonal catalogs within six months. Forecasted sales are the principal factor we use in estimating the amortization rate. We expense other advertising and promotional costs as incurred. The amount recorded as advertising expense was \$7.7 million and \$9.9 million for the years ended December 31, 2015 and 2014, respectively, and we included that amounts in selling, general, and administration expense. As we have significantly reduced production of catalogs in 2015, there were no remaining deferred costs at December 31, 2014.

Property and Equipment

We state property and equipment at cost less accumulated depreciation and amortization. We include in property and equipment the cost of internal-use software, including software used in connection with our websites. We expense all costs related to the development of internal-use software other than those incurred during the application development stage. We capitalize the costs we incur during the application development stage and amortize them over the estimated useful life of the software, which is typically three years. We compute depreciation of property and equipment on the straight-line method over estimated useful lives, generally three to seven years. We amortize leasehold improvements

over the shorter of the estimated useful lives of the assets or the remaining term of the lease. Depreciation expense is included in selling, general and administration expense in the accompanying statement of operations.

F-17

Purchase Accounting

We account for the attainment of a controlling interest in a business using the acquisition method. In determining the estimated fair value of certain acquired assets and liabilities, we make assumptions based upon many different factors, such as historical and other relevant information and analyses performed by independent parties. Assumptions may be incomplete, and unanticipated events and circumstances may occur that could affect the validity of such assumptions, estimates, or actual results.

Media Library

Our media library asset represents the fair value of libraries of media acquired through business combinations, the purchase price of media rights to both video and audio titles, and the capitalized cost to produce media products, all of which we market to retailers and to e-commerce customers. Our media library is shown in the accompanying balance sheet net of accumulated amortization of \$14.3 million and \$13.8 million at December 31, 2015 and 2014, and is amortized over the estimated useful lives of the titles, which range from five to seven years.

Capitalized media library production costs consist of costs incurred to produce the media content, net of accumulated amortization. We recognize these costs, as well as participation costs, as expenses on an individual title basis equal to the ratio that the current year s gross revenues bear to our estimate of total ultimate gross revenues from all sources to be earned over a maximum seven-year period. We state capitalized production costs at the lower of unamortized cost or estimated fair value on an individual title basis. We continually review revenue forecasts, based primarily on historical sales statistics, and revise these forecasts when warranted by changing conditions. When estimates of total revenues and other events or changes in circumstances indicate that a title has an estimated fair value that is less than its unamortized cost, we recognize an impairment loss in the current period for the amount by which the unamortized cost exceeds the title s estimated fair value.

Amortization expense for capitalized produced media content and acquired media rights is shown in the table below.

(in thousands)	2015	2014
Total media amortization expense	\$611	\$327

Based on total media library costs at December 31, 2015 and assuming no subsequent impairment of the underlying assets or a material increase in the video productions or media acquired, we expect the amortization expense for 2016, 2017, 2018, and 2019 to be \$0.6 million, \$0.3 million, \$0.1 million and \$0.1 million, respectively.

Goodwill and Other Intangibles

Goodwill represents the excess of the purchase consideration over the estimated fair value of assets acquired less liabilities assumed in a business acquisition. Our other intangibles consist of customer related assets. We review goodwill for impairment annually or more frequently if impairment indicators arise on a goodwill reporting unit level. We have the option of first assessing qualitative factors to determine whether events and circumstances indicate that it is more likely than not that the fair value of a goodwill reporting unit is less than its carrying amount. If it is determined that the fair value for a goodwill reporting unit is more likely than not greater than the carrying amount for that goodwill reporting unit, then the two-step impairment test is unnecessary. If it is determined that the two-step impairment test is necessary, then for step one, we compare the estimated fair value of a goodwill reporting unit with its carrying amount, including goodwill. If the estimated fair value of a goodwill reporting unit exceeds its carrying

amount, we consider the goodwill of the reporting unit not impaired. If the carrying amount of a goodwill reporting unit exceeds its estimated fair value, we perform the second step of the goodwill impairment test to measure the amount of impairment loss. We use either a comparable market

F-18

approach or a traditional present value method to test for potential impairment. The process of evaluating the potential impairment of goodwill is highly subjective and requires significant judgment at many points during the analysis. Application of alternative assumptions and definitions could yield significantly different results.

Long-Lived Assets

We evaluate the carrying value of long-lived assets held and used, other than goodwill, when events or changes in circumstances indicate the carrying value may not be recoverable. We consider the carrying value of a long-lived asset impaired when the total projected undiscounted cash flows from such asset are separately identifiable and are less than the carrying value. We recognize a loss based on the amount by which the carrying value exceeds the estimated fair value of the long-lived asset. We determine the estimated fair value primarily using the projected cash flows from the asset discounted at a rate commensurate with the risk involved.

Income Taxes

Our results are included in the Gaiam consolidated federal income tax returns and certain combined state income tax returns. Gaiam allocates income tax expenses to GAIAM Brand Company based on a separate return allocation method which results in income tax expense that approximates the expense that would result if GAIAM Brand Company were a stand-alone entity. Our reported deferred tax assets and liabilities, as discussed below and in Note 11 Income Taxes, are primarily determined as a result of the application of the separate return allocation method and therefore the settlement of these amounts is dependent upon Gaiam rather than tax authorities. Our current expectation is that the vast majority of deferred tax assets and liabilities will be settled through Gaiam s general intercompany obligations. Gaiam has the right to change their policy regarding settlement of these assets and liabilities at any time.

The provision for income taxes consists of an amount for taxes currently payable and an amount for tax consequences deferred to future periods. We record deferred income tax assets and liabilities reflecting future tax consequences attributable to differences between the financial statement carrying value of assets and liabilities and the tax bases of those assets and liabilities. Deferred taxes are computed using enacted tax rates expected to apply in the year in which the differences are expected to affect taxable income. The effect on deferred income tax assets and liabilities of the change in the tax rate is recognized in earnings in the period that includes the enactment date.

We establish valuation allowances when necessary to reduce deferred income tax assets to the amounts that we believe are more likely than not to be recovered. Considerable judgment is required in determining when these events may occur and whether recovery of an asset, including the utilization of a net operating loss or other carryforward prior to its expiration, is more likely than not. Due to aggregate losses in the last three years, we established a full valuation allowance on our deferred tax assets at the end of 2013. See Note 11 Income Taxes for additional information.

Revenue

We recognize revenue when the goods are shipped to the customer and collection is either probable or has occurred. The amount of revenue recognized is net of estimated returns and other chargebacks (or channel credits), which are estimated using historical return and credit rates. If the actual amount of returns and chargebacks were to vary significantly from our estimates, it could materially impact our results of operations in subsequent periods. We identify and record any cooperative advertising expenses we pay, which are for advertisements meeting the separable benefit and fair value tests, as part of selling, general, and administration expense. We recognize amounts billed to customers for postage and handling as revenue at the same time we recognize the revenue arising from the product sale. Costs for postage and handling are included in selling, general, and administration expenses in the accompanying

financial statements. We present revenue net of taxes collected from customers.

F-19

Defined Contribution Plan

Gaiam has adopted a defined contribution retirement plan under Section 401(k) of the Internal Revenue Code of 1986, as amended (Internal Revenue Code), which covers substantially all our employees. Eligible employees may contribute amounts to the plan, via payroll withholding, subject to certain limitations. The 401(k) plan permits, but does not require, us to make additional matching contributions to the 401(k) plan on behalf of all participants in the 401(k) plan. We match 50% of an employee s contribution, up to an annual maximum matching contribution of \$1,500, we made matching contributions to the 401(k) plan of \$0.1 million and \$0.1 million during 2015 and 2014, respectively.

Foreign Currency Translation

Our foreign subsidiaries in Australia and the United Kingdom use their local currency as their functional currency. We translate assets and liabilities into U.S. dollars at exchange rates in effect at the balance sheet date. We translate income and expense accounts at the average monthly exchange rates during the year. We record resulting translation adjustments, net of income taxes, as a separate component of accumulated other comprehensive income.

Comprehensive Income

Our comprehensive income is comprised of our net income, noncontrolling interest net income, and foreign currency translation adjustments, net of tax.

The tax effects allocated to our accumulated other comprehensive income components for the years ended December 31, 2015 and 2014 were as follows (in thousands):

(in thousands)	2015	2014
Before-tax amount	\$ (320)	\$ (209)
Tax expense (benefit)	(96)	(61)
Net-of-tax amount	\$ (224)	\$ (148)

Fair Value of Financial Instruments

The carrying amounts of our cash, accounts receivable, accounts payable and other current liabilities approximate their fair values.

Recently Issued Accounting Pronouncement

In February of 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update No. 2016-02, *Leases* (Topic 842). The new standard establishes a right-of-use (ROU) model that requires a lessee to record a ROU asset and a lease liability on the balance sheet for all leases with terms longer than 12 months. Leases will be classified as either finance or operating, with classification affecting the pattern of expense recognition in the income statement. The new standard is effective for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. A modified retrospective transition approach is required for lessees for capital and operating leases existing at, or entered into after, the beginning of the earliest comparative period presented in the financial statements, with certain practical expedients available. We are currently evaluating the impact of our pending

adoption of the new standard on our consolidated financial statements.

In May of 2014, the Financial Accounting Standards Board (FASB) issued ASU No. 2014-09, *Revenue from Contracts with Customers (Topic 606)* (ASU 2014-09). The new standard supersedes most previously existing revenue recognition rules, and will become effective for us in the first quarter of 2017. The core principle of the guidance is that an entity should recognize revenue to depict the transfer of promised goods or services to

F-20

customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those goods or services. Our revenue transactions typically consist of one, distinct, fixed-price performance obligation which is delivered to the customer at a single point in time. We are monitoring the evolving interpretations and implementation guidance. Our preliminary assessment is that we do not expect the new standard to have a material impact on our reported financial position or results of operations.

Use of Estimates

The preparation of financial statements in accordance with GAAP requires us to make estimates and assumptions that affect the amounts reported in the accompanying financial statements and disclosures. Although we base these estimates on our best knowledge of current events and actions that we may undertake in the future, actual results may be different from the estimates.

3. Related Party Transactions

We provide human resources, facility, accounting, and other services to our Parent. Certain of our employees have been granted options under Gaiam s 1999 and 2009 Long-Term Incentive Plans. Our employees participate in Gaiam s medical, dental, 401(k) and other benefit programs. We obtain our business insurance under Gaiam s policies. The accompanying financial statements include allocations of all of these expenses. The allocation method varies based on the type of expense and is generally based on either proportionate revenue, efforts expanded, number of employees, space occupied, or benefits realized. We believe the allocation methodologies used are reasonable, have been consistently applied, and result in an appropriate allocation of costs incurred. However, these allocations may not be indicative of the cost had we been a stand-alone entity or of future services. Gaiam allocated \$2.7 million and \$2.4 million of costs to us, and we allocated \$0.6 million and \$1.2 million of costs to Gaiam during the years ended December 31, 2015 and 2014, respectively.

We lease our corporate headquarters from our Parent under a triple net operating lease ending in December of 2024, respectively. We incurred rent expense of \$0.5 million and \$0.5 million and allocated operating costs of \$0.4 million and \$0.3 million under these leases for the years ended December 31, 2015 and 2014, respectively.

As a public company, Gaiam incurs corporate expenses and certain affiliate charges for directors, reporting, stock exchange listing, certain legal costs, and other related costs. None of these costs have been allocated to us in the accompanying financial statements.

One of our executive officers has an outstanding loan with the company totaling \$0.1 million at December 31, 2015. The loan is due December 31, 2016 and is forgiven in annual installments of \$0.1 million each year on December 31, provided the executive continues employment with the Company.

4. Property and Equipment

Property and equipment, stated at lower of cost or estimated fair value, consists of the following as of December 31, 2015 and 2014 (in thousands):

	2015	2014
Furniture, fixtures and equipment	5,064	4,710
Leasehold improvements	996	1,477

Edgar Filing: TRUPANION INC. - Form 3

Website development costs and other software	9,605	8,409
Studios, computer and telephone equipment	8,718	8,608
Warehouse and distribution equipment	1,765	1,765
	26,148	24,969
Accumulated depreciation and amortization	(23,934)	(23,707)
	\$ 2,214	\$ 1,262

5. Acquisitions

Yoga Studio

In September 2014, we acquired all the outstanding stock of Modern Lotus Limited (Yoga Studio), the number-one-ranked paid yoga app on the U.S. App Store. Our Parent issued 115,369 shares of stock with a market value of \$7.28 per share and paid net cash of \$1.4 million to acquire Yoga Studio. The aggregate consideration paid by us exceeded the aggregate estimated fair value of the assets acquired and the liabilities assumed by \$1.5 million which we have recognized as goodwill. We attribute the goodwill to the recognized market position of the app.

The following table sets forth the changes in goodwill during 2015 and 2014.

(in thousands)	
Balance at December 31, 2013	
Acquisitions	1,455
Balance at December 31, 2014 Acquisitions	\$ 1,455
Balance at December 31, 2015	\$ 1,455

The following table represents our intangibles assets by major class as of December 31, 2015 and 2014.

	As of December 31,			
(in thousands)	20	015	2	014
Indefinite Lived Intangibles	\$	340	\$	340
Intangibles Subject to Amortization				
Customer related:				
Gross carrying amount	\$	200	\$	200
Accumulated amortization	((200)		(200)
	\$		\$	
Marketing related:				
Gross carrying amount	\$	618	\$	618
Accumulated amortization	((554)		(539)
	\$	64	\$	79

The amortization periods range from 24 to 72 months.

6. Accrued Liabilities

Accrued liabilities consist of the following as of December 31, 2015 and 2014 (in thousands):

(in thousands)	2015	2014
Accrued compensation	\$ 1,028	\$ 1,646
Participations payable	570	1,079
Other accrued liabilities	304	439
	\$ 1,902	\$3,164

F-22

7. Commitments and Contingencies

Risks and Uncertainties

From time to time, we are involved in legal proceedings that we consider to be in the normal course of business. Claimed amounts against us may be substantial but may not bear any reasonable relationship to the merits of the claim or the extent of any real risk of court or arbitral awards. We record accruals for losses related to those matters against us that we consider to be probable and that can be reasonably estimated. Although it is not feasible to predict the outcome of these matters with certainty, it is reasonably possible that some legal proceedings may be disposed of or decided unfavorably to us and in excess of the amounts currently accrued. Based on available information, in the opinion of management, settlements, arbitration awards and final judgments, if any, which are considered probable of being rendered against us in litigation or arbitration in existence at December 31, 2015 and can be reasonably estimated are reserved against or would not have a material adverse effect on our financial condition, results of operations or cash flows. At December 31, 2015, management does not believe there are any related matters considered probable.

Operating Leases

We lease office and warehouse space through operating leases. One of the leases has a three-year renewal option. The following schedule represents the annual future minimum payments under these commitments, as of December 31, 2015 (in thousands):

	Operating Leases With Related Party (see Note 3)		Leas	erating ses with I Parties	Ope	Fotal erating eases
2016	\$	516	\$	828	\$	1,344
2017		531		755		1,286
2018		547		828		1,375
2019		564		431		995
2020		581				581
Thereafter		2,501				2,501
Total minimum lease payments	\$	5,240	\$	2,842	\$	8,082

We incurred rent expense of \$1.4 million and \$1.3 million for the years ended December 31, 2015 and 2014, respectively.

Media Distribution Payments

From time to time, we enter into exclusive media distribution agreements which require that we make periodic minimum payments against future distribution liabilities. The following schedule shows the annual future minimum payments under these agreements, as of December 31, 2015:

Edgar Filing: TRUPANION INC. - Form 3

(in thousands)	Distribution Payments
2016	\$ 2,698
2017	528
2018	96
2019	25
Total media distribution payments	\$ 3,347

8. Non-Controlling Interests

We own 50.01% of our Australian subsidiary, Gaiam Pty. The amount of the non-controlling interest is reflected separately in our consolidated financial statements, and all intercompany transactions have been eliminated.

9. Share-Based Compensation

Share-based compensation expenses are included in selling, general, and administration expenses in the accompanying statement of operations. We recognize compensation expenses allocated to us by our Parent for options granted to our employees under Gaiam equity incentive plans. Our share-based compensation cost charged against income was \$0.6 million and \$0.5 million during 2015 and 2014, respectively.

As of December 31, 2015, there was \$1.0 million of unrecognized cost related to unvested options granted by Gaiam. We expect that cost to be recognized over a weighted-average period of 2.96 years.

10. Income Taxes

Our provision for income tax expense is comprised of the following:

(in thousands)	2015	2014
Current:		
State	11	303
International	57	175
Total income tax expense	\$ 68	\$478

Because our eco-travel subsidiary is not part of the consolidated tax group, the provision includes Federal taxes associated with its income. The state provision consists of the taxes due for subsidiaries of Gaiam which file taxes on a separate basis, and are not able to utilize combined group net operating losses.

Variations from the federal statutory rate are as follows:

(in thousands)	201	5	20	014
Expected federal income tax expense at statutory rate of 34%	\$ 2,7	20	\$ 1	1,161
Effect of permanent other differences		57		42
State income tax expense, net of federal benefit	1	19		51
Establishment of valuation allowance on net deferred tax assets	(3,2)	222)	(1	1,109)
Other		53		371
Effect of differences between U.S. taxation and foreign taxation		(3)		(38)
Effect of repatriation	3	344		
Total income tax expense	\$	68	\$	478

Deferred income taxes reflect net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes. The components of the net accumulated deferred income tax assets as of December 31, 2015 and 2014 are as follows:

(in thousands)	2015	2014	
Deferred tax assets (liabilities):			
Current:			
Provision for doubtful accounts	\$ 338	\$ 146	
Inventory-related expense	567	543	
Accrued liabilities	930	898	
Worthless stock deduction	206	206	
Other	9	(38)	
Total current deferred tax assets	2,050	1,755	
Valuation allowance	(2,050)	(1,755)	
Total current deferred tax assets, net of valuation allowance	\$	\$	
Non-current:			
Depreciation and amortization	\$ (66)	\$ (142)	
Section 181 qualified production expense	(1,481)	(1,081)	
Net operating loss carryforward	18,280	21,509	
Charitable carryforward	1,517	1,432	
Gain from foreign business acquisition	(103)	(54)	
Tax credits	862	862	
Other	14	10	
Total non-current deferred tax assets	19,023	22,536	
Valuation allowance	(19,023)	(22,536)	
Total non-current deferred tax assets, net of valuation			
allowance			
Total net deferred tax assets	\$	\$	

As of December 31, 2015, our gross net operating losses were \$49.6 million and \$14.2 million for federal and state, respectively. These tax attributes along with charitable contributions and tax credits are expected to remain with the Parent company. The sources of income before income taxes are as follows:

(in thousands) Domestic	2015 \$ 7,573	2014 \$ 2,728
International	426	686
	\$7,999	\$3,413

Certain of our subsidiaries, namely those for which we own less than 80% of their shares and voting rights and/or are foreign entities, file tax returns separately from Gaiam s consolidated tax group. At December 31, 2015, we had made a provision for U.S. federal and state income taxes on approximately \$0.4 million of undistributed foreign earnings, which are not expected to remain outside of the U.S. indefinitely. Deferred tax liabilities have been established for future taxes on distribution of foreign earnings in the form of dividends or otherwise, in order to derive, for financial statement purposes, the U.S. income taxes (net of tax on foreign tax credits), state income taxes, and withholding taxes payable to the various foreign countries.

Periodically, we perform assessments of the realization of our net deferred tax assets considering all available evidence, both positive and negative. On the basis of this assessment, we recorded a charge of \$22.6 million to income tax expense to record a full valuation allowance against our deferred tax assets as of December 31, 2013. A significant piece of evidence evaluated was the consolidated cumulative loss incurred over the three-year

F-25

period ended December 31, 2013. Because the valuation allowance will remain in place until we return to profitability, we did not record any tax benefit in 2014 associated with our net loss or other deferred tax assets. We continue to be optimistic about our future, and expect to return to operating profitability. When that happens, we expect to reverse the valuation allowance and record the related tax benefit for future use of our net operating loss carryforwards we expect to realize.

We recognize the tax benefit from an uncertain tax position only if it is more likely than not that the tax position will be sustained on examination by the taxing authorities, based on the technical merits of the position. We measure the tax benefits recognized in the consolidated financial statements from such a position based on the largest benefit that has a greater than 50% likelihood of being realized upon ultimate resolution. The application of income tax law is inherently complex. Laws and regulations in this area are voluminous and are often ambiguous. As such, we are required to make many subjective assumptions and judgments regarding our income tax exposures. Interpretations of and guidance surrounding income tax law and regulations change over time and may result in changes to our subjective assumptions and judgments which can materially affect amounts recognized in our consolidated balance sheets and consolidated statements of operations. The result of our assessment of our uncertain tax positions did not have a material impact on our consolidated financial statements. Our federal and state tax returns for all years after 2010 are subject to future examination by tax authorities for all our tax jurisdictions. We recognize interest and penalties related to income tax matters in other expense, net and selling, general and administration expenses, respectively.

F-26